CS-22-213

Contract Tracking No.: CM3333 Bid No.:

CONTRACT FOR CONSTRUCTION OF CITRONA DRIVE PEDESTRIAN IMPROVEMENTS

THIS AGREEMENT is dated this <u>22nd</u> day of <u>May</u>, 20<u>23</u>, by and between **NASSAU COUNTY BOARD OF COUNTY COMMISSIONERS**, hereinafter referred to as the "County" and **CGC**, Inc., hereinafter referred to as the "Contractor".

The County and the Contractor, in consideration of the mutual covenants hereinafter set forth, agree as follows:

ARTICLE 1 - WORK

1.01 The Contractor shall complete all Work as specified or indicated in the Contract Documents as defined in Article 10 below. The Work is generally described as follows:

Citrona Drive Pedestrian Improvements

All Work shall be in accordance with the construction drawings, specifications, and Contract Documents. Where FDOT and County requirements differ, the more stringent shall apply.

ARTICLE 2 – THE PROJECT

2.01 The Project is for the Work referenced in Article 1 and the Contract Documents and is generally described as follows:

This project consists of construction of concrete, asphalt, signage, striping, and additional improvements as depicted in the construction plans to improve pedestrian facilities along Citrona Drive from Hickory Street and Beech Street.

ARTICLE 3 – ENGINEER

3.01 Managing authority for this project shall be the Nassau County Engineer, Robert T. Companion, P.E. or his authorized designee.

ARTICLE 4 – ENGINEER OF RECORD

4.01 The Project has been designed by the Florida Department of Transportation who is to assume all duties and responsibilities and have the rights and authority assigned to Engineer of Records in the Contract Documents in connection with completion of the Work in accordance with the Contract Documents.

ARTICLE 5 – CONTRACT TIMES

5.01 Time is of the Essence



Page 1 of 12

- A. All time limits for Milestones, if any, Substantial Completion, and Completion and readiness for Final Payment as stated in the Contract Documents are of the essence of the Contract.
- B. The Contractor hereby agrees to commence work under this Contract on a date to be specified in written "Notice to Proceed" of the County and to fully complete the Project as specified in Paragraph 5.02 of this Contract.
- C. Under no condition, shall written Notice to Proceed be issued until Contractor has provided the Performance Bonds specified in Section 6.01 of the General Conditions of Exhibit A.
- 5.02 Days to Achieve Substantial Completion and Final Payment

The Work will be substantially completed within <u>50</u> days after the date when the Contract Times commence to run as provided in Paragraph 4.01 of the General Conditions, and completed and ready for Final Payment in accordance with Paragraph 15.06 of the General Conditions of Exhibit A within <u>15</u> calendar days from the date of Substantial Completion. Total Contract Time shall be <u>65</u> calendar days; for everyday the Work goes beyond Substantial Completion, a day will be removed from Final Completion so the total days equal <u>65</u> calendar days.

- 5.03 Liquidated Damages
 - A. The Contractor and the County recognize that time is of the essence of this Contract and that the County will suffer financial loss if the Work is not completed within the times specified in Paragraph 5.02 above, plus any extensions thereof allowed in accordance with Article 11 of the General Conditions of Exhibit A. The parties also recognize the delays, expense, and difficulties involved in proving in legal or arbitration proceeding the actual loss suffered by the County if the Work is not completed on time. Accordingly, instead of requiring any such proof, the County and the Contractor agree that as liquidated damages for delay (but not as a penalty) the Contractor shall pay the County <u>\$1000.00</u> for each day that expires after the time specified in Paragraph 5.02 above for Substantial Completion and until the Work is substantially complete. After Substantial Completion, if the Contractor shall neglect, refuse, or fail to complete the remaining Work within the Contract Time or any proper extension thereof granted by the County, the Contractor shall pay the County <u>\$500.00</u> for each day that expires after the time specified in Paragraph 5.02 above for Completion and readiness for Final Payment until the Work is completed and ready

for)Final Payment. Initials



ARTICLE 6 - CONTRACT PRICE

- 6.01 The County shall pay the Contractor for completion of the Work in accordance with the General Conditions and Contract Documents an amount in current funds equal to the sum of the amounts determined pursuant to Article 15 "Payments to Contractor" of the General Conditions of Exhibit A and Paragraphs 6.01A and 6.01B below:
 - A. For all Work, at the prices stated in the Contractor's Bid.
 - B. For all Unit Price Work, an amount equal to the sum of the established unit price for each separately identified item of Unit Price Work times the estimated quantity of that item as indicated in Contractor's Bid for a Total of All Unit Prices of:

Two Hundred Twenty-One Thousand Seven Hundred Forty-Seven Dollars and Zero <u>Cents</u>

\$221,747.00

As provided in Paragraph 13.03 of the General Conditions of Exhibit A, estimated quantities are not guaranteed, and determinations of actual quantities and classification are to be made by the Engineer of Record as provided in Paragraph 13.03 of the General Conditions of Exhibit A. The Final Payment for all Unit Price Work shall be an amount equal to the sum of the established unit price for each separately identified item of Unit Price Work times the actual completed and accepted quantity of each item. Unit Prices have been computed as provided in Paragraph 13.03 of the General Conditions of Exhibit A.

ARTICLE 7 - PAYMENT PROCEDURES

- 7.01 Submittal and Processing of Payments
 - A. The Contractor shall submit Applications for Payment in accordance with Article 15 of the General Conditions of Exhibit A. Applications for Payment will be processed by the County as provided in the General Conditions.

7.02 Progress Payments; Retainage



Page 3 of 12

- A. The County shall make progress payments on account of the Contract Price on the basis of the Contractor's Applications for Payment once each month during performance of the Work as provided in Paragraphs 7.02.A.1 and 7.02.A.2 below. All such payments will be measured by the schedule of values established as provided in Paragraph 2.05.A of the General Conditions (and in the case of Unit Price Work based on the number of units completed) or, in the event there is no schedule of values, as provided in the General Requirements:
 - Prior to Substantial Completion, progress payments will be made in an amount equal to the percentage indicated below, but, in each case, less the aggregate of payments previously made and less such amounts as Engineer may determine or the County may withhold, including but not limited to liquidated damages, in accordance with Paragraph 15.01 of the General Conditions of Exhibit A:
 - a. <u>95%</u> percent of the Work completed (with the balance being retainage)
 - b. <u>95%</u> percent of cost of materials and equipment not incorporated in the Work (with the balance being retainage).
 - 2. Upon Substantial Completion, the County shall pay an amount sufficient to increase total payments to the Contractor to <u>95%</u> percent of the Contract Price (with the balance being retainage), less such amounts as the Engineer of Record shall determine or the County may withhold, for incomplete Work and for other items in accordance with Paragraph 15.01of the General Conditions of Exhibit A.
- 7.03 Final Payment
 - A. Upon Final Completion and acceptance of the Work in accordance with Paragraph 15.06 of the General Conditions of Exhibit A, the County shall pay the remainder of the Contract Price as recommended by the Engineer of Record as provided in said Paragraph 15.06.

ARTICLE 8 - INTEREST

8.01 All moneys not paid when due will be paid according to the Florida Department of Transportation (FDOT) Supplemental LAP Specifications for payment requirements.

ARTICLE 9 - CONTRACTOR'S REPRESENTATIONS



Page 4 of 12

- 9.01 In order to induce the County to enter into this Contract, the Contractor makes the following representations:
 - A. The Contractor is or has been prequalified by FDOT for the Work required by this Contract.
 - B. The Contractor has examined and carefully studied the Contract Documents including the General Conditions and the other related data identified in the Bidding Documents.
 - C. The Contractor has visited the Site and become familiar with and is satisfied as to the general, local, and Site conditions that may affect cost, progress, and performance of the Work.
 - D. The Contractor is familiar with and is satisfied as to all federal, state, and local Laws and Regulations that may affect cost, progress, and performance of the Work.
 - E. The Contractor has carefully studied all: (1) reports of explorations and tests of subsurface conditions at or contiguous to the Site and all drawings of physical conditions in or relating to existing surface or subsurface structures at or contiguous to the Site (except Underground Facilities) which have been identified in the Supplementary Conditions as provided in Paragraph 5.02 of the General Conditions of Exhibit A and (2) reports and drawings of a Hazardous Environmental Condition, if any, at the Site which has been identified in the Supplementary Conditions of Exhibit A.
 - F. The Contractor has obtained and carefully studied (or assumes responsibility for having done so) all additional or supplementary examinations, investigations, explorations, tests, studies, and data concerning conditions (surface, subsurface, and Underground Facilities) at or contiguous to the Site which may affect cost, progress, or performance of the Work or which relate to any aspect of the means, methods, techniques, sequences, and procedures of construction to be employed by the Contractor, including applying the specific means, methods, techniques, sequences, and procedures of construction, if any, expressly required by the Bidding Documents, and safety precautions and programs incident thereto.
 - G. The Contractor does not consider that any additional examinations, investigations, $\sqrt{90}$ protocorrelations, tests, studies, or data are necessary for the performance of the Work at

Initials

Initials

the Contract Price, within the Contract Times, and in accordance with the other terms and conditions of the Contract Documents.

- H. The Contractor is aware of the general nature of Work to be performed by the County and others at the Site that relates to the Work as indicated in the Contract Documents.
- The Contractor has correlated the information known to Contractor, information and observations obtained from visits to the Site, reports and drawings identified in the Contract Documents, and all additional examinations, investigations, explorations, tests, studies, and data with the Contract Documents.
- J. The Contractor has given the Engineer and the Engineer of Record written notice of all conflicts, errors, ambiguities, or discrepancies that the Contractor has discovered in the Contract Documents, and the written resolution thereof by the Engineer and the Engineer of Record is acceptable to the Contractor.
- K. The Contract Documents are generally sufficient to indicate and convey understanding of all terms and conditions for performance of the Work.

ARTICLE 10 - CONTRACT DOCUMENTS

10.01 Contents

- A. The Contract Documents consist of the following:
 - 1. This Contract
 - 2. Standard General Conditions of the Construction Contract (Attached hereto as Exhibit "A")
 - 3. Supplementary Conditions (Attached hereto as Exhibit "B")
 - 4. Florida Department of Transportation (FDOT) Supplemental LAP Specifications (Attached hereto as Exhibit "C")
 - 5. Plans and Drawings (Attached hereto as Exhibit "D")
 - 6. As-Built Requirement Checklist (Attached hereto as Exhibit "E")
 - 7. Road Closure Policy (Attached hereto as Exhibit "F")
 - 8. Contractor's Bid (Attached hereto as Exhibit "G")
 - 9. Certificate of Liability Insurance (Attached hereto as Exhibit "H")
 - 10. Title VI Appendices A and E (Attached hereto as Exhibit "J")

) A1. LAP Agreement (Attached hereto as Exhibit "K")



Page 6 of 12

- 12. FHWA 1273 (Attached hereto as Exhibit "L")
- 13. Federal Wage Rates (Attached hereto as Exhibit "M")
- 14. The following which may be delivered or issued on or after the Effective Date of the Contract and are not attached hereto:
 - a. Performance and Payment Bonds
 - b. Notice to Proceed
 - c. Work Change Directives
 - d. Change Orders
 - e. Certificate of Substantial Completion
 - f. Certificate of Final Inspection
 - g. Certificate of Engineer
 - h. Certificate of Final Completion
 - i. CONTRACTOR'S release
 - j. Drawings and plans
 - k. Supplemental Agreements/Contracts
 - I. The Contractor's Waiver of Lien (Partial)
 - m. The Contractor's Waiver of Lien (Final and Complete)
 - n. Subcontractor/Vendor's Waiver of Lien (Final and Complete)
 - o. Consent of Surety to Final Payment
 - p. Instructions to Bidders
 - q. The Contractor's Insurance Requirements, as set forth in the Bid Documents
- B. The documents listed in Paragraph 10.01.A are attached to this Contract (except as expressly noted otherwise above).
- C. There are no Contract Documents other than those listed above in this Article 10.
- D. The Contract Documents may only be amended, modified, or supplemented as provided in Paragraph 3.04 of the General Conditions of Exhibit A.

ARTICLE 11 - MISCELLANEOUS

11.01 Terms

A. Terms used in this Contract will have the meanings indicated in the General Conditions and the Supplementary Conditions.



Page 7 of 12



11.02 Assignment of Contract

A. No assignment by a party hereto of any rights under or interests in the Contract will be binding on another party hereto without the written consent of the party sought to be bound; and, specifically but without limitation, moneys that may become due and moneys that are due may not be assigned without such consent (except to the extent that the effect of this restriction may be limited by law), and unless specifically stated to the contrary in any written consent to an assignment, no assignment will release or discharge the assignor from any duty or responsibility under the Contract Documents.

11.03 Successors and Assigns

A. The County and Contractor each binds itself, its partners, successors, assigns, and legal representatives to the other party hereto, its partners, successors, assigns, and legal representatives in respect to all covenants, agreements, and obligations contained in the Contract Documents.

11.04 Severability

A. Any provision or part of the Contract Documents held to be void or unenforceable under any Law or Regulation shall be deemed stricken, and all remaining provisions shall continue to be valid and binding upon the County and the Contractor, who agree that the Contract Documents shall be reformed to replace such stricken provision or part thereof with a valid and enforceable provision that comes as close as possible to expressing the intention of the stricken provision.

11.05 Other Provisions

A. The County entered into an Agreement between the County and the State of Florida Department of Transportation, dated March 28, 2022, Contract Nos. <u>441241-1-58-01</u> and <u>441241-1-68-02</u> (the "LAP Agreement"). The Contractor shall comply with all of the provisions of the LAP Agreement, including Appendices A and E attached hereto, that are assignable to the Contractor under this Agreement. In addition, the Contractor understands and agrees that there are requirements in the LAP Agreement that require the County to include specific provisions of such LAP Agreement in this Agreement, and to require the Contractor to include those specific provisions in any Subcontract with a subcontractor ("LAP Provisions"). The parties agree that such LAP



Page 8 of 12



Provisions are incorporated herein by this reference. The Contractor agrees to comply with all of the LAP Provisions and to set out the LAP Provisions in any subcontract. The LAP Agreement is attached hereto as Exhibit "K".

- B. Florida Statute 255.099 (Preference to State residents)
 - 1. The Contractor agrees to give PREFERENCE to the employment of STATE residents in the performance of the Work on the project if STATE residents have substantially equal qualifications to those of nonresidents.
 - a) As used in this section, the term "substantially equal qualifications" means the qualifications of two or more persons among whom the employer cannot make a reasonable determination that the qualifications held by one person are better suited for the position that the qualifications held by the other person or persons.
 - b) A Contractor required to employ STATE residents must contact the Agency for Workforce Innovation to post the Contractor's employment needs in the STATE's job bank system.
- C. Public Records Requirement:

The County is a public agency subject to Chapter 119, Florida Statutes. IF THE CONTRACTOR HAS QUESTIONS REGARDING THE APPLICATION OF CHAPTER 119, FLORIDA STATUTES, TO THE CONTRACTOR'S DUTY TO PROVIDE PUBLIC RECORDS RELATING TO THIS CONTRACT, CONTACT THE CUSTODIAN OF PUBLIC RECORDS AT (904) 530-6090, RECORDS@NASSAUCOUNTYFL.COM, 96135 NASSAU PLACE, SUITE 6, YULEE, FLORIDA 32097. Under this agreement, to the extent that the Contractor is providing services to the County, and pursuant to Section 119.0701, Florida Statutes, the Contractor shall:

- 1. Keep and maintain public records required by the public agency to perform the service.
- 2. Upon request from the public agency's custodian of public records, provide the public agency with a copy of the requested records or allow the records to be inspected or copied within a reasonable time at a cost that does not exceed the cost provided in this chapter or as otherwise provided by law.



Ensure that public records that are exempt or confidential and exempt from



public records disclosure requirements are not disclosed except as authorized by law for the duration of the contract term and following completion of the Contract if the Contractor does not transfer the records to the public agency.

- 4. Upon completion of the Contract, transfer, at no cost, to the public agency all public records in possession of the Contractor or keep and maintain public records required by the public agency to perform the service. If the Contractor transfers all public records to the public agency upon completion of the Contract, the Contractor shall destroy any duplicate public records that are exempt or confidential and exempt from public records disclosure requirements. If the Contractor keeps and maintains public records upon completion of the Contract, the Contract, the Contract, the Contractor keeps and maintains public records upon completion of the Contract, the Contract, the Contractor shall meet all applicable requirements for retaining public records. All records stored electronically must be provided to the public agency, upon request from the public agency's custodian of public records, in a format that is compatible with the information technology systems of the public agency.
- 5. A request to inspect or copy public records relating to a County contract for services must be made directly to the County's Custodian of Public Records. If the County does not possess the requested records due to the Contractor maintaining the public records, the County shall immediately notify the Contractor of the request for records. The Contractor must provide the records to the County or allow the records to be inspected or copied within a reasonable time. If the Contractor does not comply with the County's request for records, the County shall be entitled to enforce the contract provisions herein for failure to comply with the terms of the contract. Any Contractor which fails to provide public records to the County within a reasonable time may also be subject to penalties as provided under Section 119.10, Florida Statutes, including punishment by fine or may be guilty of committing a misdemeanor of the first degree for any willful and knowing violation.
- D. Employment Eligibility. Contractor must comply with F.S. 448.095 and use the United States Department of Homeland Security's E-Verify system ("E-Verify") to verify the employment eligibility of all persons hired by Contractor during the term of this Agreement to work in Florida. Additionally, if Contractor uses subcontractors to perform any portion of the Work (under this Agreement), Contractor must include a requirement in the subcontractor's contract that the subcontractor use E-Verify to verify the employment eligibility of all persons hired by subcontractor to perform any such portion of the Work. Contractor must include a requirement in the subcontractor's contract that the subcontractor to perform any eligibility of all persons hired by subcontractor use E-Verify the employment eligibility of all persons hired by subcontractor to perform of the





Contract Tracking No.: CM3333 Bid No.:

work. Answers to questions regarding E-Verify as well as instructions on enrollment may be found at the E-Verify website: www.uscis.gov/e-verify.

IN WITNESS WHEREOF, the County and Contractor have signed this Contract in triplicate. One counterpart each has been delivered to the County, the Contractor, the Engineer and the Engineer of Record. All portions of the Contract Documents have been signed or identified by the County and the Contractor or identified by the Engineer and the Engineer of Record on their behalf.

This Contract will be effective on the date on which the Contract is signed and delivered by the last of the two parties to sign and deliver (which is the Effective Date of the Contract).

OWNER	CONTRACTOR
Nassau County Board of Coupty Commissioners Signed: Klynt A. Farmer Title: Chairman Date: May 17, 2023	CGC, Inc. Signed: Richard C. Gaskin, Jr. Title: President Date:
[CORPORATE SEAL] Attest: <u>p.p. 1/</u> Title: <u>John A. Crawford, Ex-Officio Cle</u> r	[CORPORATE SEAL] Attest: Richard C. Gaskin k Title: Vice President
Address for giving notices:	Address for giving notices:
96135 Nassau Place	7036 W. 12th Street
Yulee, FL 32097	Jacksonville, FL 32220
Phone: _904–530–6010 FAX:	Phone:(904) 783-4119 FAX:(904) 783-3401 License CGC1524573 / CUC1225351
	(Where applicable)
Approved as to form by County Attorney	
Agent for service of process:	

iso CMey

Signature



Page 11 of 12

/

Contract Tracking No.: CM3333 Bid No.:

(If the Owner is a corporation, attach evidence of authority to sign. If the Owner is a public body, attach evidence of authority to sign and resolution or other documents authorizing execution of Owner-Contractor Contract.)

(If the Contractor is a corporation or a partnership, attach evidence of authority to sign.)





Page 12 of 12



(seal

Supprimiting

STATE OF FLORIDA COUNTY OF <u>DUVAL</u>

I HEREBY CERTIFY

THAT <u>Richard C. Gaskin, Jr.</u>, is <u>President/Secretary</u> of <u>CGC, Inc.</u>, located at <u>7036 W 12th St., Jacksonville, FL 32220</u> and he is hereby authorized to execute the CONTRACT DOCUMENTS with cover letter dated, <u>March 28</u>, 2023 (Contract to be dated by County when fully executed), and that his execution thereof, attested by the Vice President/Assistant Secretary of the corporation and with corporate seal affixed, shall be the official act and deed of this corporation.

IN WITNESS WHEREOF, I have hereunto set my hand and affixed the official seal of the corporation this <u>27th</u> day of <u>April</u>, 2023.

Richard C. Gaskin, Vice President/Assistant Secretary

Sworn and subscribed to me this <u>27th</u> day of <u>April</u>, 2023

Notary Public Suller Signature

Tammy Burkard

Notary Public State of Florida Tammy Burkard My Commission HH 280898 Exp.10/11/2026

My Commission Expires: 10/11/2026

This document has important legal consequences; consultation with an attorney is encouraged with respect to its use or modification. This document should be adapted to the particular circumstances of the contemplated Project and the controlling Laws and Regulations.

STANDARD GENERAL CONDITIONS OF THE CONSTRUCTION CONTRACT

Prepared By









Endorsed By



EJCDC® C-700, Standard General Conditions of the Construction Contract. Copyright[®] 2018 National Society of Professional Engineers, American Council of Engineering Companies, and American Society of Civil Engineers. All rights reserved.

Copyright[©] 2018

National Society of Professional Engineers 1420 King Street, Alexandria, VA 22314-2794 (703) 684-2882

www.nspe.org

American Council of Engineering Companies 1015 15th Street N.W., Washington, DC 20005 (202) 347-7474

www.acec.org

American Society of Civil Engineers 1801 Alexander Bell Drive, Reston, VA 20191-4400 (800) 548-2723 www.asce.org

The copyright for this EJCDC document is owned jointly by the three sponsoring organizations listed above. The National Society of Professional Engineers is the Copyright Administrator for the EJCDC documents; please direct all inquiries regarding EJCDC copyrights to NSPE.

NOTE: EJCDC publications may be purchased at <u>www.ejcdc.org</u>, or from any of the sponsoring organizations above.

GUIDELINES FOR USE OF EJCDC® C-700, STANDARD GENERAL CONDITIONS OF THE CONSTRUCTION CONTRACT

1.0 PURPOSE AND INTENDED USE OF THE DOCUMENT

EJCDC[®] C-700, Standard General Conditions of the Construction Contract (2018), is the foundation document for the EJCDC Construction Series. The General Conditions define the basic rights, responsibilities, risk allocations, and contractual relationship of the Owner and Contractor, and establish how the Contract is to be administered.

2.0 OTHER DOCUMENTS

EJCDC documents are intended to be used as a system and changes in one EJCDC document may require a corresponding change in other documents. Other EJCDC documents may also serve as a reference to provide insight or guidance for the preparation of this document.

These General Conditions have been prepared for use with either EJCDC® C-520, Agreement Between Owner and Contractor for Construction Contract (Stipulated Price), or EJCDC® C-525, Agreement Between Owner and Contractor for Construction Contract (Cost-Plus-Fee) (2018 Editions). The provisions of the General Conditions and the Agreement are interrelated, and a change in one may necessitate a change in the other.

To prepare supplementary conditions that are coordinated with the General Conditions, use EJCDC® C-800, Supplementary Conditions of the Construction Contract (2018).

The full EJCDC Construction series of documents is discussed in the EJCDC[®] C-001, Commentary on the 2018 EJCDC Construction Documents (2018).

3.0 ORGANIZATION OF INFORMATION

All parties involved in a construction project benefit significantly from a standardized approach in the location of subject matter throughout the documents. Experience confirms the danger of addressing the same subject matter in more than one location; doing so frequently leads to confusion and unanticipated legal consequences. Careful attention should be given to the guidance provided in EJCDC® N-122/AIA® A521, Uniform Location of Subject Matter (2012 Edition) when preparing documents. EJCDC® N-122/AIA® A521 is available at no charge from the EJCDC website, <u>www.ejcdc.org</u>, and from the websites of EJCDC's sponsoring organizations.

If CSI MasterFormat[™] is used for organizing the Project Manual, consult CSI MasterFormat[™] for the appropriate document number (e.g., under 00 11 00, Advertisements and Invitations), and accordingly number the document and its pages.

4.0 EDITING THIS DOCUMENT

Remove these Guidelines for Use. Some users may also prefer to remove the two cover pages.

Although it is permissible to revise the Standard EJCDC Text of C-700 (the content beginning at page 1 and continuing to the end), it is common practice to leave the Standard EJCDC Text of C-700 intact and unaltered, with modifications and supplementation of C-700's provisions set forth in EJCDC[®] C-800, Supplementary Conditions of the Construction Contract (2018). If the Standard Text itself is revised, the

user must comply with the terms of the License Agreement, Paragraph 4.0, Document-Specific Provisions, concerning the tracking or highlighting of revisions. The following is a summary of the relevant License Agreement provisions:

- The term "Standard EJCDC Text" for C-700 refers to all text prepared by EJCDC in the main body of the document. Document covers, logos, footers, instructions, or copyright notices are not Standard EJCDC Text for this purpose.
- 2. During the drafting or negotiating process for C-700, it is important that the two contracting parties are both aware of any changes that have been made to the Standard EJCDC Text. Thus, if a draft or version of C-700 purports to be or appears to be an EJCDC document, the user must plainly show all changes to the Standard EJCDC Text, using "Track Changes" (redline/strikeout), highlighting, or other means of clearly indicating additions and deletions.
- 3. If C-700 has been revised or altered and is subsequently presented to third parties (such as potential bidders, grant agencies, lenders, or sureties) as an EJCDC document, then the changes to the Standard EJCDC Text must be shown, or the third parties must receive access to a version that shows the changes.
- 4. Once the document is ready to be finalized (and if applicable executed by the contracting parties), it is no longer necessary to continue to show changes to the Standard EJCDC Text. The user may produce a final version of the document in a format in which all changes are accepted, and the document at that point does not need to include any "Track Changes," redline/strikeout, highlighting, or other indication of additions and deletions to the Standard EJCDC Text.

5.0 LICENSE AGREEMENT

This document is subject to the terms and conditions of the License Agreement, 2018 EJCDC[®] Construction Series Documents. A copy of the License Agreement was furnished at the time of purchase of this document, and is available for review at <u>www.ejcdc.org</u> and the websites of EJCDC's sponsoring organizations.

STANDARD GENERAL CONDITIONS OF THE CONSTRUCTION CONTRACT

TABLE OF CONTENTS

Page	e
cle 1—Definitions and Terminology1	1
01 Defined Terms	1
02 Terminology	6
cle 2—Preliminary Matters	7
01 Delivery of Performance and Payment Bonds; Evidence of Insurance	7
02 Copies of Documents	7
03 Before Starting Construction	7
04 Preconstruction Conference; Designation of Authorized Representatives	8
05 Acceptance of Schedules	8
06 Electronic Transmittals	9
cle 3—Contract Documents: Intent, Requirements, Reuse	9
01 Intent	9
02 Reference Standards	0
.03 Reporting and Resolving Discrepancies	0
.04 Requirements of the Contract Documents	1
.05 Reuse of Documents	1
cle 4—Commencement and Progress of the Work12	2
.01 Commencement of Contract Times; Notice to Proceed	2
.02 Starting the Work	2
.03 Reference Points	2
.04 Progress Schedule	2
.05 Delays in Contractor's Progress12	2
cle 5—Site; Subsurface and Physical Conditions; Hazardous Environmental Conditions	4
.01 Availability of Lands	4
.02 Use of Site and Other Areas	4
.03 Subsurface and Physical Conditions1	5
.04 Differing Subsurface or Physical Conditions10	6

5.05	Underground Facilities
5.06	Hazardous Environmental Conditions at Site
Article 6	-Bonds and Insurance
6.01	Performance, Payment, and Other Bonds
6.02	Insurance—General Provisions
6.03	Contractor's Insurance
6.04	Builder's Risk and Other Property Insurance
6.05	Property Losses; Subrogation
6.06	Receipt and Application of Property Insurance Proceeds
Article 7	- Contractor's Responsibilities
7.01	Contractor's Means and Methods of Construction
7.02	Supervision and Superintendence
7.03	Labor; Working Hours
7.04	Services, Materials, and Equipment
7.05	"Or Equals"
7.06	Substitutes
7.07	Concerning Subcontractors and Suppliers
7.08	Patent Fees and Royalties
7.09	Permits
7.10	Taxes
7.11	Laws and Regulations
7.12	Record Documents
7.13	Safety and Protection
7.14	Hazard Communication Programs
7.15	Emergencies
7.16	Submittals
7.17	Contractor's General Warranty and Guarantee40
7.18	Indemnification
7.19	Delegation of Professional Design Services41
Article 8	-Other Work at the Site
8.01	Other Work
8.02	Coordination
8.03	Legal Relationships

EJCDC[®] C-700, Standard General Conditions of the Construction Contract. Copyright[®] 2018 National Society of Professional Engineers, American Council of Engineering Companies, and American Society of Civil Engineers. All rights reserved. TOC Page 2 of 5

Article 9-	-Owner's Responsibilities
9.01	Communications to Contractor
9.02	Replacement of Engineer
9.03	Furnish Data
9.04	Pay When Due
9.05	Lands and Easements; Reports, Tests, and Drawings
9.06	Insurance
9.07	Change Orders
9.08	Inspections, Tests, and Approvals
9.09	Limitations on Owner's Responsibilities
9.10	Undisclosed Hazardous Environmental Condition
9.11	Evidence of Financial Arrangements
9.12	Safety Programs
Article 10	0—Engineer's Status During Construction
10.01	Owner's Representative
10.02	Visits to Site
10.03	Resident Project Representative
10.04	Engineer's Authority
10.05	Determinations for Unit Price Work
10.06	Decisions on Requirements of Contract Documents and Acceptability of Work
10.07	Limitations on Engineer's Authority and Responsibilities47
10.08	Compliance with Safety Program47
Article 1	1—Changes to the Contract
11.01	Amending and Supplementing the Contract
11.02	Change Orders
11.03	Work Change Directives
11.04	Field Orders
11.05	Owner-Authorized Changes in the Work49
11.06	Unauthorized Changes in the Work
11.07	Change of Contract Price
11.08	Change of Contract Times51
11.09	Change Proposals
11.10	Notification to Surety

EJCDC® C-700, Standard General Conditions of the Construction Contract. Copyright[©] 2018 National Society of Professional Engineers, American Council of Engineering Companies, and American Society of Civil Engineers. All rights reserved. TOC Page 3 of 5

Article 12-	Claims	52
12.01	Claims	
Article 13-	-Cost of the Work; Allowances; Unit Price Work	
13.01	Cost of the Work	53
13.02	Allowances	57
13.03	Unit Price Work	57
Article 14-	-Tests and Inspections; Correction, Removal, or Acceptance of Defective Work	58
14.01	Access to Work	58
14.02	Tests, Inspections, and Approvals	58
14.03	Defective Work	60
14.04	Acceptance of Defective Work	60
14.05	Uncovering Work	60
14.06	Owner May Stop the Work	61
14.07	Owner May Correct Defective Work	61
Article 15	5—Payments to Contractor; Set-Offs; Completion; Correction Period	62
15.01	Progress Payments	
15.02	Contractor's Warranty of Title	65
15.03	Substantial Completion	65
15.04	Partial Use or Occupancy	
15.05	Final Inspection	67
15.06	Final Payment	67
15.07	Waiver of Claims	68
15.08	Correction Period	68
Article 16	5—Suspension of Work and Termination	69
16.01	Owner May Suspend Work	69
16.02	Owner May Terminate for Cause	
16.03	Owner May Terminate for Convenience	71
16.04	Contractor May Stop Work or Terminate	71
Article 17	7—Final Resolution of Disputes	72
17.01	Methods and Procedures	72
Article 18	8—Miscellaneous	72
18.01	Giving Notice	72
18.02	Computation of Times	72

EJCDC® C-700, Standard General Conditions of the Construction Contract. Copyright[©] 2018 National Society of Professional Engineers, American Council of Engineering Companies, and American Society of Civil Engineers. All rights reserved. TOC Page 4 of 5

18.03	Cumulative Remedies
18.04	Limitation of Damages
18.05	No Waiver
18.06	Survival of Obligations
18.07	Controlling Law
18.08	Assignment of Contract
18.09	Successors and Assigns
18.10	Headings73

STANDARD GENERAL CONDITIONS OF THE CONSTRUCTION CONTRACT

ARTICLE 1—DEFINITIONS AND TERMINOLOGY

1.01 Defined Terms

- A. Wherever used in the Bidding Requirements or Contract Documents, a term printed with initial capital letters, including the term's singular and plural forms, will have the meaning indicated in the definitions below. In addition to terms specifically defined, terms with initial capital letters in the Contract Documents include references to identified articles and paragraphs, and the titles of other documents or forms.
 - Addenda—Written or graphic instruments issued prior to the opening of Bids which clarify, correct, or change the Bidding Requirements or the proposed Contract Documents.
 - 2. Agreement—The written instrument, executed by Owner and Contractor, that sets forth the Contract Price and Contract Times, identifies the parties and the Engineer, and designates the specific items that are Contract Documents.
 - 3. Application for Payment—The document prepared by Contractor, in a form acceptable to Engineer, to request progress or final payments, and which is to be accompanied by such supporting documentation as is required by the Contract Documents.
 - 4. *Bid*—The offer of a Bidder submitted on the prescribed form setting forth the prices for the Work to be performed.
 - 5. Bidder-An individual or entity that submits a Bid to Owner.
 - 6. Bidding Documents—The Bidding Requirements, the proposed Contract Documents, and all Addenda.
 - 7. Bidding Requirements—The Advertisement or invitation to bid, Instructions to Bidders, Bid Bond or other Bid security, if any, the Bid Form, and the Bid with any attachments.
 - Change Order—A document which is signed by Contractor and Owner and authorizes an addition, deletion, or revision in the Work or an adjustment in the Contract Price or the Contract Times, or other revision to the Contract, issued on or after the Effective Date of the Contract.
 - 9. Change Proposal—A written request by Contractor, duly submitted in compliance with the procedural requirements set forth herein, seeking an adjustment in Contract Price or Contract Times; contesting an initial decision by Engineer concerning the requirements of the Contract Documents or the acceptability of Work under the Contract Documents; challenging a set-off against payments due; or seeking other relief with respect to the terms of the Contract.
 - 10. Claim
 - a. A demand or assertion by Owner directly to Contractor, duly submitted in compliance with the procedural requirements set forth herein, seeking an adjustment of Contract Price or Contract Times; contesting an initial decision by Engineer concerning the

EJCDC® C-700, Standard General Conditions of the Construction Contract. Copyright[®] 2018 National Society of Professional Engineers, American Council of Engineering Companies, and American Society of Civil Engineers. All rights reserved.

requirements of the Contract Documents or the acceptability of Work under the Contract Documents; contesting Engineer's decision regarding a Change Proposal; seeking resolution of a contractual issue that Engineer has declined to address; or seeking other relief with respect to the terms of the Contract.

- b. A demand or assertion by Contractor directly to Owner, duly submitted in compliance with the procedural requirements set forth herein, contesting Engineer's decision regarding a Change Proposal, or seeking resolution of a contractual issue that Engineer has declined to address.
- c. A demand or assertion by Owner or Contractor, duly submitted in compliance with the procedural requirements set forth herein, made pursuant to Paragraph 12.01.A.4, concerning disputes arising after Engineer has issued a recommendation of final payment.
- d. A demand for money or services by a third party is not a Claim.
- 11. Constituent of Concern—Asbestos, petroleum, radioactive materials, polychlorinated biphenyls (PCBs), lead-based paint (as defined by the HUD/EPA standard), hazardous waste, and any substance, product, waste, or other material of any nature whatsoever that is or becomes listed, regulated, or addressed pursuant to Laws and Regulations regulating, relating to, or imposing liability or standards of conduct concerning, any hazardous, toxic, or dangerous waste, substance, or material.
- 12. Contract—The entire and integrated written contract between Owner and Contractor concerning the Work.
- 13. Contract Documents—Those items so designated in the Agreement, and which together comprise the Contract.
- 14. Contract Price—The money that Owner has agreed to pay Contractor for completion of the Work in accordance with the Contract Documents.
- 15. Contract Times—The number of days or the dates by which Contractor shall: (a) achieve Milestones, if any; (b) achieve Substantial Completion; and (c) complete the Work.
- 16. *Contractor*—The individual or entity with which Owner has contracted for performance of the Work.
- 17. Cost of the Work—See Paragraph 13.01 for definition.
- 18. Drawings—The part of the Contract that graphically shows the scope, extent, and character of the Work to be performed by Contractor.
- 19. Effective Date of the Contract—The date, indicated in the Agreement, on which the Contract becomes effective.
- 20. *Electronic Document*—Any Project-related correspondence, attachments to correspondence, data, documents, drawings, information, or graphics, including but not limited to Shop Drawings and other Submittals, that are in an electronic or digital format.
- 21. Electronic Means—Electronic mail (email), upload/download from a secure Project website, or other communications methods that allow: (a) the transmission or communication of Electronic Documents; (b) the documentation of transmissions, including sending and receipt; (c) printing of the transmitted Electronic Document by the

recipient; (d) the storage and archiving of the Electronic Document by sender and recipient; and (e) the use by recipient of the Electronic Document for purposes permitted by this Contract. Electronic Means does not include the use of text messaging, or of Facebook, Twitter, Instagram, or similar social media services for transmission of Electronic Documents.

- 22. Engineer-The individual or entity named as such in the Agreement.
- 23. Field Order—A written order issued by Engineer which requires minor changes in the Work but does not change the Contract Price or the Contract Times.
- 24. *Hazardous Environmental Condition*—The presence at the Site of Constituents of Concern in such quantities or circumstances that may present a danger to persons or property exposed thereto.
 - a. The presence at the Site of materials that are necessary for the execution of the Work, or that are to be incorporated into the Work, and that are controlled and contained pursuant to industry practices, Laws and Regulations, and the requirements of the Contract, is not a Hazardous Environmental Condition.
 - b. The presence of Constituents of Concern that are to be removed or remediated as part of the Work is not a Hazardous Environmental Condition.
 - c. The presence of Constituents of Concern as part of the routine, anticipated, and obvious working conditions at the Site, is not a Hazardous Environmental Condition.
- 25. Laws and Regulations; Laws or Regulations—Any and all applicable laws, statutes, rules, regulations, ordinances, codes, and binding decrees, resolutions, and orders of any and all governmental bodies, agencies, authorities, and courts having jurisdiction.
- Liens—Charges, security interests, or encumbrances upon Contract-related funds, real property, or personal property.
- 27. Milestone—A principal event in the performance of the Work that the Contract requires Contractor to achieve by an intermediate completion date, or by a time prior to Substantial Completion of all the Work.
- Notice of Award—The written notice by Owner to a Bidder of Owner's acceptance of the Bid.
- Notice to Proceed—A written notice by Owner to Contractor fixing the date on which the Contract Times will commence to run and on which Contractor shall start to perform the Work.
- 30. Owner—The individual or entity with which Contractor has contracted regarding the Work, and which has agreed to pay Contractor for the performance of the Work, pursuant to the terms of the Contract.
- 31. Progress Schedule—A schedule, prepared and maintained by Contractor, describing the sequence and duration of the activities comprising Contractor's plan to accomplish the Work within the Contract Times.
- 32. *Project*—The total undertaking to be accomplished for Owner by engineers, contractors, and others, including planning, study, design, construction, testing, commissioning, and start-up, and of which the Work to be performed under the Contract Documents is a part.

- 33. Resident Project Representative—The authorized representative of Engineer assigned to assist Engineer at the Site. As used herein, the term Resident Project Representative (RPR) includes any assistants or field staff of Resident Project Representative.
- 34. Samples—Physical examples of materials, equipment, or workmanship that are representative of some portion of the Work and that establish the standards by which such portion of the Work will be judged.
- 35. Schedule of Submittals—A schedule, prepared and maintained by Contractor, of required submittals and the time requirements for Engineer's review of the submittals.
- 36. Schedule of Values—A schedule, prepared and maintained by Contractor, allocating portions of the Contract Price to various portions of the Work and used as the basis for reviewing Contractor's Applications for Payment.
- 37. Shop Drawings—All drawings, diagrams, illustrations, schedules, and other data or information that are specifically prepared or assembled by or for Contractor and submitted by Contractor to illustrate some portion of the Work. Shop Drawings, whether approved or not, are not Drawings and are not Contract Documents.
- 38. Site—Lands or areas indicated in the Contract Documents as being furnished by Owner upon which the Work is to be performed, including rights-of-way and easements, and such other lands or areas furnished by Owner which are designated for the use of Contractor.
- 39. Specifications—The part of the Contract that consists of written requirements for materials, equipment, systems, standards, and workmanship as applied to the Work, and certain administrative requirements and procedural matters applicable to the Work.
- 40. Subcontractor—An individual or entity having a direct contract with Contractor or with any other Subcontractor for the performance of a part of the Work.
- 41. Submittal—A written or graphic document, prepared by or for Contractor, which the Contract Documents require Contractor to submit to Engineer, or that is indicated as a Submittal in the Schedule of Submittals accepted by Engineer. Submittals may include Shop Drawings and Samples; schedules; product data; Owner-delegated designs; sustainable design information; information on special procedures; testing plans; results of tests and evaluations, source quality-control testing and inspections, and field or Site quality-control testing and inspections; warranties and certifications; Suppliers' instructions and reports; records of delivery of spare parts and tools; operations and maintenance data; Project photographic documentation; record documents; and other such documents required by the Contract Documents. Submittals, whether or not approved or accepted by Engineer, are not Contract Documents. Change Proposals, Change Orders, Claims, notices, Applications for Payment, and requests for interpretation or clarification are not Submittals.
- 42. Substantial Completion—The time at which the Work (or a specified part thereof) has progressed to the point where, in the opinion of Engineer, the Work (or a specified part thereof) is sufficiently complete, in accordance with the Contract Documents, so that the Work (or a specified part thereof) can be utilized for the purposes for which it is intended. The terms "substantially complete" and "substantially completed" as applied to all or part of the Work refer to Substantial Completion of such Work.

- 43. Successful Bidder-The Bidder to which the Owner makes an award of contract.
- 44. Supplementary Conditions—The part of the Contract that amends or supplements these General Conditions.
- 45. Supplier—A manufacturer, fabricator, supplier, distributor, or vendor having a direct contract with Contractor or with any Subcontractor to furnish materials or equipment to be incorporated in the Work by Contractor or a Subcontractor.
- 46. Technical Data
 - a. Those items expressly identified as Technical Data in the Supplementary Conditions, with respect to either (1) existing subsurface conditions at or adjacent to the Site, or existing physical conditions at or adjacent to the Site including existing surface or subsurface structures (except Underground Facilities) or (2) Hazardous Environmental Conditions at the Site.
 - b. If no such express identifications of Technical Data have been made with respect to conditions at the Site, then Technical Data is defined, with respect to conditions at the Site under Paragraphs 5.03, 5.04, and 5.06, as the data contained in boring logs, recorded measurements of subsurface water levels, assessments of the condition of subsurface facilities, laboratory test results, and other factual, objective information regarding conditions at the Site that are set forth in any geotechnical, environmental, or other Site or facilities conditions report prepared for the Project and made available to Contractor.
 - c. Information and data regarding the presence or location of Underground Facilities are not intended to be categorized, identified, or defined as Technical Data, and instead Underground Facilities are shown or indicated on the Drawings.
- 47. Underground Facilities—All active or not-in-service underground lines, pipelines, conduits, ducts, encasements, cables, wires, manholes, vaults, tanks, tunnels, or other such facilities or systems at the Site, including but not limited to those facilities or systems that produce, transmit, distribute, or convey telephone or other communications, cable television, fiber optic transmissions, power, electricity, light, heat, gases, oil, crude oil products, liquid petroleum products, water, steam, waste, wastewater, storm water, other liquids or chemicals, or traffic or other control systems. An abandoned facility or system is not an Underground Facility.
- 48. Unit Price Work—Work to be paid for on the basis of unit prices.
- 49. Work—The entire construction or the various separately identifiable parts thereof required to be provided under the Contract Documents. Work includes and is the result of performing or providing all labor, services, and documentation necessary to produce such construction; furnishing, installing, and incorporating all materials and equipment into such construction; and may include related services such as testing, start-up, and commissioning, all as required by the Contract Documents.
- 50. Work Change Directive—A written directive to Contractor issued on or after the Effective Date of the Contract, signed by Owner and recommended by Engineer, ordering an addition, deletion, or revision in the Work.
- 51. Engineer of Record The individual or entity named as such, or the Design Engineer, in the Agreement and has the authority set forth in the Standard General Conditions.

EJCDC® C-700, Standard General Conditions of the Construction Contract. Copyright[©] 2018 National Society of Professional Engineers, American Council of Engineering Companies, and American Society of Civil Engineers. All rights reserved.

Page 5 of 73

52. Surety – The person, firm, or corporation which is bound by the contract bonds with and for Contractor (Principal); and which is held and firmly bound unto Owner for the conditions of obligations set forth in said bonds.

1.02 Terminology

- A. The words and terms discussed in Paragraphs 1.02.B, C, D, and E are not defined terms that require initial capital letters, but, when used in the Bidding Requirements or Contract Documents, have the indicated meaning.
- B. Intent of Certain Terms or Adjectives: The Contract Documents include the terms "as allowed," "as approved," "as ordered," "as directed" or terms of like effect or import to authorize an exercise of professional judgment by Engineer. In addition, the adjectives "reasonable," "suitable," "acceptable," "proper," "satisfactory," or adjectives of like effect or import are used to describe an action or determination of Engineer as to the Work. It is intended that such exercise of professional judgment, action, or determination will be solely to evaluate, in general, the Work for compliance with the information in the Contract Documents and with the design concept of the Project as a functioning whole as shown or indicated in the Contract Documents (unless there is a specific statement indicating otherwise). The use of any such term or adjective is not intended to and shall not be effective to assign to Engineer any duty or authority to supervise or direct the performance of the Work, or any duty or authority to undertake responsibility contrary to the provisions of Article 10 or any other provision of the Contract Documents.
- C. Day: The word "day" means a calendar day of 24 hours measured from midnight to the next midnight.
- D. Defective: The word "defective," when modifying the word "Work," refers to Work that is unsatisfactory, faulty, or deficient in that it:
 - 1. does not conform to the Contract Documents;
 - does not meet the requirements of any applicable inspection, reference standard, test, or approval referred to in the Contract Documents; or
 - 3. has been damaged prior to Engineer's recommendation of final payment (unless responsibility for the protection thereof has been assumed by Owner at Substantial Completion in accordance with Paragraph 15.03 or Paragraph 15.04).
- E. Furnish, Install, Perform, Provide
 - 1. The word "furnish," when used in connection with services, materials, or equipment, means to supply and deliver said services, materials, or equipment to the Site (or some other specified location) ready for use or installation and in usable or operable condition.
 - 2. The word "install," when used in connection with services, materials, or equipment, means to put into use or place in final position said services, materials, or equipment complete and ready for intended use.
 - The words "perform" or "provide," when used in connection with services, materials, or equipment, means to furnish and install said services, materials, or equipment complete and ready for intended use.
 - 4. If the Contract Documents establish an obligation of Contractor with respect to specific services, materials, or equipment, but do not expressly use any of the four words

Copyright® 2018 National Society of Professional Engineers, American Council of Engineering Companies,

and American Society of Civil Engineers. All rights reserved.

EJCDC® C-700, Standard General Conditions of the Construction Contract.

"furnish," "install," "perform," or "provide," then Contractor shall furnish and install said services, materials, or equipment complete and ready for intended use.

- F. Contract Price or Contract Times: References to a change in "Contract Price or Contract Times" or "Contract Times or Contract Price" or similar, indicate that such change applies to (1) Contract Price, (2) Contract Times, or (3) both Contract Price and Contract Times, as warranted, even if the term "or both" is not expressed.
- G. Unless stated otherwise in the Contract Documents, words or phrases that have a well-known technical or construction industry or trade meaning are used in the Contract Documents in accordance with such recognized meaning.

ARTICLE 2—PRELIMINARY MATTERS

- 2.01 Delivery of Performance and Payment Bonds; Evidence of Insurance
 - A. Performance and Payment Bonds: When Contractor delivers the signed counterparts of the Agreement to Owner, Contractor shall also deliver to Owner the performance bond and payment bond (if the Contract requires Contractor to furnish such bonds).
 - B. Evidence of Contractor's Insurance: When Contractor delivers the signed counterparts of the Agreement to Owner, Contractor shall also deliver to Owner, with copies to each additional insured (as identified in the Contract), the certificates, endorsements, and other evidence of insurance required to be provided by Contractor in accordance with Article 6, except to the extent the Supplementary Conditions expressly establish other dates for delivery of specific insurance policies.
 - C. Evidence of Owner's Insurance: After receipt of the signed counterparts of the Agreement and all required bonds and insurance documentation, Owner shall promptly deliver to Contractor, with copies to each additional insured (as identified in the Contract), the certificates and other evidence of insurance required to be provided by Owner under Article 6.
- 2.02 Copies of Documents
 - A. Owner shall furnish to Contractor four printed copies of the Contract (including one fully signed counterpart of the Agreement), and one copy in electronic portable document format (PDF). Additional printed copies will be furnished upon request at the cost of reproduction.
 - B. Owner shall maintain and safeguard at least one original printed record version of the Contract, including Drawings and Specifications signed and sealed by Engineer and other design professionals. Owner shall make such original printed record version of the Contract available to Contractor for review. Owner may delegate the responsibilities under this provision to Engineer.

2.03 Before Starting Construction

- A. Preliminary Schedules: Within 10 days after the Effective Date of the Contract (or as otherwise required by the Contract Documents), Contractor shall submit to Engineer for timely review:
 - a preliminary Progress Schedule indicating the times (numbers of days or dates) for starting and completing the various stages of the Work, including any Milestones specified in the Contract;
 - 2. a preliminary Schedule of Submittals; and

- 3. a preliminary Schedule of Values for all of the Work which includes quantities and prices of items which when added together equal the Contract Price and subdivides the Work into component parts in sufficient detail to serve as the basis for progress payments during performance of the Work. Such prices will include an appropriate amount of overhead and profit applicable to each item of Work.
- B. Before any Work at the Site is started, Contractor shall deliver to the Owner, certificates of insurance that Contractor is required to purchase and maintain in accordance with Article 5.
- 2.04 Preconstruction Conference; Designation of Authorized Representatives
 - A. Before any Work at the Site is started, a conference attended by Owner, Contractor, Engineer, and others as appropriate will be held to establish a working understanding among the parties as to the Work, and to discuss the schedules referred to in Paragraph 2.03.A, procedures for handling Shop Drawings, Samples, and other Submittals, processing Applications for Payment, electronic or digital transmittals, and maintaining required records.
 - B. At this conference Owner and Contractor each shall designate, in writing, a specific individual to act as its authorized representative with respect to the services and responsibilities under the Contract. Such individuals shall have the authority to transmit and receive information, render decisions relative to the Contract, and otherwise act on behalf of each respective party.
 - C. Owner may issue Notice to Proceed at the Preconstruction Conference. Contractor shall begin the Work within twenty-four (24) hours of the date given in the Notice to Proceed. If the Contractor does not start the Work within fourteen (14) calendar days after this date, Owner may, at his discretion, terminate Contractor in accordance with Article 16.

2.05 Acceptance of Schedules

- A. <u>At least 10 days before submission of the first Application for Payment a conference A</u> <u>Preconstruction Conference</u>, attended by Contractor, Engineer, and others as appropriate, will be held to review the schedules submitted in accordance with Paragraph 2.03.A. No progress payment will be made to Contractor until acceptable schedules are submitted to Engineer.
 - The Progress Schedule will be acceptable to Engineer if it provides an orderly progression of the Work to completion within the Contract Times. Such acceptance will not impose on Engineer responsibility for the Progress Schedule, for sequencing, scheduling, or progress of the Work, nor interfere with or relieve Contractor from Contractor's full responsibility therefor.
 - 2. Contractor's Schedule of Submittals will be acceptable to Engineer if it provides a workable arrangement for reviewing and processing the required submittals.
 - Contractor's Schedule of Values will be acceptable to Engineer as to form and substance if it provides a reasonable allocation of the Contract Price to the component parts of the Work.
 - If a schedule is not acceptable, Contractor will have an additional 10 days to revise and resubmit the schedule.
 - A schedule in accordance with the Contract shall be submitted and accepted by the Owner and engineer within thirty (30) days of NTP. Should a schedule not be accepted

EJCDC® C-700, Standard General Conditions of the Construction Contract. Copyright® 2018 National Society of Professional Engineers, American Council of Engineering Companies, and American Society of Civil Engineers. All rights reserved.

within thirty (30) days of the NTP, it will be considered a breach of contract and reason for termination of the contract for cause in accordance with the Standard General Conditions.

2.06 Electronic Transmittals

- A. Except as otherwise stated elsewhere in the Contract, the Owner, Engineer, and Contractor may send, and shall accept, Electronic Documents transmitted by Electronic Means.
- B. If the Contract does not establish protocols for Electronic Means, then Owner, Engineer, and Contractor shall jointly develop such protocols.
- C. Subject to any governing protocols for Electronic Means, when transmitting Electronic Documents by Electronic Means, the transmitting party makes no representations as to longterm compatibility, usability, or readability of the Electronic Documents resulting from the recipient's use of software application packages, operating systems, or computer hardware differing from those used in the drafting or transmittal of the Electronic Documents.

ARTICLE 3-CONTRACT DOCUMENTS: INTENT, REQUIREMENTS, REUSE

- 3.01 Intent
 - A. The Contract Documents are complementary; what is required by one Contract Document is as binding as if required by all.
 - B. It is the intent of the Contract Documents to describe a functionally complete Project (or part thereof) to be constructed in accordance with the Contract Documents.
 - C. Unless otherwise stated in the Contract Documents, if there is a discrepancy between the electronic versions of the Contract Documents (including any printed copies derived from such electronic versions) and the printed record version, the printed record version will govern.
 - D. The Contract supersedes prior negotiations, representations, and agreements, whether written or oral.
 - E. Engineer will issue clarifications and interpretations of the Contract Documents as provided herein.
 - F. Any provision or part of the Contract Documents held to be void or unenforceable under any Law or Regulation will be deemed stricken, and all remaining provisions will continue to be valid and binding upon Owner and Contractor, which agree that the Contract Documents will be reformed to replace such stricken provision or part thereof with a valid and enforceable provision that comes as close as possible to expressing the intention of the stricken provision.
 - G. Nothing in the Contract Documents creates:
 - 1. any contractual relationship between Owner or Engineer and any Subcontractor, Supplier, or other individual or entity performing or furnishing any of the Work, for the benefit of such Subcontractor, Supplier, or other individual or entity; or
 - any obligation on the part of Owner or Engineer to pay or to see to the payment of any money due any such Subcontractor, Supplier, or other individual or entity, except as may otherwise be required by Laws and Regulations.

3.02 Reference Standards

A. Standards Specifications, Codes, Laws and Regulations

- Reference in the Contract Documents to standard specifications, manuals, reference standards, or codes of any technical society, organization, or association, or to Laws or Regulations, whether such reference be specific or by implication, means the standard specification, manual, reference standard, code, or Laws or Regulations in effect at the time of opening of Bids (or on the Effective Date of the Contract if there were no Bids), except as may be otherwise specifically stated in the Contract Documents.
- 2. No provision of any such standard specification, manual, reference standard, or code, and no instruction of a Supplier, will be effective to change the duties or responsibilities of Owner, Contractor, or Engineer from those set forth in the part of the Contract Documents prepared by or for Engineer. No such provision or instruction shall be effective to assign to Owner or Engineer any duty or authority to supervise or direct the performance of the Work, or any duty or authority to undertake responsibility inconsistent with the provisions of the part of the Contract Documents prepared by or for Engineer.

3.03 Reporting and Resolving Discrepancies

- A. Reporting Discrepancies
 - Contractor's Verification of Figures and Field Measurements: Before undertaking each part of the Work, Contractor shall carefully study the Contract Documents, and check and verify pertinent figures and dimensions therein, particularly with respect to applicable field measurements. Contractor shall promptly report in writing to Engineer any conflict, error, ambiguity, or discrepancy that Contractor discovers, or has actual knowledge of, and shall not proceed with any Work affected thereby until the conflict, error, ambiguity, or discrepancy is resolved by a clarification or interpretation by Engineer, or by an amendment or supplement to the Contract issued pursuant to Paragraph 11.01.
 - 2. Contractor's Review of Contract Documents: If, before or during the performance of the Work, Contractor discovers any conflict, error, ambiguity, or discrepancy within the Contract Documents, or between the Contract Documents and (a) any applicable Law or Regulation, (b) actual field conditions, (c) any standard specification, manual, reference standard, or code, or (d) any instruction of any Supplier, then Contractor shall promptly report it to Engineer in writing. Contractor shall not proceed with the Work affected thereby (except in an emergency as required by Paragraph 7.15) until the conflict, error, ambiguity, or discrepancy is resolved, by a clarification or interpretation by Engineer, or by an amendment or supplement to the Contract issued pursuant to Paragraph 11.01.
 - Contractor shall not be liable to Owner or Engineer for failure to report any conflict, error, ambiguity, or discrepancy in the Contract Documents unless Contractor had actual knowledge thereof.
- B. Resolving Discrepancies
 - Except as may be otherwise specifically stated in the Contract Documents, the provisions
 of the part of the Contract Documents prepared by or for Engineer take precedence in

resolving any conflict, error, ambiguity, or discrepancy between such provisions of the Contract Documents and:

- the provisions of any standard specification, manual, reference standard, or code, or the instruction of any Supplier (whether or not specifically incorporated by reference as a Contract Document); or
- b. the provisions of any Laws or Regulations applicable to the performance of the Work (unless such an interpretation of the provisions of the Contract Documents would result in violation of such Law or Regulation).

3.04 Requirements of the Contract Documents

- A. During the performance of the Work and until final payment, Contractor and Owner shall submit to the Engineer in writing all matters in question concerning the requirements of the Contract Documents (sometimes referred to as requests for information or interpretation— RFIs), or relating to the acceptability of the Work under the Contract Documents, as soon as possible after such matters arise. Engineer will be the initial interpreter of the requirements of the Contract Documents, and judge of the acceptability of the Work.
- B. Engineer will, with reasonable promptness, render a written clarification, interpretation, or decision on the issue submitted, or initiate an amendment or supplement to the Contract Documents. Engineer's written clarification, interpretation, or decision will be final and binding on Contractor, unless it appeals by submitting a Change Proposal, and on Owner, unless it appeals by filing a Claim.
- C. If a submitted matter in question concerns terms and conditions of the Contract Documents that do not involve (1) the performance or acceptability of the Work under the Contract Documents, (2) the design (as set forth in the Drawings, Specifications, or otherwise), or (3) other engineering or technical matters, then Engineer will promptly notify Owner and Contractor in writing that Engineer is unable to provide a decision or interpretation. If Owner and Contractor are unable to agree on resolution of such a matter in question, either party may pursue resolution as provided in Article 12.

3.05 Reuse of Documents

- A. Contractor and its Subcontractors and Suppliers shall not:
 - have or acquire any title to or ownership rights in any of the Drawings, Specifications, or other documents (or copies of any thereof) prepared by or bearing the seal of Engineer or its consultants, including electronic media versions, or reuse any such Drawings, Specifications, other documents, or copies thereof on extensions of the Project or any other project without written consent of Owner and Engineer and specific written verification or adaptation by Engineer; or
 - have or acquire any title or ownership rights in any other Contract Documents, reuse any such Contract Documents for any purpose without Owner's express written consent, or violate any copyrights pertaining to such Contract Documents.
- B. The prohibitions of this Paragraph 3.05 will survive final payment, or termination of the Contract. Nothing herein precludes Contractor from retaining copies of the Contract Documents for record purposes.

ARTICLE 4-COMMENCEMENT AND PROGRESS OF THE WORK

4.01 Commencement of Contract Times; Notice to Proceed

- A. The Contract Times will commence to run on the 30th day after the Effective Date of the Contract or, if a Notice to Proceed is given, on the day indicated in the Notice to Proceed. A Notice to Proceed may be given at any time within 30 days after the Effective Date of the Contract. In no event will the Contract Times commence to run later than the 60th day after the day of Bid opening or the 30th day after the Effective Date of the Contract, whichever date is earlier.
- 4.02 Starting the Work
 - A. Contractor shall start to perform the Work on the date when the Contract Times commence to run. No Work may be done at the Site prior to such date.

4.03 Reference Points

A. Owner shall provide engineering surveys to establish reference points for construction which in Engineer's judgment are necessary to enable Contractor to proceed with the Work. Contractor shall be responsible for laying out the Work, shall protect and preserve the established reference points and property monuments, and shall make no changes or relocations without the prior written approval of Owner. Contractor shall report to Engineer whenever any reference point or property monument is lost or destroyed or requires relocation because of necessary changes in grades or locations, and shall be responsible for the accurate replacement or relocation of such reference points or property monuments by professionally qualified personnel.

4.04 Progress Schedule

- A. Contractor shall adhere to the Progress Schedule established in accordance with Paragraph 2.05 as it may be adjusted from time to time as provided below.
 - 1. Contractor shall submit to Engineer for acceptance (to the extent indicated in Paragraph 2.05) proposed adjustments in the Progress Schedule that will not result in changing the Contract Times.
 - 2. Proposed adjustments in the Progress Schedule that will change the Contract Times must be submitted in accordance with the requirements of Article 11.
- B. Contractor shall carry on the Work and adhere to the Progress Schedule during all disputes or disagreements with Owner. No Work will be delayed or postponed pending resolution of any disputes or disagreements, or during any appeal process, except as permitted by Paragraph 16.04, or as Owner and Contractor may otherwise agree in writing.

4.05 Delays in Contractor's Progress

- A. If Owner, Engineer, or anyone for whom Owner is responsible, delays, disrupts, or interferes with the performance or progress of the Work, then Contractor shall be entitled to an equitable adjustment in Contract Price or Contract Times.
- B. Contractor shall not be entitled to an adjustment in Contract Price or Contract Times for delay, disruption, or interference caused by or within the control of Contractor. Delay, disruption, and interference attributable to and within the control of a Subcontractor or Supplier shall be deemed to be within the control of Contractor.

- C. If Contractor's performance or progress is delayed, disrupted, or interfered with by unanticipated causes not the fault of and beyond the control of Owner, Contractor, and those for which they are responsible, then Contractor shall be entitled to an equitable adjustment in Contract Times. Such an adjustment will be Contractor's sole and exclusive remedy for the delays, disruption, and interference described in this paragraph. Causes of delay, disruption, or interference that may give rise to an adjustment in Contract Times under this paragraph include but are not limited to the following:
 - Severe and unavoidable natural catastrophes such as fires, floods, epidemics, and earthquakes;
 - 2. Abnormal or adverse weather conditions. On the 21st of each month, or the next following business day, the Contractor shall submit to the Owner and Engineer a request for weather days. Failure to submit the request forfeits the right to claim weather days for that period and any previous period. The Owner will grant time extensions, on a day for day basis, for delays caused by the effects of rains or other inclement weather conditions or related adverse soil conditions that prevent the Contractor from productively performing controlling items of work resulting in:
 - a. The Contractor being unable to work at lease 50% of the normal work day on predetermined controlling work items due to adverse weather conditions.

No additional compensation will be made for delays caused by the effects of inclement weather.

- Acts or failures to act of third-party utility owners or other third-party entities (other than those third-party utility owners or other third-party entities performing other work at or adjacent to the Site as arranged by or under contract with Owner, as contemplated in Article 8); and
- 4. Acts of war or terrorism.
- D. Contractor's entitlement to an adjustment of Contract Times or Contract Price is limited as follows:
 - Contractor's entitlement to an adjustment of the Contract Times is conditioned on the delay, disruption, or interference adversely affecting an activity on the critical path to completion of the Work, as of the time of the delay, disruption, or interference.
 - 2. Contractor shall not be entitled to an adjustment in Contract Price for any delay, disruption, or interference if such delay is concurrent with a delay, disruption, or interference caused by or within the control of Contractor. Such a concurrent delay by Contractor shall not preclude an adjustment of Contract Times to which Contractor is otherwise entitled.
 - 3. Adjustments of Contract Times or Contract Price are subject to the provisions of Article 11.
- E. Each Contractor request or Change Proposal seeking an increase in Contract Times or Contract Price must be supplemented by supporting data that sets forth in detail the following:
 - 1. The circumstances that form the basis for the requested adjustment;
 - The date upon which each cause of delaγ, disruption, or interference began to affect the progress of the Work;

Copyright[©] 2018 National Society of Professional Engineers, American Council of Engineering Companies, and American Society of Civil Engineers. All rights reserved.

EJCDC® C-700, Standard General Conditions of the Construction Contract.

- The date upon which each cause of delay, disruption, or interference ceased to affect the progress of the Work;
- 4. The number of days' increase in Contract Times claimed as a consequence of each such cause of delay, disruption, or interference; and
- 5. The impact on Contract Price, in accordance with the provisions of Paragraph 11.07.

Contractor shall also furnish such additional supporting documentation as Owner or Engineer may require including, where appropriate, a revised progress schedule indicating all the activities affected by the delay, disruption, or interference, and an explanation of the effect of the delay, disruption, or interference on the critical path to completion of the Work.

- F. Delays, disruption, and interference to the performance or progress of the Work resulting from the existence of a differing subsurface or physical condition, an Underground Facility that was not shown or indicated by the Contract Documents, or not shown or indicated with reasonable accuracy, and those resulting from Hazardous Environmental Conditions, are governed by Article 5, together with the provisions of Paragraphs 4.05.D and 4.05.E.
- G. Paragraph 8.03 addresses delays, disruption, and interference to the performance or progress of the Work resulting from the performance of certain other work at or adjacent to the Site.

ARTICLE 5—SITE; SUBSURFACE AND PHYSICAL CONDITIONS; HAZARDOUS ENVIRONMENTAL CONDITIONS

- 5.01 Availability of Lands
 - A. Owner shall furnish the Site. Owner shall notify Contractor in writing of any encumbrances or restrictions not of general application but specifically related to use of the Site with which Contractor must comply in performing the Work.
 - B. Upon reasonable written request, Owner shall furnish Contractor with a current statement of record legal title and legal description of the lands upon which permanent improvements are to be made and Owner's interest therein as necessary for giving notice of or filing a mechanic's or construction lien against such lands in accordance with applicable Laws and Regulations.
 - C. Contractor shall provide for all additional lands and access thereto that may be required for temporary construction facilities or storage of materials and equipment.
 - D. Contractor shall provide for all additional lands and access thereto that may be required for temporary construction facilities or storage of materials and equipment. Contractor shall obtain said land rights at his own expense and without liability to the Owner. Contractor shall not enter upon private property without first obtaining written permission from the rightful property owner.
- 5.02 Use of Site and Other Areas
 - A. Limitation on Use of Site and Other Areas
 - Contractor shall confine construction equipment, temporary construction facilities, the storage of materials and equipment, and the operations of workers to the Site, adjacent areas that Contractor has arranged to use through construction easements or otherwise, and other adjacent areas permitted by Laws and Regulations, and shall not unreasonably encumber the Site and such other adjacent areas with construction equipment or other
materials or equipment. Contractor shall assume full responsibility for (a) damage to the Site; (b) damage to any such other adjacent areas used for Contractor's operations; (c) damage to any other adjacent land or areas, or to improvements, structures, utilities, or similar facilities located at such adjacent lands or areas; and (d) for injuries and losses sustained by the owners or occupants of any such land or areas; provided that such damage or injuries result from the performance of the Work or from other actions or conduct of the Contractor or those for which Contractor is responsible.

- If a damage or injury claim is made by the owner or occupant of any such land or area 2. because of the performance of the Work, or because of other actions or conduct of the Contractor or those for which Contractor is responsible, Contractor shall (a) take immediate corrective or remedial action as required by Paragraph 7.13, or otherwise; (b) promptly attempt to settle the claim as to all parties through negotiations with such owner or occupant, or otherwise resolve the claim by arbitration or other dispute resolution proceeding, or in a court of competent jurisdiction; and (c) to the fullest extent permitted by Laws and Regulations, indemnify and hold harmless Owner and Engineer, and the officers, directors, members, partners, employees, agents, consultants and subcontractors of each and any of them, from and against any such claim, and against all costs, losses, and damages (including but not limited to all fees and charges of engineers, architects, attorneys, and other professionals and all court or arbitration or other dispute resolution costs) arising out of or relating to any claim or action, legal or equitable, brought by any such owner or occupant against Owner, Engineer, or any other party indemnified hereunder to the extent caused directly or indirectly, in whole or in part by, or based upon, Contractor's performance of the Work, or because of other actions or conduct of the Contractor or those for which Contractor is responsible.
- B. Removal of Debris During Performance of the Work: During the progress of the Work the Contractor shall keep the Site and other adjacent areas free from accumulations of waste materials, rubbish, and other debris. Removal and disposal of such waste materials, rubbish, and other debris will conform to applicable Laws and Regulations.
- C. Cleaning: Prior to Substantial Completion of the Work Contractor shall clean the Site and the Work and make it ready for utilization by Owner. At the completion of the Work Contractor shall remove from the Site and adjacent areas all tools, appliances, construction equipment and machinery, and surplus materials and shall restore to original condition all property not designated for alteration by the Contract Documents.
- D. Loading of Structures: Contractor shall not load nor permit any part of any structure to be loaded in any manner that will endanger the structure, nor shall Contractor subject any part of the Work or adjacent structures or land to stresses or pressures that will endanger them.
- 5.03 Subsurface and Physical Conditions
 - A. Reports and Drawings: The Supplementary Conditions identify:
 - Those reports of explorations and tests of subsurface conditions at or adjacent to the Site that contain Technical Data;
 - Those drawings of existing physical conditions at or adjacent to the Site, including those drawings depicting existing surface or subsurface structures at or adjacent to the Site (except Underground Facilities), that contain Technical Data; and
 - 3. Technical Data contained in such reports and drawings.

EJCDC® C-700, Standard General Conditions of the Construction Contract. Copyright[©] 2018 National Society of Professional Engineers, American Council of Engineering Companies, and American Society of Civil Engineers. All rights reserved.

- B. Underground Facilities: Underground Facilities are shown or indicated on the Drawings, pursuant to Paragraph 5.05, and not in the drawings referred to in Paragraph 5.03.A. Information and data regarding the presence or location of Underground Facilities are not intended to be categorized, identified, or defined as Technical Data.
- C. Reliance by Contractor on Technical Data: Contractor may rely upon the accuracy of the Technical Data expressly identified in the Supplementary Conditions with respect to such reports and drawings, but such reports and drawings are not Contract Documents. If no such express identification has been made, then Contractor may rely upon the accuracy of the Technical Data as defined in Paragraph 1.01.A.46.b.
- D. Limitations of Other Data and Documents: Except for such reliance on Technical Data, Contractor may not rely upon or make any claim against Owner or Engineer, or any of their officers, directors, members, partners, employees, agents, consultants, or subcontractors, with respect to:
 - the completeness of such reports and drawings for Contractor's purposes, including, but not limited to, any aspects of the means, methods, techniques, sequences, and procedures of construction to be employed by Contractor, and safety precautions and programs incident thereto;
 - other data, interpretations, opinions, and information contained in such reports or shown or indicated in such drawings;
 - the contents of other Site-related documents made available to Contractor, such as record drawings from other projects at or adjacent to the Site, or Owner's archival documents concerning the Site; or
 - 4. any Contractor interpretation of or conclusion drawn from any Technical Data or any such other data, interpretations, opinions, or information.

5.04 Differing Subsurface or Physical Conditions

- A. Notice by Contractor: If Contractor believes that any subsurface or physical condition that is uncovered or revealed at the Site:
 - 1. is of such a nature as to establish that any Technical Data on which Contractor is entitled to rely as provided in Paragraph 5.03 is materially inaccurate;
 - 2. is of such a nature as to require a change in the Drawings or Specifications;
 - 3. differs materially from that shown or indicated in the Contract Documents; or
 - is of an unusual nature, and differs materially from conditions ordinarily encountered and generally recognized as inherent in work of the character provided for in the Contract Documents;

then Contractor shall, promptly after becoming aware thereof and before further disturbing the subsurface or physical conditions or performing any Work in connection therewith (except in an emergency as required by Paragraph 7.15), notify Owner and Engineer in writing about such condition. Contractor shall not further disturb such condition or perform any Work in connection therewith (except with respect to an emergency) until receipt of a written statement permitting Contractor to do so.

- B. Engineer's Review: After receipt of written notice as required by the preceding paragraph, Engineer will promptly review the subsurface or physical condition in question; determine whether it is necessary for Owner to obtain additional exploration or tests with respect to the condition; conclude whether the condition falls within any one or more of the differing site condition categories in Paragraph 5.04.A; obtain any pertinent cost or schedule information from Contractor; prepare recommendations to Owner regarding the Contractor's resumption of Work in connection with the subsurface or physical condition in question and the need for any change in the Drawings or Specifications; and advise Owner in writing of Engineer's findings, conclusions, and recommendations.
- C. Owner's Statement to Contractor Regarding Site Condition: After receipt of Engineer's written findings, conclusions, and recommendations, Owner shall issue a written statement to Contractor (with a copy to Engineer) regarding the subsurface or physical condition in question, addressing the resumption of Work in connection with such condition, indicating whether any change in the Drawings or Specifications will be made, and adopting or rejecting Engineer's written findings, conclusions, and recommendations, in whole or in part.
- D. Early Resumption of Work: If at any time Engineer determines that Work in connection with the subsurface or physical condition in question may resume prior to completion of Engineer's review or Owner's issuance of its statement to Contractor, because the condition in question has been adequately documented, and analyzed on a preliminary basis, then the Engineer may at its discretion instruct Contractor to resume such Work.
- E. Possible Price and Times Adjustments
 - Contractor shall be entitled to an equitable adjustment in Contract Price or Contract Times, to the extent that the existence of a differing subsurface or physical condition, or any related delay, disruption, or interference, causes an increase or decrease in Contractor's cost of, or time required for, performance of the Work; subject, however, to the following:
 - a. Such condition must fall within any one or more of the categories described in Paragraph 5.04.A;
 - b. With respect to Work that is paid for on a unit price basis, any adjustment in Contract Price will be subject to the provisions of Paragraph 13.03; and,
 - c. Contractor's entitlement to an adjustment of the Contract Times is subject to the provisions of Paragraphs 4.05.D and 4.05.E.
 - Contractor shall not be entitled to any adjustment in the Contract Price or Contract Times with respect to a subsurface or physical condition if:
 - a. Contractor knew of the existence of such condition at the time Contractor made a commitment to Owner with respect to Contract Price and Contract Times by the submission of a Bid or becoming bound under a negotiated contract, or otherwise;
 - b. The existence of such condition reasonably could have been discovered or revealed as a result of any examination, investigation, exploration, test, or study of the Site and contiguous areas expressly required by the Bidding Requirements or Contract Documents to be conducted by or for Contractor prior to Contractor's making such commitment; or
 - c. Contractor failed to give the written notice required by Paragraph 5.04.A.

EJCDC[®] C-700, Standard General Conditions of the Construction Contract. Copyright[®] 2018 National Society of Professional Engineers, American Council of Engineering Companies, and American Society of Civil Engineers. All rights reserved.

- 3. If Owner and Contractor agree regarding Contractor's entitlement to and the amount or extent of any adjustment in the Contract Price or Contract Times, then any such adjustment will be set forth in a Change Order.
- 4. Contractor may submit a Change Proposal regarding its entitlement to or the amount or extent of any adjustment in the Contract Price or Contract Times, no later than 30 days after Owner's issuance of the Owner's written statement to Contractor regarding the subsurface or physical condition in question.
- F. Underground Facilities; Hazardous Environmental Conditions: Paragraph 5.05 governs rights and responsibilities regarding the presence or location of Underground Facilities. Paragraph 5.06 governs rights and responsibilities regarding Hazardous Environmental Conditions. The provisions of Paragraphs 5.03 and 5.04 are not applicable to the presence or location of Underground Facilities, or to Hazardous Environmental Conditions.

5.05 Underground Facilities

- A. Contractor's Responsibilities: Unless it is otherwise expressly provided in the Supplementary Conditions, the cost of all of the following are included in the Contract Price, and Contractor shall have full responsibility for:
 - 1. reviewing and checking all information and data regarding existing Underground Facilities at the Site;
 - complying with applicable state and local utility damage prevention Laws and Regulations;
 - verifying the actual location of those Underground Facilities shown or indicated in the Contract Documents as being within the area affected by the Work, by exposing such Underground Facilities during the course of construction;
 - 4. coordination of the Work with the owners (including Owner) of such Underground Facilities, during construction; and
 - 5. the safety and protection of all existing Underground Facilities at the Site, and repairing any damage thereto resulting from the Work.
- B. Notice by Contractor: If Contractor believes that an Underground Facility that is uncovered or revealed at the Site was not shown or indicated on the Drawings, or was not shown or indicated on the Drawings with reasonable accuracy, then Contractor shall, promptly after becoming aware thereof and before further disturbing conditions affected thereby or performing any Work in connection therewith (except in an emergency as required by Paragraph 7.15), notify Owner and Engineer in writing regarding such Underground Facility.
- C. Engineer's Review: Engineer will:
 - promptly review the Underground Facility and conclude whether such Underground Facility was not shown or indicated on the Drawings, or was not shown or indicated with reasonable accuracy;
 - identify and communicate with the owner of the Underground Facility; prepare recommendations to Owner (and if necessary issue any preliminary instructions to Contractor) regarding the Contractor's resumption of Work in connection with the Underground Facility in question;

- 3. obtain any pertinent cost or schedule information from Contractor; determine the extent, if any, to which a change is required in the Drawings or Specifications to reflect and document the consequences of the existence or location of the Underground Facility; and
- 4. advise Owner in writing of Engineer's findings, conclusions, and recommendations.

During such time, Contractor shall be responsible for the safety and protection of such Underground Facility.

- D. Owner's Statement to Contractor Regarding Underground Facility: After receipt of Engineer's written findings, conclusions, and recommendations, Owner shall issue a written statement to Contractor (with a copy to Engineer) regarding the Underground Facility in question addressing the resumption of Work in connection with such Underground Facility, indicating whether any change in the Drawings or Specifications will be made, and adopting or rejecting Engineer's written findings, conclusions, and recommendations in whole or in part.
- E. Early Resumption of Work: If at any time Engineer determines that Work in connection with the Underground Facility may resume prior to completion of Engineer's review or Owner's issuance of its statement to Contractor, because the Underground Facility in question and conditions affected by its presence have been adequately documented, and analyzed on a preliminary basis, then the Engineer may at its discretion instruct Contractor to resume such Work.
- F. Possible Price and Times Adjustments
 - Contractor shall be entitled to an equitable adjustment in the Contract Price or Contract Times, to the extent that any existing Underground Facility at the Site that was not shown or indicated on the Drawings, or was not shown or indicated with reasonable accuracy, or any related delay, disruption, or interference, causes an increase or decrease in Contractor's cost of, or time required for, performance of the Work; subject, however, to the following:
 - With respect to Work that is paid for on a unit price basis, any adjustment in Contract Price will be subject to the provisions of Paragraph 13.03;
 - b. Contractor's entitlement to an adjustment of the Contract Times is subject to the provisions of Paragraphs 4.05.D and 4.05.E; and
 - c. Contractor gave the notice required in Paragraph 5.05.B.
 - If Owner and Contractor agree regarding Contractor's entitlement to and the amount or extent of any adjustment in the Contract Price or Contract Times, then any such adjustment will be set forth in a Change Order.
 - Contractor may submit a Change Proposal regarding its entitlement to or the amount or extent of any adjustment in the Contract Price or Contract Times, no later than 30 days after Owner's issuance of the Owner's written statement to Contractor regarding the Underground Facility in question.
 - 4. The information and data shown or indicated on the Drawings with respect to existing Underground Facilities at the Site is based on information and data (a) furnished by the owners of such Underground Facilities, or by others, (b) obtained from available records, or (c) gathered in an investigation conducted in accordance with the current edition of ASCE 38, Standard Guideline for the Collection and Depiction of Existing Subsurface Utility

Data, by the American Society of Civil Engineers. If such information or data is incorrect or incomplete, Contractor's remedies are limited to those set forth in this Paragraph 5.05.F.

5.06 Hazardous Environmental Conditions at Site

A. Reports and Drawings: The Supplementary Conditions identify:

- 1. those reports known to Owner relating to Hazardous Environmental Conditions that have been identified at or adjacent to the Site;
- 2. drawings known to Owner relating to Hazardous Environmental Conditions that have been identified at or adjacent to the Site; and
- 3. Technical Data contained in such reports and drawings.
- B. Reliance by Contractor on Technical Data Authorized: Contractor may rely upon the accuracy of the Technical Data expressly identified in the Supplementary Conditions with respect to such reports and drawings, but such reports and drawings are not Contract Documents. If no such express identification has been made, then Contractor may rely on the accuracy of the Technical Data as defined in Paragraph 1.01.A.46.b. Except for such reliance on Technical Data, Contractor may not rely upon or make any claim against Owner or Engineer, or any of their officers, directors, members, partners, employees, agents, consultants, or subcontractors, with respect to:
 - the completeness of such reports and drawings for Contractor's purposes, including, but not limited to, any aspects of the means, methods, techniques, sequences and procedures of construction to be employed by Contractor, and safety precautions and programs incident thereto;
 - other data, interpretations, opinions, and information contained in such reports or shown or indicated in such drawings; or
 - 3. any Contractor interpretation of or conclusion drawn from any Technical Data or any such other data, interpretations, opinions or information.
- C. Contractor shall not be responsible for removing or remediating any Hazardous Environmental Condition encountered, uncovered, or revealed at the Site unless such removal or remediation is expressly identified in the Contract Documents to be within the scope of the Work.
- D. Contractor shall be responsible for controlling, containing, and duly removing all Constituents of Concern brought to the Site by Contractor, Subcontractors, Suppliers, or anyone else for whom Contractor is responsible, and for any associated costs; and for the costs of removing and remediating any Hazardous Environmental Condition created by the presence of any such Constituents of Concern.
- E. If Contractor encounters, uncovers, or reveals a Hazardous Environmental Condition whose removal or remediation is not expressly identified in the Contract Documents as being within the scope of the Work, or if Contractor or anyone for whom Contractor is responsible creates a Hazardous Environmental Condition, then Contractor shall immediately: (1) secure or otherwise isolate such condition; (2) stop all Work in connection with such condition and in any area affected thereby (except in an emergency as required by Paragraph 7.15); and (3) notify Owner and Engineer (and promptly thereafter confirm such notice in writing). Owner

Page 20 of 73

shall promptly consult with Engineer concerning the necessity for Owner to retain a qualified expert to evaluate such condition or take corrective action, if any. Promptly after consulting with Engineer, Owner shall take such actions as are necessary to permit Owner to timely obtain required permits and provide Contractor the written notice required by Paragraph 5.06.F. If Contractor or anyone for whom Contractor is responsible created the Hazardous Environmental Condition in question, then Owner may remove and remediate the Hazardous Environmental Condition, and impose a set-off against payments to account for the associated costs.

- F. Contractor shall not resume Work in connection with such Hazardous Environmental Condition or in any affected area until after Owner has obtained any required permits related thereto, and delivered written notice to Contractor either (1) specifying that such condition and any affected area is or has been rendered safe for the resumption of Work, or (2) specifying any special conditions under which such Work may be resumed safely.
- G. If Owner and Contractor cannot agree as to entitlement to or on the amount or extent, if any, of any adjustment in Contract Price or Contract Times, as a result of such Work stoppage, such special conditions under which Work is agreed to be resumed by Contractor, or any costs or expenses incurred in response to the Hazardous Environmental Condition, then within 30 days of Owner's written notice regarding the resumption of Work, Contractor may submit a Change Proposal, or Owner may impose a set-off. Entitlement to any such adjustment is subject to the provisions of Paragraphs 4.05.D, 4.05.E, 11.07, and 11.08.
- H. If, after receipt of such written notice, Contractor does not agree to resume such Work based on a reasonable belief it is unsafe, or does not agree to resume such Work under such special conditions, then Owner may order the portion of the Work that is in the area affected by such condition to be deleted from the Work, following the contractual change procedures in Article 11. Owner may have such deleted portion of the Work performed by Owner's own forces or others in accordance with Article 8.
- I. To the fullest extent permitted by Laws and Regulations, Owner shall indemnify and hold harmless Contractor, Subcontractors, and Engineer, and the officers, directors, members, partners, employees, agents, consultants, and subcontractors of each and any of them, from and against all claims, costs, losses, and damages (including but not limited to all fees and charges of engineers, architects, attorneys, and other professionals, and all court, arbitration, or other dispute resolution costs) arising out of or relating to a Hazardous Environmental Condition, provided that such Hazardous Environmental Condition (1) was not shown or indicated in the Drawings, Specifications, or other Contract Documents, identified as Technical Data entitled to limited reliance pursuant to Paragraph 5.06.B, or identified in the Contract Documents to be included within the scope of the Work, and (2) was not created by Contractor or by anyone for whom Contractor is responsible. Nothing in this Paragraph 5.06.I obligates Owner to indemnify any individual or entity from and against the consequences of that individual's or entity's own negligence.
- J. To the fullest extent permitted by Laws and Regulations, Contractor shall indemnify and hold harmless Owner and Engineer, and the officers, directors, members, partners, employees, agents, consultants, and subcontractors of each and any of them, from and against all claims, costs, losses, and damages (including but not limited to all fees and charges of engineers, architects, attorneys, and other professionals and all court or arbitration or other dispute resolution costs) arising out of or relating to the failure to control, contain, or remove a

Page 21 of 73

Constituent of Concern brought to the Site by Contractor or by anyone for whom Contractor is responsible, or to a Hazardous Environmental Condition created by Contractor or by anyone for whom Contractor is responsible. Nothing in this Paragraph 5.06.J obligates Contractor to indemnify any individual or entity from and against the consequences of that individual's or entity's own negligence.

K. The provisions of Paragraphs 5.03, 5.04, and 5.05 do not apply to the presence of Constituents of Concern or to a Hazardous Environmental Condition uncovered or revealed at the Site.

ARTICLE 6—BONDS AND INSURANCE

- 6.01 Performance, Payment, and Other Bonds
 - A. Contractor shall furnish a performance bond and a payment bond, each in an amount at least equal to the Contract Price, as security for the faithful performance and payment of Contractor's obligations under the Contract. These bonds must remain in effect until one year after the date when final payment becomes due or until completion of the correction period specified in Paragraph 15.08, whichever is later, except as provided otherwise by Laws or Regulations, the terms of a prescribed bond form, the Supplementary Conditions, or other provisions of the Contract.
 - B. Contractor shall also furnish such other bonds (if any) as are required by the Supplementary Conditions or other provisions of the Contract.
 - C. All bonds must be in the form included in the Bidding Documents or otherwise specified by Owner prior to execution of the Contract, except as provided otherwise by Laws or Regulations, and must be issued and signed by a surety named in "Companies Holding Certificates of Authority as Acceptable Sureties on Federal Bonds and as Acceptable Reinsuring Companies" as published in Department Circular 570 (as amended and supplemented) by the Bureau of the Fiscal Service, U.S. Department of the Treasury. A bond signed by an agent or attorney in fact must be accompanied by a certified copy of that individual's authority to bind the surety. The evidence of authority must show that it is effective on the date the agent or attorney in fact signed the accompanying bond All Bonds shall be in the form prescribed by the Contract Documents or other form approved by Owner. All else notwithstanding, the terms of all Bonds shall be substantially in the form prescribed by Chapter 225, Florida Statutes. All Bonds shall be executed by Contractor and a corporate bonding company licensed to transact such business in the State of Florida and named on the current list of "Surety Companies Acceptable on Federal Bonds" as published in the Treasury Department Circular Number 570. Contactor will cause the Bonds to be recorded with the Clerk of the Circuit Court in the county in which the Work is to be performed. Failure to execute bonds in the form prescribed may constitute non-responsiveness on the part of the Contractor. The expense for all Bonds shall be the Contractor's responsibility.
 - D. Contractor shall obtain the required bonds from surety companies that are duly licensed or authorized, in the state or jurisdiction in which the Project is located, to issue bonds in the required amounts.
 - E. If the surety on a bond furnished by Contractor is declared bankrupt or becomes insolvent, or the surety ceases to meet the requirements above, then Contractor shall promptly notify Owner and Engineer in writing and shall, within 20 days after the event giving rise to such

notification, provide another bond and surety, both of which must comply with the bond and surety requirements above.

- F. If Contractor has failed to obtain a required bond, Owner may exclude the Contractor from the Site and exercise Owner's termination rights under Article 16.
- G. Upon request to Owner from any Subcontractor, Supplier, or other person or entity claiming to have furnished labor, services, materials, or equipment used in the performance of the Work, Owner shall provide a copy of the payment bond to such person or entity.
- H. Upon request to Contractor from any Subcontractor, Supplier, or other person or entity claiming to have furnished labor, services, materials, or equipment used in the performance of the Work, Contractor shall provide a copy of the payment bond to such person or entity.

6.02 Insurance—General Provisions

- A. Owner and Contractor shall obtain and maintain insurance as required in this article and in the Supplementary Conditions.
- B. All insurance required by the Contract to be purchased and maintained by Owner or Contractor shall be obtained from insurance companies that are duly licensed or authorized in the state or jurisdiction in which the Project is located to issue insurance policies for the required limits and coverages. Unless a different standard is indicated in the Supplementary Conditions, <u>or unless approved in writing by the Owner</u>, all companies that provide insurance policies required under this Contract shall have an A.M. Best rating of A-VII or better a current <u>Rating Classification/Financial Category as published in the latest edition of "Best's Key Rating Guide" of at least an A- or above</u>.
- C. Alternative forms of insurance coverage, including but not limited to self-insurance and "Occupational Accident and Excess Employer's Indemnity Policies," are not sufficient to meet the insurance requirements of this Contract, unless expressly allowed in the Supplementary Conditions.
- D. Contractor shall deliver to Owner, with copies to each additional insured identified in the Contract, certificates of insurance and endorsements establishing that Contractor has obtained and is maintaining the policies and coverages required by the Contract. Upon request by Owner or any other insured, Contractor shall also furnish other evidence of such required insurance, including but not limited to copies of policies, documentation of applicable self-insured retentions (if allowed) and deductibles, full disclosure of all relevant exclusions, and evidence of insurance required to be purchased and maintained by Subcontractors or Suppliers. In any documentation furnished under this provision, Contractor, Subcontractors, and Suppliers may block out (redact) (1) any confidential premium or pricing information and (2) any wording specific to a project or jurisdiction other than those applicable to this Contract.
- E. Owner shall deliver to Contractor, with copies to each additional insured identified in the Contract, certificates of insurance and endorsements establishing that Owner has obtained and is maintaining the policies and coverages required of Owner by the Contract (if any). All certificates of insurance shall be dated and shall show the name of the insured Contractor, the specific job by name and job number, the name of the insurer, the policy number assigned, its effective date and its termination date and a list of any exclusionary endorsements. Upon request by Contractor or any other insured, Owner shall also provide other evidence of such required insurance (if any), including but not limited to copies of

EJCDC® C-700, Standard General Conditions of the Construction Contract.

Copyright[©] 2018 National Society of Professional Engineers, American Council of Engineering Companies, and American Society of Civil Engineers. All rights reserved. policies, documentation of applicable self-insured retentions (if allowed) and deductibles, and full disclosure of all relevant exclusions. In any documentation furnished under this provision, Owner may block out (redact) (1) any confidential premium or pricing information and (2) any wording specific to a project or jurisdiction other than those relevant to this Contract.

- F. Failure of Owner or Contractor to demand such certificates or other evidence of the other party's full compliance with these insurance requirements, or failure of Owner or Contractor to identify a deficiency in compliance from the evidence provided, will not be construed as a waiver of the other party's obligation to obtain and maintain such insurance.
- G. In addition to the liability insurance required to be provided by Contractor, the Owner, at Owner's option, may purchase and maintain Owner's own liability insurance. Owner's liability policies, if any, operate separately and independently from policies required to be provided by Contractor, and Contractor cannot rely upon Owner's liability policies for any of Contractor's obligations to the Owner, Engineer, or third parties.
- H. Contractor shall require:
 - Subcontractors to purchase and maintain worker's compensation, commercial general liability, and other insurance that is appropriate for their participation in the Project, and to name as additional insureds Owner and Engineer (and any other individuals or entities identified in the Supplementary Conditions as additional insureds on Contractor's liability policies) on each Subcontractor's commercial general liability insurance policy; and
 - Suppliers to purchase and maintain insurance that is appropriate for their participation in the Project.
- If either party does not purchase or maintain the insurance required of such party by the Contract, such party shall notify the other party in writing of such failure to purchase prior to the start of the Work, or of such failure to maintain prior to any change in the required coverage.
- J. If Contractor has failed to obtain and maintain required insurance, Contractor's entitlement to enter or remain at the Site will end immediately, and Owner may impose an appropriate set-off against payment for any associated costs (including but not limited to the cost of purchasing necessary insurance coverage), and exercise Owner's termination rights under Article 16.
- K. Without prejudice to any other right or remedy, if a party has failed to obtain required insurance, the other party may elect (but is in no way obligated) to obtain equivalent insurance to protect such other party's interests at the expense of the party who was required to provide such coverage, and the Contract Price will be adjusted accordingly.
- L. Owner does not represent that insurance coverage and limits established in this Contract necessarily will be adequate to protect Contractor or Contractor's interests. Contractor is responsible for determining whether such coverage and limits are adequate to protect its interests, and for obtaining and maintaining any additional insurance that Contractor deems necessary.
- M. The insurance and insurance limits required herein will not be deemed as a limitation on Contractor's liability, or that of its Subcontractors or Suppliers, under the indemnities granted to Owner and other individuals and entities in the Contract or otherwise.

N. All the policies of insurance required to be purchased and maintained under this Contract will contain a provision or endorsement that the coverage afforded will not be canceled, or renewal refused, until at least 10 thirty (30) days prior written notice has been given to the purchasing policyholder, <u>Owner and other individuals and entities in the Contract or otherwise listed in the Contract Documents</u>, Within three days of receipt of any such written notice, the purchasing policyholder shall provide a copy of the notice to each other insured and Engineer.

6.03 Contractor's Insurance

- A. Required Insurance: Contractor shall purchase and maintain Worker's Compensation, Commercial General Liability, and other insurance pursuant to the specific requirements of the Supplementary Conditions.
- B. *General Provisions*: The policies of insurance required by this Paragraph 6.03 as supplemented must:
 - 1. include at least the specific coverages required;
 - 2. be written for not less than the limits provided, or those required by Laws or Regulations, whichever is greater;
 - 3. remain in effect at least until the Work is complete (as set forth in Paragraph 15.06.D), and longer if expressly required elsewhere in this Contract, and at all times thereafter when Contractor may be correcting, removing, or replacing defective Work as a warranty or correction obligation, or otherwise, or returning to the Site to conduct other tasks arising from the Contract;
 - 4. apply with respect to the performance of the Work, whether such performance is by Contractor, any Subcontractor or Supplier, or by anyone directly or indirectly employed by any of them to perform any of the Work, or by anyone for whose acts any of them may be liable; and
 - 5. include all necessary endorsements to support the stated requirements.
- C. Additional Insureds: The Contractor's commercial general liability, automobile liability, employer's liability, umbrella or excess, pollution liability, and unmanned aerial vehicle liability policies, if required by this Contract, must:
 - 1. include and list as additional insureds Owner and Engineer, and any individuals or entities identified as additional insureds in the Supplementary Conditions;
 - 2. include coverage for the respective officers, directors, members, partners, employees, and consultants of all such additional insureds;
 - afford primary coverage to these additional insureds for all claims covered thereby (including as applicable those arising from both ongoing and completed operations);
 - 4. not seek contribution from insurance maintained by the additional insured; and
 - as to commercial general liability insurance, apply to additional insureds with respect to liability caused in whole or in part by Contractor's acts or omissions, or the acts and omissions of those working on Contractor's behalf, in the performance of Contractor's operations.

6.04 Builder's Risk and Other Property Insurance

- A. Builder's Risk: Unless otherwise provided in the Supplementary Conditions, Contractor shall purchase and maintain builder's risk insurance upon the Work on a completed value basis, in the amount of the Work's full insurable replacement cost (subject to such deductible amounts as may be provided in the Supplementary Conditions or required by Laws and Regulations). The specific requirements applicable to the builder's risk insurance are set forth in the Supplementary Conditions.
- B. Property Insurance for Facilities of Owner Where Work Will Occur: Owner is responsible for obtaining and maintaining property insurance covering each existing structure, building, or facility in which any part of the Work will occur, or to which any part of the Work will attach or be adjoined. Such property insurance will be written on a special perils (all-risk) form, on a replacement cost basis, providing coverage consistent with that required for the builder's risk insurance, and will be maintained until the Work is complete, as set forth in Paragraph 15.06.D.
- C. Property Insurance for Substantially Complete Facilities: Promptly after Substantial Completion, and before actual occupancy or use of the substantially completed Work, Owner will obtain property insurance for such substantially completed Work, and maintain such property insurance at least until the Work is complete, as set forth in Paragraph 15.06.D. Such property insurance will be written on a special perils (all-risk) form, on a replacement cost basis, and provide coverage consistent with that required for the builder's risk insurance. The builder's risk insurance may terminate upon written confirmation of Owner's procurement of such property insurance.
- D. Partial Occupancy or Use by Owner: If Owner will occupy or use a portion or portions of the Work prior to Substantial Completion of all the Work, as provided in Paragraph 15.04, then Owner (directly, if it is the purchaser of the builder's risk policy, or through Contractor) will provide advance notice of such occupancy or use to the builder's risk insurer, and obtain an endorsement consenting to the continuation of coverage prior to commencing such partial occupancy or use.
- E. Insurance of Other Property; Additional Insurance: If the express insurance provisions of the Contract do not require or address the insurance of a property item or interest, then the entity or individual owning such property item will be responsible for insuring it. If Contractor elects to obtain other special insurance to be included in or supplement the builder's risk or property insurance policies provided under this Paragraph 6.04, it may do so at Contractor's expense.

6.05 Property Losses; Subrogation

- A. The builder's risk insurance policy purchased and maintained in accordance with Paragraph 6.04 (or an installation floater policy if authorized by the Supplementary Conditions), will contain provisions to the effect that in the event of payment of any loss or damage the insurer will have no rights of recovery against any insureds thereunder, or against Engineer or its consultants, or their officers, directors, members, partners, employees, agents, consultants, or subcontractors.
 - Owner and Contractor waive all rights against each other and the respective officers, directors, members, partners, employees, agents, consultants, and subcontractors of each and any of them, for all losses and damages caused by, arising out of, or resulting from any of the perils, risks, or causes of loss covered by such policies and any other

Page 26 of 73

property insurance applicable to the Work; and, in addition, waive all such rights against Engineer, its consultants, all individuals or entities identified in the Supplementary Conditions as builder's risk or installation floater insureds, and the officers, directors, members, partners, employees, agents, consultants, and subcontractors of each and any of them, under such policies for losses and damages so caused.

- None of the above waivers extends to the rights that any party making such waiver may have to the proceeds of insurance held by Owner or Contractor as trustee or fiduciary, or otherwise payable under any policy so issued.
- B. Any property insurance policy maintained by Owner covering any loss, damage, or consequential loss to Owner's existing structures, buildings, or facilities in which any part of the Work will occur, or to which any part of the Work will attach or adjoin; to adjacent structures, buildings, or facilities of Owner; or to part or all of the completed or substantially completed Work, during partial occupancy or use pursuant to Paragraph 15.04, after Substantial Completion pursuant to Paragraph 15.03, or after final payment pursuant to Paragraph 15.06, will contain provisions to the effect that in the event of payment of any loss or damage the insurer will have no rights of recovery against any insureds thereunder, or against Contractor, Subcontractors, or Engineer, or the officers, directors, members, partners, employees, agents, consultants, or subcontractors of each and any of them, and that the insured is allowed to waive the insurer's rights of subrogation in a written contract executed prior to the loss, damage, or consequential loss.
 - Owner waives all rights against Contractor, Subcontractors, and Engineer, and the officers, directors, members, partners, employees, agents, consultants and subcontractors of each and any of them, for all losses and damages caused by, arising out of, or resulting from fire or any of the perils, risks, or causes of loss covered by such policies.
- C. The waivers in this Paragraph 6.05 include the waiver of rights due to business interruption, loss of use, or other consequential loss extending beyond direct physical loss or damage to Owner's property or the Work caused by, arising out of, or resulting from fire or other insured peril, risk, or cause of loss.
- D. Contractor shall be responsible for assuring that each Subcontract contains provisions whereby the Subcontractor waives all rights against Owner, Contractor, all individuals or entities identified in the Supplementary Conditions as insureds, the Engineer and its consultants, and the officers, directors, members, partners, employees, agents, consultants, and subcontractors of each and any of them, for all losses and damages caused by, arising out of, relating to, or resulting from fire or other peril, risk, or cause of loss covered by builder's risk insurance, installation floater, and any other property insurance applicable to the Work.

6.06 Receipt and Application of Property Insurance Proceeds

A. Any insured loss under the builder's risk and other policies of property insurance required by Paragraph 6.04 will be adjusted and settled with the named insured that purchased the policy. Such named insured shall act as fiduciary for the other insureds, and give notice to such other insureds that adjustment and settlement of a claim is in progress. Any other insured may state its position regarding a claim for insured loss in writing within 15 days after notice of such claim.

- B. Proceeds for such insured losses may be made payable by the insurer either jointly to multiple insureds, or to the named insured that purchased the policy in its own right and as fiduciary for other insureds, subject to the requirements of any applicable mortgage clause. A named insured receiving insurance proceeds under the builder's risk and other policies of insurance required by Paragraph 6.04 shall maintain such proceeds in a segregated account, and distribute such proceeds in accordance with such agreement as the parties in interest may reach, or as otherwise required under the dispute resolution provisions of this Contract or applicable Laws and Regulations.
- C. If no other special agreement is reached, Contractor shall repair or replace the damaged Work, using allocated insurance proceeds.

ARTICLE 7—CONTRACTOR'S RESPONSIBILITIES

- 7.01 Contractor's Means and Methods of Construction
 - A. Contractor shall be solely responsible for the means, methods, techniques, sequences, and procedures of construction. <u>Owner reserves the right to reject any means, methods,</u> <u>techniques, sequences or procedures proposed by Contractor which:</u>
 - 1. will constitute or create a hazard to the Work or to the persons or damage to property or existing utilities; or
 - 2. will not produce finished Work in accordance with the terms of the Contract Documents.

Owner's failure to exercise his right to reject such means, methods, techniques, sequences or procedures shall not relieve the Contractor of his obligation to accomplish the result intended in the Contract Documents nor shall the exercise of such right create a cause of action for damages.

- B. If the Contract Documents note, or Contractor determines, that professional engineering or other design services are needed to carry out Contractor's responsibilities for construction means, methods, techniques, sequences, and procedures, or for Site safety, then Contractor shall cause such services to be provided by a properly licensed design professional, at Contractor's expense. Such services are not Owner-delegated professional design services under this Contract, and neither Owner nor Engineer has any responsibility with respect to (1) Contractor's determination of the need for such services, (2) the qualifications or licensing of the design professionals retained or employed by Contractor, (3) the performance of such services, or (4) any errors, omissions, or defects in such services.
- 7.02 Supervision and Superintendence
 - A. Contractor shall supervise, inspect, and direct the Work competently and efficiently, devoting such attention thereto and applying such skills and expertise as may be necessary to perform the Work in accordance with the Contract Documents.
 - B. At all times during the progress of the Work, Contractor shall assign a competent resident superintendent who will not be replaced without written notice to Owner and Engineer except under extraordinary circumstances.

7.03 Labor; Working Hours

- A. Contractor shall provide competent, suitably qualified personnel to survey and lay out the Work and perform construction as required by the Contract Documents. Contractor shall maintain good discipline and order at the Site.
- B. Contractor shall be fully responsible to Owner and Engineer for all acts and omissions of Contractor's employees; of Suppliers and Subcontractors, and their employees; and of any other individuals or entities performing or furnishing any of the Work, just as Contractor is responsible for Contractor's own acts and omissions.
- C. Except as otherwise required for the safety or protection of persons or the Work or property at the Site or adjacent thereto, and except as otherwise stated in the Contract Documents, all Work at the Site will be performed during regular working hours, Monday through Friday. Contractor will not perform Work on a Saturday, Sunday, or any legal holiday. Contractor may perform Work outside regular working hours or on Saturdays, Sundays, or legal holidays only with Owner's written consent, which will not be unreasonably withheld.
- D. Engineer shall record time and costs required by Engineer and Engineer's Consultants to provide inspection services due to Contractor's working beyond regular working hours as defined in the General Requirements. Owner's reimbursement for the charges shall be a deduction from Contractor's Partial Payment(s) in accordance with Section 01 11 00 Part 2.02.
- 7.04 Services, Materials, and Equipment
 - A. Unless otherwise specified in the Contract Documents, Contractor shall provide and assume full responsibility for all services, materials, equipment, labor, transportation, construction equipment and machinery, tools, appliances, fuel, power, light, heat, telephone, water, sanitary facilities, temporary facilities, and all other facilities and incidentals necessary for the performance, testing, start up, and completion of the Work, whether or not such items are specifically called for in the Contract Documents.
 - B. All materials and equipment incorporated into the Work must be new and of good quality, except as otherwise provided in the Contract Documents. All special warranties and guarantees required by the Specifications will expressly run to the benefit of Owner. If required by Engineer, Contractor shall furnish satisfactory evidence (including reports of required tests) as to the source, kind, and quality of materials and equipment.
 - C. All materials and equipment must be stored, applied, installed, connected, erected, protected, used, cleaned, and conditioned in accordance with instructions of the applicable Supplier, except as otherwise may be provided in the Contract Documents.

7.05 "Or Equals"

A. Contractor's Request; Governing Criteria: Whenever an item of equipment or material is specified or described in the Contract Documents by using the names of one or more proprietary items or specific Suppliers, the Contract Price has been based upon Contractor furnishing such item as specified. The specification or description of such an item is intended to establish the type, function, appearance, and quality required. Unless the specification or description contains or is followed by words reading that no like, equivalent, or "or equal" item is permitted, Contractor may request that Engineer authorize the use of other items of equipment or material, or items from other proposed Suppliers, under the circumstances described below.

- If Engineer in its sole discretion determines that an item of equipment or material proposed by Contractor is functionally equal to that named and sufficiently similar so that no change in related Work will be required, Engineer will deem it an "or equal" item. For the purposes of this paragraph, a proposed item of equipment or material will be considered functionally equal to an item so named if:
 - a. in the exercise of reasonable judgment Engineer determines that the proposed item:
 - 1) is at least equal in materials of construction, quality, durability, appearance, strength, and design characteristics;
 - will reliably perform at least equally well the function and achieve the results imposed by the design concept of the completed Project as a functioning whole;
 - 3) has a proven record of performance and availability of responsive service; and
 - 4) is not objectionable to Owner.
 - Contractor certifies that, if the proposed item is approved and incorporated into the Work:
 - 1) there will be no increase in cost to the Owner or increase in Contract Times; and
 - 2) the item will conform substantially to the detailed requirements of the item named in the Contract Documents.
- B. Contractor's Expense: Contractor shall provide all data in support of any proposed "or equal" item at Contractor's expense.
- C. Engineer's Evaluation and Determination: Engineer will be allowed a reasonable time to evaluate each "or-equal" request. Engineer may require Contractor to furnish additional data about the proposed "or-equal" item. Engineer will be the sole judge of acceptability. No "orequal" item will be ordered, furnished, installed, or utilized until Engineer's review is complete and Engineer determines that the proposed item is an "or-equal," which will be evidenced by an approved Shop Drawing or other written communication. Engineer will advise Contractor in writing of any negative determination.
- D. Effect of Engineer's Determination: Neither approval nor denial of an "or-equal" request will result in any change in Contract Price. The Engineer's denial of an "or-equal" request will be final and binding, and may not be reversed through an appeal under any provision of the Contract.
- E. Treatment as a Substitution Request: If Engineer determines that an item of equipment or material proposed by Contractor does not qualify as an "or-equal" item, Contractor may request that Engineer consider the item a proposed substitute pursuant to Paragraph 7.06.

7.06 Substitutes

A. Contractor's Request; Governing Criteria: Unless the specification or description of an item of equipment or material required to be furnished under the Contract Documents contains or is followed by words reading that no substitution is permitted, Contractor may request that Engineer authorize the use of other items of equipment or material under the circumstances described below. To the extent possible such requests must be made before commencement of related construction at the Site.

- Contractor shall submit sufficient information as provided below to allow Engineer to determine if the item of material or equipment proposed is functionally equivalent to that named and an acceptable substitute therefor. Engineer will not accept requests for review of proposed substitute items of equipment or material from anyone other than Contractor.
- The requirements for review by Engineer will be as set forth in Paragraph 7.06.B, as supplemented by the Specifications, and as Engineer may decide is appropriate under the circumstances.
- 3. Contractor shall make written application to Engineer for review of a proposed substitute item of equipment or material that Contractor seeks to furnish or use. The application:
 - a. will certify that the proposed substitute item will:
 - perform adequately the functions and achieve the results called for by the general design;
 - 2) be similar in substance to the item specified; and
 - 3) be suited to the same use as the item specified.
 - b. will state:
 - the extent, if any, to which the use of the proposed substitute item will necessitate a change in Contract Times;
 - 2) whether use of the proposed substitute item in the Work will require a change in any of the Contract Documents (or in the provisions of any other direct contract with Owner for other work on the Project) to adapt the design to the proposed substitute item; and
 - 3) whether incorporation or use of the proposed substitute item in connection with the Work is subject to payment of any license fee or royalty.
 - c. will identify:
 - 1) all variations of the proposed substitute item from the item specified; and
 - 2) available engineering, sales, maintenance, repair, and replacement services.
 - d. will contain an itemized estimate of all costs or credits that will result directly or indirectly from use of such substitute item, including but not limited to changes in Contract Price, shared savings, costs of redesign, and claims of other contractors affected by any resulting change.
- B. Engineer's Evaluation and Determination: Engineer will be allowed a reasonable time to evaluate each substitute request, and to obtain comments and direction from Owner. Engineer may require Contractor to furnish additional data about the proposed substitute item. Engineer will be the sole judge of acceptability. No substitute will be ordered, furnished, installed, or utilized until Engineer's review is complete and Engineer determines that the proposed item is an acceptable substitute. Engineer's determination will be evidenced by a Field Order or a proposed Change Order accounting for the substitution itself and all related impacts, including changes in Contract Price or Contract Times. Engineer will advise Contractor in writing of any negative determination.

- C. Special Guarantee: Owner may require Contractor to furnish at Contractor's expense a special performance guarantee or other surety with respect to any substitute.
- D. Reimbursement of Engineer's Cost: Engineer will record Engineer's costs in evaluating a substitute proposed or submitted by Contractor. Whether or not Engineer approves a substitute so proposed or submitted by Contractor, Contractor shall reimburse Owner for the reasonable charges of Engineer for evaluating each such proposed substitute. Contractor shall also reimburse Owner for the reasonable charges of Engineer for evaluating of Engineer for making changes in the Contract Documents (or in the provisions of any other direct contract with Owner) resulting from the acceptance of each proposed substitute.
- E. Contractor's Expense: Contractor shall provide all data in support of any proposed substitute at Contractor's expense.
- F. Effect of Engineer's Determination: If Engineer approves the substitution request, Contractor shall execute the proposed Change Order and proceed with the substitution. The Engineer's denial of a substitution request will be final and binding, and may not be reversed through an appeal under any provision of the Contract. Contractor may challenge the scope of reimbursement costs imposed under Paragraph 7.06.D, by timely submittal of a Change Proposal.
- 7.07 Concerning Subcontractors and Suppliers
 - A. Contractor may retain Subcontractors and Suppliers for the performance of parts of the Work. Such Subcontractors and Suppliers must be acceptable to Owner. The Contractor's retention of a Subcontractor or Supplier for the performance of parts of the Work will not relieve Contractor's obligation to Owner to perform and complete the Work in accordance with the Contract Documents.
 - B. Contractor shall retain specific Subcontractors and Suppliers for the performance of designated parts of the Work if required by the Contract to do so.
 - C. Subsequent to the submittal of Contractor's Bid or final negotiation of the terms of the Contract, Owner may not require Contractor to retain any Subcontractor or Supplier to furnish or perform any of the Work against which Contractor has reasonable objection.
 - D. Prior to entry into any binding subcontract or purchase order, Contractor shall submit to Owner the identity of the proposed Subcontractor or Supplier (unless Owner has already deemed such proposed Subcontractor or Supplier acceptable during the bidding process or otherwise). Such proposed Subcontractor or Supplier shall be deemed acceptable to Owner unless Owner raises a substantive, reasonable objection within 5 days.
 - E. Owner may require the replacement of any Subcontractor or Supplier. Owner also may require Contractor to retain specific replacements; provided, however, that Owner may not require a replacement to which Contractor has a reasonable objection. If Contractor has submitted the identity of certain Subcontractors or Suppliers for acceptance by Owner, and Owner has accepted it (either in writing or by failing to make written objection thereto), then Owner may subsequently revoke the acceptance of any such Subcontractor or Supplier so identified solely on the basis of substantive, reasonable objection after due investigation. Contractor shall submit an acceptable replacement for the rejected Subcontractor or Supplier.

Page 32 of 73

- F. If Owner requires the replacement of any Subcontractor or Supplier retained by Contractor to perform any part of the Work, then Contractor shall be entitled to an adjustment in Contract Price or Contract Times, with respect to the replacement; and Contractor shall initiate a Change Proposal for such adjustment within 30 days of Owner's requirement of replacement.
- G. No acceptance by Owner of any such Subcontractor or Supplier, whether initially or as a replacement, will constitute a waiver of the right of Owner to the completion of the Work in accordance with the Contract Documents.
- H. On a monthly basis, Contractor shall submit to Engineer a complete list of all Subcontractors and Suppliers having a direct contract with Contractor, and of all other Subcontractors and Suppliers known to Contractor at the time of submittal.
- Contractor shall be solely responsible for scheduling and coordinating the work of Subcontractors and Suppliers.
- J. The divisions and sections of the Specifications and the identifications of any Drawings do not control Contractor in dividing the Work among Subcontractors or Suppliers, or in delineating the Work to be performed by any specific trade.
- K. All Work performed for Contractor by a Subcontractor or Supplier must be pursuant to an appropriate contractual agreement that specifically binds the Subcontractor or Supplier to the applicable terms and conditions of the Contract for the benefit of Owner and Engineer.
- L. Owner may furnish to any Subcontractor or Supplier, to the extent practicable, information about amounts paid to Contractor for Work performed for Contractor by the Subcontractor or Supplier.
- M. Contractor shall restrict all Subcontractors and Suppliers from communicating with Engineer or Owner, except through Contractor or in case of an emergency, or as otherwise expressly allowed in this Contract.
- N. Contractor shall not subcontract part(s) of the Work, the aggregate cost of which is greater than fifty (50) percent of the contract price, without prior written approval by Owner. Contractor shall cause appropriate provisions to be inserted in all subcontracts relative to the Work that bind Subcontractors to Contractor by the terms of the Contract Documents insofar as applicable to the work of Subcontractors and to give Contractor the same power as regards terminating any subcontract that Owner may exercise over Contractor under any provision of the Contract Documents. Nothing contained in the subcontract shall create any contractual relationship between Subcontractor and Owner. Each Subcontractor shall discharge all duties and responsibilities of Contractor to Owner covered by his subcontract.

7.08 Patent Fees and Royalties

A. Contractor shall pay all license fees and royalties and assume all costs incident to the use in the performance of the Work or the incorporation in the Work of any invention, design, process, product, or device which is the subject of patent rights or copyrights held by others. If an invention, design, process, product, or device is specified in the Contract Documents for use in the performance of the Work and if, to the actual knowledge of Owner or Engineer, its use is subject to patent rights or copyrights calling for the payment of any license fee or royalty to others, the existence of such rights will be disclosed in the Contract Documents.

Page 33 of 73

- B. To the fullest extent permitted by Laws and Regulations, Owner shall indemnify and hold harmless Contractor, and its officers, directors, members, partners, employees, agents, consultants, and subcontractors, from and against all claims, costs, losses, and damages (including but not limited to all fees and charges of engineers, architects, attorneys, and other professionals, and all court or arbitration or other dispute resolution costs) arising out of or relating to any infringement of patent rights or copyrights incident to the use in the performance of the Work or resulting from the incorporation in the Work of any invention, design, process, product, or device specified in the Contract Documents, but not identified as being subject to payment of any license fee or royalty to others required by patent rights or copyrights.
- C. To the fullest extent permitted by Laws and Regulations, Contractor shall indemnify and hold harmless Owner and Engineer, and the officers, directors, members, partners, employees, agents, consultants and subcontractors of each and any of them, from and against all claims, costs, losses, and damages (including but not limited to all fees and charges of engineers, architects, attorneys, and other professionals and all court or arbitration or other dispute resolution costs) arising out of or relating to any infringement of patent rights or copyrights incident to the use in the performance of the Work or resulting from the incorporation in the Work of any invention, design, process, product, or device not specified in the Contract Documents.

7.09 Permits

A. Unless otherwise provided in the Contract Documents, Contractor shall obtain and pay for all construction permits, licenses, and certificates of occupancy. Owner shall assist Contractor, when necessary, in obtaining such permits and licenses. Contractor shall pay all governmental charges and inspection fees necessary for the prosecution of the Work which are applicable at the time of the submission of Contractor's Bid (or when Contractor became bound under a negotiated contract). Owner shall pay all charges of utility owners for connections for providing permanent service to the Work. Easements for permanent structures or permanent changes in existing facilities shall be secured and paid for by Owner, unless otherwise specified.

7.10 Taxes

A. Contractor shall pay all sales, consumer, use, and other similar taxes required to be paid by Contractor in accordance with the Laws and Regulations of the place of the Project which are applicable during the performance of the Work.

7.11 Laws and Regulations

- A. Contractor shall give all notices required by and shall comply with all Laws and Regulations applicable to the performance of the Work. Neither Owner nor Engineer shall be responsible for monitoring Contractor's compliance with any Laws or Regulations. <u>Contractor shall also</u> <u>cause all Subcontractors to comply with all federal, state, county, and municipal laws,</u> <u>ordinances, rules, and regulations bearing on the conduct of Work.</u>
- B. If Contractor performs any Work or takes any other action knowing or having reason to know that it is contrary to Laws or Regulations, Contractor shall bear all resulting costs and losses, and shall indemnify and hold harmless Owner and Engineer, and the officers, directors, members, partners, employees, agents, consultants, and subcontractors of each and any of them, from and against all claims, costs, losses, and damages (including but not limited to all

fees and charges of engineers, architects, attorneys, and other professionals and all court or arbitration or other dispute resolution costs) arising out of or relating to such Work or other action. It is not Contractor's responsibility to make certain that the Work described in the Contract Documents is in accordance with Laws and Regulations, but this does not relieve Contractor of its obligations under Paragraph 3.03. If Contractor observes that the Contract Documents are at variance with said laws, ordinances, rates, or regulations, Contractor shall promptly notify Engineer of Record and the County in writing, and any necessary changes shall be adjusted as provided in the Contract Documents. Contractor shall not proceed with the Work until so instructed by Owner.

- C. Owner or Contractor may give written notice to the other party of any changes after the submission of Contractor's Bid (or after the date when Contractor became bound under a negotiated contract) in Laws or Regulations having an effect on the cost or time of performance of the Work, including but not limited to changes in Laws or Regulations having an effect on procuring permits and on sales, use, value-added, consumption, and other similar taxes. If Owner and Contractor are unable to agree on entitlement to or on the amount or extent, if any, of any adjustment in Contract Price or Contract Times resulting from such changes, then within 30 days of such written notice Contractor may submit a Change Proposal, or Owner may initiate a Claim.
- 7.12 Record Documents
 - A. Contractor shall maintain in a safe place at the Site one printed record copy of all Drawings, Specifications, Addenda, Change Orders, Work Change Directives, Field Orders, written interpretations and clarifications, and approved Shop Drawings. Contractor shall keep such record documents in good order and annotate them to show changes made during construction. These record documents, together with all approved Samples, will be available to Engineer for reference. Upon completion of the Work, Contractor shall deliver these record documents to Engineer. <u>Annotations of record documents shall be legible, precise, and complete as determined by Engineer of Record and the County.</u>

7.13 Safety and Protection

- A. Contractor shall be solely responsible for initiating, maintaining, and supervising all safety precautions and programs in connection with the Work. Such responsibility does not relieve Subcontractors of their responsibility for the safety of persons or property in the performance of their work, nor for compliance with applicable safety Laws and Regulations.
- B. Contractor shall designate a qualified and experienced safety representative whose duties and responsibilities are the prevention of Work-related accidents and the maintenance and supervision of safety precautions and programs.
- C. Contractor shall take all necessary precautions for the safety of, and shall provide the necessary protection to prevent damage, injury, or loss to:
 - 1. all persons on the Site or who may be affected by the Work;
 - 2. all the Work and materials and equipment to be incorporated therein, whether in storage on or off the Site; and
 - 3. other property at the Site or adjacent thereto, including trees, shrubs, lawns, walks, pavements, roadways, structures, other work in progress, utilities, and Underground

Facilities not designated for removal, relocation, or replacement in the course of construction.

- D. All damage, injury, or loss to any property referred to in Paragraph 7.13.C.2 or 7.13.C.3 caused, directly or indirectly, in whole or in part, by Contractor, any Subcontractor, Supplier, or any other individual or entity directly or indirectly employed by any of them to perform any of the Work, or anyone for whose acts any of them may be liable, shall be remedied by Contractor at its expense (except damage or loss attributable to the fault of Drawings or Specifications or to the acts or omissions of Owner or Engineer or anyone employed by any of them, or anyone for whose acts any of them may be liable, and not attributable, directly or indirectly, in whole or in part, to the fault or negligence of Contractor or any Subcontractor, Supplier, or other individual or entity directly or indirectly employed by any of them).
- E. Contractor shall comply with all applicable Laws and Regulations relating to the safety of persons or property, or to the protection of persons or property from damage, injury, or loss; and shall erect and maintain all necessary safeguards for such safety and protection.
- F. Contractor shall notify Owner; the owners of adjacent property; the owners of Underground Facilities and other utilities (if the identity of such owners is known to Contractor); and other contractors and utility owners performing work at or adjacent to the Site, in writing, when Contractor knows that prosecution of the Work may affect them, and shall cooperate with them in the protection, removal, relocation, and replacement of their property or work in progress.
- G. Contractor shall comply with the applicable requirements of Owner's safety programs, if any. Any Owner's safety programs that are applicable to the Work are identified or included in the Supplementary Conditions or Specifications.
- H. Contractor shall inform Owner and Engineer of the specific requirements of Contractor's safety program with which Owner's and Engineer's employees and representatives must comply while at the Site.
- Contractor's duties and responsibilities for safety and protection will continue until all the Work is completed, Engineer has issued a written notice to Owner and Contractor in accordance with Paragraph 15.06.C that the Work is acceptable, and Contractor has left the Site (except as otherwise expressly provided in connection with Substantial Completion).
- J. Contractor's duties and responsibilities for safety and protection will resume whenever Contractor or any Subcontractor or Supplier returns to the Site to fulfill warranty or correction obligations, or to conduct other tasks arising from the Contract Documents.
- K. Contractor shall be responsible at all times for precautions to achieve the protection of all persons, including employees and property. Contractor shall make special effort to detect hazardous conditions and shall take prompt action where necessary to avoid accident, injury, or property damage. OSHA and all other applicable safety laws and ordinances shall be followed as well as American National Standards Institute Safety Standards. All accidents, injuries, claims, or potential claims shall be reported promptly to Owner.

7.14 Hazard Communication Programs

A. Contractor shall be responsible for coordinating any exchange of safety data sheets (formerly known as material safety data sheets) or other hazard communication information required

to be made available to or exchanged between or among employers at the Site in accordance with Laws or Regulations.

7.15 Emergencies

A. In emergencies affecting the safety or protection of persons or the Work or property at the Site or adjacent thereto, Contractor is obligated to act to prevent damage, injury, or loss. Contractor shall give Engineer prompt written notice if Contractor believes that any significant changes in the Work or variations from the Contract Documents have been caused by an emergency, or are required as a result of Contractor's response to an emergency. If Engineer determines that a change in the Contract Documents is required because of an emergency or Contractor's response, a Work Change Directive or Change Order will be issued.

7.16 Submittals

- A. Shop Drawing and Sample Requirements
 - 1. Before submitting a Shop Drawing or Sample, Contractor shall:
 - a. review and coordinate the Shop Drawing or Sample with other Shop Drawings and Samples and with the requirements of the Work and the Contract Documents;
 - b. determine and verify:
 - all field measurements, quantities, dimensions, specified performance and design criteria, installation requirements, materials, catalog numbers, and similar information with respect to the Submittal;
 - the suitability of all materials and equipment offered with respect to the indicated application, fabrication, shipping, handling, storage, assembly, and installation pertaining to the performance of the Work; and
 - all information relative to Contractor's responsibilities for means, methods, techniques, sequences, and procedures of construction, and safety precautions and programs incident thereto;
 - c. confirm that the Submittal is complete with respect to all related data included in the Submittal.
 - Each Shop Drawing or Sample must bear a stamp or specific written certification that Contractor has satisfied Contractor's obligations under the Contract Documents with respect to Contractor's review of that Submittal, and that Contractor approves the Submittal.
 - 3. With each Shop Drawing or Sample, Contractor shall give Engineer specific written notice of any variations that the Submittal may have from the requirements of the Contract Documents. This notice must be set forth in a written communication separate from the Submittal; and, in addition, in the case of a Shop Drawing by a specific notation made on the Shop Drawing itself.
- B. Submittal Procedures for Shop Drawings and Samples: Contractor shall label and submit Shop Drawings and Samples to Engineer for review and approval in accordance with the accepted Schedule of Submittals.

- 1. Shop Drawings
 - a. Contractor shall submit the number of copies required in the Specifications.
 - b. Data shown on the Shop Drawings must be complete with respect to quantities, dimensions, specified performance and design criteria, materials, and similar data to show Engineer the services, materials, and equipment Contractor proposes to provide, and to enable Engineer to review the information for the limited purposes required by Paragraph 7.16.C.
- 2. Samples
 - a. Contractor shall submit the number of Samples required in the Specifications.
 - b. Contractor shall clearly identify each Sample as to material, Supplier, pertinent data such as catalog numbers, the use for which intended and other data as Engineer may require to enable Engineer to review the Submittal for the limited purposes required by Paragraph 7.16.C.
- 3. Where a Shop Drawing or Sample is required by the Contract Documents or the Schedule of Submittals, any related Work performed prior to Engineer's review and approval of the pertinent submittal will be at the sole expense and responsibility of Contractor.
- C. Engineer's Review of Shop Drawings and Samples
 - Engineer will provide timely review of Shop Drawings and Samples in accordance with the accepted Schedule of Submittals. Engineer's review and approval will be only to determine if the items covered by the Submittals will, after installation or incorporation in the Work, comply with the requirements of the Contract Documents, and be compatible with the design concept of the completed Project as a functioning whole as indicated by the Contract Documents.
 - Engineer's review and approval will not extend to means, methods, techniques, sequences, or procedures of construction, or to safety precautions or programs incident thereto.
 - 3. Engineer's review and approval of a separate item as such will not indicate approval of the assembly in which the item functions.
 - 4. Engineer's review and approval of a Shop Drawing or Sample will not relieve Contractor from responsibility for any variation from the requirements of the Contract Documents unless Contractor has complied with the requirements of Paragraph 7.16.A.3 and Engineer has given written approval of each such variation by specific written notation thereof incorporated in or accompanying the Shop Drawing or Sample. Engineer will document any such approved variation from the requirements of the Contract Documents in a Field Order or other appropriate Contract modification.
 - 5. Engineer's review and approval of a Shop Drawing or Sample will not relieve Contractor from responsibility for complying with the requirements of Paragraphs 7.16.A and B.
 - 6. Engineer's review and approval of a Shop Drawing or Sample, or of a variation from the requirements of the Contract Documents, will not, under any circumstances, change the Contract Times or Contract Price, unless such changes are included in a Change Order.

- 7. Neither Engineer's receipt, review, acceptance, or approval of a Shop Drawing or Sample will result in such item becoming a Contract Document.
- 8. Contractor shall perform the Work in compliance with the requirements and commitments set forth in approved Shop Drawings and Samples, subject to the provisions of Paragraph 7.16.C.4.
- D. Resubmittal Procedures for Shop Drawings and Samples
 - Contractor shall make corrections required by Engineer and shall return the required number of corrected copies of Shop Drawings and submit, as required, new Samples for review and approval. Contractor shall direct specific attention in writing to revisions other than the corrections called for by Engineer on previous Submittals.
 - 2. Contractor shall furnish required Shop Drawing and Sample submittals with sufficient information and accuracy to obtain required approval of an item with no more than two resubmittals. Engineer will record Engineer's time for reviewing a third or subsequent resubmittal of a Shop Drawing or Sample, and Contractor shall be responsible for Engineer's charges to Owner for such time. Owner may impose a set-off against payments due Contractor to secure reimbursement for such charges.
 - 3. If Contractor requests a change of a previously approved Shop Drawing or Sample, Contractor shall be responsible for Engineer's charges to Owner for its review time, and Owner may impose a set-off against payments due Contractor to secure reimbursement for such charges, unless the need for such change is beyond the control of Contractor.
- E. Submittals Other than Shop Drawings, Samples, and Owner-Delegated Designs
 - 1. The following provisions apply to all Submittals other than Shop Drawings, Samples, and Owner-delegated designs:
 - a. Contractor shall submit all such Submittals to the Engineer in accordance with the Schedule of Submittals and pursuant to the applicable terms of the Contract Documents.
 - b. Engineer will provide timely review of all such Submittals in accordance with the Schedule of Submittals and return such Submittals with a notation of either Accepted or Not Accepted. Any such Submittal that is not returned within the time established in the Schedule of Submittals will be deemed accepted.
 - c. Engineer's review will be only to determine if the Submittal is acceptable under the requirements of the Contract Documents as to general form and content of the Submittal.
 - d. If any such Submittal is not accepted, Contractor shall confer with Engineer regarding the reason for the non-acceptance, and resubmit an acceptable document.
 - 2. Procedures for the submittal and acceptance of the Progress Schedule, the Schedule of Submittals, and the Schedule of Values are set forth in Paragraphs 2.03. 2.04, and 2.05.
- F. Owner-delegated Designs: Submittals pursuant to Owner-delegated designs are governed by the provisions of Paragraph 7.19.

7.17 Contractor's General Warranty and Guarantee

- A. Contractor warrants and guarantees to Owner that all Work will be in accordance with the Contract Documents and will not be defective. Engineer is entitled to rely on Contractor's warranty and guarantee.
- B. Owner's rights under this warranty and guarantee are in addition to, and are not limited by, Owner's rights under the correction period provisions of Paragraph 15.08. The time in which Owner may enforce its warranty and guarantee rights under this Paragraph 7.17 is limited only by applicable Laws and Regulations restricting actions to enforce such rights; provided, however, that after the end of the correction period under Paragraph 15.08:
 - 1. Owner shall give Contractor written notice of any defective Work within 60 days of the discovery that such Work is defective; and
 - 2. Such notice will be deemed the start of an event giving rise to a Claim under Paragraph 12.01.B, such that any related Claim must be brought within 30 days of the notice.
- C. Contractor's warranty and guarantee hereunder excludes defects or damage caused by:
 - 1. abuse, or improper modification, maintenance, or operation, by persons other than Contractor, Subcontractors, Suppliers, or any other individual or entity for whom Contractor is responsible; or
 - 2. normal wear and tear under normal usage.
- D. Contractor's obligation to perform and complete the Work in accordance with the Contract Documents is absolute. None of the following will constitute an acceptance of Work that is not in accordance with the Contract Documents, a release of Contractor's obligation to perform the Work in accordance with the Contract Documents, or a release of Owner's warranty and guarantee rights under this Paragraph 7.17:
 - 1. Observations by Engineer;
 - 2. Recommendation by Engineer or payment by Owner of any progress or final payment;
 - 3. The issuance of a certificate of Substantial Completion by Engineer or any payment related thereto by Owner;
 - 4. Use or occupancy of the Work or any part thereof by Owner;
 - 5. Any review and approval of a Shop Drawing or Sample submittal;
 - 6. The issuance of a notice of acceptability by Engineer;
 - 7. The end of the correction period established in Paragraph 15.08;
 - 8. Any inspection, test, or approval by others; or
 - 9. Any correction of defective Work by Owner.
- E. If the Contract requires the Contractor to accept the assignment of a contract entered into by Owner, then the specific warranties, guarantees, and correction obligations contained in the assigned contract will govern with respect to Contractor's performance obligations to Owner for the Work described in the assigned contract.

7.18 Indemnification

- A. To the fullest extent permitted by Laws and Regulations, and in addition to any other obligations of Contractor under the Contract or otherwise, Contractor shall indemnify and hold harmless Owner and Engineer, and the officers, directors, members, partners, employees, agents, consultants and subcontractors of each and any of them, from losses, damages, costs, and judgments (including but not limited to all fees and charges of engineers, architects, attorneys, and other professionals, and all court or arbitration or other dispute resolution costs) arising from third-party claims or actions relating to or resulting from the performance or furnishing of the Work, provided that any such claim, action, loss, cost, judgment or damage is attributable to bodily injury, sickness, disease, or death, or to damage to or destruction of tangible property (other than the Work itself), including the loss of use resulting therefrom, but only to the extent caused by any negligent act or omission of Contractor, any Subcontractor, any Supplier, or any individual or entity directly or indirectly employed by any of them to perform any of the Work, or anyone for whose acts any of them may be liable. Five (5) percent of the Contract Price is given as consideration for this indemnification.
- B. In any and all claims against Owner or Engineer, or any of their officers, directors, members, partners, employees, agents, consultants, or subcontractors, by any employee (or the survivor or personal representative of such employee) of Contractor, any Subcontractor, any Supplier, or any individual or entity directly or indirectly employed by any of them to perform any of the Work, or anyone for whose acts any of them may be liable, the indemnification obligation under Paragraph 7.18.A will not be limited in any way by any limitation on the amount or type of damages, compensation, or benefits payable by or for Contractor or any such Subcontractor, Supplier, or other individual or entity under workers' compensation acts, disability benefit acts, or other employee benefit acts.
- C. The indemnification obligations of Contractor under Paragraph 7.18 shall not extend to the liability of Engineer and Engineer's officers, directors, members, partners, employees, agents, consultants and subcontractors arising out of:
 - 1. the preparation or approval of, or the failure to prepare or approve maps, Drawings, opinions, reports, surveys, Change Orders, designs, or Specifications; or
 - 2. giving directions or instructions, or failing to give them, if that is the primary cause of the injury or damage.
- D. Neither Contractor nor Owner shall sell, transfer, assign, or otherwise dispose of the contract or any portion thereof, or any right, title, or interest therein, or any obligations thereunder, without written consent of the other party.

7.19 Delegation of Professional Design Services

- A. Owner may require Contractor to provide professional design services for a portion of the Work by express delegation in the Contract Documents. Such delegation will specify the performance and design criteria that such services must satisfy, and the Submittals that Contractor must furnish to Engineer with respect to the Owner-delegated design.
- B. Contractor shall cause such Owner-delegated professional design services to be provided pursuant to the professional standard of care by a properly licensed design professional, whose signature and seal must appear on all drawings, calculations, specifications,

certifications, and Submittals prepared by such design professional. Such design professional must issue all certifications of design required by Laws and Regulations.

- C. If a Shop Drawing or other Submittal related to the Owner-delegated design is prepared by Contractor, a Subcontractor, or others for submittal to Engineer, then such Shop Drawing or other Submittal must bear the written approval of Contractor's design professional when submitted by Contractor to Engineer.
- D. Owner and Engineer shall be entitled to rely upon the adequacy, accuracy, and completeness of the services, certifications, and approvals performed or provided by the design professionals retained or employed by Contractor under an Owner-delegated design, subject to the professional standard of care and the performance and design criteria stated in the Contract Documents.
- E. Pursuant to this Paragraph 7.19, Engineer's review, approval, and other determinations regarding design drawings, calculations, specifications, certifications, and other Submittals furnished by Contractor pursuant to an Owner-delegated design will be only for the following limited purposes:
 - 1. Checking for conformance with the requirements of this Paragraph 7.19;
 - Confirming that Contractor (through its design professionals) has used the performance and design criteria specified in the Contract Documents; and
 - 3. Establishing that the design furnished by Contractor is consistent with the design concept expressed in the Contract Documents.
- F. Contractor shall not be responsible for the adequacy of performance or design criteria specified by Owner or Engineer.
- G. Contractor is not required to provide professional services in violation of applicable Laws and Regulations.

ARTICLE 8—OTHER WORK AT THE SITE

8.01 Other Work

- A. In addition to and apart from the Work under the Contract Documents, the Owner may perform other work at or adjacent to the Site. Such other work may be performed by Owner's employees, or through contracts between the Owner and third parties. Owner may also arrange to have third-party utility owners perform work on their utilities and facilities at or adjacent to the Site.
- B. If Owner performs other work at or adjacent to the Site with Owner's employees, or through contracts for such other work, then Owner shall give Contractor written notice thereof prior to starting any such other work. If Owner has advance information regarding the start of any third-party utility work that Owner has arranged to take place at or adjacent to the Site, Owner shall provide such information to Contractor.
- C. Contractor shall afford proper and safe access to the Site to each contractor that performs such other work, each utility owner performing other work, and Owner, if Owner is performing other work with Owner's employees, and provide a reasonable opportunity for the introduction and storage of materials and equipment and the execution of such other work.

- D. Contractor shall do all cutting, fitting, and patching of the Work that may be required to properly connect or otherwise make its several parts come together and properly integrate with such other work. Contractor shall not endanger any work of others by cutting, excavating, or otherwise altering such work; provided, however, that Contractor may cut or alter others' work with the written consent of Engineer and the others whose work will be affected.
- E. If the proper execution or results of any part of Contractor's Work depends upon work performed by others, Contractor shall inspect such other work and promptly report to Engineer in writing any delays, defects, or deficiencies in such other work that render it unavailable or unsuitable for the proper execution and results of Contractor's Work. Contractor's failure to so report will constitute an acceptance of such other work as fit and proper for integration with Contractor's Work except for latent defects and deficiencies in such other work.
- F. The provisions of this article are not applicable to work that is performed by third-party utilities or other third-party entities without a contract with Owner, or that is performed without having been arranged by Owner. If such work occurs, then any related delay, disruption, or interference incurred by Contractor is governed by the provisions of Paragraph 4.05.C.3.

8.02 Coordination

- A. If Owner intends to contract with others for the performance of other work at or adjacent to the Site, to perform other work at or adjacent to the Site with Owner's employees, or to arrange to have utility owners perform work at or adjacent to the Site, the following will be set forth in the Supplementary Conditions or provided to Contractor prior to the start of any such other work:
 - The identity of the individual or entity that will have authority and responsibility for coordination of the activities among the various contractors;
 - An itemization of the specific matters to be covered by such authority and responsibility; and
 - 3. The extent of such authority and responsibilities.
- B. Unless otherwise provided in the Supplementary Conditions, Owner shall have sole authority and responsibility for such coordination.

8.03 Legal Relationships

A. If, in the course of performing other work for Owner at or adjacent to the Site, the Owner's employees, any other contractor working for Owner, or any utility owner that Owner has arranged to perform work, causes damage to the Work or to the property of Contractor or its Subcontractors, or delays, disrupts, interferes with, or increases the scope or cost of the performance of the Work, through actions or inaction, then Contractor shall be entitled to an equitable adjustment in the Contract Price or the Contract Times. Contractor must submit any Change Proposal seeking an equitable adjustment in the Contract Price or the Contract Price or the Contract Times under this paragraph within 30 days of the damaging, delaying, disrupting, or interfering event. The entitlement to, and extent of, any such equitable adjustment will take into account information (if any) regarding such other work that was provided to Contractor in the Contract Documents prior to the submittal of the Bid or the final negotiation of the terms of the

Contract, and any remedies available to Contractor under Laws or Regulations concerning utility action or inaction. When applicable, any such equitable adjustment in Contract Price will be conditioned on Contractor assigning to Owner all Contractor's rights against such other contractor or utility owner with respect to the damage, delay, disruption, or interference that is the subject of the adjustment. Contractor's entitlement to an adjustment of the Contract Times or Contract Price is subject to the provisions of Paragraphs 4.05.D and 4.05.E.

- B. Contractor shall take reasonable and customary measures to avoid damaging, delaying, disrupting, or interfering with the work of Owner, any other contractor, or any utility owner performing other work at or adjacent to the Site.
 - If Contractor fails to take such measures and as a result damages, delays, disrupts, or interferes with the work of any such other contractor or utility owner, then Owner may impose a set-off against payments due Contractor, and assign to such other contractor or utility owner the Owner's contractual rights against Contractor with respect to the breach of the obligations set forth in this Paragraph 8.03.8.
 - 2. When Owner is performing other work at or adjacent to the Site with Owner's employees, Contractor shall be liable to Owner for damage to such other work, and for the reasonable direct delay, disruption, and interference costs incurred by Owner as a result of Contractor's failure to take reasonable and customary measures with respect to Owner's other work. In response to such damage, delay, disruption, or interference, Owner may impose a set-off against payments due Contractor.
- C. If Contractor damages, delays, disrupts, or interferes with the work of any other contractor, or any utility owner performing other work at or adjacent to the Site, through Contractor's failure to take reasonable and customary measures to avoid such impacts, or if any claim arising out of Contractor's actions, inactions, or negligence in performance of the Work at or adjacent to the Site is made by any such other contractor or utility owner against Contractor, Owner, or Engineer, then Contractor shall (1) promptly attempt to settle the claim as to all parties through negotiations with such other contractor or utility owner, or otherwise resolve the claim by arbitration or other dispute resolution proceeding or at law, and (2) indemnify and hold harmless Owner and Engineer, and the officers, directors, members, partners, employees, agents, consultants and subcontractors of each and any of them from and against any such claims, and against all costs, losses, and damages (including but not limited to all fees and charges of engineers, architects, attorneys, and other professionals and all court or arbitration or other dispute resolution costs) arising out of or relating to such damage, delay, disruption, or interference.

ARTICLE 9—OWNER'S RESPONSIBILITIES

- 9.01 Communications to Contractor
 - A. Except as otherwise provided in these General Conditions, Owner shall issue all communications to Contractor through Engineer.

9.02 Replacement of Engineer

A. Owner may at its discretion appoint an engineer to replace Engineer, provided Contractor makes no reasonable objection to the replacement engineer. The replacement engineer's status under the Contract Documents will be that of the former Engineer.

9.03 Furnish Data

A. Owner shall promptly furnish the data required of Owner under the Contract Documents.

- 9.04 Pay When Due
 - A. Owner shall make payments to Contractor when they are due as provided in the Agreement.
- 9.05 Lands and Easements; Reports, Tests, and Drawings
 - A. Owner's duties with respect to providing lands and easements are set forth in Paragraph 5.01.
 - B. Owner's duties with respect to providing engineering surveys to establish reference points are set forth in Paragraph 4.03.
 - C. Article 5 refers to Owner's identifying and making available to Contractor copies of reports of explorations and tests of conditions at the Site, and drawings of physical conditions relating to existing surface or subsurface structures at the Site.
- 9.06 Insurance
 - A. Owner's responsibilities, if any, with respect to purchasing and maintaining liability and property insurance are set forth in Article 6.
- 9.07 Change Orders
 - A. Owner's responsibilities with respect to Change Orders are set forth in Article 11.
- 9.08 Inspections, Tests, and Approvals
 - A. Owner's responsibility with respect to certain inspections, tests, and approvals is set forth in Paragraph 14.02.B.
- 9.09 Limitations on Owner's Responsibilities
 - A. The Owner shall not supervise, direct, or have control or authority over, nor be responsible for, Contractor's means, methods, techniques, sequences, or procedures of construction, or the safety precautions and programs incident thereto, or for any failure of Contractor to comply with Laws and Regulations applicable to the performance of the Work. Owner will not be responsible for Contractor's failure to perform the Work in accordance with the Contract Documents.
- 9.10 Undisclosed Hazardous Environmental Condition
 - A. Owner's responsibility in respect to an undisclosed Hazardous Environmental Condition is set forth in Paragraph 5.06.
- 9.11 Evidence of Financial Arrangements
 - A. Upon request of Contractor, Owner shall furnish Contractor reasonable evidence that financial arrangements have been made to satisfy Owner's obligations under the Contract (including obligations under proposed changes in the Work).
- 9.12 Safety Programs
 - A. While at the Site, Owner's employees and representatives shall comply with the specific applicable requirements of Contractor's safety programs of which Owner has been informed.
 - B. Owner shall furnish copies of any applicable Owner safety programs to Contractor.

ARTICLE 10-ENGINEER'S STATUS DURING CONSTRUCTION

- 10.01 Owner's Representative
 - A. Engineer will be Owner's representative during the construction period. The duties and responsibilities and the limitations of authority of Engineer as Owner's representative during construction are set forth in the Contract.
- 10.02 Visits to Site
 - A. Engineer will make visits to the Site at intervals appropriate to the various stages of construction as Engineer deems necessary in order to observe, as an experienced and qualified design professional, the progress that has been made and the quality of the various aspects of Contractor's executed Work. Based on information obtained during such visits and observations, Engineer, for the benefit of Owner, will determine, in general, if the Work is proceeding in accordance with the Contract Documents. Engineer will not be required to make exhaustive or continuous inspections on the Site to check the quality or quantity of the Work. Engineer's efforts will be directed toward providing for Owner a greater degree of confidence that the completed Work will conform generally to the Contract Documents. On the basis of such visits and observations, Engineer will nedeavor to guard Owner against defective Work.
 - B. Engineer's visits and observations are subject to all the limitations on Engineer's authority and responsibility set forth in Paragraph 10.07. Particularly, but without limitation, during or as a result of Engineer's visits or observations of Contractor's Work, Engineer will not supervise, direct, control, or have authority over or be responsible for Contractor's means, methods, techniques, sequences, or procedures of construction, or the safety precautions and programs incident thereto, or for any failure of Contractor to comply with Laws and Regulations applicable to the performance of the Work.

10.03 Resident Project Representative

- A. If Owner and Engineer have agreed that Engineer will furnish a Resident Project Representative to represent Engineer at the Site and assist Engineer in observing the progress and quality of the Work, then the authority and responsibilities of any such Resident Project Representative will be as provided in the Supplementary Conditions, and limitations on the responsibilities thereof will be as provided in the Supplementary Conditions and in Paragraph 10.07.
- B. If Owner designates an individual or entity who is not Engineer's consultant, agent, or employee to represent Owner at the Site, then the responsibilities and authority of such individual or entity will be as provided in the Supplementary Conditions.

10.04 Engineer's Authority

- A. Engineer has the authority to reject Work in accordance with Article 14.
- B. Engineer's authority as to Submittals is set forth in Paragraph 7.16.
- C. Engineer's authority as to design drawings, calculations, specifications, certifications and other Submittals from Contractor in response to Owner's delegation (if any) to Contractor of professional design services, is set forth in Paragraph 7.19.
- D. Engineer's authority as to changes in the Work is set forth in Article 11.

E. Engineer's authority as to Applications for Payment is set forth in Article 15.

10.05 Determinations for Unit Price Work

- A. Engineer will determine the actual quantities and classifications of Unit Price Work performed by Contractor as set forth in Paragraph 13.03.
- 10.06 Decisions on Requirements of Contract Documents and Acceptability of Work
 - A. Engineer will render decisions regarding the requirements of the Contract Documents, and judge the acceptability of the Work, pursuant to the specific procedures set forth herein for initial interpretations, Change Proposals, and acceptance of the Work. In rendering such decisions and judgments, Engineer will not show partiality to Owner or Contractor, and will not be liable to Owner, Contractor, or others in connection with any proceedings, interpretations, decisions, or judgments conducted or rendered in good faith.

10.07 Limitations on Engineer's Authority and Responsibilities

- A. Neither Engineer's authority or responsibility under this Article 10 or under any other provision of the Contract, nor any decision made by Engineer in good faith either to exercise or not exercise such authority or responsibility or the undertaking, exercise, or performance of any authority or responsibility by Engineer, will create, impose, or give rise to any duty in contract, tort, or otherwise owed by Engineer to Contractor, any Subcontractor, any Supplier, any other individual or entity, or to any surety for or employee or agent of any of them.
- B. Engineer will not supervise, direct, control, or have authority over or be responsible for Contractor's means, methods, techniques, sequences, or procedures of construction, or the safety precautions and programs incident thereto, or for any failure of Contractor to comply with Laws and Regulations applicable to the performance of the Work. Engineer will not be responsible for Contractor's failure to perform the Work in accordance with the Contract Documents.
- C. Engineer will not be responsible for the acts or omissions of Contractor or of any Subcontractor, any Supplier, or of any other individual or entity performing any of the Work.
- D. Engineer's review of the final Application for Payment and accompanying documentation, and all maintenance and operating instructions, schedules, guarantees, bonds, certificates of inspection, tests and approvals, and other documentation required to be delivered by Contractor under Paragraph 15.06.A, will only be to determine generally that their content complies with the requirements of, and in the case of certificates of inspections, tests, and approvals, that the results certified indicate compliance with the Contract Documents.
- E. The limitations upon authority and responsibility set forth in this Paragraph 10.07 also apply to the Resident Project Representative, if any.

10.08 Compliance with Safety Program

A. While at the Site, Engineer's employees and representatives will comply with the specific applicable requirements of Owner's and Contractor's safety programs of which Engineer has been informed.

ARTICLE 11-CHANGES TO THE CONTRACT

11.01 Amending and Supplementing the Contract

- A. The Contract may be amended or supplemented by a Change Order, a Work Change Directive, or a Field Order.
- B. If an amendment or supplement to the Contract includes a change in the Contract Price or the Contract Times, such amendment or supplement must be set forth in a Change Order.
- C. All changes to the Contract that involve (1) the performance or acceptability of the Work, (2) the design (as set forth in the Drawings, Specifications, or otherwise), or (3) other engineering or technical matters, must be supported by Engineer's recommendation. Owner and Contractor may amend other terms and conditions of the Contract without the recommendation of the Engineer.

11.02 Change Orders

- A. Owner and Contractor shall execute appropriate Change Orders covering:
 - Changes in Contract Price or Contract Times which are agreed to by the parties, including any undisputed sum or amount of time for Work actually performed in accordance with a Work Change Directive;
 - Changes in Contract Price resulting from an Owner set-off, unless Contractor has duly contested such set-off;
 - 3. Changes in the Work which are: (a) ordered by Owner pursuant to Paragraph 11.05, (b) required because of Owner's acceptance of defective Work under Paragraph 14.04 or Owner's correction of defective Work under Paragraph 14.07, or (c) agreed to by the parties, subject to the need for Engineer's recommendation if the change in the Work involves the design (as set forth in the Drawings, Specifications, or otherwise) or other engineering or technical matters; and
 - 4. Changes that embody the substance of any final and binding results under: Paragraph 11.03.B, resolving the impact of a Work Change Directive; Paragraph 11.09, concerning Change Proposals; Article 12, Claims; Paragraph 13.02.D, final adjustments resulting from allowances; Paragraph 13.03.D, final adjustments relating to determination of quantities for Unit Price Work; and similar provisions.
- B. If Owner or Contractor refuses to execute a Change Order that is required to be executed under the terms of Paragraph 11.02.A, it will be deemed to be of full force and effect, as if fully executed.

11.03 Work Change Directives

A. A Work Change Directive will not change the Contract Price or the Contract Times but is evidence that the parties expect that the modification ordered or documented by a Work Change Directive will be incorporated in a subsequently issued Change Order, following negotiations by the parties as to the Work Change Directive's effect, if any, on the Contract Price and Contract Times; or, if negotiations are unsuccessful, by a determination under the terms of the Contract Documents governing adjustments, expressly including Paragraph 11.07 regarding change of Contract Price.

- B. If Owner has issued a Work Change Directive and:
 - Contractor believes that an adjustment in Contract Times or Contract Price is necessary, then Contractor shall submit any Change Proposal seeking such an adjustment no later than 30 days after the completion of the Work set out in the Work Change Directive.
 - Owner believes that an adjustment in Contract Times or Contract Price is necessary, then Owner shall submit any Claim seeking such an adjustment no later than 60 days after issuance of the Work Change Directive.

11.04 Field Orders

- A. Engineer may authorize minor changes in the Work if the changes do not involve an adjustment in the Contract Price or the Contract Times and are compatible with the design concept of the completed Project as a functioning whole as indicated by the Contract Documents. Such changes will be accomplished by a Field Order and will be binding on Owner and also on Contractor, which shall perform the Work involved promptly.
- B. If Contractor believes that a Field Order justifies an adjustment in the Contract Price or Contract Times, then before proceeding with the Work at issue, Contractor shall submit a Change Proposal as provided herein.
- 11.05 Owner-Authorized Changes in the Work
 - A. Without invalidating the Contract and without notice to any surety, Owner may, at any time or from time to time, order additions, deletions, or revisions in the Work. Changes involving the design (as set forth in the Drawings, Specifications, or otherwise) or other engineering or technical matters will be supported by Engineer's recommendation.
 - B. Such changes in the Work may be accomplished by a Change Order, if Owner and Contractor have agreed as to the effect, if any, of the changes on Contract Times or Contract Price; or by a Work Change Directive. Upon receipt of any such document, Contractor shall promptly proceed with the Work involved; or, in the case of a deletion in the Work, promptly cease construction activities with respect to such deleted Work. Added or revised Work must be performed under the applicable conditions of the Contract Documents.
 - C. Nothing in this Paragraph 11.05 obligates Contractor to undertake work that Contractor reasonably concludes cannot be performed in a manner consistent with Contractor's safety obligations under the Contract Documents or Laws and Regulations.

11.06 Unauthorized Changes in the Work

A. Contractor shall not be entitled to an increase in the Contract Price or an extension of the Contract Times with respect to any work performed that is not required by the Contract Documents, as amended, modified, or supplemented, except in the case of an emergency as provided in Paragraph 7.15 or in the case of uncovering Work as provided in Paragraph 14.05.C.2.

11.07 Change of Contract Price

- A. The Contract Price may only be changed by a Change Order. Any Change Proposal for an adjustment in the Contract Price must comply with the provisions of Paragraph 11.09. Any Claim for an adjustment of Contract Price must comply with the provisions of Article 12.
- B. An adjustment in the Contract Price will be determined as follows:

- Where the Work involved is covered by unit prices contained in the Contract Documents, then by application of such unit prices to the quantities of the items involved (subject to the provisions of Paragraph 13.03);
- 2. Where the Work involved is not covered by unit prices contained in the Contract Documents, then by a mutually agreed lump sum (which may include an allowance for overhead and profit not necessarily in accordance with Paragraph 11.07.C.2); or
- 3. Where the Work involved is not covered by unit prices contained in the Contract Documents and the parties do not reach mutual agreement to a lump sum, then on the basis of the Cost of the Work (determined as provided in Paragraph 13.01) plus a Contractor's fee for overhead and profit (determined as provided in Paragraph 11.07.C).
- C. *Contractor's Fee*: When applicable, the Contractor's fee for overhead and profit will be determined as follows:
 - 1. A mutually acceptable fixed fee; or
 - 2. If a fixed fee is not agreed upon, then a fee based on the following percentages of the various portions of the Cost of the Work:
 - For costs incurred under Paragraphs 13.01.B.1 and 13.01.B.2, the Contractor's fee will be 15 percent;
 - b. For costs incurred under Paragraph 13.01.B.3, the Contractor's fee will be 5 percent;
 - c. Where one or more tiers of subcontracts are on the basis of Cost of the Work plus a fee and no fixed fee is agreed upon, the intent of Paragraphs 11.07.C.2.a and 11.07.C.2.b is that the Contractor's fee will be based on: (1) a fee of 15 percent of the costs incurred under Paragraphs 13.01.B.1 and 13.01.B.2 by the Subcontractor that actually performs the Work, at whatever tier, and (2) with respect to Contractor itself and to any Subcontractors of a tier higher than that of the Subcontractor that actually performs the Work, a fee of 5 percent of the amount (fee plus underlying costs incurred) attributable to the next lower tier Subcontractor; provided, however, that for any such subcontracted Work the maximum total fee to be paid by Owner will be no greater than 27 percent of the costs incurred by the Subcontractor that actually performs the Work;
 - d. No fee will be payable on the basis of costs itemized under Paragraphs 13.01.B.4, 13.01.B.5, and 13.01.C;
 - e. The amount of credit to be allowed by Contractor to Owner for any change which results in a net decrease in Cost of the Work will be the amount of the actual net decrease in Cost of the Work and a deduction of an additional amount equal to 5 percent of such actual net decrease in Cost of the Work; and
 - f. When both additions and credits are involved in any one change or Change Proposal, the adjustment in Contractor's fee will be computed by determining the sum of the costs in each of the cost categories in Paragraph 13.01.B (specifically, payroll costs, Paragraph 13.01.B.1; incorporated materials and equipment costs, Paragraph 13.01.B.2; Subcontract costs, Paragraph 13.01.B.3; special consultants costs, Paragraph 13.01.B.4; and other costs, Paragraph 13.01.B.5) and applying to each such cost category sum the appropriate fee from Paragraphs 11.07.C.2.a through 11.07.C.2.e, inclusive.
11.08 Change of Contract Times

- A. The Contract Times may only be changed by a Change Order. Any Change Proposal for an adjustment in the Contract Times must comply with the provisions of Paragraph 11.09. Any Claim for an adjustment in the Contract Times must comply with the provisions of Article 12.
- B. Delay, disruption, and interference in the Work, and any related changes in Contract Times, are addressed in and governed by Paragraph 4.05.
- 11.09 Change Proposals
 - A. Purpose and Content: Contractor shall submit a Change Proposal to Engineer to request an adjustment in the Contract Times or Contract Price; contest an initial decision by Engineer concerning the requirements of the Contract Documents or relating to the acceptability of the Work under the Contract Documents; challenge a set-off against payment due; or seek other relief under the Contract. The Change Proposal will specify any proposed change in Contract Times or Contract Price, or other proposed relief, and explain the reason for the proposed change, with citations to any governing or applicable provisions of the Contract Documents. Each Change Proposal will address only one issue, or a set of closely related issues.
 - B. Change Proposal Procedures
 - 1. Submittal: Contractor shall submit each Change Proposal to Engineer within 30 days after the start of the event giving rise thereto, or after such initial decision.
 - 2. Supporting Data: The Contractor shall submit supporting data, including the proposed change in Contract Price or Contract Time (if any), to the Engineer and Owner within 15 days after the submittal of the Change Proposal.
 - a. Change Proposals based on or related to delay, interruption, or interference must comply with the provisions of Paragraphs 4.05.D and 4.05.E.
 - b. Change proposals related to a change of Contract Price must include full and detailed accounts of materials incorporated into the Work and labor and equipment used for the subject Work.

The supporting data must be accompanied by a written statement that the supporting data are accurate and complete, and that any requested time or price adjustment is the entire adjustment to which Contractor believes it is entitled as a result of said event.

- 3. Engineer's Initial Review: Engineer will advise Owner regarding the Change Proposal, and consider any comments or response from Owner regarding the Change Proposal. If in its discretion Engineer concludes that additional supporting data is needed before conducting a full review and making a decision regarding the Change Proposal, then Engineer may request that Contractor submit such additional supporting data by a date specified by Engineer, prior to Engineer beginning its full review of the Change Proposal.
- 4. Engineer's Full Review and Action on the Change Proposal: Upon receipt of Contractor's supporting data (including any additional data requested by Engineer), Engineer will conduct a full review of each Change Proposal and, within 30 days after such receipt of the Contractor's supporting data, either approve the Change Proposal in whole, deny it in whole, or approve it in part and deny it in part. Such actions must be in writing, with a copy provided to Owner and Contractor. If Engineer does not take action on the Change

Page 51 of 73

Proposal within 30 days, then either Owner or Contractor may at any time thereafter submit a letter to the other party indicating that as a result of Engineer's inaction the Change Proposal is deemed denied, thereby commencing the time for appeal of the denial under Article 12.

- 5. *Binding Decision*: Engineer's decision is final and binding upon Owner and Contractor, unless Owner or Contractor appeals the decision by filing a Claim under Article 12.
- C. Resolution of Certain Change Proposals: If the Change Proposal does not involve the design (as set forth in the Drawings, Specifications, or otherwise), the acceptability of the Work, or other engineering or technical matters, then Engineer will notify the parties in writing that the Engineer is unable to resolve the Change Proposal. For purposes of further resolution of such a Change Proposal, such notice will be deemed a denial, and Contractor may choose to seek resolution under the terms of Article 12.
- D. Post-Completion: Contractor shall not submit any Change Proposals after Engineer issues a written recommendation of final payment pursuant to Paragraph 15.06.B.

11.10 Notification to Surety

A. If the provisions of any bond require notice to be given to a surety of any change affecting the general scope of the Work or the provisions of the Contract Documents (including, but not limited to, Contract Price or Contract Times), the giving of any such notice will be Contractor's responsibility. The amount of each applicable bond will be adjusted to reflect the effect of any such change.

ARTICLE 12-CLAIMS

12.01 Claims

- A. *Claims Process*: The following disputes between Owner and Contractor are subject to the Claims process set forth in this article:
 - 1. Appeals by Owner or Contractor of Engineer's decisions regarding Change Proposals;
 - Owner demands for adjustments in the Contract Price or Contract Times, or other relief under the Contract Documents;
 - Disputes that Engineer has been unable to address because they do not involve the design (as set forth in the Drawings, Specifications, or otherwise), the acceptability of the Work, or other engineering or technical matters; and
 - 4. Subject to the waiver provisions of Paragraph 15.07, any dispute arising after Engineer has issued a written recommendation of final payment pursuant to Paragraph 15.06.B.
- B. Submittal of Claim: The party submitting a Claim shall deliver it directly to the other party to the Contract promptly (but in no event later than 30 days) after the start of the event giving rise thereto; in the case of appeals regarding Change Proposals within 30 days of the decision under appeal. The party submitting the Claim shall also furnish a copy to the Engineer, for its information only. The responsibility to substantiate a Claim rests with the party making the Claim. In the case of a Claim by Contractor seeking an increase in the Contract Times or Contract Price, Contractor shall certify that the Claim is made in good faith, that the supporting data are accurate and complete, and that to the best of Contractor's knowledge

and belief the amount of time or money requested accurately reflects the full amount to which Contractor is entitled.

- C. Review and Resolution: The party receiving a Claim shall review it thoroughly, giving full consideration to its merits. The two parties shall seek to resolve the Claim through the exchange of information and direct negotiations. The parties may extend the time for resolving the Claim by mutual agreement. All actions taken on a Claim will be stated in writing and submitted to the other party, with a copy to Engineer.
- D. Mediation
 - At any time after initiation of a Claim, Owner and Contractor may mutually agree to mediation of the underlying dispute. The agreement to mediate will stay the Claim submittal and response process.
 - 2. If Owner and Contractor agree to mediation, then after 60 days from such agreement, either Owner or Contractor may unilaterally terminate the mediation process, and the Claim submittal and decision process will resume as of the date of the termination. If the mediation proceeds but is unsuccessful in resolving the dispute, the Claim submittal and decision process will resume as of the date of the mediation, as determined by the mediator.
 - 3. Owner and Contractor shall each pay one-half of the mediator's fees and costs.
- E. *Partial Approval*: If the party receiving a Claim approves the Claim in part and denies it in part, such action will be final and binding unless within 30 days of such action the other party invokes the procedure set forth in Article 17 for final resolution of disputes.
- F. Denial of Claim: If efforts to resolve a Claim are not successful, the party receiving the Claim may deny it by giving written notice of denial to the other party. If the receiving party does not take action on the Claim within 90 days, then either Owner or Contractor may at any time thereafter submit a letter to the other party indicating that as a result of the inaction, the Claim is deemed denied, thereby commencing the time for appeal of the denial. A denial of the Claim will be final and binding unless within 30 days of the denial the other party invokes the procedure set forth in Article 17 for the final resolution of disputes.
- G. Final and Binding Results: If the parties reach a mutual agreement regarding a Claim, whether through approval of the Claim, direct negotiations, mediation, or otherwise; or if a Claim is approved in part and denied in part, or denied in full, and such actions become final and binding; then the results of the agreement or action on the Claim will be incorporated in a Change Order or other written document to the extent they affect the Contract, including the Work, the Contract Times, or the Contract Price.

ARTICLE 13-COST OF THE WORK; ALLOWANCES; UNIT PRICE WORK

13.01 Cost of the Work

- A. Purposes for Determination of Cost of the Work: The term Cost of the Work means the sum of all costs necessary for the proper performance of the Work at issue, as further defined below. The provisions of this Paragraph 13.01 are used for two distinct purposes:
 - 1. To determine Cost of the Work when Cost of the Work is a component of the Contract Price, under cost-plus-fee, time-and-materials, or other cost-based terms; or

- 2. When needed to determine the value of a Change Order, Change Proposal, Claim, set-off, or other adjustment in Contract Price. When the value of any such adjustment is determined on the basis of Cost of the Work, Contractor is entitled only to those additional or incremental costs required because of the change in the Work or because of the event giving rise to the adjustment.
- B. Costs Included: Except as otherwise may be agreed to in writing by Owner, costs included in the Cost of the Work will be in amounts no higher than those commonly incurred in the locality of the Project, will not include any of the costs itemized in Paragraph 13.01.C, and will include only the following items:
 - 1. Payroll costs for employees in the direct employ of Contractor in the performance of the Work under schedules of job classifications agreed upon by Owner and Contractor in advance of the subject Work. Such employees include, without limitation, superintendents, foremen, safety managers, safety representatives, and other personnel employed full time on the Work. Payroll costs for employees not employed full time on the Work will be apportioned on the basis of their time spent on the Work. Payroll costs include, but are not limited to, salaries and wages plus the cost of fringe benefits, which include social security contributions, unemployment, excise, and payroll taxes, workers' compensation, health and retirement benefits, sick leave, and vacation and holiday pay applicable thereto. The expenses of performing Work outside of regular working hours, on Saturday, Sunday, or legal holidays, will be included in the above to the extent authorized by Owner.
 - 2. Cost of all materials and equipment furnished and incorporated in the Work, including costs of transportation and storage thereof, and Suppliers' field services required in connection therewith. All cash discounts accrue to Contractor unless Owner deposits funds with Contractor with which to make payments, in which case the cash discounts will accrue to Owner. All trade discounts, rebates, and refunds and returns from sale of surplus materials and equipment will accrue to Owner, and Contractor shall make provisions so that they may be obtained.
 - 3. Payments made by Contractor to Subcontractors for Work performed by Subcontractors. If required by Owner, Contractor shall obtain competitive bids from subcontractors acceptable to Owner and Contractor and shall deliver such bids to Owner, which will then determine, with the advice of Engineer, which bids, if any, will be acceptable. If any subcontract provides that the Subcontractor is to be paid on the basis of Cost of the Work plus a fee, the Subcontractor's Cost of the Work and fee will be determined in the same manner as Contractor's Cost of the Work and fee as provided in this Paragraph 13.01.
 - Costs of special consultants (including but not limited to engineers, architects, testing laboratories, surveyors, attorneys, and accountants) employed or retained for services specifically related to the Work.
 - 5. Other costs consisting of the following:
 - a. The proportion of necessary transportation, travel, and subsistence expenses of Contractor's employees incurred in discharge of duties connected with the Work.
 - b. Cost, including transportation and maintenance, of all materials, supplies, equipment, machinery, appliances, office, and temporary facilities at the Site, which are

consumed in the performance of the Work, and cost, less market value, of such items used but not consumed which remain the property of Contractor.

- In establishing included costs for materials such as scaffolding, plating, or sheeting, consideration will be given to the actual or the estimated life of the material for use on other projects; or rental rates may be established on the basis of purchase or salvage value of such items, whichever is less. Contractor will not be eligible for compensation for such items in an amount that exceeds the purchase cost of such item.
- c. Construction Equipment Rental
 - 1) Rentals of all construction equipment and machinery, and the parts thereof, in accordance with rental agreements approved by Owner as to price (including any surcharge or special rates applicable to overtime use of the construction equipment or machinery), and the costs of transportation, loading, unloading, assembly, dismantling, and removal thereof. All such costs will be in accordance with the terms of said rental agreements. The rental of any such equipment, machinery, or parts must cease when the use thereof is no longer necessary for the Work.
 - 2) Costs for equipment and machinery owned by Contractor or a Contractor-related entity will be paid at a rate shown for such equipment in the equipment rental rate book specified in the Supplementary Conditions. An hourly rate will be computed by dividing the monthly rates by 176. These computed rates will include all operating costs.
 - 3) With respect to Work that is the result of a Change Order, Change Proposal, Claim, set-off, or other adjustment in Contract Price ("changed Work"), included costs will be based on the time the equipment or machinery is in use on the changed Work and the costs of transportation, loading, unloading, assembly, dismantling, and removal when directly attributable to the changed Work. The cost of any such equipment or machinery, or parts thereof, must cease to accrue when the use thereof is no longer necessary for the changed Work.
- d. Sales, consumer, use, and other similar taxes related to the Work, and for which Contractor is liable, as imposed by Laws and Regulations.
- e. Deposits lost for causes other than negligence of Contractor, any Subcontractor, or anyone directly or indirectly employed by any of them or for whose acts any of them may be liable, and royalty payments and fees for permits and licenses.
- f. Losses and damages (and related expenses) caused by damage to the Work, not compensated by insurance or otherwise, sustained by Contractor in connection with the performance of the Work (except losses and damages within the deductible amounts of builder's risk or other property insurance established in accordance with Paragraph 6.04), provided such losses and damages have resulted from causes other than the negligence of Contractor, any Subcontractor, or anyone directly or indirectly employed by any of them or for whose acts any of them may be liable. Such losses include settlements made with the written consent and approval of Owner. No such losses, damages, and expenses will be included in the Cost of the Work for the purpose of determining Contractor's fee.

- g. The cost of utilities, fuel, and sanitary facilities at the Site.
- h. Minor expenses such as communication service at the Site, express and courier services, and similar petty cash items in connection with the Work.
- i. The costs of premiums for all bonds and insurance that Contractor is required by the Contract Documents to purchase and maintain.
- j. The cost of specific consideration for the indemnifications set forth in paragraph 7.18.
- k. The cost of compliance with current local, state and federal safety regulations.
- C. Costs Excluded: The term Cost of the Work does not include any of the following items:
 - Payroll costs and other compensation of Contractor's officers, executives, principals, general managers, engineers, architects, estimators, attorneys, auditors, accountants, purchasing and contracting agents, expediters, timekeepers, clerks, and other personnel employed by Contractor, whether at the Site or in Contractor's principal or branch office for general administration of the Work and not specifically included in the agreed upon schedule of job classifications referred to in Paragraph 13.01.B.1 or specifically covered by Paragraph 13.01.B.4. The payroll costs and other compensation excluded here are to be considered administrative costs covered by the Contractor's fee.
 - 2. The cost of purchasing, renting, or furnishing small tools and hand tools.
 - Expenses of Contractor's principal and branch offices other than Contractor's office at the Site.
 - 4. Any part of Contractor's capital expenses, including interest on Contractor's capital employed for the Work and charges against Contractor for delinquent payments.
 - 5. Costs due to the negligence of Contractor, any Subcontractor, or anyone directly or indirectly employed by any of them or for whose acts any of them may be liable, including but not limited to, the correction of defective Work, disposal of materials or equipment wrongly supplied, and making good any damage to property.
 - 6. Expenses incurred in preparing and advancing Claims.
 - 7. Other overhead or general expense costs of any kind and the costs of any item not specifically and expressly included in Paragraph 13.01.B.
- D. Contractor's Fee
 - 1. When the Work as a whole is performed on the basis of cost-plus-a-fee, then:
 - a. Contractor's fee for the Work set forth in the Contract Documents as of the Effective Date of the Contract will be determined as set forth in the Agreement.
 - b. for any Work covered by a Change Order, Change Proposal, Claim, set-off, or other adjustment in Contract Price on the basis of Cost of the Work, Contractor's fee will be determined as follows:
 - When the fee for the Work as a whole is a percentage of the Cost of the Work, the fee will automatically adjust as the Cost of the Work changes.
 - 2) When the fee for the Work as a whole is a fixed fee, the fee for any additions or deletions will be determined in accordance with Paragraph 11.07.C.2.

- 2. When the Work as a whole is performed on the basis of a stipulated sum, or any other basis other than cost-plus-a-fee, then Contractor's fee for any Work covered by a Change Order, Change Proposal, Claim, set-off, or other adjustment in Contract Price on the basis of Cost of the Work will be determined in accordance with Paragraph 11.07.C.2.
- E. Documentation and Audit: Whenever the Cost of the Work for any purpose is to be determined pursuant to this Article 13, Contractor and pertinent Subcontractors will establish and maintain records of the costs in accordance with generally accepted accounting practices. Subject to prior written notice, Owner will be afforded reasonable access, during normal business hours, to all Contractor's accounts, records, books, correspondence, instructions, drawings, receipts, vouchers, memoranda, and similar data relating to the Cost of the Work and Contractor's fee. Contractor shall preserve all such documents for a period of three years after the final payment by Owner. Pertinent Subcontractors will afford such access to Owner, and preserve such documents, to the same extent required of Contractor.

13.02 Allowances

- A. It is understood that Contractor has included in the Contract Price all allowances so named in the Contract Documents and shall cause the Work so covered to be performed for such sums and by such persons or entities as may be acceptable to Owner and Engineer.
- B. Cash Allowances: Contractor agrees that:
 - the cash allowances include the cost to Contractor (less any applicable trade discounts) of materials and equipment required by the allowances to be delivered at the Site, and all applicable taxes; and
 - Contractor's costs for unloading and handling on the Site, labor, installation, overhead, profit, and other expenses contemplated for the cash allowances have been included in the Contract Price and not in the allowances, and no demand for additional payment for any of the foregoing will be valid.
- C. Owner's Contingency Allowance: Contractor agrees that an Owner's contingency allowance, if any, is for the sole use of Owner to cover unanticipated costs.
- D. Prior to final payment, an appropriate Change Order will be issued as recommended by Engineer to reflect actual amounts due Contractor for Work covered by allowances, and the Contract Price will be correspondingly adjusted.

13.03 Unit Price Work

- A. Where the Contract Documents provide that all or part of the Work is to be Unit Price Work, initially the Contract Price will be deemed to include for all Unit Price Work an amount equal to the sum of the unit price for each separately identified item of Unit Price Work times the estimated quantity of each item as indicated in the Agreement.
- B. The estimated quantities of items of Unit Price Work are not guaranteed and are solely for the purpose of comparison of Bids and determining an initial Contract Price. Payments to Contractor for Unit Price Work will be based on actual quantities.
- C. Each unit price will be deemed to include an amount considered by Contractor to be adequate to cover Contractor's overhead and profit for each separately identified item.
- D. Engineer will determine the actual quantities and classifications of Unit Price Work performed by Contractor. Engineer will review with Contractor the Engineer's preliminary

Copyright[©] 2018 National Society of Professional Engineers, American Council of Engineering Companies,

and American Society of Civil Engineers. All rights reserved.

Page 57 of 73

EJCDC® C-700, Standard General Conditions of the Construction Contract.

determinations on such matters before rendering a written decision thereon (by recommendation of an Application for Payment or otherwise). Engineer's written decision thereon will be final and binding (except as modified by Engineer to reflect changed factual conditions or more accurate data) upon Owner and Contractor, and the final adjustment of Contract Price will be set forth in a Change Order, subject to the provisions of the following paragraph.

- E. Adjustments in Unit Price
 - 1. Contractor or Owner shall be entitled to an adjustment in the unit price with respect to an item of Unit Price Work if:
 - a. the quantity of the item of Unit Price Work performed by Contractor differs materially and significantly from the estimated quantity of such item indicated in the Agreement; and
 - b. Contractor's unit costs to perform the item of Unit Price Work have changed materially and significantly as a result of the quantity change.
 - 2. The adjustment in unit price will account for and be coordinated with any related changes in quantities of other items of Work, and in Contractor's costs to perform such other Work, such that the resulting overall change in Contract Price is equitable to Owner and Contractor.
 - 3. Adjusted unit prices will apply to all units of that item.
 - E. If Owner objects to Contractor's quoted adjustment in Unit Price Work, Owner may assign such Work to its own forces or another contractor.
 - F. Owner reserves the right to delete any Unit Price Work without financial penalty incurred from Contractor.

ARTICLE 14-TESTS AND INSPECTIONS; CORRECTION, REMOVAL, OR ACCEPTANCE OF DEFECTIVE WORK

- 14.01 Access to Work
 - A. Owner, Engineer, their consultants and other representatives and personnel of Owner, independent testing laboratories, and authorities having jurisdiction have access to the Site and the Work at reasonable times for their observation, inspection, and testing. Contractor shall provide them proper and safe conditions for such access and advise them of Contractor's safety procedures and programs so that they may comply with such procedures and programs as applicable.
- 14.02 Tests, Inspections, and Approvals
 - A. Contractor shall give Engineer timely notice of readiness of the Work (or specific parts thereof) for all required inspections and tests, and shall cooperate with inspection and testing personnel to facilitate required inspections and tests. <u>Contractor shall reimburse Owner for the charges of Engineer and Engineer's Consultants for all costs due to work not being ready for tests and/or inspections when the Contractor has notified Engineer that work is ready for tests and/or inspections. Contractor shall reimburse Owner for all failed tests and subsequent retests. Reimbursement for the charges shall be a deduction from the Contractor's Partial Payment(s).</u>

B. Owner shall retain and pay for the services of an independent inspector, testing laboratory, or other qualified individual or entity to perform all inspections and tests expressly required by the Contract Documents to be furnished and paid for by Owner, except that costs incurred in connection with tests or inspections of covered Work will be governed by the provisions of Paragraph 14.05 The Contractor shall obtain and employ an independent testing laboratory to provide the services specified herein and shall include all costs for these services in the contract price. The Contractor shall submit for approval by the Owner and Engineer, the independent testing laboratory, including qualifications.

Copies of all testing agency invoices submitted to the Contractor for payment shall be forwarded with the Contractor's request for partial payment. Invoices shall be submitted routinely throughout the project and shall be pertinent to the partial payment period under consideration. Testing agency invoices in excess of sixty (60) days old shall not be considered for payment, Invoices shall clearly indicate if the invoiced testing cost is a result of retests required due to the Contractor's failure to achieve specified requirements. The cost of retesting due to the test failure will be borne by the Contractor. Payment to the Contractor for testing shall not be made without the required itemized invoicing.

- C. If Laws or Regulations of any public body having jurisdiction require any Work (or part thereof) specifically to be inspected, tested, or approved by an employee or other representative of such public body, Contractor shall assume full responsibility for arranging and obtaining such inspections, tests, or approvals, pay all costs in connection therewith, and furnish Engineer the required certificates of inspection or approval.
- D. Contractor shall be responsible for arranging, obtaining, and paying for all inspections and tests required:
 - 1. by the Contract Documents, unless the Contract Documents expressly allocate responsibility for a specific inspection or test to Owner;
 - to attain Owner's and Engineer's acceptance of materials or equipment to be incorporated in the Work;
 - 3. by manufacturers of equipment furnished under the Contract Documents;
 - 4. for testing, adjusting, and balancing of mechanical, electrical, and other equipment to be incorporated into the Work; and
 - 5. for acceptance of materials, mix designs, or equipment submitted for approval prior to Contractor's purchase thereof for incorporation in the Work.

Such inspections and tests will be performed by independent inspectors, testing laboratories, or other gualified individuals or entities acceptable to Owner and Engineer.

- E. If the Contract Documents require the Work (or part thereof) to be approved by Owner, Engineer, or another designated individual or entity, then Contractor shall assume full responsibility for arranging and obtaining such approvals.
- F. If any Work (or the work of others) that is to be inspected, tested, or approved is covered by Contractor without written concurrence of Engineer, Contractor shall, if requested by Engineer, uncover such Work for observation. Such uncovering will be at Contractor's expense unless Contractor had given Engineer timely notice of Contractor's intention to cover the same and Engineer had not acted with reasonable promptness in response to such notice.

14.03 Defective Work

- A. Contractor's Obligation: It is Contractor's obligation to assure that the Work is not defective.
- B. Engineer's Authority: Engineer has the authority to determine whether Work is defective, and to reject defective Work.
- C. Notice of Defects: Prompt written notice of all defective Work of which Owner or Engineer has actual knowledge will be given to Contractor.
- D. Correction, or Removal and Replacement: Promptly after receipt of written notice of defective Work, Contractor shall correct all such defective Work, whether or not fabricated, installed, or completed, or, if Engineer has rejected the defective Work, remove it from the Project and replace it with Work that is not defective.
- E. Preservation of Warranties: When correcting defective Work, Contractor shall take no action that would void or otherwise impair Owner's special warranty and guarantee, if any, on said Work.
- F. Costs and Damages: In addition to its correction, removal, and replacement obligations with respect to defective Work, Contractor shall pay all claims, costs, losses, and damages arising out of or relating to defective Work, including but not limited to the cost of the inspection, testing, correction, removal, replacement, or reconstruction of such defective Work, fines levied against Owner by governmental authorities because the Work is defective, and the costs of repair or replacement of work of others resulting from defective Work. Prior to final payment, if Owner and Contractor are unable to agree as to the measure of such claims, costs, losses, and damages resulting from defective Work, then Owner may impose a reasonable set-off against payments due under Article 15.
- 14.04 Acceptance of Defective Work
 - A. If, instead of requiring correction or removal and replacement of defective Work, Owner prefers to accept it, Owner may do so (subject, if such acceptance occurs prior to final payment, to Engineer's confirmation that such acceptance is in general accord with the design intent and applicable engineering principles, and will not endanger public safety). Contractor shall pay all claims, costs, losses, and damages attributable to Owner's evaluation of and determination to accept such defective Work (such costs to be approved by Engineer as to reasonableness), and for the diminished value of the Work to the extent not otherwise paid by Contractor. If any such acceptance occurs prior to final payment, the necessary revisions in the Contract Documents with respect to the Work will be incorporated in a Change Order. If the parties are unable to agree as to the decrease in the Contract Price, reflecting the diminished value of Work so accepted, then Owner may impose a reasonable set-off against payments due under Article 15. If the acceptance of defective Work occurs after final payment, Contractor shall pay an appropriate amount to Owner.

14.05 Uncovering Work

- A. Engineer has the authority to require additional inspection or testing of the Work, whether or not the Work is fabricated, installed, or completed.
- B. If any Work is covered contrary to the written request of Engineer, then Contractor shall, if requested by Engineer, uncover such Work for Engineer's observation, and then replace the covering, all at Contractor's expense.

- C. If Engineer considers it necessary or advisable that covered Work be observed by Engineer or inspected or tested by others, then Contractor, at Engineer's request, shall uncover, expose, or otherwise make available for observation, inspection, or testing as Engineer may require, that portion of the Work in question, and provide all necessary labor, material, and equipment.
 - If it is found that the uncovered Work is defective, Contractor shall be responsible for all claims, costs, losses, and damages arising out of or relating to such uncovering, exposure, observation, inspection, and testing, and of satisfactory replacement or reconstruction (including but not limited to all costs of repair or replacement of work of others); and pending Contractor's full discharge of this responsibility the Owner shall be entitled to impose a reasonable set-off against payments due under Article 15.
 - 2. If the uncovered Work is not found to be defective, Contractor shall be allowed an increase in the Contract Price or an extension of the Contract Times, directly attributable to such uncovering, exposure, observation, inspection, testing, replacement, and reconstruction. If the parties are unable to agree as to the amount or extent thereof, then Contractor may submit a Change Proposal within 30 days of the determination that the Work is not defective.

14.06 Owner May Stop the Work

A. If the Work is defective, or Contractor fails to supply sufficient skilled workers or suitable materials or equipment, or fails to perform the Work in such a way that the completed Work will conform to the Contract Documents, then Owner may order Contractor to stop the Work, or any portion thereof, until the cause for such order has been eliminated; however, this right of Owner to stop the Work will not give rise to any duty on the part of Owner to exercise this right for the benefit of Contractor, any Subcontractor, any Supplier, any other individual or entity, or any surety for, or employee or agent of any of them.

14.07 Owner May Correct Defective Work

- A. If Contractor fails within a reasonable time after written notice from Engineer to correct defective Work, or to remove and replace defective Work as required by Engineer, then Owner may, after 7 days' written notice to Contractor, correct or remedy any such deficiency.
- B. In exercising the rights and remedies under this Paragraph 14.07, Owner shall proceed expeditiously. In connection with such corrective or remedial action, Owner may exclude Contractor from all or part of the Site, take possession of all or part of the Work and suspend Contractor's services related thereto, and incorporate in the Work all materials and equipment stored at the Site or for which Owner has paid Contractor but which are stored elsewhere. Contractor shall allow Owner, Owner's representatives, agents and employees, Owner's other contractors, and Engineer and Engineer's consultants access to the Site to enable Owner to exercise the rights and remedies under this paragraph.
- C. All claims, costs, losses, and damages incurred or sustained by Owner in exercising the rights and remedies under this Paragraph 14.07 will be charged against Contractor as set-offs against payments due under Article 15. Such claims, costs, losses and damages will include but not be limited to all costs of repair, or replacement of work of others destroyed or damaged by correction, removal, or replacement of Contractor's defective Work.

D. Contractor shall not be allowed an extension of the Contract Times because of any delay in the performance of the Work attributable to the exercise by Owner of Owner's rights and remedies under this Paragraph 14.07.

ARTICLE 15-PAYMENTS TO CONTRACTOR; SET-OFFS; COMPLETION; CORRECTION PERIOD

15.01 Progress Payments

- A. Basis for Progress Payments: The Schedule of Values established as provided in Article 2 will serve as the basis for progress payments and will be incorporated into a form of Application for Payment acceptable to Engineer. Progress payments for Unit Price Work will be based on the number of units completed during the pay period, as determined under the provisions of Paragraph 13.03. Progress payments for cost-based Work will be based on Cost of the Work completed by Contractor during the pay period.
- B. Applications for Payments
 - At least 20 days before the date established in the Agreement for each progress payment (but not more often than once a month), Contractor shall submit to Engineer for review an Application for Payment filled out and signed by Contractor covering the Work completed as of the date of the Application and accompanied by such supporting documentation as is required by the Contract Documents.
 - 2. If payment is requested on the basis of materials and equipment not incorporated in the Work but delivered and suitably stored at the Site or at another location agreed to in writing, the Application for Payment must also be accompanied by: (a) a bill of sale, invoice, copies of subcontract or purchase order payments, or other documentation establishing full payment by Contractor for the materials and equipment; (b) at Owner's request, documentation warranting that Owner has received the materials and equipment free and clear of all Liens; and (c) evidence that the materials and equipment are covered by appropriate property insurance, a warehouse bond, or other arrangements to protect Owner's interest therein, all of which must be satisfactory to Owner.
 - 3. Beginning with the second Application for Payment, each Application <u>shall include an</u> <u>affidavit of Contractor certifying that Contractor has disbursed to all subcontractors and</u> <u>suppliers having an interest in the contract their pro rata shares of the payment out of</u> <u>previous progress payments received by Contractor for all work completed and materials</u> <u>furnished in the previous work period, less a retainage withheld by Contractor pursuant</u> <u>to an agreement with a subcontractor</u>. Within thirty (30) days of receipt for the final progress payment or any other payments received thereafter except the final payment, <u>Contractor shall pay all subcontractors and suppliers having an interest in the contract their pro rata shares of the payment for all work completed and materials <u>furnished</u>. must include an affidavit of Contractor stating that all previous progress payments received by Contractor's legitimate obligations associated with prior Applications for Payment.</u></u>
 - The amount of retainage with respect to progress payments will be as stipulated in the Agreement.

C. Review of Applications

- Engineer will, within 10 days after receipt of each Application for Payment, including each
 resubmittal, either indicate in writing a recommendation of payment and present the
 Application to Owner, or return the Application to Contractor indicating in writing
 Engineer's reasons for refusing to recommend payment. In the latter case, Contractor
 may make the necessary corrections and resubmit the Application <u>or direct Engineer to
 present the Application to Owner with Engineer's recommendation of partial payment</u>.
- 2. Engineer's recommendation of any payment requested in an Application for Payment will constitute a representation by Engineer to Owner, based on Engineer's observations of the executed Work as an experienced and qualified design professional, and on Engineer's review of the Application for Payment and the accompanying data and schedules and the <u>Contract Documents</u>, that to the best of Engineer's knowledge, information and belief:
 - a. the Work has progressed to the point indicated;
 - b. the quality of the Work is generally in accordance with the Contract Documents (subject to an evaluation of the Work as a functioning whole prior to or upon Substantial Completion, the results of any subsequent tests called for in the Contract Documents, a final determination of quantities and classifications for Unit Price Work under Paragraph 13.03, and any other qualifications stated in the recommendation); and
 - c. the conditions precedent to Contractor's being entitled to such payment appear to have been fulfilled in so far as it is Engineer's responsibility to observe the Work.
 - d. Contractor's other obligations under the Contract Documents have been fulfilled.
- 3. By recommending any such payment Engineer will not thereby be deemed to have represented that:
 - a. inspections made to check the quality or the quantity of the Work as it has been performed have been exhaustive, extended to every aspect of the Work in progress, or involved detailed inspections of the Work beyond the responsibilities specifically assigned to Engineer in the Contract; or
 - there may not be other matters or issues between the parties that might entitle Contractor to be paid additionally by Owner or entitle Owner to withhold payment to Contractor.
- 4. Neither Engineer's review of Contractor's Work for the purposes of recommending payments nor Engineer's recommendation of any payment, including final payment, will impose responsibility on Engineer:
 - a. to supervise, direct, or control the Work;
 - b. for the means, methods, techniques, sequences, or procedures of construction, or the safety precautions and programs incident thereto;
 - c. for Contractor's failure to comply with Laws and Regulations applicable to Contractor's performance of the Work;
 - d. to make any examination to ascertain how or for what purposes Contractor has used the money paid by Owner; or

EJCDC[®] C-700, Standard General Conditions of the Construction Contract. Copyright[®] 2018 National Society of Professional Engineers, American Council of Engineering Companies,

- e. to determine that title to any of the Work, materials, or equipment has passed to Owner free and clear of any Liens.
- 5. Engineer may refuse to recommend the whole or any part of any payment if, in Engineer's opinion, it would be incorrect to make the representations to Owner stated in Paragraph 15.01.C.2.
- 6. Engineer will recommend reductions in payment (set-offs) necessary in Engineer's opinion to protect Owner from loss because:
 - a. the Work is defective, requiring correction or replacement;
 - b. the Contract Price has been reduced by Change Orders;
 - c. Owner has been required to correct defective Work in accordance with Paragraph 14.07, or has accepted defective Work pursuant to Paragraph 14.04;
 - d. Owner has been required to remove or remediate a Hazardous Environmental Condition for which Contractor is responsible; or
 - e. Engineer has actual knowledge of the occurrence of any of the events that would constitute a default by Contractor and therefore justify termination for cause under the Contract Documents.
- D. Payment Becomes Due
 - Ten days after presentation of the Application for Payment to Owner with Engineer's recommendation, the amount recommended (subject to any Owner set-offs) will become due, and when due will be paid by Owner to Contractor.
- E. Reductions in Payment by Owner
 - 1. In addition to any reductions in payment (set-offs) recommended by Engineer, Owner is entitled to impose a set-off against payment based on any of the following:
 - a. Claims have been made against Owner based on Contractor's conduct in the performance or furnishing of the Work, or Owner has incurred costs, losses, or damages resulting from Contractor's conduct in the performance or furnishing of the Work, including but not limited to claims, costs, losses, or damages from workplace injuries, adjacent property damage, non-compliance with Laws and Regulations, and patent infringement;
 - Contractor has failed to take reasonable and customary measures to avoid damage, delay, disruption, and interference with other work at or adjacent to the Site;
 - c. Contractor has failed to provide and maintain required bonds or insurance;
 - d. Owner has been required to remove or remediate a Hazardous Environmental Condition for which Contractor is responsible;
 - e. Owner has incurred extra charges or engineering costs related to submittal reviews, evaluations of proposed substitutes, tests and inspections, or return visits to manufacturing or assembly facilities;
 - f. The Work is defective, requiring correction or replacement;
 - g. Owner has been required to correct defective Work in accordance with Paragraph 14.07, or has accepted defective Work pursuant to Paragraph 14.04;

EJCDC® C-700, Standard General Conditions of the Construction Contract.

Copyright[®] 2018 National Society of Professional Engineers, American Council of Engineering Companies, and American Society of Civil Engineers. All rights reserved.

- h. The Contract Price has been reduced by Change Orders;
- An event has occurred that would constitute a default by Contractor and therefore justify a termination for cause;
- j. Liquidated or other damages have accrued as a result of Contractor's failure to achieve Milestones, Substantial Completion, or final completion of the Work;
- k. Liens have been filed in connection with the Work, except where Contractor has delivered a specific bond satisfactory to Owner to secure the satisfaction and discharge of such Liens; or
- I. Other items entitle Owner to a set-off against the amount recommended.
- 2. If Owner imposes any set-off against payment, whether based on its own knowledge or on the written recommendations of Engineer, Owner will give Contractor immediate written notice (with a copy to Engineer) stating the reasons for such action and the specific amount of the reduction, and promptly pay Contractor any amount remaining after deduction of the amount so withheld. Owner shall promptly pay Contractor the amount so withheld, or any adjustment thereto agreed to by Owner and Contractor, if Contractor remedies the reasons for such action. The reduction imposed will be binding on Contractor unless it duly submits a Change Proposal contesting the reduction.
- 3. Upon a subsequent determination that Owner's refusal of payment was not justified, the amount wrongfully withheld will be treated as an amount due as determined by Paragraph 15.01.D.1 and subject to interest as provided in the Agreement.
- 15.02 Contractor's Warranty of Title
 - A. Contractor warrants and guarantees that title to all Work, materials, and equipment furnished under the Contract will pass to Owner free and clear of (1) all Liens and other title defects, and (2) all patent, licensing, copyright, or royalty obligations, no later than 7 days after the time of payment by Owner.
- 15.03 Substantial Completion
 - A. When Contractor considers the entire Work ready for its intended use Contractor shall notify Owner and Engineer in writing that the entire Work is substantially complete and request that Engineer issue a certificate of Substantial Completion. Contractor shall at the same time submit to Owner and Engineer an initial draft of punch list items to be completed or corrected before final payment.
 - B. Promptly after Contractor's notification, Owner, Contractor, and Engineer shall make an inspection of the Work to determine the status of completion. If Engineer does not consider the Work substantially complete, Engineer will notify Contractor in writing giving the reasons therefor.
 - C. If Engineer considers the Work substantially complete, Engineer will deliver to Owner a preliminary certificate of Substantial Completion which will fix the date of Substantial Completion. Engineer shall attach to the certificate a punch list of items to be completed or corrected before final payment. Owner shall have 7 days after receipt of the preliminary certificate during which to make written objection to Engineer as to any provisions of the certificate or attached punch list. If, after considering the objections to the provisions of the preliminary certificate, Engineer concludes that the Work is not substantially complete,

Engineer will, within 14 days after submission of the preliminary certificate to Owner, notify Contractor in writing that the Work is not substantially complete, stating the reasons therefor. If Owner does not object to the provisions of the certificate, or if despite consideration of Owner's objections Engineer concludes that the Work is substantially complete, then Engineer will, within said 14 days, execute and deliver to Owner and Contractor a final certificate of Substantial Completion (with a revised punch list of items to be completed or corrected) reflecting such changes from the preliminary certificate as Engineer believes justified after consideration of any objections from Owner.

- D. At the time of receipt of the preliminary certificate of Substantial Completion, Owner and Contractor will confer regarding Owner's use or occupancy of the Work following Substantial Completion, review the builder's risk insurance policy with respect to the end of the builder's risk coverage, and confirm the transition to coverage of the Work under a permanent property insurance policy held by Owner. Unless Owner and Contractor agree otherwise in writing, Owner shall bear responsibility for security, operation, protection of the Work, property insurance, maintenance, heat, and utilities upon Owner's use or occupancy of the Work.
- E. After Substantial Completion the Contractor shall promptly begin work on the punch list of items to be completed or corrected prior to final payment. In appropriate cases Contractor may submit monthly Applications for Payment for completed punch list items, following the progress payment procedures set forth above.
- F. Owner shall have the right to exclude Contractor from the Site after the date of Substantial Completion subject to allowing Contractor reasonable access to remove its property and complete or correct items on the punch list.

15.04 Partial Use or Occupancy

- A. Prior to Substantial Completion of all the Work, Owner may use or occupy any substantially completed part of the Work which has specifically been identified in the Contract Documents, or which Owner, Engineer, and Contractor agree constitutes a separately functioning and usable part of the Work that can be used by Owner for its intended purpose without significant interference with Contractor's performance of the remainder of the Work, subject to the following conditions:
 - At any time, Owner may request in writing that Contractor permit Owner to use or occupy any such part of the Work that Owner believes to be substantially complete. If and when Contractor agrees that such part of the Work is substantially complete, Contractor, Owner, and Engineer will follow the procedures of Paragraph 15.03.A through 15.03.E for that part of the Work.
 - At any time, Contractor may notify Owner and Engineer in writing that Contractor considers any such part of the Work substantially complete and request Engineer to issue a certificate of Substantial Completion for that part of the Work.
 - 3. Within a reasonable time after either such request, Owner, Contractor, and Engineer shall make an inspection of that part of the Work to determine its status of completion. If Engineer does not consider that part of the Work to be substantially complete, Engineer will notify Owner and Contractor in writing giving the reasons therefor. If Engineer considers that part of the Work to be substantially complete, the provisions of

Paragraph 15.03 will apply with respect to certification of Substantial Completion of that part of the Work and the division of responsibility in respect thereof and access thereto.

4. No use or occupancy or separate operation of part of the Work may occur prior to compliance with the requirements of Paragraph 6.04 regarding builder's risk or other property insurance.

15.05 Final Inspection

A. Upon written notice from Contractor that the entire Work or an agreed portion thereof is complete, Engineer will promptly make a final inspection with Owner and Contractor and will notify Contractor in writing of all particulars in which this inspection reveals that the Work, or agreed portion thereof, is incomplete or defective. Contractor shall immediately take such measures as are necessary to complete such Work or remedy such deficiencies.

15.06 Final Payment

- A. Application for Payment
 - After Contractor has, in the opinion of Engineer, satisfactorily completed all corrections identified during the final inspection and has delivered, in accordance with the Contract Documents, all maintenance and operating instructions, schedules, guarantees, bonds, certificates or other evidence of insurance, certificates of inspection, annotated record documents (as provided in Paragraph 7.12), and other documents, Contractor may make application for final payment.
 - The final Application for Payment must be accompanied (except as previously delivered) by:
 - a. all documentation called for in the Contract Documents;
 - b. consent of the surety, if any, to final payment;
 - c. satisfactory evidence that all title issues have been resolved such that title to all Work, materials, and equipment has passed to Owner free and clear of any Liens or other title defects, or will so pass upon final payment.
 - d. a list of all duly pending Change Proposals and Claims; and
 - e. complete and legally effective releases or waivers (satisfactory to Owner) of all Lien rights arising out of the Work, and of Liens filed in connection with the Work.
 - 3. In lieu of the releases or waivers of Liens specified in Paragraph 15.06.A.2 and as approved by Owner, Contractor may furnish receipts or releases in full and an affidavit of Contractor that: (a) the releases and receipts include all labor, services, material, and equipment for which a Lien could be filed; and (b) all payrolls, material and equipment bills, and other indebtedness connected with the Work for which Owner might in any way be responsible, or which might in any way result in liens or other burdens on Owner's property, have been paid or otherwise satisfied. If any Subcontractor or Supplier fails to furnish such a release or receipt in full, Contractor may furnish a bond or other collateral satisfactory to Owner to indemnify Owner against any Lien, or Owner at its option may issue joint checks payable to Contractor and specified Subcontractors and Suppliers.
- B. Engineer's Review of Final Application and Recommendation of Payment: If, on the basis of Engineer's observation of the Work during construction and final inspection, and Engineer's

review of the final Application for Payment and accompanying documentation as required by the Contract Documents, Engineer is satisfied that the Work has been completed and Contractor's other obligations under the Contract have been fulfilled, Engineer will, within 10 days after receipt of the final Application for Payment, indicate in writing Engineer's recommendation of final payment and present the final Application for Payment to Owner for payment. Such recommendation will account for any set-offs against payment that are necessary in Engineer's opinion to protect Owner from loss for the reasons stated above with respect to progress payments. Otherwise, Engineer will return the Application for Payment to Contractor, indicating in writing the reasons for refusing to recommend final payment, in which case Contractor shall make the necessary corrections and resubmit the Application for Payment The acceptance of Final Payment by Contractor designated and identified by Engineer as Final Payment shall be and shall operate as a release to Owner of all claims and all liability to Contractor other than claims in stated amounts as may be specifically excepted by Contractor for all things done or finished in connection it the Work and for every act of Owner and others relating to or arising out of the Work. Any payment, however, final or otherwise, shall not release Contractor or Contractor's Surety(ies) from any obligations under the Contract Documents or Bonds.

- C. Notice of Acceptability: In support of its recommendation of payment of the final Application for Payment, Engineer will also give written notice to Owner and Contractor that the Work is acceptable, subject to stated limitations in the notice and to the provisions of Paragraph 15.07.
- D. Completion of Work: The Work is complete (subject to surviving obligations) when it is ready for final payment as established by the Engineer's written recommendation of final payment and issuance of notice of the acceptability of the Work.
- E. Final Payment Becomes Due: Upon receipt from Engineer of the final Application for Payment and accompanying documentation, Owner shall set off against the amount recommended by Engineer for final payment any further sum to which Owner is entitled, including but not limited to set-offs for liquidated damages and set-offs allowed under the provisions of this Contract with respect to progress payments. Owner shall pay the resulting balance due to Contractor within 30 days of Owner's receipt of the final Application for Payment from Engineer.

15.07 Waiver of Claims

- A. By making final payment, Owner waives its claim or right to liquidated damages or other damages for late completion by Contractor, except as set forth in an outstanding Claim, appeal under the provisions of Article 17, set-off, or express reservation of rights by Owner. Owner reserves all other claims or rights after final payment.
- B. The acceptance of final payment by Contractor will constitute a waiver by Contractor of all claims and rights against Owner other than those pending matters that have been duly submitted as a Claim, or appealed under the provisions of Article 17.

15.08 Correction Period

A. If within one year after the date of Substantial Completion (or such longer period of time as may be prescribed by the Supplementary Conditions or the terms of any applicable special guarantee required by the Contract Documents), Owner gives Contractor written notice that any Work has been found to be defective, or that Contractor's repair of any damages to the Site or adjacent areas has been found to be defective, then after receipt of such notice of defect Contractor shall promptly, without cost to Owner and in accordance with Owner's written instructions:

- 1. correct the defective repairs to the Site or such adjacent areas;
- 2. correct such defective Work;
- remove the defective Work from the Project and replace it with Work that is not defective, if the defective Work has been rejected by Owner, and
- satisfactorily correct or repair or remove and replace any damage to other Work, to the work of others, or to other land or areas resulting from the corrective measures.
- B. Owner shall give any such notice of defect within 60 days of the discovery that such Work or repairs is defective. If such notice is given within such 60 days but after the end of the correction period, the notice will be deemed a notice of defective Work under Paragraph 7.17.B.
- C. If, after receipt of a notice of defect within 60 days and within the correction period, Contractor does not promptly comply with the terms of Owner's written instructions, or in an emergency where delay would cause serious risk of loss or damage, Owner may have the defective Work corrected or repaired or may have the rejected Work removed and replaced. Contractor shall pay all costs, losses, and damages (including but not limited to all fees and charges of engineers, architects, attorneys, and other professionals and all court or arbitration or other dispute resolution costs) arising out of or relating to such correction or repair or such removal and replacement (including but not limited to all costs of repair or replacement of work of others). Contractor's failure to pay such costs, losses, and damages within 10 days of invoice from Owner will be deemed the start of an event giving rise to a Claim under Paragraph 12.01.B, such that any related Claim must be brought within 30 days of the failure to pay.
- D. In special circumstances where a particular item of equipment is placed in continuous service before Substantial Completion of all the Work, the correction period for that item may start to run from an earlier date if so provided in the Specifications.
- E. Where defective Work (and damage to other Work resulting therefrom) has been corrected or removed and replaced under this paragraph, the correction period hereunder with respect to such Work will be extended for an additional period of one year after such correction or removal and replacement has been satisfactorily completed.
- F. Contractor's obligations under this paragraph are in addition to all other obligations and warranties. The provisions of this paragraph are not to be construed as a substitute for, or a waiver of, the provisions of any applicable statute of limitation or repose.

ARTICLE 16-SUSPENSION OF WORK AND TERMINATION

16.01 Owner May Suspend Work

A. At any time and without cause, Owner may suspend the Work or any portion thereof for a period of not more than 90 consecutive days by written notice to Contractor and Engineer. Such notice will fix the date on which Work will be resumed. Contractor shall resume the Work on the date so fixed. Contractor shall be entitled to an adjustment in the Contract Price or an

extension of the Contract Times directly attributable to any such suspension. Any Change Proposal seeking such adjustments must be submitted no later than 30 days after the date fixed for resumption of Work.

16.02 Owner May Terminate for Cause

- A. The occurrence of any one or more of the following events will constitute a default by Contractor and justify termination for cause:
 - Contractor's <u>failure to start the Work in accordance with the Notice to Proceed or</u> <u>Contractor's</u> persistent failure to perform the Work in accordance with the Contract Documents (including, but not limited to, failure to supply sufficient skilled workers or suitable materials or equipment, or failure to adhere to the Progress Schedule);
 - Failure of Contractor to perform or otherwise to comply with a material term of the Contract Documents;
 - 3. Contractor's disregard of Laws or Regulations of any public body having jurisdiction; or
 - 4. Contractor's repeated disregard of the authority of Owner or Engineer.
- B. If one or more of the events identified in Paragraph 16.02.A occurs, then after giving Contractor (and any surety) 10 days' written notice that Owner is considering a declaration that Contractor is in default and termination of the Contract, Owner may proceed to:
 - 1. declare Contractor to be in default, and give Contractor (and any surety) written notice that the Contract is terminated; and
 - 2. enforce the rights available to Owner under any applicable performance bond.
- C. Subject to the terms and operation of any applicable performance bond, if Owner has terminated the Contract for cause, Owner may exclude Contractor from the Site, take possession of the Work, incorporate in the Work all materials and equipment stored at the Site or for which Owner has paid Contractor but which are stored elsewhere, and complete the Work as Owner may deem expedient.
- D. Owner may not proceed with termination of the Contract under Paragraph 16.02.B if Contractor within 7 days of receipt of notice of intent to terminate begins to correct its failure to perform and proceeds diligently to cure such failure.
- E. If Owner proceeds as provided in Paragraph 16.02.B, Contractor shall not be entitled to receive any further payment until the Work is completed. If the unpaid balance of the Contract Price exceeds the cost to complete the Work, including all related claims, costs, losses, and damages (including but not limited to all fees and charges of engineers, architects, attorneys, and other professionals) sustained by Owner, such excess will be paid to Contractor. If the cost to complete the Work including such related claims, costs, losses, and damages exceeds such unpaid balance, Contractor shall pay the difference to Owner. Such claims, costs, losses, and damages incurred by Owner will be reviewed by Engineer as to their reasonableness and, when so approved by Engineer, incorporated in a Change Order. When exercising any rights or remedies under this paragraph, Owner shall not be required to obtain the lowest price for the Work performed.
- F. Where Contractor's services have been so terminated by Owner, the termination will not affect any rights or remedies of Owner against Contractor then existing or which may thereafter accrue, or any rights or remedies of Owner against Contractor or any surety under

any payment bond or performance bond. Any retention or payment of money due Contractor by Owner will not release Contractor from liability.

G. If and to the extent that Contractor has provided a performance bond under the provisions of Paragraph 6.01.A, the provisions of that bond will govern over any inconsistent provisions of Paragraphs 16.02.B and 16.02.D.

16.03 Owner May Terminate for Convenience

- A. Upon 7 days' written notice to Contractor and Engineer, Owner may, without cause and without prejudice to any other right or remedy of Owner, terminate the Contract. In such case, Contractor shall be paid for (without duplication of any items):
 - completed and acceptable Work executed in accordance with the Contract Documents prior to the effective date of termination, including fair and reasonable sums for overhead and profit on such Work;
 - expenses sustained prior to the effective date of termination in performing services and furnishing labor, materials, or equipment as required by the Contract Documents in connection with uncompleted Work, plus fair and reasonable sums for overhead and profit on such expenses; and
 - 3. other reasonable expenses directly attributable to termination, including costs incurred to prepare a termination for convenience cost proposal.
- B. Contractor shall not be paid for any loss of anticipated profits or revenue, post-termination overhead costs, or other economic loss arising out of or resulting from such termination.

16.04 Contractor May Stop Work or Terminate

- A. If, through no act or fault of Contractor, (1) the Work is suspended for more than 90 consecutive days by Owner or under an order of court or other public authority, or (2) Engineer fails to act on any Application for Payment within 30 days after it is submitted, or (3) Owner fails for 30 days to pay Contractor any sum finally determined to be due, then Contractor may, upon 7 days' written notice to Owner and Engineer, and provided Owner or Engineer do not remedy such suspension or failure within that time, terminate the contract and recover from Owner payment on the same terms as provided in Paragraph 16.03.
- B. In lieu of terminating the Contract and without prejudice to any other right or remedy, if Engineer has failed to act on an Application for Payment within 30 days after it is submitted, or Owner has failed for 30 days to pay Contractor any sum finally determined to be due, Contractor may, 7 days after written notice to Owner and Engineer, stop the Work until payment is made of all such amounts due Contractor, including interest thereon. The provisions of this paragraph are not intended to preclude Contractor from submitting a Change Proposal for an adjustment in Contract Price or Contract Times or otherwise for expenses or damage directly attributable to Contractor's stopping the Work as permitted by this paragraph.

ARTICLE 17—FINAL RESOLUTION OF DISPUTES

17.01 Methods and Procedures

- A. Disputes Subject to Final Resolution: The following disputed matters are subject to final resolution under the provisions of this article:
 - 1. A timely appeal of an approval in part and denial in part of a Claim, or of a denial in full, pursuant to Article 12; and
 - 2. Disputes between Owner and Contractor concerning the Work, or obligations under the Contract Documents, that arise after final payment has been made.
- B. Final Resolution of Disputes: For any dispute subject to resolution under this article, Owner or Contractor may:
 - elect in writing to invoke the dispute resolution process provided for in the Supplementary Conditions;
 - agree with the other party to submit the dispute to another dispute resolution process; or
 - 3. if no dispute resolution process is provided for in the Supplementary Conditions or mutually agreed to, give written notice to the other party of the intent to submit the dispute to a court of competent jurisdiction. <u>Any litigation action shall be filed in a state Court in Nassau County, Florida, with the exception of any federal litigation action which shall be filed with a Court for the Middle District of Florida. Any mediation shall be held in Nassau County, Florida.</u>

ARTICLE 18-MISCELLANEOUS

18.01 Giving Notice

- A. Whenever any provision of the Contract requires the giving of written notice to Owner, Engineer, or Contractor, it will be deemed to have been validly given only if delivered:
 - in person, by a commercial courier service or otherwise, to the recipient's place of business;
 - 2. by registered or certified mail, postage prepaid, to the recipient's place of business; or
 - 3. by e-mail to the recipient, with the words "Formal Notice" or similar in the e-mail's subject line.

18.02 Computation of Times

A. When any period of time is referred to in the Contract by days, it will be computed to exclude the first and include the last day of such period. If the last day of any such period falls on a Saturday or Sunday or on a day made a legal holiday by the law of the applicable jurisdiction, such day will be omitted from the computation.

18.03 Cumulative Remedies

A. The duties and obligations imposed by these General Conditions and the rights and remedies available hereunder to the parties hereto are in addition to, and are not to be construed in any way as a limitation of, any rights and remedies available to any or all of them which are

Page 72 of 73

otherwise imposed or available by Laws or Regulations, by special warranty or guarantee, or by other provisions of the Contract. The provisions of this paragraph will be as effective as if repeated specifically in the Contract Documents in connection with each particular duty, obligation, right, and remedy to which they apply.

18.04 Limitation of Damages

A. With respect to any and all Change Proposals, Claims, disputes subject to final resolution, and other matters at issue, neither Owner nor Engineer, nor any of their officers, directors, members, partners, employees, agents, consultants, or subcontractors, shall be liable to Contractor for any claims, costs, losses, or damages sustained by Contractor on or in connection with any other project or anticipated project.

18.05 No Waiver

A. A party's non-enforcement of any provision will not constitute a waiver of that provision, nor will it affect the enforceability of that provision or of the remainder of this Contract.

18.06 Survival of Obligations

A. All representations, indemnifications, warranties, and guarantees made in, required by, or given in accordance with the Contract, as well as all continuing obligations indicated in the Contract, will survive final payment, completion, and acceptance of the Work or termination of the Contract or of the services of Contractor.

18.07 Controlling Law

A. This Contract is to be governed by the law of the state in which the Project is located.

18.08 Assignment of Contract

A. Unless expressly agreed to elsewhere in the Contract, no assignment by a party to this Contract of any rights under or interests in the Contract will be binding on the other party without the written consent of the party sought to be bound; and, specifically but without limitation, money that may become due and money that is due may not be assigned without such consent (except to the extent that the effect of this restriction may be limited by law), and unless specifically stated to the contrary in any written consent to an assignment, no assignment will release or discharge the assignor from any duty or responsibility under the Contract.

18.09 Successors and Assigns

A. Owner and Contractor each binds itself, its successors, assigns, and legal representatives to the other party hereto, its successors, assigns, and legal representatives in respect to all covenants, agreements, and obligations contained in the Contract Documents.

18.10 Headings

A. Article and paragraph headings are inserted for convenience only and do not constitute parts of these General Conditions.

EXHIBIT "B"

SECTION 00 73 15

SUPPLEMENTARY CONDITIONS

These Supplementary Conditions amend or supplement the Standard General Conditions of the Construction Contract (EJCDC C-700, 2007 Edition). All provisions which are not so amended or supplemented remain in full force and effect.

The terms used in these Supplementary Conditions which are defined in the Standard General Conditions of the Construction Contract have the meanings assigned to them in the General Conditions.

Local Agency Program Agreement 441241-1-58-01 Section 14.f

Neither the Recipient nor any of Its contractors or their subcontractors shall enter into any contract, subcontract or arrangement in connection. with the Project or any property Included or planned to be included in the Project in which any member, officer or employee of the Recipient or the locality during tenure or for 2 years thereafter has any interest, direct or indirect. If any such present or former member, officer or employee Involuntarily acquires or had acquired prior to the beginning of tenure any such Interest, and if such interest Is immediately disclosed to the Recipient, the Recipient, with prior approval of the Department, may waive the prohibition contained In this paragraph provided that any such present member, officer or employee shall not participate in any action by the Recipient or the locality relating to such contract, subcontract or arrangement. The Recipient shall insert in all contracts entered into in connection with the Project or any property included or planned to be included in any Project, and shall require its contractors to insert In each of their subcontracts, the following provision:

"No member, officer or employee of the Recipient or of the locality during his tenure or for 2 years thereafter shall have any interest, direct or indirect, in this Contract or the proceeds thereof."

The provisions of this paragraph shall not be applicable to any agreement between the Recipient and its fiscal depositories or to any agreement for utility services the rates for which are fixed or controlled by a governmental agency.

Local Agency Program Agreement 441241-1-58-01 Section 15.c

Recipient agrees to include the following indemnification in all contracts with contractors, subcontractors, consultants, or subconsultants (each referred to as "Entity" for the purposes of the below indemnification) who perform work in connection with this Agreement:

"To the extent provided by law, [ENTITY] shall indemnify, defend, and hold harmless the

[RECIPIENT] and the State of Florida, Department of Transportation, Including the Department's officers, agents, and employees, against any actions, claims, or damages arising out of, relating to, or resulting from negligent or wrongful act(s) of [ENTITY], or any of its officers, agents, or employees, acting within the scope of their office or employment, in connection with the rights granted to or exercised by [ENTITY] hereunder, to the extent and within the limitations of Section 768.28, Florida Statutes.

The foregoing indemnification shall not constitute a waiver of sovereign Immunity beyond the limits set forth in Florida Statutes, Section 768.28. Nor shall the same be construed to constitute agreement by [ENTITY] to indemnify [RECIPIENT] for the negligent acts or omissions of [RECIPIENT], its officers, agents, or employees, or third parties. Nor shall the same be construed to constitute agreement by [ENTITY] to indemnify the Department for the negligent acts or omissions of the Department, its officers, agents, or employees, or third parties, or third parties. This Indemnification shall survive the termination of this Agreement."

Local Agency Program Agreement 441241-1-58-01

Section 17.0

The Parties agree to comply with s.20.055(5), Florida Statutes, and to Incorporate in all subcontracts the obligation to comply with s.20.055(5), Florida Statutes.

February 4, 2022 PREPARED BY: D. Gil, PE, Beyer, PE, R. Endrzejewski, PE

LAP SPECIFICATIONS PACKAGE FINANCIAL PROJECT ID(S). 441241-1-58-01

NASSAU COUNTY

The July 2020 Edition of the Florida Department of Transportation Standard Specifications is revised as follows:

I hereby certify that this specifications package has been properly prepared by me, or under my responsible charge, in accordance with procedures adopted by the Florida Department of Transportation.

The official record of this package is the electronic file digitally signed and sealed under Rule 61G 15-23.004, F.A.C.

Derek M. Gil, PE	
February 4, 2022	
54798	
Element Engineering Group, LLC	
1713 E. 9th Avenue	
Tampa, Florida, 33605	
zation Number: 26921	
1-39	



FPID(S): 441241-1-58-01

Table of Contents

SUPPLEMENTAL SPECIFICATIONS
LAP DIVISION 1 SPECIFICATIONS
SECTION 1 – Definitions and Terms:
FROM SECTION 4 (Alteration of Work).
FROM SECTION 5 - CONTROL OF THE WORK (Claims) 14
FROM SECTION 6 - CONTROL OF MATERIALS (Convict Labor and Buy
America)
FROM SECTION 7 – LEGAL REQUIREMENTS AND RESPONSIBILITIES TO
THE PUBLIC (FHWA 1273, Wage Rates, E-Verify, Title VI, DBE, and On-
The-Job Training)
FROM SECTION 8 (Subletting, Contract Time Extensions, and Liquidated Damages) 32
FROM SECTION 9 (Partial Payments)
THIS COMPLETES THIS SPECIFICATIONS PACKAGE

SUPPLEMENTAL SPECIFICATIONS

FPID(S): 441241-1-58-01

LAP DIVISION 1 SPECIFICATIONS

Construction Checklist Specifications from Department of Transportation Standard Specifications for Road and Bridge Construction

The following excerpts from the Standard Specifications and Special Provisions are provided for use in LAP Specifications as needed in accordance with the Local Agency Program Checklist for Construction Contracts (Phase 58) – Federal and State Requirements (525-070-44)

SECTION 1 – DEFINITIONS AND TERMS:

Department Name Nassau County

Engineer <u>Nassau County Engineer, acting directly or through duly authorized</u> represuentatives; such representatives acting within the scope of the duties and authority assigned to them.

Contractor's Engineer of Record.

A Professional Engineer registered in the State of Florida, other than the Engineer of Record or his subcontracted consultant, who undertakes the design and drawing of components of the permanent structure as part of a redesign or Cost Savings Initiative Proposal, or for repair designs and details of the permanent work. The Contractor's Engineer of Record may also serve as the Specialty Engineer.

The Contractor's Engineer of Record must be an employee of a pre-qualified firm. The firm shall be pre-qualified in accordance with the Rules of the Department of Transportation, Chapter 14-75. Any Corporation or Partnership offering engineering services must hold a Certificate of Authorization from the Florida Department of Business and Professional Regulation.

As an alternate to being an employee of a pre-qualified firm, the Contractor's Engineer of Record may be a pre-qualified Specialty Engineer. For items of the permanent work declared by the State Construction Office to be "major" or "structural", the work performed by a prequalified Specialty Engineer must be checked by another pre-qualified Specialty Engineer. An individual Engineer may become pre-qualified in the work groups listed in the Rules of the Department of Transportation, Chapter 14-75, if the requirements for the Professional Engineer are met for the individual work groups. Pre-qualified Specialty Engineers are listed on the State Construction Website. Pre-qualified Specialty Engineers will not be authorized to perform redesigns or Cost Savings Initiative Proposal designs of items fully detailed in the plans.

Specialty Engineer.

A Professional Engineer registered in the State of Florida, other than the Engineer of Record or his subcontracted consultant, who undertakes the design and drawing preparation of components, systems, or installation methods and equipment for specific temporary portions of the project work or for special items of the permanent works not fully detailed in the plans and required to be furnished by the Contractor. The Specialty Engineer may also provide designs and details, repair designs and details, or perform Engineering Analyses for items of the permanent work declared by the State Construction Office to be "minor" or "non-structural".

For items of work not specifically covered by the Rules of the Department of Transportation, a Specialty Engineer is qualified if he has the following qualifications:

(1) Registration as a Professional Engineer in the State of Florida.

(2) The education and experience necessary to perform the submitted design as required by the Florida Department of Business and Professional Regulation.

FROM SECTION 4 (ALTERATION OF WORK).

4-3 Alteration of Plans or of Character of Work.

4-3.1 General: The Engineer reserves the right to make, at any time prior to or during the progress of the work, such increases or decreases in quantities, whether a significant change or not, and such alterations in the details of construction, whether a substantial change or not, including but not limited to alterations in the grade or alignment of the road or structure or both, as may be found necessary or desirable by the Engineer. Such increases, decreases or alterations shall not constitute a breach of Contract, shall not invalidate the Contract, nor release the Surety from any liability arising out of this Contract or the Surety bond. The Contractor agrees to perform the work, as altered, the same as if it had been a part of the original Contract.

The term "significant change" applies only when:

1. The Engineer determines that the character of the work as altered differs materially in kind or nature from that involved or included in the original proposed construction, or

2. A major item of work, as defined in 1-3, is increased in excess of 125% or decreased below 75% of the original Contract quantity. The Department will apply any price adjustment for an increase in quantity only to that portion in excess of 125% of the original Contract item quantity in accordance with 4-3.2 below. In the case of a decrease below 75% the Department will only apply a price adjustment for the additional costs that are a direct result of the reduction in quantity.

In (1) above, the determination by the Engineer shall be conclusive. If the determination is challenged by the Contractor in any proceeding, the Contractor must establish by clear and convincing proof that the determination by the Engineer was without any reasonable basis.

4-3.2 Increase, Decrease or Alteration in the Work: The Engineer reserves the right to make alterations in the character of the work which involve a substantial change in the nature of the design or in the type of construction or which materially increases or decreases the cost or time of performance. Such alteration shall not constitute a breach of Contract, shall not invalidate the Contract or release the Surety.

Notwithstanding that the Contractor shall have no formal right whatsoever to any extra compensation or time extension deemed due by the Contractor for any cause unless and until the Contractor follows the procedures set forth in 5-12.2 for preservation, presentation and resolution of the claim, the Contractor may at any time, after having otherwise timely submitted a notice of intent to claim or preliminary time extension request pursuant to 5-12.2 and 8-7.3.2, submit to the Department a request for equitable adjustment of compensation or time or other dispute resolution proposal. The Contractor shall in any request for equitable adjustment of compensation, time, or other dispute resolution proposal certify under oath and in writing, in accordance with the formalities required by Florida law, that the request is made in good faith, that any supportive data submitted is accurate and complete to the Contractor's best knowledge and belief, and that the amount of the request accurately reflects what the Contractor in good faith believes to be the Department's responsibility. Such certification must be made by an officer or director of the Contractor with the authority to bind the Contractor. Any such certified statements of entitlement and costs shall be subject to the audit provisions set forth in 5-12.14. While the submittal or review of a duly certified request for equitable adjustment shall neither create, modify, nor activate any legal rights or obligations as to the Contractor or the Department, the Department will review the content of any duly certified request for equitable

FPID(S): 441241-1-58-01

adjustment or other dispute resolution proposal, with any further action or inaction by the Department thereafter being in its sole discretion. Any request for equitable adjustment that fails to fully comply with the certification requirements will not be reviewed by the Department.

The monetary compensation provided for below constitutes full and complete payment for such additional work and the Contractor shall have no right to any additional monetary compensation for any direct or indirect costs or profit for any such additional work beyond that expressly provided below. The Contractor shall be entitled to a time extension only to the extent that the performance of any portion of the additional work is a controlling work item and the performance of such controlling work item actually extends completion of the project due to no fault of the Contractor. All time related costs for actual performance of such additional work are included in the compensation already provided below and any time extension entitlement hereunder will be without additional monetary compensation. The Contractor shall have no right to any monetary compensation or damages whatsoever for any direct or indirect delay to a controlling work item arising out of or in any way related to the circumstances leading up to or resulting from additional work (but not relating to the actual performance of the additional work, which is paid for as otherwise provided herein), except only as provided for under 5-12.6.2.1.

4-3.2.1 Allowable Costs for Extra Work: The Engineer may direct in writing that extra work be done and, at the Engineer's sole discretion, the Contractor will be paid pursuant to an agreed Supplemental Agreement or in the following manner:

1. Labor and Burden: The Contractor will receive payment for actual costs of direct labor and burden for the additional or unforeseen work. Labor includes foremen actually engaged in the work; and will not include project supervisory personnel nor necessary on-site clerical staff, except when the additional or unforeseen work is a controlling work item and the performance of such controlling work item actually extends completion of the project due to no fault of the Contractor. Compensation for project supervisory personnel, but in no case higher than a Project Manager's position, shall only be for the pro-rata time such supervisory personnel spent on the contract. In no case shall an officer or director of the Company, nor those persons who own more than 1% of the Company, be considered as project supervisory personnel, direct labor or foremen hereunder.

	Table 4-3.2.1	
Item	Rate	
FICA	Rate established by Law	
FUTA/SUTA	Rate established by Law	
Medical Insurance	Actual	
Holidays, Sick & Vacation benefits	Actual	
Retirement benefits	Actual	
Workers Compensation	Rates based on the National Council on Compensation Insur basic rate tables adjusted by Contractor's actual experience modification factor in effect at the time of the additional wo unforeseen work.	
Per Diem	Actual but not to exceed State of Florida's rate	
Insurance*	Actual	

Payment for burden shall be limited solely to the following:

17.5	Table 4-3.2.1	
ltem	Rate	
	nce is limited solely to General Liability Coverage and does not b, Umbrella Coverage, Automobile Insurance, etc.).	include any other insurance coverage

At the Pre-construction conference, certify to the Engineer the

following:

a. A listing of on-site clerical staff, supervisory personnel and their pro-rated time assigned to the contract,

b. Actual Rate for items listed in Table 4-3.2.1,

c. Existence of employee benefit plan for Holiday, Sick and Vacation benefits and a Retirement Plan, and,

d. Payment of Per Diem is a company practice for instances when compensation for Per Diem is requested.

Such certification must be made by an officer or director of the Contractor with authority to bind the Contractor. Timely certification is a condition precedent to any right of the Contractor to recover compensations for such costs, and failure to timely submit the certification will constitute a full, complete, absolute and irrevocable waiver by the Contractor of any right to recover such costs. Any subsequent changes shall be certified to the Engineer as part of the cost proposal or seven calendar days in advance of performing such extra work.

2. Materials and Supplies: For materials accepted by the Engineer and used on the project, the Contractor will receive the actual cost of such materials incorporated into the work, including Contractor paid transportation charges (exclusive of equipment as hereinafter set forth). For supplies reasonably needed for performing the work, the Contractor will receive the actual cost of such supplies.

3. Equipment: For any machinery or special equipment (other than small tools), including fuel and lubricant, the Contractor will receive 100% of the "Rental Rate Blue Book" for the actual time that such equipment is in operation on the work, and 50% of the "Rental Rate Blue Book" for the time the equipment is directed to standby and remain on the project site, to be calculated as indicated below. The equipment rates will be based on the latest edition (as of the date the work to be performed begins) of the "Rental Rate Blue Book for Construction Equipment" or the "Rental Rate Blue Book for Older Construction Equipment," whichever is applicable, as published by Machinery Information Division of PRIMEDIA Information, Inc. (version current at the time of bid), using all instructions and adjustments contained therein and as modified below. On all projects, the Engineer will adjust the rates using regional adjustments and Rate Adjustment Tables according to the instructions in the Blue Book.

Allowable Equipment Rates will be established as set out below: a. Allowable Hourly Equipment Rate = Monthly Rate/176

x Adjustment Factors x 100%. b. Allowable Hourly Operating Cost = Hourly Operating Cost x 100%. c. Allowable Rate Per Hour = Allowable Hourly Equipment Rate + Allowable Hourly Operating Cost. d. Standby Rate = Allowable Hourly Equipment Rate x 50%. above.

above.

The Monthly Rate is The Basic Machine Rate Plus Any

Attachments. Standby rates will apply when equipment is not in operation and is directed by the Engineer to standby at the project site when needed again to complete work and the cost of moving the equipment will exceed the accumulated standby cost. Standby rates will not apply on any day the equipment operates for eight or more hours. Standby payment will be limited to only that number of hours which, when added to the operating time for that day equals eight hours. Standby payment will not be made on days that are not normally considered work days on the project.

The Department will allow for the cost of transporting the equipment to and from the location at which it will be used. If the equipment requires assembly or disassembly for transport, the Department will pay for the time to perform this work at the rate for standby equipment.

Equipment may include vehicles utilized only by Labor, as defined

4. Indirect Costs, Expenses, and Profit: Compensation for all indirect costs, expenses, and profit of the Contractor, including but not limited to overhead of any kind, whether jobsite, field office, division office, regional office, home office, or otherwise, is expressly limited to the greater of either (a) or (b) below:

a. Solely a mark-up of 17.5% on the payments in (1) through (3),

1. Bond: The Contractor will receive compensation for any premium for acquiring a bond for such additional or unforeseen work at the original Contract bond rate paid by the Contractor. No compensation for bond premium will be allowed for additional or unforeseen work paid by the Department via initial contingency pay item.

2. The Contractor will be allowed a markup of 10% on the first \$50,000 and a markup of 5% on any amount over \$50,000 on any subcontract directly related to the additional or unforeseen work. Any such subcontractor mark-up will be allowed only by the prime Contractor and a first tier subcontractor, and the Contractor must elect the markup for any eligible first tier subcontractor to do so.

b. Solely the formula set forth below and only as applied solely as to such number of calendar days of entitlement that are in excess of ten cumulative calendar days as defined below.

$$\mathbf{D} = \frac{\mathbf{A} \times \mathbf{C}}{\mathbf{B}}$$

Where A = Original Contract Amount B = Original Contract Time C = 8% D = Average Overhead Per Day

Cumulative Calendar Days is defined as the combined total number of calendar days granted as time extensions due to either extra work, excluding overruns to existing contract items, that extend the duration of the project or delay of a controlling work item caused solely by the Department, or the combined total number of calendar days for which a claim of entitlement to a time extension due to delay of a controlling work item caused solely by the Department is otherwise ultimately determined to be in favor of the Contractor.

No compensation, whatsoever, will be paid to the Contractor for any jobsite overhead and other indirect impacts when the total number of calendar days granted for time extension due to delay of a controlling work item caused solely by the Department is, or the total number of calendar days for which entitlement to a time extension due to delay of a controlling work item caused solely by the Department is otherwise ultimately determined in favor of the Contractor to be, equal to or less than ten calendar days and the Contractor also fully assumes all monetary risk of any and all partial or single calendar day delay periods, due to delay of a controlling work item caused solely by the Department, that when combined together are equal to or less than ten calendar days and regardless of whether monetary compensation is otherwise provided for hereunder for one or more calendar days of time extension entitlement for each calendar day exceeding ten calendar days. All calculations under this provision shall exclude weather days, Holidays, and Special Events.

Further, for (a) and (b) above, in the event there are concurrent delays to one or more controlling work items, one or more being caused by the Department and one or more being caused by the Contractor, the Contractor shall be entitled to a time extension for each day that a controlling work item is delayed by the Department but shall have no right to nor receive any monetary compensation for any indirect costs for any days of concurrent delay.

4-3.2.2 Subcontracted Work: Compensation for the additional or unforeseen work performed by a subcontractor shall be limited solely to that provided for in 4-3.2.1 (1), (2), (3) and (4)(a). In addition, the Contractor compensation is expressly limited to the greater of the total provided in either 4-3.2.1(4)(a) or (4)(b), except that the Average Overhead Per-Day calculation is as follows:

$$\mathbf{Ds} = \frac{\mathbf{As} \times \mathbf{C}}{\mathbf{B}}$$

Where As = Original Contract Amount minus Original

Subcontract amounts(s)*

B = Original Contract Time C = 8% Ds = Average Overhead Per-Day

* deduct Original Subcontract Amount(s) of

subcontractor(s) performing the work

The subcontractor may receive compensation for any premium for acquiring a bond for the additional or unforeseen work; provided, however, that such payment for additional subcontractor bond will only be paid upon presentment to the Department of clear and convincing proof that the subcontractor has actually submitted and paid for separate bond premiums for such additional or unforeseen work in such amount and that the subcontractor was required by the Contractor to acquire a bond.

The Contractor shall require the subcontractor to submit a certification, in accordance with 4-3.2.1 (1), as part of the cost proposal and submit such to the Engineer. Such certification must be made by an officer or director of the subcontractor with authority to bind

the subcontractor. Timely certification is a condition precedent to any right of the Contractor to recover compensation for such subcontractor costs, and failure to timely submit the certification will constitute a full, complete, absolute and irrevocable waiver by the Contractor of any right to recover such subcontractor costs.

4-3.3 No Waiver of Contract: Changes made by the Engineer will not be considered to waive any of the provisions of the Contract, nor may the Contractor make any claim for loss of anticipated profits because of the changes, or by reason of any variation between the approximate quantities and the quantities of work actually performed. All work shall be performed as directed by the Engineer and in accordance with the Contract Documents.

4-3.4 Conditions Requiring a Supplemental Agreement or Unilateral Payment: A Supplemental Agreement or Unilateral Payment will be used to clarify the Plans and Specifications of the Contract; to provide for unforeseen work, grade changes, or alterations in the Plans which could not reasonably have been contemplated or foreseen in the original Plans and Specifications; to change the limits of construction to meet field conditions; to provide a safe and functional connection to an existing pavement; to settle documented Contract claims; to make the project functionally operational in accordance with the intent of the original Contract and subsequent amendments thereto.

A Supplemental Agreement or Unilateral Payment may be used to expand the physical limits of the project only to the extent necessary to make the project functionally operational in accordance with the intent of the original Contract. The cost of any such agreement extending the physical limits of the project shall not exceed \$100,000 or 10% of the original Contract price, whichever is greater.

Perform no work to be covered by a Supplemental Agreement or Unilateral Payment before written authorization is received from the Engineer. The Engineer's written authorization will set forth sufficient work information to allow the work to begin. The work activities, terms and conditions will be reduced to written Supplemental Agreement or Unilateral Payment form promptly thereafter. No payment will be made on a Supplemental Agreement or Unilateral Payment prior to the Department's approval of the document.

4-3.5 Extra Work: Extra work authorized in writing by the Engineer will be paid in accordance with the formula in 4-3.2. Such payment will be the full extent of all monetary compensation entitlement due to the Contractor for such extra work. Any entitlement to a time extension due to extra work will be limited solely to that provided for in 4-3.2 for additional work.

4-3.6 Connections to Existing Pavement, Drives and Walks: Generally adhere to the limits of construction at the beginning and end of the project as detailed in the Plans. However, if the Engineer determines that it is necessary to extend the construction in order to make suitable connections to existing pavement, the Engineer will authorize such a change in writing.

For necessary connections to existing walks and drives that are not indicated in the Plans, the Engineer will submit direction regarding the proper connections in accordance with the Design Standards.

4-3.7 Differing Site Conditions: During the progress of the work, if subsurface or latent physical conditions are encountered at the site differing materially from those indicated in the Contract, or if unknown physical conditions of an unusual nature differing materially from those ordinarily encountered and generally recognized as inherent in the work provided for in the Contract are encountered at the site, the party discovering such conditions shall promptly notify
the other party in writing of the specific differing conditions before the Contractor disturbs the conditions or performs the affected work.

Upon receipt of written notification of differing site conditions from the Contractor, the Engineer will investigate the conditions, and if it is determined that the conditions materially differ and cause an increase or decrease in the cost or time required for the performance of any work under the Contract, an adjustment will be made, excluding loss of anticipated profits, and the Contract will be modified in writing accordingly. The Engineer will notify the Contractor whether or not an adjustment of the Contract is warranted.

The Engineer will not allow a Contract adjustment for a differing site condition unless the Contractor has submitted the required written notice.

The Engineer will not allow a Contract adjustment under this clause for any effects caused to any other Department or non-Department projects on which the Contractor may be working.

4-3.8 Changes Affecting Utilities: The Contractor shall be responsible for identifying and assessing any potential impacts to a utility that may be caused by the changes proposed by the Contractor, and the Contractor shall at the time of making the request for a change notify the Department in writing of any such potential impacts to utilities.

Department approval of a Contractor proposed change does not relieve the Contractor of sole responsibility for all utility impacts, costs, delays or damages, whether direct or indirect, resulting from Contractor initiated changes in the design or construction activities from those in the original Contract Specifications, Design Plans (including Traffic Control Plans) or other Contract Documents and which effect a change in utility work different from that shown in the Utility Plans, joint project agreements or utility relocation schedules.

4-3.9 Cost Savings Initiative Proposal:

4-3.9.1 Intent and Objective:

1. This Subarticle applies to any cost reduction proposal (hereinafter referred to as a Proposal) that the Contractor initiates and develops for the purpose of refining the Contract to increase cost effectiveness or significantly improve the quality of the end result. A mandatory Cost Savings Initiative Workshop will be held prior to Contract Time beginning for the Contractor and Department to discuss potential Proposals. This mandatory workshop can only be eliminated if agreed to in writing by both the Contractor and Department. This Subarticle does not, however, apply to any such proposal unless the Contractor identifies it at the time of its submission to the Department as a proposal submitted pursuant to this Subarticle.

2. The Department will consider Proposals that would result in net savings to the Department by providing a decrease in the cost of the Contract. Proposals must result in savings without impairing essential functions and characteristics such as safety, service, life, reliability, economy of operation, ease of maintenance, aesthetics and necessary standard design features. The Department will not recognize the Contractor's correction of plan errors that result in a cost reduction, as a Proposal. Deletions of work, approved by the Engineer which are not directly associated with or integral to a Proposal will be handled as full credit to the Department for the work deleted.

3. The Department shall have the right to reject, at its discretion, any Proposal submitted that proposes a change in the design of the pavement system or that would require additional right-of-way. Pending the Department's execution of a formal supplemental agreement implementing an approved Proposal, the Contractor shall remain obligated to perform the work in accordance with the terms of the existing Contract. The Department may grant time extensions to allow for the time required to develop and review a Proposal.

4. For potential Proposals not discussed at the Cost Savings Initiative Workshop, a mandatory concept meeting will be held for the Contractor and Department to discuss the potential Proposal prior to development of the Proposal. This mandatory meeting can only be eliminated if agreed to in writing by both the Contractor and Department.

4-3.9.2 Subcontractors: The Department encourages the Contractor to include the provisions of this Subarticle in Contracts with subcontractors and to encourage submission of Proposals from subcontractors. However, it is not mandatory to submit Proposals to the Department or to accept or transmit subcontractor proposed Proposals to the Department.

4-3.9.3 Data Requirements: As a minimum, submit the following information with each Proposal:

1. a description of the difference between the existing Contract requirement, including any time extension request, and the proposed change, and the comparative advantages and disadvantages.

2. separate detailed cost estimates for both the existing Contract requirement and the proposed change. Break down the cost estimates by pay item numbers indicating quantity increases or decreases and deleted pay items. Identify additional proposed work not covered by pay items within the Contract, by using pay item numbers in the Basis of Estimates Manual. In preparing the estimates, include overhead, profit, and bond within pay items in the Contract. Separate pay item(s) for the cost of overhead, profit, and bond will not be allowed.

3. an itemization of the changes, deletions or additions to plan details, plan sheets, design standards and Specifications that are required to implement the Proposal if the Department adopts it. Submit preliminary plan drawings sufficient to describe the proposed changes.

4. engineering or other analysis in sufficient detail to identify and describe specific features of the Contract that must be changed if the Department accepts the Proposal with a proposal as to how these changes can be accomplished and an assessment of their effect on other project elements. The Department may require that engineering analyses be performed by a prequalified consultant in the applicable class of work. Support all design changes that result from the Proposal with drawings and computations signed and sealed by the Contractor's Engineer of Record. Written documentation or drawings will be submitted clearly delineating the responsibility of the Contractor's Engineer of Record.

5. the date by which the Department must approve the Proposal to obtain the total estimated cost reduction during the remainder of the Contract, noting any effect on the Contract completion time or delivery schedule.

6. a revised project schedule that would be followed upon approval of the Proposal. This schedule would include submittal dates and review time for the Department and Peer reviews.

4-3.9.4 Processing Procedures: Submit Proposals to the Engineer or his duly authorized representative. The Department will process Proposals expeditiously; however, the Department is not liable for any delay in acting upon a Proposal submitted pursuant to this Subarticle. The Contractor may withdraw, in whole or in part, a Proposal not accepted by the Department within the period specified in the Proposal. The Department is not liable for any Proposal development cost in the case where the Department rejects or the Contractor withdraws a Proposal.

The Engineer is the sole judge of the acceptability of a Proposal and of the estimated net savings in construction costs from the adoption of all or any part of such proposal. In determining the estimated net savings, the Department reserves the right to disregard the Contract bid prices if, in the judgment of the Engineer, such prices do not represent a fair measure of the value of work to be performed or to be deleted.

Prior to approval, the Engineer may modify a Proposal, with the concurrence of the Contractor, to make it acceptable. If any modification increases or decreases the net savings resulting from the Proposal, the Department will determine the Contractor's fair share upon the basis of the Proposal as modified and upon the final quantities. The Department will compute the net savings by subtracting the revised total cost of all bid items affected by the Proposal from the total cost of the same bid items as represented in the original Contract.

Prior to approval of the Proposal that initiates the supplemental agreement, submit acceptable Contract-quality plan sheets revised to show all details consistent with the Proposal design.

4-3.9.5 Computations for Change in Contract Cost of Performance: If the Proposal is adopted, the Contractor's share of the net savings as defined hereinafter represents full compensation to the Contractor for the Proposal.

The Department will not include its costs to process and implement a Proposal in the estimate. However, the Department reserves the right, where it deems such action appropriate, to require the Contractor to pay the Department's cost of investigating and implementing a Proposal as a condition of considering such proposal. When the Department imposes such a condition, the Contractor shall accept this condition in writing, authorizing the Department to deduct amounts payable to the Department from any monies due or that may become due to the Contractor under the Contract.

4-3.9.6 Conditions of Acceptance for Major Design Modifications of Category 2 Bridges: A Proposal that proposes major design modifications of a category 2 bridge, as determined by the Engineer, shall have the following conditions of acceptance:

All bridge Plans relating to the Proposal shall undergo an independent peer review conducted by a single independent engineering firm referred to for the purposes of this article as the Independent Review Engineer who is not the originator of the Proposal design, and is pre-qualified by the Department in accordance with Rule 14-75, Florida Administrative Code. The independent peer review is intended to be a comprehensive, thorough verification of the original work, giving assurance that the design is in compliance with all Department requirements. The Independent Review Engineer's comments, along with the resolution of each comment, shall be submitted to the Department. The Independent Review Engineer shall sign and seal the submittal cover letter stating that all comments have been adequately addressed and the design is in compliance with the Department requirements. If there are any unresolved comments the Independent Review Engineer shall specifically list all unresolved issues in the signed and sealed cover letter.

The Contractor shall designate a primary engineer responsible for the Proposal design and as such will be designated as the Contractors Engineer of Record for the Proposal design. The Department reserves the right to require the Contractor's Engineer of Record to assume responsibility for design of the entire structure. New designs and independent peer reviews shall be in compliance with all applicable Department, FHWA and AASHTO criteria requirements including bridge load ratings.

4-3.9.7 Sharing Arrangements: If the Department approves a Proposal, the Contractor shall receive 50% of the net reduction in the cost of performance of the Contract as determined by the final negotiated agreement between the Contractor and the Department. The net reduction will be determined by subtracting from the savings of the construction costs the reasonable documented engineering costs incurred by the contractor to design and develop a Proposal. The reasonable documented engineering costs will be paid by the Department. Engineering costs will be based on the consultant's certified invoice and may include the costs of the Independent Review Engineer in 4-3.9.6. The total engineering costs to be subtracted from the savings to determine the net reduction will be limited to 25% of the construction savings and shall not include any markup by the Contractor or the costs for engineering services performed by the Contractor.

4-3.9.8 Notice of Intellectual Property Interests and Department's Future Rights to a Proposal:

4-3.9.8.1 Notice of Intellectual Property Interests: The Contractor's Proposal submittal shall identify with specificity any and all forms of intellectual property rights that either the Contractor or any officer, shareholder, employee, consultant, or affiliate, of the Contractor, or any other entity who contributed in any measure to the substance of the Contractor's Proposal development, have or may have that are in whole or in part implicated in the Proposal. Such required intellectual property rights notice includes, but is not limited to, disclosure of any issued patents, copyrights, or licenses; pending patent, copyright or license applications; and any intellectual property rights that though not yet issued, applied for or intended to be pursued, could nevertheless otherwise be subsequently the subject of patent, copyright or license protection by the Contractor or others in the future. This notice requirement does not extend to intellectual property rights as to stand-alone or integral components of the Proposal that are already on the Department's Approved Product List (APL) or Standard Plans, or are otherwise generally known in the industry as being subject to patent or copyright protection.

4-3.9.8.2 Department's Future Rights to a Proposal: Notwithstanding 7-3 nor any other provision of the Standard Specifications, upon acceptance of a Proposal, the Contractor hereby grants to the Department and its contractors (such grant being expressly limited solely to any and all existing or future Department construction projects and any other Department projects that are partially or wholly funded by or for the Department) a royalty-free and perpetual license under all forms of intellectual property rights to manufacture, to use, to design, to construct, to disclose, to reproduce, to prepare and fully utilize derivative works, to distribute, display and publish, in whole or in part, and to permit others to do any of the above, and to otherwise in any manner and for any purpose whatsoever do anything reasonably necessary to fully utilize any and all aspects of such Proposal on any and all existing and future construction projects and any other Department projects.

Contractor shall hold harmless, indemnify and defend the Department and its contractors and others in privity therewith from and against any and all claims, liabilities, other obligations or losses, and reasonable expenses related thereto (including reasonable attorneys' fees), which are incurred or are suffered by any breach of the foregoing grants, and regardless of whether such intellectual property rights were or were not disclosed by the Contractor pursuant to 4-3.9.8.1, unless the Department has by express written exception in the Proposal acceptance process specifically released the Contractor from such obligation to hold harmless, indemnify and defend as to one or more disclosed intellectual property rights.

FROM SECTION 5 - CONTROL OF THE WORK (CLAIMS).

5-12 Claims by Contractor.

5-12.1 General: When the Contractor deems that extra compensation or a time extension is due beyond that agreed to by the Engineer, whether due to delay, additional work, altered work, differing site conditions, breach of Contract, or for any other cause, the Contractor shall follow the procedures set forth herein for preservation, presentation and resolution of the claim.

Submission of timely notice of intent to file a claim, preliminary time extension request, time extension request, and the certified written claim, together with full and complete claim documentation, are each a condition precedent to the Contractor bringing any circuit court, arbitration, or other formal claims resolution proceeding against the Department for the items and for the sums or time set forth in the Contractor's certified written claim. The failure to provide such notice of intent, preliminary time extension request, time extension request, certified written claim and full and complete claim documentation within the time required shall constitute a full, complete, absolute and irrevocable waiver by the Contractor of any right to additional compensation or a time extension for such claim.

5-12.2 Notice of Claim:

5-12.2.1 Claims For Extra Work: Where the Contractor deems that additional compensation or a time extension is due for work or materials not expressly provided for in the Contract or which is by written directive expressly ordered by the Engineer pursuant to 4-3, the Contractor shall submit written notification to the Engineer of the intention to make a claim for additional compensation before beginning the work on which the claim is based, and if seeking a time extension, the Contractor shall also submit a preliminary request for time extension pursuant to 8-7.3.2 within ten calendar days after commencement of a delay and a request for Contract Time extension pursuant to 8-7.3.2 within thirty calendar days after the elimination of the delay. If such written notification is not submitted and the Engineer is not afforded the opportunity for keeping strict account of actual labor, material, equipment, and time, the Contractor waives the claim for additional compensation or a time extension. Such notice by the Contractor, and the fact that the Engineer has kept account of the labor, materials and equipment, and time, shall not in any way be construed as establishing the validity of the claim or method for computing any compensation or time extension for such claim. On projects with an original Contract amount of \$3,000,000 or less within 90 calendar days after final acceptance of the project and on projects with an original Contract amount greater than \$3,000,000 within 180 calendar days after final acceptance of the project the Contractor shall submit full and complete claim documentation as described in 5-12.3 and duly certified pursuant to 5-12.9. However, for any claim or part of a claim that pertains solely to final estimate quantities disputes the Contractor shall submit full and complete claim documentation as described in 5-12.3 and duly certified pursuant to 5-12.9, as to such final estimate claim dispute issues, within 90 or 180 calendar days, respectively, of the Contractor's receipt of the Department's final estimate. If the Contractor fails to submit a certificate of claim as described in

5-12.9, the Department will so notify the Contractor in writing. The Contractor shall have ten

calendar days from receipt of the notice to resubmit the claim documentation, without change, with a certificate of claim as described in 5-12.9, without regard to whether the resubmission is within the applicable 90 or 180 calendar day deadline for submission of full and complete claim documentation. Failure by the Contractor to comply with the ten calendar day notice shall constitute a waiver of the claim.

5-12.2.2 Claims For Delay: Where the Contractor deems that additional compensation or a time extension is due on account of delay, differing site conditions, breach of Contract, or any other cause other than for work or materials not expressly provided for in the Contract (Extra Work) or which is by written directive of the Engineer expressly ordered by the Engineer pursuant to 4-3, the Contractor shall submit a written notice of intent to the Engineer within ten days after commencement of a delay to a controlling work item expressly notifying the Engineer that the Contractor intends to seek additional compensation, and if seeking a time extension, the Contractor shall also submit a preliminary request for time extension pursuant to 8-7.3.2 within ten calendar days after commencement of a delay to a controlling work item, as to such delay and providing a reasonably complete description as to the cause and nature of the delay and the possible impacts to the Contractor's work by such delay, and a request for Contract Time extension pursuant to 8-7.3.2 within thirty calendar days after the elimination of the delay. On projects with an original Contract amount of \$3,000,000 or less within 90 calendar days after final acceptance of the project and on projects with an original Contract amount greater than \$3,000,000 within 180 calendar days after final acceptance of the project the Contractor shall submit full and complete documentation as described in 5-12.3 and duly certified pursuant to 5-12.9.

If the Contractor fails to submit a certificate of claim as described in 5-12.9, the Department will so notify the Contractor in writing. The Contractor shall have ten calendar days from receipt of the notice to resubmit the claim documentation, without change, with a certificate of claim as described in 5-12.9, without regard to whether the resubmission is within the applicable 90 or 180 calendar day deadline for submission of full and complete claim documentation. Failure by the Contractor to comply with the ten calendar day notice shall constitute a waiver of the claim.

There shall be no Contractor entitlement to any monetary compensation or time extension for any delays or delay impacts, whatsoever, that are not to a controlling work item, and then as to any such delay to a controlling work item entitlement to any monetary compensation or time extension shall only be to the extent such is otherwise provided for expressly under 4-3 or 5-12, except that in the instance of delay to a non-controlling item of work the Contractor may be compensated for the direct costs of idle labor or equipment only, at the rates set forth in 4-3.2.1(1) and (3), and then only to the extent the Contractor could not reasonably mitigate such idleness.

5-12.3 Content of Written Claim: As a condition precedent to the Contractor being entitled to additional compensation or a time extension under the Contract, for any claim, the Contractor shall submit a certified written claim to the Department which will include for each individual claim, at a minimum, the following information:

1. A detailed factual statement of the claim providing all necessary dates, locations, and items of work affected and included in each claim;

 The date or dates on which actions resulting in the claim occurred or conditions resulting in the claim became evident; support thereof.

3. Identification of all pertinent documents and the substance of any material oral communications relating to such claim and the name of the persons making such material oral communications;

4. Identification of the provisions of the Contract which support the claim and a statement of the reasons why such provisions support the claim, or alternatively, the provisions of the Contract which allegedly have been breached and the actions constituting such breach;

5. A detailed compilation of the amount of additional compensation sought and a breakdown of the amount sought as follows:

a. documented additional job site labor expenses;

b. documented additional cost of materials and supplies;

c. a list of additional equipment costs claimed, including each piece of equipment and the rental rate claimed for each;

d. any other additional direct costs or damages and the documents in support thereof;

e. any additional indirect costs or damages and all documentation in

6. A detailed compilation of the specific dates and the exact number of calendar days sought for a time extension, the basis for entitlement to time for each day, all documentation of the delay, and a breakout of the number of days claimed for each identified event, circumstance or occurrence.

Further, the Contractor shall be prohibited from amending either the bases of entitlement or the amount of any compensation or time stated for any and all issues claimed in the Contractor's written claim submitted hereunder, and any circuit court, arbitration, or other formal claims resolution proceeding shall be limited solely to the bases of entitlement and the amount of any compensation or time stated for any and all issues claimed in the Contractor's written claim submitted hereunder. This shall not, however, preclude a Contractor from withdrawing or reducing any of the bases of entitlement and the amount of any compensation or time stated for any and all issues claimed in the Contractor's written claim submitted hereunder at any time.

5-12.4 Action on Claim: The Engineer will respond in writing on projects with an original Contract amount of \$3,000,000 or less within 90 calendar days of receipt of a complete claim submitted by a Contractor in compliance with 5-12.3, and on projects with an original Contract amount greater than \$3,000,000 within 120 calendar days of receipt of a complete claim submitted by a Contractor in compliance with 5-12.3. Failure by the Engineer to respond to a claim in writing within 90 or 120 days, respectively, after receipt of a complete claim submitted by the Contractor in compliance with 5-12.3 constitutes a denial of the claim by the Engineer. If the Engineer finds the claim or any part thereof to be valid, such partial or whole claim will be allowed and paid for to the extent deemed valid and any time extension granted, if applicable, as provided in the Contract. No circuit court or arbitration proceedings on any claim, or a part thereof, may be filed until after final acceptance of all Contract work by the Department or denial hereunder, whichever occurs last.

5-12.5 Pre-Settlement and Pre-Judgment Interest: Entitlement to any pre-settlement or pre-judgment interest on any claim amount determined to be valid subsequent to the Department's receipt of a certified written claim in full compliance with 5-12.3, whether determined by a settlement or a final ruling in formal proceedings, the Department shall pay to the Contractor simple interest calculated at the Prime Rate (as reported by the Wall Street Journal as the base rate on corporate loans posted by at least 75% of the nations 30 largest banks) as of the 60th calendar day following the Department's receipt of a certified written claim in full compliance with 5-12.3, such interest to accrue beginning 60 calendar days following the Department's receipt of a certified written claim in full compliance with 5-12.3 and ending on the date of final settlement or formal ruling.

5-12.6 Compensation for Extra Work or Delay:

5-12.6.1 Compensation for Extra Work: Notwithstanding anything to the contrary contained in the Contract Documents, the Contractor shall not be entitled to any compensation beyond that provided for in 4-3.2.

5-12.6.2 Compensation for Delay: Notwithstanding anything to the contrary contained in the Contract Documents, the additional compensation set forth in 5-12.6.2.1 shall be the Contractor's sole monetary remedy for any delay other than to perform extra work caused by the Department unless the delay shall have been caused by acts constituting willful or intentional interference by the Department with the Contractor's performance of the work and then only where such acts continue after Contractor's written notice to the Department of such interference. The parties anticipate that delays may be caused by or arise from any number of events during the term of the Contract, including, but not limited to, work performed, work deleted, supplemental agreements, work orders, disruptions, differing site conditions, utility conflicts, design changes or defects, time extensions, extra work, right-of-way issues, permitting issues, actions of suppliers, subcontractors or other contractors, actions by third parties, suspensions of work by the Engineer shop drawing approval process delays, expansion of the physical limits of the project to make it functional, weather, weekends, holidays, special events, suspension of Contract Time, or other events, forces or factors sometimes experienced in construction work. Such delays or events and their potential impacts on the performance by the Contractor are specifically contemplated and acknowledged by the parties in entering into this Contract, and shall not be deemed to constitute willful or intentional interference with the Contractor's performance of the work without clear and convincing proof that they were the result of a deliberate act, without reasonable and good-faith basis, and specifically intended to disrupt the Contractor's performance.

5-12.6.2.1 Compensation for Direct Costs, Indirect Costs, Expenses, and Profit thereon, of or from Delay: For any delay claim, the Contractor shall be entitled to monetary compensation for the actual idle labor and equipment, and indirect costs, expenses, and profit thereon, as provided for in 4-3.2.1(4) and solely for costs incurred beyond what reasonable mitigation thereof the Contractor could have undertaken.

5-12.7 Mandatory Claim Records: After submitting to the Engineer a notice of intent to file a claim for extra work or delay, the Contractor must keep daily records of all labor, material and equipment costs incurred for operations affected by the extra work or delay. These daily records must identify each operation affected by the extra work or delay and the specific locations where work is affected by the extra work or delay, as nearly as possible. The Engineer may also keep records of all labor, material and equipment used on the operations affected by the extra work or delay. The Contractor shall, once a notice of intent to claim has been timely filed, and not less than weekly thereafter as long as appropriate, submit the Contractor's daily records to the Engineer and be likewise entitled to receive the Department's daily records. The daily records to be submitted hereunder shall be done at no cost to the recipient.

5-12.8 Claims For Acceleration: The Department shall have no liability for any constructive acceleration of the work, nor shall the Contractor have any right to make any claim

for constructive acceleration nor include the same as an element of any claim the Contractor may otherwise submit under this Contract. If the Engineer gives express written direction for the Contractor to accelerate its efforts, such written direction will set forth the prices and other pertinent information and will be reduced to a written Contract Document promptly. No payment will be made on a Supplemental Agreement for acceleration prior to the Department's approval of the documents.

5-12.9 Certificate of Claim: When submitting any claim, the Contractor shall certify under oath and in writing, in accordance with the formalities required by Florida law, that the claim is made in good faith, that the supportive data are accurate and complete to the Contractor's best knowledge and belief, and that the amount of the claim accurately reflects what the Contractor in good faith believes to be the Department's liability. Such certification must be made by an officer or director of the Contractor with the authority to bind the Contractor.

5-12.10 Non-Recoverable Items: The parties agree that for any claim the Department will not have liability for the following items of damages or expense:

1. Loss of profit, incentives or bonuses;

2. Any claim for other than extra work or delay;

3. Consequential damages, including, but not limited to, loss of bonding capacity, loss of bidding opportunities, loss of credit standing, cost of financing, interest paid, loss of other work or insolvency;

4. Acceleration costs and expenses, except where the Department has expressly and specifically directed the Contractor in writing "to accelerate at the Department's expense"; nor

5. Attorney fees, claims preparation expenses and costs of litigation.

5-12.11 Exclusive Remedies: Notwithstanding any other provision of this Contract, the parties agree that the Department shall have no liability to the Contractor for expenses, costs, or items of damages other than those which are specifically identified as payable under 5-12. In the event any legal action for additional compensation, whether on account of delay, acceleration, breach of contract, or otherwise, the Contractor agrees that the Department's liability will be limited to those items which are specifically identified as payable in 5-12.

5-12.12 Settlement Discussions: The content of any discussions or meetings held between the Department and the Contractor to settle or resolve any claims submitted by the Contractor against the Department shall be inadmissible in any legal, equitable, arbitration or administrative proceedings brought by the Contractor against the Department for payment of such claim. Dispute Resolution Board, State Arbitration Board and Claim Review Committee proceedings are not settlement discussions, for purposes of this provision.

5-12.13 Personal Liability of Public Officials: In carrying out any of the provisions of the Contract or in exercising any power or authority granted to the Secretary of Transportation, Engineer or any of their respective employees or agents, there shall be no liability on behalf of any employee, officer or official of the Department for which such individual is responsible, either personally or as officials or representatives of the Department. It is understood that in all such matters such individuals act solely as agents and representatives of the Department.

5-12.14 Auditing of Claims: All claims filed against the Department shall be subject to audit at any time following the filing of the claim, whether or not such claim is part of a suit pending in the Courts of this State. The audit may be performed, at the Department's sole discretion, by employees of the Department or by any independent auditor appointed by the Department, or both. The audit may begin after ten days written notice to the Contractor,

subcontractor, or supplier. The Contractor, subcontractor, or supplier shall make a good faith effort to cooperate with the auditors. As a condition precedent to recovery on any claim, the Contractor, subcontractor, or supplier must retain sufficient records, and provide full and reasonable access to such records, to allow the Department's auditors to verify the claim and failure to retain sufficient records of the claim or failure to provide full and reasonable access to such records of the claim or failure to provide full and reasonable access to such records shall constitute a waiver of that portion of such claim that cannot be verified and shall bar recovery thereunder. Further, and in addition to such audit access, upon the Contractor submitting a written claim, the Department shall have the right to request and receive, and the Contractor shall have the affirmative obligation to submit to the Department any and all documents in the possession of the Contractor or its subcontractors, materialmen or suppliers as may be deemed relevant by the Department in its review of the basis, validity or value of the Contractor's claim.

Without limiting the generality of the foregoing, the Contractor shall upon written request of the Department make available to the Department's auditors, or upon the Department's written request, submit at the Department's expense, any or all of the following documents:

- 1. Daily time sheets and foreman's daily reports and diaries;
- 2. Insurance, welfare and benefits records;
- 3. Payroll register;
- 4. Earnings records;
- 5. Payroll tax return;
- 6. Material invoices, purchase orders, and all material and supply

acquisition contracts;

7. Material cost distribution worksheet;

8. Equipment records (list of company owned, rented or other equipment

used);

9. Vendor rental agreements and subcontractor invoices;

10. Subcontractor payment certificates;

- 11. Canceled checks for the project, including, payroll and vendors;
- 12. Job cost report;
- 13. Job payroll ledger;

14. General ledger, general journal, (if used) and all subsidiary ledgers and journals together with all supporting documentation pertinent to entries made in these ledgers and journals;

15. Cash disbursements journal;

16. Financial statements for all years reflecting the operations on this

project;

17. Income tax returns for all years reflecting the operations on this

project;

18. All documents which reflect the Contractor's actual profit and overhead during the years this Contract was being performed and for each of the five years prior to the commencement of this Contract;

19. All documents related to the preparation of the Contractor's bid including the final calculations on which the bid was based;

20. All documents which relate to each and every claim together with all documents which support the amount of damages as to each claim;

21. Worksheets used to prepare the claim establishing the cost components for items of the claim including, but not limited to, labor, benefits and insurance, materials, equipment, subcontractors, and all documents that establish which time periods and individuals were involved, and the hours and rates for such individuals.

FROM SECTION 6 – CONTROL OF MATERIALS (CONVICT LABOR AND BUY AMERICA).

6-5 Products and Source of Supply.

6-5.1 Source of Supply–Convict Labor (Federal-Aid Contracts Only): Do not use materials that were produced after July 1, 1991, by convict labor for Federal-aid highway construction projects unless the prison facility has been producing convict-made materials for Federal-aid highway construction projects before July 1, 1987.

Use materials that were produced prior to July 2, 1991, by convicts on Federal-aid highway construction projects free from the restrictions placed on the use of these materials by 23 U.S.C. 114. The Department will limit the use of materials produced by convict labor for use in Federal-aid highway construction projects to:

1. Materials produced by convicts on parole, supervised release, or probation from a prison or,

2. Materials produced in a qualified prison facility.

The amount of such materials produced for Federal-aid highway construction during any 12-month period shall not exceed the amount produced in such facility for use in such construction during the 12-month period ending July 1, 1987.

6-5.2 Source of Supply-Steel: Use steel and iron manufactured in the United States, in accordance with the Buy America provisions of 23 CFR 635.410, as amended. Ensure that all manufacturing processes for this material occur in the United States. As used in this specification, a manufacturing process is any process that modifies the chemical content, physical shape or size, or final finish of a product, beginning with the initial melting and continuing through the final shaping and coating. If a steel or iron product is taken outside the United States for any manufacturing process, it becomes foreign source material. When using steel or iron materials as a component of any manufactured product (e.g., concrete pipe, prestressed beams, corrugated steel pipe, etc.), these same provisions apply. Foreign steel and iron may be used when the total actual cost of such foreign materials does not exceed 0.1% of the total Contract amount or \$2,500, whichever is greater. These requirements are applicable to all steel and iron materials incorporated into the finished work, but are not applicable to steel and iron items that the Contractor uses but does not incorporate into the finished work. Submit a certification from the manufacturer of steel or iron, or any product containing steel or iron, stating that all steel or iron furnished or incorporated into the furnished product was produced and manufactured in the United States or a statement that the product was produced within the United States except for minimal quantities of foreign steel and iron valued at \$ (actual cost). Submit each such certification to the Engineer prior to incorporating the material or product into the project. Prior to the use of foreign steel or iron materials on a project, submit invoices to document the actual cost of such material, and obtain the Engineer's written approval prior to incorporating the material into the project

FROM SECTION 7 – LEGAL REQUIREMENTS AND RESPONSIBILITIES TO THE PUBLIC (FHWA 1273, WAGE RATES, E-VERIFY, TITLE VI, DBE, AND ON-THE-JOB TRAINING).

Compliance with FHWA 1273: The FHWA-1273 Electronic version, dated May 1, 2012 is posted on the Department's website at the following URL address https://fdotwww.blob.core.windows.net/sitefinity/docs/defaultsource/programmanagement/implemented/urlinspecs/files/deo1124628d797c0ec0f48a4a05c478e

<u>4fba9e20.pdf?sfvrsn=83d8df44_2</u>. Take responsibility to obtain this information and comply with all requirements posted on this website up through five calendar days before the opening of bids.

Comply with the provisions contained in FHWA-1273.

If the Department's website cannot be accessed, contact the Department's Specifications Office Web Coordinator at (850) 414-4101.

7-1.4 Compliance with Federal Endangered Species Act and other Wildlife Regulations: The Federal Endangered Species Act requires that the Department investigate the potential impact to a threatened or endangered species prior to initiating an activity performed in conjunction with a highway construction project. If the Department's investigation determines that there is a potential impact to a protected, threatened or an endangered species, the Department will conduct an evaluation to determine what measures may be necessary to mitigate such impact. When mitigation measures and/or special conditions are necessary, these measures and conditions will be addressed in the Contract Documents or permits.

In addition, in cases where certain protected, threatened or endangered species are found or appear within close proximity to the project boundaries, the Department has established guidelines that will apply when interaction with certain species occurs, absent of any special mitigation measures or permit conditions otherwise identified for the project.

These guidelines are posted at the following URL address:<u>https://fdotwww.blob.core.windows.net/sitefinity/docs/default-</u> source/programmanagement/implemented/urlinspecs/files/endangeredwildlifeguidelines.pdf?sfvr sn=e27baf3f_2.

Take responsibility to obtain this information and take all actions and precautions necessary to comply with the conditions of these guidelines during all project activities.

Prior to establishing any off-project activity in conjunction with a project, notify the Engineer of the proposed activity. Covered activities include but are not necessarily limited to borrow pits, concrete or asphalt plant sites, disposal sites, field offices, and material or equipment storage sites. Include in the notification the Financial Project ID, a description of the activity, the location of the site by township, range, section, county, and city, a site location map including the access route, the name of the property owner, and a person to contact to arrange a site inspection. Submit this notification at least 30 days in advance of planned commencement of the off-site activity, to allow for the Department to conduct an investigation without delaying job progress.

Do not perform any off-project activity without obtaining written clearance from the Engineer. In the event the Department's investigation determines a potential impact to a protected, threatened or endangered species and mitigation measures or permits are necessary, coordinate with the appropriate resource agencies for clearance, obtain permits and perform mitigation measures as necessary. Immediately notify the Engineer in writing of the results of this coordination with the appropriate resource agencies. Additional compensation or time will not be allowed for permitting or mitigation, associated with Contractor initiated off-project activities.

7-1.8 Compliance with Section 4(f) of the USDOT Act: Section 4(f) of the USDOT Act prohibits the U. S. Secretary of Transportation from approving a project which requires the use of publicly owned land of a public park, recreation area or a wildlife and waterfowl refuge, or of any historic site of national, state, or local significance unless there is no prudent or feasible alternative to using that land and the program or project includes all possible planning to minimize the harm to the site resulting from the use.

Before undertaking any off-project activity associated with any federally assisted undertaking, ensure that the proposed site does not represent a public park, recreation area, wildlife or waterfowl refuge, or a historic site (according to the results of the Cultural Resources Survey discussed in 120-6.2). If such a site is proposed, notify the Engineer and provide a description of the proposed off-site activity, the Financial Project ID, the location of the site by township, range, section, a county or city map showing the site location, including the access route and the name of the property. It is the Contractor's responsibility to submit justification for use of Section 4(f) property that is sufficient for the Florida Department of Transportation and the Federal Highway Administration to make a Section 4(f) determination. Submit this notification sufficiently in advance of planned commencement of the off-site activity to allow a reasonable time for the Engineer to conduct an investigation without delaying job progress. Do not begin any off-project activity without obtaining written clearance from the Engineer.

7-16 Wage Rates for Federal-Aid Projects.

For this Contract, payment of predetermined minimum wages applies.

The U.S. Department of Labor (USDOL) Wage Rates applicable to this Contract are listed in table below, as modified up through ten days prior to the opening of bids.

Wage Rate Decision Number FL20220193	Associated Work	
	Highway	

Obtain the applicable General Decision(s) (Wage Tables) through the Department's Office of Construction website and ensure that employees receive the minimum compensation applicable. Review the General Decisions for all classifications necessary to complete the project. Request additional classifications through the Engineer's office when needed.

For guidance on the requirements for the payment of wages and benefits and the submittal of certified payrolls, and for general guidance and examples of multiple wage rates when assigned to a Contract, refer to the Department's Office of Construction website. Questions

regarding wage rates and the applicability of wage tables should be submitted in accordance with 2-4.

7-24 Disadvantaged Business Enterprise Program.

7-24.1 Disadvantaged Business Enterprise Affirmative Action Plan: Prior to award of the Contract, have an approved Disadvantaged Business Enterprise (DBE) Affirmative Action Program Plan filed with the Equal Opportunity Office. Update and resubmit the plan every three years. No Contract will be awarded until the Department approves the Plan. The DBE Affirmative Action Program Plan is incorporated into and made a part of the Contract.

7-24.2 Required Contract and Subcontract DBE Assurance Language: In accordance with 49 CFR 26.13 (b), the Contract FDOT signs with the Contractor (and each subcontract the prime contractor signs with a subcontractor) must include the following assurance: "The Contractor, sub-recipient or subcontractor shall not discriminate on the basis of race, color, national origin, or sex in the performance of this contract. The Contractor shall carry out applicable requirements of 49 CFR Part 26 in the award and administration of DOT-assisted Contracts. Failure by the Contractor to carry out these requirements is a material breach of this Contract, which may result in the termination of this Contract or such other remedy as the recipient deems appropriate, which may include, but is not limited to,

1. Withholding monthly progress payments;

2. Assessing sanctions;

3. Liquidated damages; and/or

4. Disqualifying the Contractor from future bidding as non-responsible."

1. A policy statement, signed by an authorized representative (president, chief executive officer, or chairman of the contractor), expressing a commitment to use DBEs in all aspects of contracting to the maximum extent feasible, outlining the various levels of responsibility, and stating the objectives of the program. Circulate the policy statement throughout the Contractor's organization.

2. The designation of a Liaison Officer within the Contractor's organization, as well as support staff, necessary and proper to administer the program, and a description of the authority, responsibility, and duties of the Liaison Officer and support staff. The Liaison Officer and staff are responsible for developing, managing, and implementing the program on a day-today basis for carrying out technical assistance activities for DBEs and for disseminating information on available business opportunities so that DBEs are provided an equitable opportunity to participate in Contracts let by the Department.

3. Utilization of techniques to facilitate DBE participation in contracting activities which include, but are not limited to:

a. Soliciting price quotations and arranging a time for the review of Plans, quantities, specifications, and delivery schedules, and for the preparation and presentation of quotations.

b. Providing assistance to DBEs in overcoming barriers such as the inability to obtain bonding, financing, or technical assistance.

c. Carrying out information and communication programs or workshops on contracting procedures and specific contracting opportunities in a timely manner, with such programs being bilingual where appropriate.

d. Encouraging eligible DBEs to apply for certification with the

Department.

e. Contacting Minority Contractor Associations and city and county agencies with programs for disadvantaged individuals for assistance in recruiting and encouraging eligible DBE contractors to apply for certification with the Department.

7-24.4 DBE Records and Reports: Submit the following through the Equal Opportunity Compliance System:

1. DBE Commitments - at or before the Pre-Construction Conference.

2. Report monthly, through the Equal Opportunity Compliance System on the Department's Website, actual payments (including retainage) made to DBEs for work performed with their own workforce and equipment in the area in which they are certified. Report payments made to all DBE and Minority Business Enterprise (MBE) subcontractors and DBE and MBE construction material and major suppliers.

The Equal Opportunity Office will provide instructions on accessing this system. Develop a record keeping system to monitor DBE affirmative action efforts which include the following:

1. the procedures adopted to comply with these Specifications;

2. the number of subordinated Contracts on Department projects awarded

to DBEs;

3. the dollar value of the Contracts awarded to DBEs;

4. the percentage of the dollar value of all subordinated Contracts awarded to DBEs as a percentage of the total Contract amount;

5. a description of the general categories of Contracts awarded to DBEs;

and

6. the specific efforts employed to identify and award Contracts to DBEs. Upon request, provide the records to the Department for review.

Maintain all such records for a period of five years following acceptance of final payment and have them available for inspection by the Department and the Federal Highway Administration.

7-24.5 Counting DBE Participation and Commercially Useful Functions: 49 CFR Part 26.55 specifies when DBE credit shall be awarded for work performed by a DBE. DBE credit can only be awarded for work actually performed by DBEs themselves for the types of work for which they are certified. When reporting DBE Commitments, only include the dollars that a DBE is expected to earn for work they perform with their own workforce and equipment. Update DBE Commitments to reflect changes to the initial amount that was previously reported or to add DBEs not initially reported.

When a DBE participates in a contract, the value of the work is determined in accordance with 49 CFR Part 26.55, for example:

I. The Department will count only the value of the work performed by the DBE toward DBE goals. The entire amount of the contract that is performed by the DBE's own forces (including the cost of supplies, equipment and materials obtained by the DBE for the contract work) will be counted as DBE credit.

2. The Department will count the entire amount of fees or commissions charged by the DBE firm for providing a bona fide service, such as professional, technical, consultant, or managerial services or for providing bonds or insurance specifically required for the performance of a Department-assisted contract, toward DBE goals, provided that the Department determines the fees to be reasonable and not excessive as compared with fees customarily followed for similar services.

3. When the DBE subcontracts part of the work of its contract to another firm, the Department will count the value of the subcontracted work only if the DBE's subcontractor is itself a DBE. Work that a DBE subcontracts to a non-DBE firm does not count toward DBE goals.

4. When a DBE performs as a participant in a joint venture, the Department will count the portion of the dollar value of the contract equal to the distinct, clearly defined portion of the work the DBE performs with its own forces toward DBE goals.

5. The Contractors shall ensure that only expenditures to DBEs that perform a commercially useful function (CUF) in the work of a contract may be counted toward the voluntary DBE goal.

6. A DBE performs a commercially useful function when it is responsible for execution of the work of the contract and is carrying out its responsibilities by actually performing, managing, and supervising the work involved. To perform a commercially useful function, the DBE must also be responsible, with respect to materials and supplies used on the contract, for negotiating price, determining quality and quantity, ordering the material, and installing (where applicable) and paying for the material itself.

7. Contractors wishing to use joint checks involving DBE credit must provide written notice to the District Contract Compliance Office prior to issuance of the joint check. The Contractor must also provide a copy of the notice to the DBE subcontractor and maintain a copy with the project records.

8. To determine whether a DBE is performing a commercially useful function, the Department will evaluate the amount of work subcontracted, industry practices, whether the amount the firm is to be paid under the contract is commensurate with the work it is actually performing and the DBE credit claimed for its performance of the work, and other relevant factors.

9. A DBE does not perform a commercially useful function if its role is limited to that of an extra participant in a transaction, contract, or project through which funds are passed in order to obtain the appearance of DBE participation.

10. If a DBE does not perform or exercise responsibility for at least 30% of the total cost of its contract with its own workforce, or if the DBE subcontracts a greater portion of the work of a contract than would be expected on the basis of normal industry practice for the type of work involved, the DBE has not performed a commercially useful function.

7-24.6 Prompt Payments: Meet the requirements of 9-5 for payments to all DBE subcontractors.

7-25 On-The-Job Training Requirements.

As part of the Contractor's equal employment opportunity affirmative action program, training shall be provided as follows:

The Contractor shall provide On-The-Job Training aimed at developing full journeymen in the type of trade or job classification involved in the work. In the event the Contractor subcontracts a portion of the contract work, it shall determine how many, if any, of the trainees are to be trained by the subcontractor provided, that the Contractor shall retain the primary responsibility for meeting the training requirements imposed by this Section. Ensure that, when feasible, 25% of trainees in each occupation are in their first year of training. The Contractor shall incorporate the requirements of this Section into such subcontract.

The number of trainees will be estimated on the number of calendar days of the contract, the dollar value, and the scope of work to be performed. The trainee goal will be finalized at a Post-Preconstruction Trainee Evaluation Meeting and the goal will be distributed among the work classifications based on the following criteria:

1. Determine the number of trainees on Federal Aid Contract:

a. No trainees will be required for contracts with a Contract Time allowance of less than 275 calendar days.

b. If the Contract Time allowance is 275 calendar days or more, the number of trainees shall be established in accordance with the following chart:

Estimated Contract Amount	Trainees Required
\$2,000,000 or less	0
Over \$2,000,000 to \$4,000,000	2
Over \$4,000,000 to \$6,000,000	3
Over \$6,000,000 to \$12,000,000	5
Over \$12,000,000 to \$18,000,000	7
Over \$18,000,000 to \$24,000,000	9
Over \$24,000,000 to \$31,000,000	12
Over \$31,000,000 to \$37,000,000	13
Over \$37,000,000 to \$43,000,000	14
Over \$43,000,000 to \$49,000,000	15
Over \$49,000,000 to \$55,000,000	16
Over \$55,000,000 to \$62,000,000	17
Over \$62,000,000 to \$68,000,000	18
Over \$68,000,000 to \$74,000,000	19
Over \$74,000,000 to \$81,000,000	20
Over \$81,000,000 to \$87,000,000	21
Over \$87,000,000 to \$93,000,000	22
Over \$93,000,000 to \$99,000,000	23
Over \$99,000,000 to \$105,000,000	24
Over \$105,000,000 to \$112,000,000	25
Over \$112,000,000 to \$118,000,000	26
Over \$118,000,000 to \$124,000,000	27
Over \$124,000,000 to \$130,000,000	28
Over \$130,000,000 to *	

Further, if the Contractor or subcontractor requests to utilize banked trainees as discussed later in this Section, a Banking Certificate will be validated at this meeting allowing credit to the Contractor for previously banked trainees. Banked credits of prime Contractors working as Subcontractors may be accepted for credit. The Contractor's Project Manager, the Construction Project Engineer and the Department's District Contract Compliance Manager will attend this meeting. Within ten days after the Post-Preconstruction Training Evaluation Meeting, the Contractor shall submit to the Department for approval an On-The-Job Training Schedule indicating the number of trainees to be trained in each selected classification and the portion of the Contract Time during which training of each trainee is to take place. This schedule may be subject to change if any of the following occur:

1. When a start date on the approved On-The-Job Training Schedule has been missed by 14 or more days;

2. When there is a change in previously approved classifications;

3. When replacement trainees are added due to voluntary or involuntary

termination

The revised schedule will be resubmitted to and approved by the Department's District Contract Compliance Manager.

The following criteria will be used in determining whether or not the Contractor has complied with this Section as it relates to the number of trainees to be trained:

1. Credit will be allowed for each trainee that is both enrolled and satisfactorily completes training on this Contract. Credit for trainees, over the established number for this Contract, will be carried in a "bank" for the Contractor and credit will be allowed for those surplus trainees in subsequent, applicable projects. A "banked" trainee is described as an employee who has been trained on a project, over and above the established goal, and for which the Contractor desires to preserve credit for utilization on a subsequent project.

2. Credit will be allowed for each trainee that has been previously enrolled in the Department's approved training program on another contract and continues training in the same job classification and completes their training on a different contract.

3. Credit will be allowed for each trainee who, due to the amount of work available in their classification, is given the greatest practical amount of training on the contract regardless of whether or not the trainee completes training.

4. Credit will be allowed for any training position indicated in the approved On-The-Job Training Schedule, if the Contractor can demonstrate that made a good faith effort to provide training in that classification was made.

5. No credit will be allowed for a trainee whose employment by the Contractor is involuntarily terminated unless the Contractor can clearly demonstrate good cause for this action.

Training and upgrading of minorities, women and economically disadvantaged persons toward journeyman status is a primary objective of this Section. Accordingly, the Contractor shall make every effort to enroll minority trainees and women (e.g., by conducting systematic and direct recruitment through public and private sources likely to yield minority and women trainees) to the extent such persons are available within a reasonable area of recruitment. If a non-minority male is enrolled into the On-The-Job Training Program, the On-The-Job Training Notification of Personnel Action Form notifying the District Contract Compliance Manager of such action shall be accompanied by a disadvantaged certification or a justification for such action acceptable to the Department's District Contract Compliance Manager. The Contractor will be given an opportunity and will be responsible for demonstrating the steps that it has taken in pursuance thereof, prior to a determination as to whether the Contractor is in compliance with this Section. This training is not intended, and shall not be used, to discriminate against any applicant for training, whether a minority, woman or disadvantaged person.

No employee shall be employed as a trainee in any classification in which they have successfully completed a training course leading to journeyman status, or have been employed as a journeyman. The Contractor may satisfy this requirement by including appropriate questions in the employee application or by other suitable means. Regardless of the method used, the Contractor's records should document the findings in each case.

The minimum length and type of training for each classification will be as established at the Post-Preconstruction Trainee Evaluation Meeting and approved by the Department. Graduation to journeyman status will be based upon satisfactory completion of a Proficiency Demonstration set up at the completion of training and established for the specific training classification, completion of the minimum hours in a training classification range, and the employer's satisfaction that the trainee does meet journeyman status in the classification of training. Upon reaching journeyman status, the following documentation must be forwarded to the District Contract Compliance Office:

1. Trainee Enrollment and Personnel Action Form

2. Proficiency Demonstration Verification Form indicating completion of each standard established for the classification signed by representatives of both the Contractor and the Department.

The Department and the Contractor shall establish a program that is tied to the scope of the work in the project and the length of operations providing it is reasonably calculated to meet the equal employment opportunity obligations of the Contractor and to qualify the average trainee for journeyman status in the classifications concerned, by at least, the minimum hours prescribed for a training classification. Furthermore, apprenticeship programs registered with the U.S. Department of Labor, Bureau of Apprenticeship and Training, or with a State apprenticeship agency recognized by the Bureau and training programs approved but not necessarily sponsored by the U.S. Department of Labor, Manpower Administration, Bureau of Apprenticeship and Training shall also be considered acceptable provided it is being administered in a manner consistent with the equal employment obligations of Federal Aid highway construction contract. Approval or acceptance of a training schedule shall be obtained from the Department prior to commencing work on the classifications covered by the program.

A voluntary On-The-Job Training Program is available to a Contractor which has been awarded a state funded project. Through this program, the Contractor will have the option to train employees on state funded projects for "banked credit" as discussed previously in this provision, to be utilized on subsequent Federal Aid Projects where training is required. Those Contractors availing themselves of this opportunity to train personnel on state funded projects and bank trainee hours for credit shall comply with all training criteria set forth in this Section for Federal Aid Projects; voluntary banking may be denied by the Department if staff is not available to monitor compliance with the training criteria.

It is the intention of these provisions that training is to be provided in the construction crafts rather than clerk-typists or secretarial type positions. Training is permissible in lower level management positions such as office engineers, estimators, etc., where the training is oriented toward construction applications. Training in the laborer classifications, except Common/General Laborer, may be permitted provided that significant and meaningful training is provided and approved by the District Contract Compliance Office.

When approved in advance by the District Contract Compliance Manager, credit will be given for training of persons in excess of the number specified herein under the current contract or a Contractor will be allowed to bank trainees who have successfully completed a training program and may apply those trainees to a training requirement in subsequent project(s) upon approval of the Department's District Contract Compliance Manager. This credit will be given even though the Contractor may receive training program funds from other sources, provided such other source do not specifically prohibit the Contractor from receiving other form of compensation. Offsite training is permissible as long as the training is an integral part of an approved training program and does not compromise a significant part of the overall training. Credit for offsite training indicated above may only be made to the Contractor when it does one or more of the following and the trainees are concurrently employed on a Federal Aid Project:

1. Contributes to the cost of the training,

2. Provides the instruction to the trainee,

3. Pays the trainee's wages during the offsite training period.

The Contractor shall compensate the trainee at no less than the laborer rate established in the Contract at the onset of training. The compensation rate will be increased to the journeyman's wage upon graduation from the training program for the remainder of the time the trainee works in the classification in which they were trained.

The Contractor shall furnish the trainee a copy of the program they will follow in providing the training. The Contractor shall provide each trainee with a certification showing the type and length of training satisfactorily completed. The Contractor shall enroll a trainee in one training classification at a time to completion before the trainee can be enrolled in another classification on the same project.

The Contractor shall maintain records to document the actual hours each trainee is engaged in training on work being performed as a part of this Contract.

The Contractor shall submit to the District Contract Compliance Manager a copy of an On-The-Job Training Notification of Personnel Action form no later than seven days after the effective date of the action when the following actions occur: a trainee is transferred on the project, transferred from the project to continue training on another contract, completes training, is upgraded to journeyman status or voluntary terminates or is involuntary terminated from the project.

The Contractor shall furnish to the District Contract Compliance Manager a copy of a Monthly Time Report for each trainee. The Monthly Time Report for each month shall be submitted no later than the tenth day of the subsequent month. The Monthly Time Report shall indicate the phases and sub-phases of the number of hours devoted to each proficiency.

Highway or Bridge Carpenter Helper, Mechanic Helper, Rodman/Chainman, and Timekeeper classifications will not be approved for the On-The-Job Training Program.

The number of trainees may be distributed among the work classifications on the basis of the Contractor's needs and the availability of journeymen in the various classifications within a reasonable area of recruitment.

The Contractor will have fulfilled the responsibilities of this Specification when acceptable training has been provided to the trainee as specified above.

7-26 Cargo Preference Act - Use of United States-Flag Vessels.

Pursuant to Title 46 CFR 381, the Contractor agrees

1. To utilize privately owned United States-flag commercial vessels to ship at least 50 percent of the gross tonnage (computed separately for dry bulk carriers, dry cargo liners, and tankers) involved, whenever shipping any equipment, material, or commodities pursuant to this Contract, to the extent such vessels are available at fair and reasonable rates for United States-flag commercial vessels.

2. To furnish within 20 days following the date of loading for shipments originating within the United States or within 30 working days following the date of loading for

shipments originating outside the United States, a legible copy of a rated, 'on-board' commercial ocean bill-of-lading in English for each shipment of cargo described in paragraph 1 of this Article to both the Contracting Officer (through the prime contractor in the case of subcontractor bills-of-lading) and to the Division of National Cargo, Office of Market Development, Maritime Administration, Washington, DC 20590.

3. To insert the substance of the provisions of this clause in all subcontracts issued pursuant to this Contract.

7-29 E-Verify.

The Contractor shall utilize the U.S. Department of Homeland Security's E-Verify system to verify the employment eligibility of all new employees hired by the Contractor during the term of the Contract and shall expressly require any subcontractors performing work or providing services pursuant to the Contract to likewise utilize the U.S. Department of Homeland Security's E-Verify system to verify the employment eligibility of all new employees hired by the subcontractor during the Contract term.

7-31 Title VI Assurance - DOT 1050.2A, Appendix A and Appendix E.

7-31.1 Appendix A: During the performance of this Contract, the Contractor, for itself, its assignees and successors in interest (hereinafter referred to as the "Contractor") agrees as follows:

1. Compliance with Regulations: The Contractor shall comply with the Regulations relative to nondiscrimination in Federally-assisted programs of the US Department of Transportation (hereinafter, "USDOT") Title 49, Code of Federal Regulations, Part 21, as they may be amended from time to time, (hereinafter referred to as the Regulations), which are herein incorporated by reference and made a part of this Contract.

2. Nondiscrimination: The Contractor, with regard to the work performed by it during the Contract, shall not discriminate on the basis of race, color, national origin or sex in the selection and retention of sub-contractors, including procurements of materials and leases of equipment. The Contractor shall not participate either directly or indirectly in the discrimination prohibited by Section 21.5 of the Regulations, including employment practices when the Contract covers a program set forth in Appendix B of the Regulations.

3. Solicitations for subcontractors, including procurements of materials and equipment: In all solicitations either by competitive bidding or negotiation made by the Contractor for work to be performed under subcontract, including procurements of materials or leases of equipment, each potential subcontractor or supplier shall be notified by the Contractor of the Contractor's obligations under this contract and the Regulations relative to nondiscrimination on the basis of race, color, national origin, or sex.

4. Information and Reports: The Contractor shall provide all information and reports required by the Regulations or directives issued pursuant thereto, and shall permit access to its books, records, accounts, other sources of information and its facilities as may be determined by the Florida Department of Transportation or the Federal Highway Administration, Federal Transit Administration, Federal Aviation Administration, and Federal Motor Carrier Safety Administration to be pertinent to ascertain compliance with such Regulations, order and instructions. Where any information required of a Contractor is in the exclusive possession of another who fails or refuses to furnish this information the Contractor shall so certify to the Florida Department of Transportation, or the Federal Highway Administration, Federal Transit Administration, Federal Aviation Administration, or Federal Motor Carrier Safety Administration as appropriate, and shall set forth what efforts it has made to obtain the information.

5. Sanctions for Noncompliance: In the event of the Contractor's noncompliance with the nondiscrimination provisions of this Contract, the Florida Department of Transportation shall impose such Contract sanctions as it or the Federal Highway Administration, Federal Transit Administration, Federal Aviation Administration, or Federal Motor Carrier Safety Administration may determine to be appropriate, including, but not limited to:

a. withholding of payments to the Contractor under the Contract until the Contractor complies, or

b. cancellation, termination or suspension of the Contract, in whole or in

part.

6. Incorporation of Provisions: The Contractor shall include the provisions of this appendix in every subcontract, including procurements of materials and leases of equipment, unless exempt by the Regulations, or directives issued pursuant thereto. The Contractor shall take such action with respect to any subcontract or procurement as the Florida Department of Transportation or the Federal Highway Administration, Federal Transit Administration, Federal Aviation Administration, or Federal Motor Carrier Safety Administration may direct as a means of enforcing such provisions including sanctions for noncompliance, provided, however, that, in the event a Contractor becomes involved in, or is threatened with, litigation with a subcontractor or supplier as a result of such direction, the Contractor may request the Florida Department of Transportation to enter into such litigation to protect the interests of the Florida Department of Transportation, and, in addition, the Contractor may request the United States to enter into such litigation to protect the interests of the United States.

7-31.2 Appendix E: During the performance of this Contract, the Contractor, for itself, its assignees, and successors in interest (hereinafter referred to as the "Contractor" agrees to comply with the following non-discrimination statutes and authorities; including but not limited to:

1. Title VI of the Civil Rights Act of 1964 (42 U.S.C. § 2000d et seq., 78 stat. 252), (prohibits discrimination on the basis of race, color, national origin); and 49 CFR Part 21;

2. The Uniform Relocation Assistance and Real Property Acquisition Policies Act of 1970, (42 U.S.C. § 4601), (prohibits unfair treatment of persons displaced or whose property has been acquired of Federal or Federal-aid programs and projects);

3. Federal-Aid Highway Act of 1973, (23 U.S.C § 324 et seq.), (prohibits discrimination on the basis of sex);

4. Section 504 of the Rehabilitation Act of 1973 (29 U.S.C. § 794 et seq.), as amended, (prohibits discrimination on the basis of disability); and 49 CFR Part 27;

5. The Age Discrimination Act of 1975, as amended, (42 U.S.C. § 6101 et seq.), (prohibits discrimination on the basis of age);

6. Airport and Airway Improvement Act of 1982, (49 U.S.C. 471, Section 47123), as amended, (prohibits discrimination based on race, creed, color national origins or sex);

7. The Civil Rights Restoration Act of 1987 (PL 100-209), (Broadened the scope, coverage and applicability of Title VI of the Civil Rights Act of 1964, The Age Discrimination Act of 1975 and Section 504 of the Rehabilitation Act of 1973, by expanding the definition of

the terms "programs or activities" to include all of the programs or activities of the Federal-aid recipients, sub-recipients and contractors, whether such programs or activities are Federally funded or not);

8. Titles II and III of the Americans with Disabilities Act, which prohibits discrimination on the basis of disability in the operation of public entities, public and private transportation systems, places of public accommodation, and certain testing entities (42 U.S.C. §§ 12131 – 12189) as implemented by Department of Transportation regulations at 49 C.F.R. parts 37 and 38;

9. The Federal Aviation Administration's Non-discrimination statute (49 U.S.C. § 47123) (prohibits discrimination on the basis of race, color, national origin, and sex);

10. Executive Order 12898, Federal Actions to Address Environmental Justice in Minority Populations and Low-Income Populations, which ensures non-discrimination against minority populations by discouraging programs, policies, and activities with disproportionately high and adverse human health or environmental effects on minority and low-income populations;

11. Executive Order 13166, Improving Access to Services for Persons with Limited English Proficiency, and resulting agency guidance, national origin discrimination includes discrimination because of limited English proficiency (LEP). To ensure compliance with Title VI, you must take reasonable steps to ensure that LEP persons have meaningful access to your programs (70 Fed. Reg. at 74087 to 74100);

12. Title IX of the Education Amendments of 1972, as amended, which prohibits discrimination based on sex in education programs, or activities (20 U.S.C. 1681 et seq.).

FROM SECTION 8 (SUBLETTING, CONTRACT TIME EXTENSIONS, AND LIQUIDATED DAMAGES).

8-1 Subletting or Assigning of Contracts.

Do not, sell, transfer, assign or otherwise dispose of the Contract or Contracts or any portion thereof, or of the right, title, or interest therein, without written consent of the Department. If the Contractor chooses to sublet any portion of the Contract, the Contractor must provide a written request to sublet work on the Certification of Sublet Work form developed by the Department for this purpose. With the Engineer's acceptance of the request, the Contractor may sublet a portion of the work, but shall perform with its own organization work amounting to not less than 40% of the total Contract amount. The Certification of Sublet Work request will be deemed acceptable by the Department, for purposes of the Department's consent, unless the Engineer notifies the Contractor within 5 business days of receipt of the Certification of Sublet Work that the Department is not consenting to the requested subletting.

Include in the total Contract amount the cost of materials and manufactured component products, and their transportation to the project site. For the purpose of meeting this requirement the Department will not consider off-site commercial production of materials and manufactured component products that the Contractor purchases, or their transportation to the project, as subcontracted work.

If the Contractor sublets a part of a Contract item, the Department will use only the sublet proportional cost in determining the percentage of subcontracted normal work.

Execute all agreements to sublet work in writing and include all pertinent provisions and requirements of the Contract. All other agreements must be in writing and reference all applicable Contract provisions. Upon request, furnish the Department with a copy of the subcontract and agreement. The subletting of work does not relieve the Contractor or the surety of their respective liabilities under the Contract.

The Department recognizes a subcontractor only in the capacity of an employee or agent of the Contractor, and the Engineer may require the Contractor to remove the subcontractor as in the case of an employee.

8-7.3.2 Contract Time Extensions: The Department may grant an extension of Contract Time when a controlling item of work is delayed by factors not reasonably anticipated or foreseeable at the time of bid. The Department may allow such extension of time only for delays occurring during the Contract Time period or authorized extensions of the Contract Time period. When failure by the Department to fulfill an obligation under the Contract results in delays to the controlling items of work, the Department will consider such delays as a basis for granting a time extension to the Contract.

Whenever the Engineer suspends the Contractor's operations, as provided in 8-6, for reasons other than the fault of the Contractor, the Engineer will grant a time extension for any delay to a controlling item of work due to such suspension. The Department will not grant time extensions to the Contract for delays due to the fault or negligence of the Contractor.

The Department does not include an allowance for delays caused by the effects of inclement weather or suspension of Contractor's operations in establishing Contract Time. The Engineer will continually monitor the effects of weather and, when found justified, grant time extensions on either a bimonthly or monthly basis. The Engineer will not require the Contractor to submit a request for additional time due to the effects of weather.

The Department will grant time extensions, on a day for day basis, for delays caused by the effects of rains or other inclement weather conditions, related adverse soil conditions or suspension of operations that prevent the Contractor from productively performing controlling items of work resulting in:

1. The Contractor being unable to work at least 50% of the normal work day on pre-determined controlling work items; or

2. The Contractor must make major repairs to work damaged by weather, provided that the damage is not attributable to the Contractor's failure to perform or neglect; and provided that the Contractor was unable to work at least 50% of the normal workday on pre-determined controlling work items.

No additional compensation will be made for delays caused by the effects of inclement weather.

The Department will consider the delays in delivery of materials or component equipment that affect progress on a controlling item of work as a basis for granting a time extension if such delays are beyond the control of the Contractor or supplier. Such delays may include an area-wide shortage, an industry-wide strike, or a natural disaster that affects all feasible sources of supply. In such cases, the Contractor shall furnish substantiating letters from a representative number of manufacturers of such materials or equipment clearly confirming that the delays in delivery were the result of an area-wide shortage, an industry-wide strike, etc. No additional compensation will be made for delays caused by delivery of materials or component equipment. The Department will not consider requests for time extension due to delay in the delivery of custom manufactured equipment such as traffic signal equipment, highway lighting equipment, etc., unless the Contractor furnishes documentation that he placed the order for such equipment in a timely manner, the delay was caused by factors beyond the manufacturer's control, and the lack of such equipment caused a delay in progress on a controlling item of work. No additional compensation will be paid for delays caused by delivery of custom manufactured equipment.

The Department will consider the affect of utility relocation and adjustment work on job progress as the basis for granting a time extension only if all the following criteria are met:

1. Delays are the result of either utility work that was not detailed in the Plans, or utility work that was detailed in the Plans but was not accomplished in reasonably close accordance with the schedule included in the Contract Documents.

controlling work items.

2. Utility work actually affected progress toward completion of

3. The Contractor took all reasonable measures to minimize the effect of utility work on job progress, including cooperative scheduling of the Contractor's operations with the scheduled utility work at the preconstruction conference and providing adequate advance notification to utility companies as to the dates to coordinate their operations with the Contractor's operations to avoid delays.

As a condition precedent to an extension of Contract Time the Contractor must submit to the Engineer:

A preliminary request for an extension of Contract Time must be made in writing to the Engineer within ten calendar days after the commencement of a delay to a controlling item of work. If the Contractor fails to submit this required preliminary request for an extension of Contract Time, the Contractor fully, completely, absolutely and irrevocably waives any entitlement to an extension of Contract Time for that delay. In the case of a continuing delay only a single preliminary request for an extension of Contract Time will be required. Each such preliminary request for an extension of Contract Time shall include as a minimum the commencement date of the delay, the cause of the delay, and the controlling item of work affected by the delay.

Furthermore, the Contractor must submit to the Engineer a request for a Contract Time extension in writing within 30 days after the elimination of the delay to the controlling item of work identified in the preliminary request for an extension of Contract Time. Each request for a Contract Time extension shall include as a minimum all documentation that the Contractor wishes the Department to consider related to the delay, and the exact number of days requested to be added to Contract Time. If the Contractor contends that the delay is compensable, then the Contractor shall also be required to submit with the request for a Contract Time extension a detailed cost analysis of the requested additional compensation. If the Contractor fails to submit this required request for a Contract Time extension, with or without a detailed cost analysis, depriving the Engineer of the timely opportunity to verify the delay and the costs of the delay, the Contractor waives any entitlement to an extension of Contract Time or additional compensation for the delay.

Upon timely receipt of the preliminary request of Contract Time from the Contractor, the Engineer will investigate the conditions, and if it is determined that a controlling item of work is being delayed for reasons beyond the control of the Contractor the Engineer will take appropriate action to mitigate the delay and the costs of the delay. Upon timely receipt of the request for a Contract Time extension the Engineer will further investigate the conditions, and if it is determined that there was an increase in the time or the cost of performance of the controlling item of work beyond the control of the Contractor, then an adjustment of Contract Time will be made, and a monetary adjustment will be made, excluding loss of anticipated profits, and the Contract will be modified in writing accordingly.

The existence of an accepted schedule, including any required update(s), is a condition precedent to the Contractor having any right to the granting of an extension of Contract Time or any monetary compensation arising out of any delay. Contractor failure to have an accepted schedule, including any required update(s), for the period of potential impact, or in the event the currently accepted schedule and applicable updates do not accurately reflect the actual status of the project or fail to accurately show the true controlling or non-controlling work activities for the period of potential impact, will result in any entitlement determination as to time or money for such period of potential impact being limited solely to the Department's analysis and identification of the actual controlling or non-controlling work activities. Further, in such instances, the Department's determination as to entitlement as to either time or compensability will be final, unless the Contractor can prove by clear and convincing evidence to a Disputes Review Board that the Department's determination was without any reasonable factual basis.

8-10 Liquidated Damages for Failure to Complete the Work.

8-10.2 Amount of Liquidated Damages: Applicable liquidated damages are the amounts established in the following schedule:

Original Contract Amount Daily Charge Per	Calendar Day
\$50,000 and under	\$868
Over \$50,000 but less than \$250,000	\$882
\$250,000 but less than \$500,000	\$1,197
\$500,000 but less than \$2,500,000	\$1,694
\$2,500,000 but less than \$5,000,000	\$2,592
\$5,000,000 but less than \$10,000,000	\$3,786
\$10,000,000 but less than \$15,000,000	\$4,769
\$15,000,000 but less than \$20,000,000	\$5,855
\$20,000,000 and over \$9,214 plus (0.00005 of any
amount over \$20 million (Round to nearest	whole dollar)

The Engineer may approve adjustments to the liquidated damages amounts in accordance with the Construction Project Administration Manual (CPAM) provided all contract work is complete.

FROM SECTION 9 (PARTIAL PAYMENTS).

9-5 Partial Payments.

9-5.1 General: The Engineer will make partial payments on monthly estimates based on the amount of work that the Contractor completes during the month (including delivery of certain materials, as specified herein below). The Engineer will make approximate monthly payments,

and the Department will correct all partial estimates and payments in the subsequent estimates and in the final estimate and payment.

The Department will base the amount of such payments on the total value of the work that the Contractor has performed to the date of the estimate, based on the quantities completed and the Contract prices, less payments previously made and less any retainage withheld.

Retainage will not be withheld until the percent of Contract Time used exceeds 75%. From that time forward, the Department will withhold retainage of 5% of the amount due on the current estimate as retainage when the percent of Contract Time used exceeds the percent of Contract amount earned by more than 15%.

Contract amount is defined as the original Contract amount adjusted by approved supplemental agreements.

Retainage will be determined for each job on multiple job Contracts. The Department will not accept Securities, Certificates of Deposit or letters of credit as a replacement for retainage. Amounts withheld will not be released until payment of the final estimate.

9-5.2 Unsatisfactory Payment Record: In accordance with Sections 255.05 and 337.16 of the Florida Statutes, and the rules of the Department, the Department may disqualify the Contractor from bidding on future Department contracts if the Contractor's payment record in connection with contract work becomes unsatisfactory.

9-5.3 Withholding Payment:

9-5.3.1 Withholding Payment for Defective Work: If the Department discovers any defective work or material prior to the final acceptance, or if the Department has a reasonable doubt as to the integrity of any part of the completed work prior to final acceptance, then the Department will not allow payment for such defective or questioned work until the Contractor has remedied the defect and removed any causes of doubt.

9-5.3.2 Withholding Payment for Failure to Comply: The Department will withhold progress payments from the Contractor if he fails to comply with any or all of the following within 60 days after beginning work:

1. comply with and submit required paperwork relating to prevailing wage rate provisions, Equal Employment Opportunity, On-The-Job Training, and Affirmative Action;

2. comply with the requirement to all necessary information, including actual payments to DBEs, all other subcontractors and major suppliers, through the Internet based Equal Opportunity Reporting System;

3. comply with or make a good faith effort to ensure employment opportunity for minorities and females in accordance with the required contract provisions for Federal Aid Construction Contracts, and

4. comply with or make a good faith effort to meet On-The-Job Training goals.

The Department will withhold progress payments until the Contractor has satisfied the above conditions.

9-5.4 Release of Retainage After Acceptance: When the Contractor has furnished the Department with all submittals required by the Contract, such as invoices, EEO reports, materials certifications, certification of materials procured, etc., (excluding Contractor's letter of acceptance of final amount due and Form 21-A release) and the Engineer has determined that the measurement and computation of pay quantities is correct, the Department may reduce the

retainage to \$1,000 plus any amount that the Department elects to deduct for defective work as provided in 9-5.3.

The Department may deduct from payment estimates any sums that the Contractor owes to the Department on any account. Where more than one project or job (separate job number) is included in the Contract, the Department will distribute the reduced retainage as provided in the first paragraph of this Subarticle to each separate project or job in the ratio that the Contract value of the work for the particular job bears to the total Contract amount.

9-5.5 Partial Payments for Delivery of Certain Materials:

9-5.5.1 General: The Department will allow partial payments for new materials that will be permanently incorporated into the project and are stockpiled in approved locations in the project vicinity. Stockpile materials so that they will not be damaged by the elements and in a manner that identifies the project on which they are to be used.

The following conditions apply to all payments for stockpiled materials:

1. There must be reasonable assurance that the stockpiled material will be incorporated into the specific project on which partial payment is made.

2. The stockpiled material must be approved as meeting applicable specifications.

3. The total quantity for which partial payment is made shall not exceed the estimated total quantity required to complete the project.

4. The Contractor shall furnish the Engineer with copies of certified invoices to document the value of the materials received. The amount of the partial payment will be determined from invoices for the material up to the unit price in the Contract.

5. Delivery charges for materials delivered to the jobsite will be included in partial payments if properly documented.

6. Partial payments will not be made for materials which were stockpiled prior to award of the Contract for a project.

9-5.5.2 Partial Payment Amounts: The following partial payment restrictions

apply:

1. Partial payments less than \$5,000 for any one month will not be

processed.

2. Partial payments for structural steel and precast prestressed items will not exceed 85% of the bid price for the item. Partial payments for all other items will not exceed 75% of the bid price of the item in which the material is to be used.

3. Partial payment will not be made for aggregate and base course material received after paving or base construction operations begin except when a construction sequence designated by the Department requires suspension of paving and base construction after the initial paving operations, partial payments will be reinstated until the paving and base construction resumes.

9-5.5.3 Off Site Storage: If the conditions of 9-5.5.1 are satisfied, partial payments will be allowed for materials stockpiled in approved in-state locations. Additionally, partial payments for materials stockpiled in approved out-of-state locations will be allowed if the conditions of 9-5.5.1 and the following conditions are met:

1. Furnish the Department a Materials Bond stating the supplier guarantees to furnish the material described in the Contract to the Contractor and Department. Under this bond, the Obligor shall be the material supplier and the Obligees shall be the Contractor and the Florida Department of Transportation. The bond shall be in the full dollar amount of the bid price for the materials described in the contract.

2. The following clauses must be added to the construction Contract between the Contractor and the supplier of the stockpiled materials:

"Notwithstanding anything to the contrary, <u><supplier></u> will be liable to the Contractor and the Florida Department of Transportation should <u><supplier></u> default in the performance of this agreement."

"Notwithstanding anything to the contrary, this agreement, and the performance bond issued pursuant to this agreement, does not alter, modify, or otherwise change the Contractor's obligation to furnish the materials described in this agreement to the Florida Department of Transportation."

3. The agreement between the Contractor and the supplier of the stockpiled materials must include provisions that the supplier will store the materials and that such materials are the property of the Contractor.

9-5.6 Certification of Payment to Subcontractors: The term "subcontractor," as used herein, includes persons or firms furnishing materials or equipment incorporated into the work or stockpiled for which the Department has made partial payment and firms working under equipment-rental agreements. The Contractor is required to pay all subcontractors for satisfactory performance of their Contracts before the Department will make a further progress (partial) payment. The Contractor shall also return all retainage withheld to the subcontractors within 30 days after the subcontractor's work is satisfactorily complete, as determined by the Department. Prior to receipt of any progress (partial) payment, the prime contractor shall certify that all subcontractors having an interest in the Contract were paid for satisfactory performance of their Contracts and that the retainage is returned to subcontractors within 30 days after satisfactory completion of the subcontractor's work. Provide this certification in the form designated by the Department.

Within 30 days of the Contractor's receipt of the final progress payment or any other payments thereafter, except the final payment, the Contractor shall pay all subcontractors and suppliers having an interest in the Contract for all work completed and materials furnished. The Department will honor an exception to the above when the Contractor demonstrates good cause for not making any required payment and furnishes written notification of any such good cause to both the Department and the affected subcontractors or suppliers within said 30 day period.

The Contractor shall indemnify and provide defense for the Department when called upon to do so for all claims or suits against the Department, by third parties, pertaining to Contractor payment or performance issues arising out of the Contract. It is expressly understood that the monetary limitation on the extent of the indemnification shall be the approved Contract amount, which shall be the original Contract amount as may be increased by subsequent Supplemental Agreements.

April 14, 2023 PREPARED BY: Paul Webb, P.E. Derek M. Gil, P.E.

SUPPLEMENTAL SPECIFICATIONS PACKAGE NUMBER ONE FINANCIAL PROJECT ID(S): 441241-1-58-01

DISTRICT TWO NASSAU COUNTY

The July 2020 Edition of the Florida Department of Transportation Standard Specifications, the Specifications Package for this project, dated <u>February 4, 2022</u>, attached thereto, are revised as follows:

I hereby certify that this specifications package has been properly prepared by me, or under my responsible charge, in accordance with procedures adopted by the Florida Department of Transportation.

This item has been digitally signed and sealed by Derek M. Gil, PE on the date adjacent to the seal. Printed copies of this document are not considered signed and sealed and the signature must be verified on any electronic copies.

Date:	April 14, 2023	
State of Florida,		1
Professional Engineer, License No .:	54798	5 /
Firm/Agency Name:	Element Engineering Group, LLC	*
Firm/Agency Address:	1713 E. 9th Avenue	31
City, State, Zip Code:	Tampa, Florida, 33605	1.P
Page(s):	1-13	11



-1-

SUPPLEMENTAL ONE

SUPPLEMENTAL SPECIFICATIONS

SUPPLEMENTAL ONE

FROM SECTION 7 – LEGAL REQUIREMENTS AND RESPONSIBILITIES TO THE PUBLIC (FHWA 1273, WAGE RATES, E-VERIFY, TITLE VI, DBE, AND ON-THE-JOB TRAINING)., page(s) 21-32, is(are) deleted and the following substituted:

FROM SECTION 7 – LEGAL REQUIREMENTS AND RESPONSIBILITIES TO THE PUBLIC (FHWA 1273, WAGE RATES, E-VERIFY, TITLE VI, DBE, AND ON-THE-JOB TRAINING)

7-1.1Compliance with FHWA 1273: The FHWA-1273 Electronic version, dated July 5, 2022 is posted on the Department's website at the following URL address <u>https://fdotwww.blob.core.windows.net/sitefinity/docs/default-</u> source/programmanagement/implemented/urlinspecs/files/fhwa1273-7-5-

<u>22.pdf?sfvrsn=726ca05d_2</u>Take responsibility to obtain this information and comply with all requirements posted on this website up through five calendar days before the opening of bids.

Comply with the provisions contained in FHWA-1273.

If the Department's website cannot be accessed, contact the Department's Specifications Office Web Coordinator at (850) 414-4101.

7-1.4 Compliance with Federal Endangered Species Act and other Wildlife Regulations: The Federal Endangered Species Act requires that the Department investigate the potential impact to a threatened or endangered species prior to initiating an activity performed in conjunction with a highway construction project. If the Department's investigation determines that there is a potential impact to a protected, threatened or an endangered species, the Department will conduct an evaluation to determine what measures may be necessary to mitigate such impact. When mitigation measures and/or special conditions are necessary, these measures and conditions will be addressed in the Contract Documents or permits.

In addition, in cases where certain protected, threatened or endangered species are found or appear within close proximity to the project boundaries, the Department has established guidelines that will apply when interaction with certain species occurs, absent of any special mitigation measures or permit conditions otherwise identified for the project.

These guidelines are posted at the following URL address:<u>https://fdotwww.blob.core.windows.net/sitefinity/docs/default-</u> source/programmanagement/implemented/urlinspecs/files/endangeredwildlifeguidelines. pdf?sfvrsn=e27baf3f_2.

Take responsibility to obtain this information and take all actions and precautions necessary to comply with the conditions of these guidelines during all project activities.

Prior to establishing any off-project activity in conjunction with a project, notify the Engineer of the proposed activity. Covered activities include but are not necessarily limited to borrow pits, concrete or asphalt plant sites, disposal sites, field offices, and material or equipment storage sites. Include in the notification the Financial Project ID, a description of the activity, the location of the site by township, range, section, county, and city, a site location map including the access route, the name of the property owner, and a person to contact to arrange a site inspection. Submit this

SUPPLEMENTAL ONE

notification at least 30 days in advance of planned commencement of the off-site activity, to allow for the Department to conduct an investigation without delaying job progress.

Do not perform any off-project activity without obtaining written clearance from the Engineer. In the event the Department's investigation determines a potential impact to a protected, threatened or endangered species and mitigation measures or permits are necessary, coordinate with the appropriate resource agencies for clearance, obtain permits and perform mitigation measures as necessary. Immediately notify the Engineer in writing of the results of this coordination with the appropriate resource agencies. Additional compensation or time will not be allowed for permitting or mitigation, associated with Contractor initiated off-project activities.

7-1.8 Compliance with Section 4(f) of the USDOT Act: Section 4(f) of the USDOT Act prohibits the U. S. Secretary of Transportation from approving a project which requires the use of publicly owned land of a public park, recreation area or a wildlife and waterfowl refuge, or of any historic site of national, state, or local significance unless there is no prudent or feasible alternative to using that land and the program or project includes all possible planning to minimize the harm to the site resulting from the use.

Before undertaking any off-project activity associated with any federally assisted undertaking, ensure that the proposed site does not represent a public park, recreation area, wildlife or waterfowl refuge, or a historic site (according to the results of the Cultural Resources Survey discussed in 120-6.2). If such a site is proposed, notify the Engineer and provide a description of the proposed off-site activity, the Financial Project ID, the location of the site by township, range, section, a county or city map showing the site location, including the access route and the name of the property. It is the Contractor's responsibility to submit justification for use of Section 4(f) property that is sufficient for the Florida Department of Transportation and the Federal Highway Administration to make a Section 4(f) determination. Submit this notification sufficiently in advance of planned commencement of the off-site activity to allow a reasonable time for the Engineer to conduct an investigation without delaying job progress. Do not begin any off-project activity without obtaining written clearance from the Engineer.

7-16 Wage Rates for Federal-Aid Projects.

For this Contract, payment of predetermined minimum wages applies.

The U.S. Department of Labor (USDOL) Wage Rates applicable to this Contract are listed in table below, as modified up through ten days prior to the opening of bids.

Wage Rate Decision Number	County	Associated Work
FL20220193	Nassau	Highway

Review the General Decisions for all classifications necessary to complete the project. Request additional classifications through the Engineer's office when needed. 7-24 Disadvantaged Business Enterprise Program.

7-24.1 Disadvantaged Business Enterprise Affirmative Action Plan: Prior to award of the Contract, have an approved Disadvantaged Business Enterprise (DBE) Affirmative Action Program Plan filed with the Equal Opportunity Office. Update and resubmit the plan every three years. No Contract will be awarded until the Department approves the Plan. The DBE Affirmative Action Program Plan is incorporated into and made a part of the Contract.

7-24.2 Required Contract and Subcontract DBE Assurance Language: In accordance with 49 CFR 26.13 (b), the Contract FDOT signs with the Contractor (and each subcontract the prime contractor signs with a subcontractor) must include the following assurance: "The Contractor, sub-recipient or subcontractor shall not discriminate on the basis of race, color, national origin, or sex in the performance of this contract. The Contractor shall carry out applicable requirements of 49 CFR Part 26 in the award and administration of DOT-assisted Contracts. Failure by the Contractor to carry out these requirements is a material breach of this Contract, which may result in the termination of this Contract or such other remedy as the recipient deems appropriate, which may include, but is not limited to,

1. Withholding monthly progress payments;

2. Assessing sanctions;

3. Liquidated damages; and/or

4. Disqualifying the Contractor from future bidding as non-responsible." 7-24.3 Plan Requirements: Include the following in the DBE Affirmative Action Program Plan:

1. A policy statement, signed by an authorized representative (president, chief executive officer, or chairman of the contractor), expressing a commitment to use DBEs in all aspects of contracting to the maximum extent feasible, outlining the various levels of responsibility, and stating the objectives of the program. Circulate the policy statement throughout the Contractor's organization.

2. The designation of a Liaison Officer within the Contractor's organization, as well as support staff, necessary and proper to administer the program, and a description of the authority, responsibility, and duties of the Liaison Officer and support staff. The Liaison Officer and staff are responsible for developing, managing, and implementing the program on a day-to-day basis for carrying out technical assistance activities for DBEs and for disseminating information on available business opportunities so that DBEs are provided an equitable opportunity to participate in Contracts let by the Department.

3. Utilization of techniques to facilitate DBE participation in contracting activities which include, but are not limited to:

a. Soliciting price quotations and arranging a time for the review of Plans, quantities, specifications, and delivery schedules, and for the preparation and presentation of quotations.

b. Providing assistance to DBEs in overcoming barriers such as the inability to obtain bonding, financing, or technical assistance.

e. Carrying out information and communication programs or workshops on contracting procedures and specific contracting opportunities in a timely manner, with such programs being bilingual where appropriate.

SUPPLEMENTAL ONE

d. Encouraging eligible DBEs to apply for certification with the

Department.

e. Contacting Minority Contractor Associations and city and county agencies with programs for disadvantaged individuals for assistance in recruiting and encouraging eligible DBE contractors to apply for certification with the Department.

7-24.4 DBE Records and Reports: Submit the following through the Equal Opportunity Compliance System:

1. DBE Commitments - at or before the Pre-Construction Conference.

2. Report monthly, through the Equal Opportunity Compliance System on the Department's Website, actual payments (including retainage) made to DBEs for work performed with their own workforce and equipment in the area in which they are certified. Report payments made to all DBE and Minority Business Enterprise (MBE) subcontractors and DBE and MBE construction material and major suppliers.

The Equal Opportunity Office will provide instructions on accessing this system. Develop a record keeping system to monitor DBE affirmative action efforts which include the following:

1. the procedures adopted to comply with these Specifications;

2. the number of subordinated Contracts on Department projects

awarded to DBEs;

3. the dollar value of the Contracts awarded to DBEs;

4. the percentage of the dollar value of all subordinated Contracts awarded to DBEs as a percentage of the total Contract amount;

5. a description of the general categories of Contracts awarded to

DBEs; and

6. the specific efforts employed to identify and award Contracts to

DBEs.

Upon request, provide the records to the Department for review.

Maintain all such records for a period of five years following acceptance of final payment and have them available for inspection by the Department and the Federal Highway Administration.

7-24.5 Counting DBE Participation and Commercially Useful Functions: 49 CFR Part 26.55 specifies when DBE credit shall be awarded for work performed by a DBE. DBE credit can only be awarded for work actually performed by DBEs themselves for the types of work for which they are certified. When reporting DBE Commitments, only include the dollars that a DBE is expected to earn for work they perform with their own workforce and equipment. Update DBE Commitments to reflect changes to the initial amount that was previously reported or to add DBEs not initially reported.

When a DBE participates in a contract, the value of the work is determined in accordance with 49 CFR Part 26.55, for example:

1. The Department will count only the value of the work performed by the DBE toward DBE goals. The entire amount of the contract that is performed by the DBE's own forces (including the cost of supplies, equipment and materials obtained by the DBE for the contract work) will be counted as DBE credit.

2. The Department will count the entire amount of fees or commissions charged by the DBE firm for providing a bona fide service, such as professional, technical, consultant, or managerial services or for providing bonds or

SUPPLEMENTAL ONE

insurance specifically required for the performance of a Department-assisted contract, toward DBE goals, provided that the Department determines the fees to be reasonable and not excessive as compared with fees customarily followed for similar services.

3. When the DBE subcontracts part of the work of its contract to another firm, the Department will count the value of the subcontracted work only if the DBE's subcontractor is itself a DBE. Work that a DBE subcontracts to a non-DBE firm does not count toward DBE goals.

4. When a DBE performs as a participant in a joint venture, the Department will count the portion of the dollar value of the contract equal to the distinct, clearly defined portion of the work the DBE performs with its own forces toward DBE goals.

5. The Contractors shall ensure that only expenditures to DBEs that perform a commercially useful function (CUF) in the work of a contract may be counted toward the voluntary DBE goal.

6. A DBE performs a commercially useful function when it is responsible for execution of the work of the contract and is carrying out its responsibilities by actually performing, managing, and supervising the work involved. To perform a commercially useful function, the DBE must also be responsible, with respect to materials and supplies used on the contract, for negotiating price, determining quality and quantity, ordering the material, and installing (where applicable) and paying for the material itself.

7. Contractors wishing to use joint checks involving DBE credit must provide written notice to the District Contract Compliance Office prior to issuance of the joint check. The Contractor must also provide a copy of the notice to the DBE subcontractor and maintain a copy with the project records.

8. To determine whether a DBE is performing a commercially useful function, the Department will evaluate the amount of work subcontracted, industry practices, whether the amount the firm is to be paid under the contract is commensurate with the work it is actually performing and the DBE credit claimed for its performance of the work, and other relevant factors.

9. A DBE does not perform a commercially useful function if its role is limited to that of an extra participant in a transaction, contract, or project through which funds are passed in order to obtain the appearance of DBE participation.

10. If a DBE does not perform or exercise responsibility for at least 30% of the total cost of its contract with its own workforce, or if the DBE subcontracts a greater portion of the work of a contract than would be expected on the basis of normal industry practice for the type of work involved, the DBE has not performed a commercially useful function.

7-24.6 Prompt Payments: Meet the requirements of 9-5 for payments to all DBE subcontractors.

7-25 On-The-Job Training Requirements.

As part of the Contractor's equal employment opportunity affirmative action program, training shall be provided as follows:

The Contractor shall provide On-The-Job Training aimed at developing full journeymen in the type of trade or job classification involved in the work. In the
event the Contractor subcontracts a portion of the contract work, it shall determine how many, if any, of the trainees are to be trained by the subcontractor provided, that the Contractor shall retain the primary responsibility for meeting the training requirements imposed by this Section. Ensure that, when feasible, 25% of trainees in each occupation are in their first year of training. The Contractor shall incorporate the requirements of this Section into such subcontract.

The number of trainees will be estimated on the number of calendar days of the contract, the dollar value, and the scope of work to be performed. The trainee goal will be finalized at a Post-Preconstruction Trainee Evaluation Meeting and the goal will be distributed among the work classifications based on the following criteria:

1. Determine the number of trainees on Federal Aid Contract:

a. No trainees will be required for contracts with a Contract Time allowance of less than 275 calendar days.

b. If the Contract Time allowance is 275 calendar days or more, the number of trainees shall be established in accordance with the following chart:

Estimated Contract Amount	Trainees Required
\$2,000,000 or less	0
Over \$2,000,000 to \$4,000,000	2
Over \$4,000,000 to \$6,000,000	3
Over \$6,000,000 to \$12,000,000	5
Over \$12,000,000 to \$18,000,000	7
Over \$18,000,000 to \$24,000,000	9
Over \$24,000,000 to \$31,000,000	12
Over \$31,000,000 to \$37,000,000	13
Over \$37,000,000 to \$43,000,000	14
Over \$43,000,000 to \$49,000,000	15
Over \$49,000,000 to \$55,000,000	16
Over \$55,000,000 to \$62,000,000	17
Over \$62,000,000 to \$68,000,000	18
Over \$68,000,000 to \$74,000,000	19
Over \$74,000,000 to \$81,000,000	20
Over \$81,000,000 to \$87,000,000	21
Over \$87,000,000 to \$93,000,000	22
Over \$93,000,000 to \$99,000,000	23
Over \$99,000,000 to \$105,000,000	24
Over \$105,000,000 to \$112,000,000	25
Over \$112,000,000 to \$118,000,000	26
Over \$118,000,000 to \$124,000,000	27
Over \$124,000,000 to \$130,000,000	28
Over \$130,000,000 to *	
*One additional trainee per \$6,000,000 of estimated Construction Contract amount over \$130,000,000	

Further, if the Contractor or subcontractor requests to utilize banked trainees as discussed later in this Section, a Banking Certificate will be validated at this meeting

-8-

SUPPLEMENTAL ONE

FPID(S):441241-1-58-01

allowing credit to the Contractor for previously banked trainees. Banked credits of prime Contractors working as Subcontractors may be accepted for credit. The Contractor's Project Manager, the Construction Project Engineer and the Department's District Contract Compliance Manager will attend this meeting. Within ten days after the Post-Preconstruction Training Evaluation Meeting, the Contractor shall submit to the Department for approval an On-The-Job Training Schedule indicating the number of trainees to be trained in each selected classification and the portion of the Contract Time during which training of each trainee is to take place. This schedule may be subject to change if any of the following occur:

1. When a start date on the approved On-The-Job Training Schedule has been missed by 14 or more days;

2. When there is a change in previously approved classifications;

3. When replacement trainees are added due to voluntary or involuntary termination

The revised schedule will be resubmitted to and approved by the Department's District Contract Compliance Manager.

The following criteria will be used in determining whether or not the Contractor has complied with this Section as it relates to the number of trainees to be trained:

1. Credit will be allowed for each trainee that is both enrolled and satisfactorily completes training on this Contract. Credit for trainees, over the established number for this Contract, will be carried in a "bank" for the Contractor and credit will be allowed for those surplus trainees in subsequent, applicable projects. A "banked" trainee is described as an employee who has been trained on a project, over and above the established goal, and for which the Contractor desires to preserve credit for utilization on a subsequent project.

2. Credit will be allowed for each trainee that has been previously enrolled in the Department's approved training program on another contract and continues training in the same job classification and completes their training on a different contract.

3. Credit will be allowed for each trainee who, due to the amount of work available in their classification, is given the greatest practical amount of training on the contract regardless of whether or not the trainee completes training.

4. Credit will be allowed for any training position indicated in the approved On-The-Job Training Schedule, if the Contractor can demonstrate that made a good faith effort to provide training in that classification was made.

5. No credit will be allowed for a trainee whose employment by the Contractor is involuntarily terminated unless the Contractor can clearly demonstrate good cause for this action.

Training and upgrading of minorities, women and economically disadvantaged persons toward journeyman status is a primary objective of this Section. Accordingly, the Contractor shall make every effort to enroll minority trainees and women (e.g., by conducting systematic and direct recruitment through public and private sources likely to yield minority and women trainees) to the extent such persons are available within a reasonable area of recruitment. If a non-minority male is enrolled into the On-The-Job Training Program, the On-The-Job Training Notification of Personnel Action Form notifying the District Contract Compliance Manager of such action shall be accompanied by a disadvantaged certification or a justification for such action acceptable to the

FPID(S):441241-1-58-01

Department's District Contract Compliance Manager. The Contractor will be given an opportunity and will be responsible for demonstrating the steps that it has taken in pursuance thereof, prior to a determination as to whether the Contractor is in compliance with this Section. This training is not intended, and shall not be used, to discriminate against any applicant for training, whether a minority, woman or disadvantaged person.

No employee shall be employed as a trainee in any classification in which they have successfully completed a training course leading to journeyman status, or have been employed as a journeyman. The Contractor may satisfy this requirement by including appropriate questions in the employee application or by other suitable means. Regardless of the method used, the Contractor's records should document the findings in each case.

The minimum length and type of training for each classification will be as established at the Post-Preconstruction Trainee Evaluation Meeting and approved by the Department. Graduation to journeyman status will be based upon satisfactory completion of a Proficiency Demonstration set up at the completion of training and established for the specific training classification, completion of the minimum hours in a training classification range, and the employer's satisfaction that the trainee does meet journeyman status in the classification of training. Upon reaching journeyman status, the following documentation must be forwarded to the District Contract Compliance Office:

1. Trainee Enrollment and Personnel Action Form

2. Proficiency Demonstration Verification Form indicating completion of each standard established for the classification signed by representatives of both the Contractor and the Department.

The Department and the Contractor shall establish a program that is tied to the scope of the work in the project and the length of operations providing it is reasonably calculated to meet the equal employment opportunity obligations of the Contractor and to qualify the average trainee for journeyman status in the classifications concerned, by at least, the minimum hours prescribed for a training classification. Furthermore, apprenticeship programs registered with the U.S. Department of Labor, Bureau of Apprenticeship and Training, or with a State apprenticeship agency recognized by the Bureau and training programs approved but not necessarily sponsored by the U.S. Department of Labor, Manpower Administration, Bureau of Apprenticeship and Training shall also be considered acceptable provided it is being administered in a manner consistent with the equal employment obligations of Federal Aid highway construction contract. Approval or acceptance of a training schedule shall be obtained from the Department prior to commencing work on the classifications covered by the program.

A voluntary On-The-Job Training Program is available to a Contractor which has been awarded a state funded project. Through this program, the Contractor will have the option to train employees on state funded projects for "banked credit" as discussed previously in this provision, to be utilized on subsequent Federal Aid Projects where training is required. Those Contractors availing themselves of this opportunity to train personnel on state funded projects and bank trainee hours for credit shall comply with all training criteria set forth in this Section for Federal Aid Projects; voluntary banking may be denied by the Department if staff is not available to monitor compliance with the training criteria.

It is the intention of these provisions that training is to be provided in the construction crafts rather than clerk-typists or secretarial type positions. Training is

FPID(S):441241-1-58-01

permissible in lower level management positions such as office engineers, estimators, etc., where the training is oriented toward construction applications. Training in the laborer classifications, except Common/General Laborer, may be permitted provided that significant and meaningful training is provided and approved by the District Contract Compliance Office.

When approved in advance by the District Contract Compliance Manager, credit will be given for training of persons in excess of the number specified herein under the current contract or a Contractor will be allowed to bank trainees who have successfully completed a training program and may apply those trainees to a training requirement in subsequent project(s) upon approval of the Department's District Contract Compliance Manager. This credit will be given even though the Contractor may receive training program funds from other sources, provided such other source do not specifically prohibit the Contractor from receiving other form of compensation. Offsite training is permissible as long as the training is an integral part of an approved training program and does not compromise a significant part of the overall training. Credit for offsite training indicated above may only be made to the Contractor when it does one or more of the following and the trainees are concurrently employed on a Federal Aid Project:

1. Contributes to the cost of the training,

2. Provides the instruction to the trainee,

3. Pays the trainee's wages during the offsite training period.

The Contractor shall compensate the trainee at no less than the laborer rate established in the Contract at the onset of training. The compensation rate will be increased to the journeyman's wage upon graduation from the training program for the remainder of the time the trainee works in the classification in which they were trained.

The Contractor shall furnish the trainee a copy of the program they will follow in providing the training. The Contractor shall provide each trainee with a certification showing the type and length of training satisfactorily completed. The Contractor shall enroll a trainee in one training classification at a time to completion before the trainee can be enrolled in another classification on the same project.

The Contractor shall maintain records to document the actual hours each trainee is engaged in training on work being performed as a part of this Contract.

The Contractor shall submit to the District Contract Compliance Manager a copy of an On-The-Job Training Notification of Personnel Action form no later than seven days after the effective date of the action when the following actions occur: a trainee is transferred on the project, transferred from the project to continue training on another contract, completes training, is upgraded to journeyman status or voluntary terminates or is involuntary terminated from the project.

The Contractor shall furnish to the District Contract Compliance Manager a copy of a Monthly Time Report for each trainee. The Monthly Time Report for each month shall be submitted no later than the tenth day of the subsequent month. The Monthly Time Report shall indicate the phases and sub-phases of the number of hours devoted to each proficiency.

Highway or Bridge Carpenter Helper, Mechanic Helper, Rodman/Chainman, and Timekeeper classifications will not be approved for the On-The-Job Training Program.

The number of trainees may be distributed among the work classifications on the basis of the Contractor's needs and the availability of journeymen in the various classifications within a reasonable area of recruitment.

The Contractor will have fulfilled the responsibilities of this Specification when acceptable training has been provided to the trainee as specified above.

7-29 E-Verify.

The Contractor shall utilize the U.S. Department of Homeland Security's E-Verify system to verify the employment eligibility of all new employees hired by the Contractor during the term of the Contract and shall expressly require any subcontractors performing work or providing services pursuant to the Contract to likewise utilize the U.S. Department of Homeland Security's E-Verify system to verify the employment eligibility of all new employees hired by the subcontractor during the Contract term.

THIS COMPLETES THIS SPECIFICATIONS PACKAGE





9	
8	
8	
8	
PP	
2	
×.	
×	
贸	
Ť	
53	
2	
Ö.	
5	
õ	
ö	
e.	
흻	
P.	
m	
Sig	
B	
ŏ	

COUNTY: NASSAU COUNTY
CHANNELIZING DEVICE - TYPES I, II, DI, VP, DRUM, DR LCD

	PROJECT SUMMARY OF PAY ITEMS - 0002 SUMMARY OF SIGNING		
PROJECT: 441241-1	COUNTY: NASSAU COUNTY		
		TIMIT	QUANT ITY
ITEM NO.	ITEM DESCRIPTION	ITNIA	TU INI
711-11-123	THERMOPLASTIC, STD, WHITE SOLID FOR CROSSWALK AND ROUNDABOUT 12"	LF	835
711-11-125	THERMOPLASTIC, STD, WHITE SOLID FOR STOP LINE AND CROSSWALK 24"	LF	520

SHEET m

SUMMARY OF PAY ITEMS

FINANCIAL PROJECT ID N23/2021 10:45-48 MM 441241-1

Elligh Color

JAMES H. PITMAN JR. CITRONA DRIVE S PFE LICENSE NUMBER 42035 CITRONA DRIVE S PEG BOX 3223 PAG BOX 3235 PAG BOX 32056 CITRONA DR NASSAU COUNTY CITRONA DR NASSAU COUNTY

DESCRIPTION

REVISIONS I DATE

DATE

CITRONA DRIVE SIDEWALK

		SHEET NO.
ILT FEWCE, AND OTHER ITEMS NECESSARY TO HITS AND/OR ENTERING INLETS & CUIVERTS. GER. (CAVATION NECESSARY FOR CONSTRUCTION OF TENERS, ETC) NEEOED FOR CUIDERALL CT THE THICKENED EDGE FOR THE PIPE CT THE THICKENED EDGE FOR THE PIPE PLASTIC MARKINGS. COST TO BE INCLUDED IN		PROJECT NOTES
THE THE MULT THE THE MULT STANDED OF SECURIENT BARRIEN - INCLUDES STATETIC BALES, STAKED SILT FEWEE, AND OTHER ITEMS WEGESSART TO RINNARDOD - SECURIENT SECURENT FROM LEAVING THE PROJECT LUNTS AND/OR EATERNING INLETS & CULVERTS. DIADATION AND PREVENT SECURENT FROM LEAVING THE PROJECT LUNTS AND/OR EATERNING INLETS & CULVERTS. DIADATION OF ENGLARE SCAWATION (3-A PROJECTS ONLY) - INCLUDES ALL EXCAVATION MEGESSART FOR CONSTRUCTION OF THE PROJECT. SIS-1-2 INCLUDES ALL MATERIALS (SCREWS, PLATES SPLICES, SLEEVES, FASTEMERS, ETC) NEEDED FOR GUISFRALL CONSTRUCTON SIS-1-2 INCLUDES ALL MATERIALS (SCREWS, PLATES SPLICES, SLEEVES, FASTEMERS, ETC) NEEDED FOR GUIDERALL CONSTRUCTION SIS-1-2 INCLUDES ALL MATERIALS (SCREWS, PLATES ALE SCAVATION THE THICKENED EDGE FOR THE PIPE CONSTRUCTION SIS-1-2 INCLUDES ALL MATERIALS (SCREWS, PLATES ALE SCAVATION THE THICKENED EDGE FOR THE PIPE CONSTRUCTION SIS-1-2 INCLUDES ALL MATERIALS (SCREWS, PLATES ALE CONSTRUCT THE THICKENED EDGE FOR THE PIPE SJ2-2 USE FELLON, WET-SET DETECTABLE WARNING PADS. SJ2-2 USE TELLON, WET-SET DETECTABLE WARNING PADS. SJ1-11 APPLY AN AKTLIC PRIMER TO AREAS THAT ARE TO RECEIVE THERMOPLASTIC MARKIMOS. COST TO BE INCLUDED IN THE COST OF THE THERMOPLASTIC (TEM.		CITRONA DRIVE SIDEWALK READ NO COUNTY FINANCLA PRDIECT ID CITRONA NASSAU COUNTY 441241-1 DR
SW .	CONTACT PHONE MUMBERS. CONTACT PHONE MUMBERS. activity of FERMANDIA Back. CITY of FERMANDIA Back. TO AN CITAGE CONTUNCTIONS CONTACTOR SHALL CORDINATE ALL WORK WITH OFFINE MORE CONTACTOR SHALL CORDINATE ALL WORK WITH OFFINE CONTACTORS WITH MASSAU COUNT SCHOOL DISTRICT FACILITIES MANAGER IS REQUIRED PHONE TO CONSTRUCTION INSPECTOR. MD MASSAU COUNT SCHOOL DISTRICT FACILITIES MANAGER IS REQUIRED PHONE TO CONSTRUCTION INSPECTOR. MD ULL FORMACTOR SHALL CONDUCTION MEETING WITH THE MASSAU COUNT ENGINEED ALLONG FRANCING. 12. NO WORK SHALL CONSTRUCT FACILITIES MANAGER IS REQUIRED PHONE TO CONSTRUCTION INSPECTOR. MD MASSAU COUNT SCHOOL DISTRICT FACILITIES MANAGER IS REQUIRED PHONE TO CONSTRUCTION INSPECTOR. MD 13. NO WORK SHALL CONSTRUCT FACILITIES MANAGER IS RECOMIRED PHONE TO CONSTRUCTION 14. Inter CONTT CONTT CONTENT FACE AND CONSTRUCTION CONSTRUCTION 15. NO WORK SHALL CONTT FACE AND COUNT REVENTION OF SIDEWALS AND CONSTRUCTION CONSTRUCTION 15. OF ASPHULT SURFACE, FLUSH WITH THE EXCITING CURR AND CONSTRUCTION CONSTRUCTION OF SIDEWALS AND CONSTRUCTION CONSTRUCTION CONSTRUCTION OF SIDEWALS AND CONSTRUCTION CONSTRUCTION CONSTRUCTION CONSTRUCTION CONSTRUCTION CONSTRUCTION OF SIDEWALS AND CONSTRUCTION CONTRUCTION CONSTRUCTION CONS	INMES H. PITMAN JR P.E. LICENSE NUMBER 42035 P.E. LICENSE NUMBER 42035 P.G. DX 3823 LAVE GTT, FLORUPA 32056
TERTING CONDITIONS AND DIMENSION TERDED ON THE PLANS. SHOULD AN THE THE MORK. CITH THE MORK. CITH THE MORK. CITH THE MORK. CITH THE MORK. THE GONSTRUCTION STRET IN A SECURE M REM ACCESS BY THE GENERAL PUBL. BY AURD FROM A BOUNDARY SURVEY BUSINESS # 2108. DIMEN FROM STRET IN ASSAU COUNT BUSINESS # 2108. DIMET PLANS ARE APPONED. COUNTY RJW SHALL BE SODDED. IN THE CONSTRUCTION LIMITS SHALL IN THE CONSTRUCTION LIMITS SHALL IN THE CONSTRUCTION LIMITS SHALL IN THE CONSTRUCTION LIMITS SHALL IN THE CONSTRUCTION LIMITS SHALL CONSTRUCTION. CONTRACTOR SHALL ACT BIT OR 1-800-432-4770 AT LEAS ACT BIT OR 1-800-432-4770 AT LEAS	 DIND FARANGGIO REX LESTER AEX LESTER TELERDIC TEAN FILERDIC TEAN CHANS HEBERG DANID JOHNSON COUNTY ERCOMIZED HOLD NUCTION OF SIDEWALKS AND CURRING AND CONSING GENE LASSERRE BUDIC CALL JOH-SIG GENE LASSERRE BUDIC CALL JOHNSON TIRCH BUTLLE RESHAPTING AND CONSTAULING AND CON THE EXISTING ADJACENT SUBFACES 	ND 1470-2530 5 80
GENERAL NOTES GENERAL NOTES THE CONTRACTOR SHALL VERIFY ALL EXISTING CONDITIONS AND DIMENSIONS AT THE JOB SITE TO ENSURE THAT ALL NEW WORK WILL IT IN THE MANER INTENDED ON THE PLANS. SHOUD ANT CONDITIONS EXIST THAT ARE ALL NEW WORK SHALL MAINTERN THE CONSTITUTE FLANS. SHOUD ANT CONDITIONS EXIST THAT ARE CONTRARY TO THOSE SHALL BE TANKEN WOTH THE WORK. CONTRACTOR SHALL MAINTEN THE CONSTITUTE IN A SECURE MANNER. ALL OPEN TRENCES CONTRARY TO THOSE SHALL BE PROTECTED FROM ACCESS BY THE GENERAL PUBLIC. 2. THE CONTRACTOR SHALL MAINTAN THE CONSTRUCTION SITE IN A SECURE NAMMER. ALL OPEN TRENCHES AND EXCANTED AREAS SHALL BE PROTECTED FROM ACCESS BY THE GENERAL PUBLIC. 3. BOUNDARY INFORMATION SHOWN ON THE PLANS ARE NOTH ANENICAL DATUM OF 1988 (NAVD BBI. 4. BENCHWARK ELEVATIONS SHOWN ON THE PLANS ARE NOTH ANENICAL DATUM OF 1988 (NAVD BBI. 5. BOUNDARY INFORMATION SHOWN ON THE PLANS ARE NOTH ANENICAL DATUM OF 1988 (NAVD BBI. 5. BENCHWARK ELEVATIONS SHOWN ON THE PLANS ARE NOTH AMENICAL VERTICAL DATUM OF 1988 (NAVD BBI. 5. CHE SIDCATED IN TOWNSHIP 3 WORTH RAVIS SHALL BESTOWN, FLURIDA. 6. ALL DISTUBBED AREAS WITHIN MASSAU COUNTY RAVIS SHALL REMAIN UNLESS NOTED OTHERMISE. 7. EXISTING DANIMAGE STRUCTURES WITHIN THE CONSTRUCTION LIMITS SHALL REMAIN UNLESS NOTED OTHERMISE. 8. THE LOCATION OF THE UTILITIES SHOWN OF CONSTRUCTION. CONTRAVATE DUCT. THE EXACT LOCATION SHALL BE FOLLOWING DATIVED STRUCTURES WITHIN THE CONSTRUCTION. CONTRAVEL'S AUGUST. THE EXANT BOUNT. THE INTERSES AND 8. THE LOCATION OF THE UTILITIES SHOWN OF CONSTRUCTION. CONTRAVEL PANNA BE DAVE. THE CONTRACTOR SHALL BENER 8. THE LOCATION OF THE UTILITIES SHOWN CONSTRUCTION. CONTRAVELED AND THE SHALL BERDINGE SHALL BERDINGED BATCHER SHALL NOT CONSTRUCTION. CONTRACTOR SHALL REAVILE ALL THENTHER 8. THE LOCATION OF THE UTILITIES SHOWN CONSTRUCTION. CONTRACTOR SHALL REAVEL WATURES AND THE THE BENERAL DESIDERED SHALLONG CONSTRUCTION. CONTRACTOR SHALL REAVEL AL UTILITIES WITHIN THE CONTRACTOR OF THE UTILITIES SHALLING CONSTRUCTION. CONTRACTOR SHALL REAV	CONTACT PHONE WINNERS. CONTACT PHONE WINNERSH. FISTORIA CTTO F FERMINION CTTO F FERMINION CTTO F FERMINION CTTO F SERVICUMIN CTTO F SERVICU	9 10 1 5 1 0 17 9 10 1 5 1 0 17 9 10 1 5 1 0 17



HE OFFICIAL RECORD OF THIS SHEEF IS THE RECTRONIC FILE DIGITALLY SIGNED AND SEALED UNDER AULE BIGIS-23.004, F.AC



					-									.21	* # **	00.65	-5191	באדע מגנוכואר שבכמשם מג גאווז פאועבע וא גאדע ערבעשמאוכ גורע סומוג. איין איין איין איין איין איין איין איין	7
																			PROJECT CONTROL SHEET
POINTS	DESCRIPTION	SET 5/8" IRC W/ALUN CAP STAMPED "64512 V1"	SET 5/8" IRC SSMC TRAV	FND S/8" IRC MANZIE LB 7039	FND 5/2" IRC NANZIE LB 7039	FND S/8" IRC NANZIE LB 7039	FND 5/8" IRC MANZIE LB 7039	FND 5/8" IRC MANZIE LB 7039	SET NGD SSMC ELEV	SET 5/8" IRC SSMC TRAV	FND 5/8" IRC MANZIE LB 7039	FND 5/8" IRC PRN LB 7039	SET 5/8" IRC SSMC TRAV	SET 5/8" IRC SSMC TRAV	SET N&D SSNC ELEV		#X# CNON W/BRASS DISK STANPED "64512 HV2"		ULI KUNA UKIYE Road at Countr
Y CONTROL POINTS	ELEVATION	15,610	17.320	0.000	0:000	0.000	0:000	0.000	19.850	19.440	0.000	0:000	20.480	18.260	18.490	21.640	20.750		5
SURVEY	EASTING	516233.7410	516306.7270	516316.0455	516311.1598	516300.2315	516295.7233	516294.7371	516297.0110	516228.9370	516297.5915	516307.2095	516338.8140	516486.8240	516505,6070	516591.1520	516643.3060	e suever	NUMBER 4203 3 LORIDA 32056
	NORTHING	2299681.2010	2299999,5890	2300033.8670	2300242.1820	2300406.7540	2300481.7380	2300551.8400	2300600.8060	2300652.9650	2300652.9740	2300767.5270	2301201.8500	2301635.2480	2301715.6520	2302306.0720	2302691.1820		P.E. LICENSE NUMBER 42035 NFPS NC. BOX 3823 LACE CITY, FLORIDA 32056
	POINT NUMBER	101	102	103	104	105	106	101	108	109	110	111	112	113	114	115	116	102	DE SCRIPTION
																		101-1-1-1-1-104 8EVISIONS	
																			DESCRIPTION
																		-2=	370

																				#50		<u>3113 21NO</u> 2	PI #51	_			SHEET NO.
RUCTIC	EASTING	\$ \$16524.3115	516529.7616	516538.0898	516543.1234		516583.7484	516593.3805	516597.9109	516597,7831	5 516597.8543	516599.2465	1 516603,6910	516612.0431	516611.2947	516618.2402				Id Lot# Id Lag# Id Lot# Id		PI #44-	Pi s42-				
OR & CONST	NORTHING	2301814.5273	2301855.4063	2301923.0943	2301964.4912	2502179.7370	2302267.4829	2302347,3603	2302377.2308	2302421.7044	2302426.6735	2302437.0919	2302453.7911	2302513.2069	2302524.2401	2302573.2926						#36 PI #37	-PI #38 PI #42 PI #42-				1
PI TABLE FOR	STATION	119+94.83	120+36.07	121+04.27	121+45.97	123+63.20	124+51.67	125+32.13	125+62.34	125+06.82	125+11.78	126+22.30	126+39.58	126+99.58	127+10.63	127+60.18				R	#33 PI #34 PI #35	5 4			ļ		ALK
	Pla	37	38	39	40	15	42	43	44	45	46	15	48	49	- 20	51				* 10	PI #33	_	+	†			SIDEW
																						#30 #30					CITRONA DRIVE SIDEWALK
CTION	EASTING	516261.0554	516270.7623	516275.6105	516289.3069	516299.1107	516305.4904	516318.2284	516334.7643	516365.1112	516408,2071	516417.8612	516432.6604	516449.4085	516453.2992	516457.8022	516476.7336	516484.1358			PI #27	PI #29	11-1-1				CITRO
TABLE FOR & CONSTRUCTION	NORTHING	2300741.7215	2300821.5752	2300852.9978	2300929.3575	2300978.1612	2301007.6689	2301057.5207	2301115.8454	2301205.3741	2301321.5456	2301345.9616	2301386.5754	2301439.5188	2301453,2639	2301539.3169	2301658.8556	2301705,1303		PPI #26	1	1-2-4		í			2035
	STATION	108+80.76	109+61.20	109+92.99	110+70.57	111+20.35	111+50.56	112+01.99	112+62.62	113+57.15	114481.06	115+07.31	115+50.54	116+06.07	116+20.35	117+06.52	118+27.55	118+74.42	0.00	12# Id	PI #22						JAMES H. PITMAN JR. P.E. LICENSE NUMBER 42035
Id	#Id	61	8	21	22	23	24	25	56	12	82	8	8	31	32	33	34	35	R	2	#18 #19 PI #20		ĺ				P.E. LICE
UCTION	EASTING	516229.7324	516241.6713	516252.5802	516259.8616	516264,6014	516271.6379	516272.3728	516272.1108	516268.1862	516257.2924	516256.7950	516255.8216	516252.3718	516251.5715	516251.2359	516252.7053	516255.9877	c/netezale	L	14 [PI #14 12 [PI #15 #16]				V DEWALKI		DESCRIPTION
PI TABLE FOR & CONSTRUCTION	NORTHING	2299877.1948	2299881.2424	2299889.8185	2299968.6990	2300048.7817	2300075.4457	2300081.1374	0519,5110052	2300191.6145	2300351.1624	2300373.2183	2300386,3944	2300474.5575	2300504.8698	2300580.5821	2300627.6214	2300685,3009	C820102 MMC2		II# Id_ II# Id_				- E CONSTRUCTION CITRONA DR (SIDEWALK)		I O N S DATE
TABLE FO	STATION	100+00:00	100+12.61	100+26.48	101+05.70	101+85.92	102+13.50	102+19.24	102+52.01	103+29.81	104+89.73	105+11.80	105+25.01	106+13.24	106+43.56	107+19.27	107+66.34	108+24.11	ac.664801				+		,		REVISIONS
	#Id	I	2	m		5	9	7	80	6	10	п	12	13	Id	15	16	17	2				- 7- 7- 7d				NELIGIK















F

			20	SUMMARY	OF LUM	P SUM ITEMS		1000	
	PAY ITEM NO.	PAY IT	EM DESCRI	PTION	P 00	ANTITY	DESIGN	CONSTRUC REMARK	S
	1 1010	MOBIFIZATION			-				
SUMMARY OF		TRAFFIC	CONTRO!	L PLAN	ITEMS				
PAY ITEM DESCRIPTION	UNIT DI	URATION	QUANT	1.4.12		1.11	DESIGN NOTE.		KS
	3	DAIS			-		SAVG 59 (151)		
MAINTENANCE OF TRAFFIC	3 B	65			2	75			
CHANNELIZING DEVICE- TYPES 1, 11, D1, VP, DRUN, OR LCD	-	65	20	-	13	00			
		SLIMM	OF	ROSION	AND SE	DIMENT CC		CES	
		OCATION	5	-	SI	EDIMENT BAR			
			T	-	_	0104 10	-	_	CTION
	ST/	4. TO STA.			10	LF	NOTES	-	KS
			1			Р	F	_	
	100+00.5	10	19.60	LT	2F2A	213.8			
	103+90.1	10	90.20	LT 7	821F	1.2			
	103+90.2	co.	02.01	LT 7	8220	20.5			
	103+95.2	to	05.70	LT	975A	28.0			
	104+10.6	to	10.70	LT 2	8221	1.0			
	104+45.6	to	20.75	LT 2	8223	27.8			
	104+50.7	0.1	55.75	-	9750	37.0			
	105+16.3	to	35.47	1	8224	21.9		-	
	105+21.5	to	30.59	+	975C	25.3			
	C'17+C01	2		+	0020	74.0			
	107+46.1	10	66.23	+	27.50 97.5E	27.0			
	112+49.5	to	84.58	+	2F31	39.4			
	112+54.5	ţa	79.58	LT [975F	57.9			
	113+25.0	40	45.00	LT	9726	20.0			
	113+30.0	10	40.00	LT .	9760	27.0			
	114+66.5	10	16.02	-	2F3C	\$3.4			
	117+59.5	10	19.65		2F3D	104.2			
	117+64.5	to.	54.67	17	9762	5. 181			
	118+84.0	0	40.11		07.0F	46.6			
	1 214351	0 0	42.41	5	2F3F	145.8			
	OFTEET		50 18	t	782C3	23.3			
					-TOTAL:	1271			
					TOTAL:	1271			
JANES H. PITMAN JR.		CITRONA D	DRIVE SIDE	WALK	-				SHEET
P.C. BOX 3823	RDAD	NO. COU	ALTY FINA	NCIAL PROJECT		UMMARY	OF QUANT	TTTES	SOI
LAKE CITY, FLORINA SCUDA	ATION 29011 CITHUN		_	441241-1	-				
	SUMMARY O SUMMARY O PAY ITEM DESCRIPTION MAINTERANCE OF TRAFFIC NORE ZONE STOR NORE ZONE STOR DOCUMENTING DEVICE: TTPES 1, 11, D1, VP, DMUR, OR LCC CHAMBELIZING DEVICE: TTPES 1, 11, D1, VP, DMUR, OR LCC DAMMELIZING DEVICE: TTPES 1, 11, D1, VP, DMUR, OR LCC DAMMELIZING DEVICE: TTPES 1, 11, D1, VP, DMUR, OR LCC DAMMELIZING DEVICE: TTPES 1, 11, D1, VP, DMUR, OR LCC DAMMELIZING DEVICE: TTPES 1, 11, D1, VP, DMUR, OR LCC DAMMELIZING DEVICE: TTPES 1, 11, D1, VP, DMUR, OR LCC DAMMELIZING DEVICE: TTPES 1, 11, D1, VP, DMUR, OR LCC DAMMELIZING DEVICE: TTPES 1, 11, D1, VP, DMUR, OR LCC DAMMELIZING DEVICE: TTPES 1, 11, D1, VP, DMUR, OR LCC DAMMELIZING DEVICE: TTPES 1, 11, D1, VP, DMUR, OR LCC DAMMELIZING DEVICE: TTPES 1, 11, D1, VP, DMUR, OR LCC DAMMELIZING DEVICE: TTPES 1, 11, D1, VP, DMUR, OR LCC DAMMELIZING DEVICE: TTPES 1, 11, D1, VP, DMUR, OR LCC DAMMELIZING DEVICE: TTPES 1, 11, D1, VP, DMUR, OR LCC DAMMELIZING DEVICE: TTPES 1, 11, D1, VP, DMUR, OR LCC DAMMELIZING DEVICE: TTPES 1, 11, D1, VP, DMUR, OR LCC DAMMERIZING DEVICE: TTPES 1, 11, D1, VP, DMUR, OR LCC DAMMERIZING DEVICE: TTPES 1, 11, D1, VP, DMUR, OR LCC DAMMERIZING DEVICE: TTPES 1, 11, D1, VP, DMUR, OR LCC DAMEE AND	IPTION IPTION DI. VP. DRUN, OR LCD DI. VP. DRUN, OR LCD LICENSE WUMBER 42005 Social 33028 ECTIVE FOR DA 33028	PAY ITEM NO. 0101 1 1 SUIMMARY OF TEMPORARY DU SUIMMARY OF TEMPORARY DU SUIMMARY OF TEMPORARY DU SUIMMARY OF TEMPORARY DU SUIMMARY OR DU DU DI. VP. DRUM. OR DU DI. VP. DRUM. OR CD DI. VP. DRUM. OR CD LO DI. VP. DRUM. DI LO LO DI DI DI DI LO LO LO LO LO LO	PAY ITEM PAY ITEM DESC 0101 1 PAY ITEM DESC 0101 1 PAY ITEM DESC 0101 1 PAY ITEM DESC 0101 1 DESC 0101 1	PAY ITEM PAY ITEM DESCRIPTION POID NOOL NOOL DESCRIPTION POID NOOL NOOL PAY IPTION NOIT NOOL PAY IPTION UNIT DEAYS P I IPTION UNIT DAYS P I I IPTION INT DAYS P I I IPTION IPTION IPTION IT I I IPTION IPTION IPTION IT I I IPTION IPTION	SUMMARY OF ITEM DESCRIPTION SUMMARY OF TAME TAFFIC CONTROL PLAN ITEMS SUMMARY OF TEMPORARY TRAFFIC CONTROL PLAN ITEMS SUMMARY OF EROSION MIN TRAFFIC CONTROL PLAN ITEMS SUMMARY OF EROSION MIN TRAFFIC CONTROL PLAN ITEMS SUMMARY OF EROSION MIN TRAFFIC SUMMARY OF EROSION MIN TRAFFIC SUMMARY OF EROSION AND S SIGNARY SIGN OF IT 2024 SUMMARY OF EROSION AND S SIGNARY O	SUMMARY OF ITEM DESCRIPTION SUMMARY ITEM DESCRIPTION O SUMMARY OF TEMPORARY TRAFFIC CONTROL PLAN ITEMS SUMMARY OF TEMPORARY TRAFFIC CONTROL PLAN ITEMS SUMMARY OF TEMPORARY TRAFFIC CONTROL PLAN ITEMS OU. VP. DOWN. OR LOD DURATION OUNTITY P DIV DURATION CONTROL PLAN ITEMS SUMMARY OF TEMPORARY TRAFFIC CONTROL NULL DURATION OUNTITY P Diversities DAYS P I D Diversities DIVERTION OUNTITY P P Diversities DIVERTION DIVERTION DIVERTION DI Diversities DIVERTION DIVERTION DIVERTION DI Diversities Diversities Diversities Diversities Diversities Diversities Diversities Diversities Diversities Diversities Diversities Diversities Diversities Diversities Diversities Diversities Diversities Diversities	SUMMARY OF ITEM DESCRIPTION SUMMARY OF TAME TAFFIC CONTROL PLAN ITEMS SUMMARY OF TEMPORARY TRAFFIC CONTROL PLAN ITEMS SUMMARY OF EROSION MIN TRAFFIC CONTROL PLAN ITEMS SUMMARY OF EROSION MIN TRAFFIC CONTROL PLAN ITEMS SUMMARY OF EROSION MIN TRAFFIC SUMMARY OF EROSION MIN TRAFFIC SUMMARY OF EROSION AND S SIGNARY SIGN OF IT 2024 SUMMARY OF EROSION AND S SIGNARY O	Park ITEM SUMMARY OF CONTINUE OF

ISign Envelope ID: 5D3013DB-2C27-40B2-8CCD-57B583C9BC42	
5	
DocuSig	

F NOTES REMARKS																						1.5553		SS	
	+																							SUMMARY OF QUANTITIES	
4		-			-	-	-	-	-	-				-	+	-	+	+	-	-	-			Y OF Q	
1.1	0.12	-	-			-	_	-		416	-		_	-	-	-		-	-	-	-			MMAR	
٩	0					_	1	_	-		_				_	-		_	-	_	_		_	SU	
L															_									LK PROJECT ID	1 196100
Ρ	0.02	10.0	0.01	0.01	0.01	0.01	10.0	0.01	0.02	125	ø	a	4	Q	14	a	9	19	50	52	66		55	SIDEWAL	
SIDE	LT	11	5	LT .	Ц	LT .	LT	LT .	17	RT	LT	U.	11	17	U	11	LT	17	LT .	17	LT		2	A DRIVE	and the second sec
TA.	102+12.85	104+05.2	C1.C0+201	107+61.17	112+79.58	113440.	115+11.14	118+54.75	120+10.66	102+15.25	104+05.2	104+65.75	105+30.8	107+61.17	112+79.58	113+40.	114+87.06	115+11.01	118+54.75	120+05.68	127+54.18		127+54.18	ROAD N	
10	10	01	10	to	to	to	to	10	10	01	07	to	40	40	10	to	to	01	03	to	to		02		
ST/	100+02.49	103+95.2	15-124501	10.12+201	112+54.5	113+30.	114+71.56	117+64.5	118+89.67	1004001	103+95.2	104+50.75	105+21.14	107+51.17	112+54.5	113+30.	114+71.62	114+71.62	117+64.5	118+89.67	126+03.62		126+03, 62	N JR. MBER 42035	22000 000
PAY ITEM DESCRIPTION	CLEARING AND GRUBBING									DE EVICT	IC EVISI													JAMES H. PITMA P.E. LICENSE NU NFPS	P.O. BOX 3823
PAY ITEM NO.	1 1 0110										0110 4 10													DESCRIPTION	
												ļ.						1	1					REVISION DESCRIPTION	
	ITEM NO. PAY ITEM DESCRIPTION STA. TO STA. 3100	ITEM NO. PAY ITEM DESCRIPTION STA. TO STA. 20 JUC 10 1 CLEARING AND GRUBBING 100+02.49 to 102+12.85 LT	ITEM NO. PAY ITEM DESCRIPTION STA. TO STA. 310LE 10 1 1 CLEARING AND GAUBBING 100+02.49 to 102+12.85 LT 10 3 104-05.2 LT 10 1 1 004-05.49 to 104-05.2 LT	ITEM NO. PAY ITEM DESCRIPTION STA. TO STA. 3105 10 1 CLEARING AND GAUGBING 100402.49 10 102412.65 LT 103495.2 10 104405.2 LT 104490.15 10 104465.75 LT	ITEM NO. PAY ITEM DESCRIPTION STA. TO STA. 3105 10 1 1 CLEARING AND GRUBBING 100402.49 to 102412.65 LT 103495.2 to 104405.2 LT 103495.5 to 104465.75 LT 104450.15 to 105430.99 LT 104450.01 to 107461.17 LT	ITEM NO. PAY ITEM DESCRIPTION STA. TO STA. DUC 10 1 CLEARING AND GRUBBING 100+02.49 to 107+12.65 LT 10 1 CLEARING AND GRUBBING 100+02.49 to 107+12.65 LT 10 1 CLEARING AND GRUBBING 103+95.2 to 104+55.2 LT 10 1 103+95.2 to 104+55.75 LT 10 105+21.61 to 105+30.99 LT 10 107+51.01 to 107+61.17 LT 11 107+51.61 to 112+79.58 LT	ITEM NO. PAY ITEM DESCRIPTION STA. TO STA. 10 1 CLEARING AND GRUBBING 100+02.49 to 102+12.65 LT 10 1 CLEARING AND GRUBBING 100+02.49 to 102+12.65 LT 10 1 CLEARING AND GRUBBING 100+02.49 to 104+05.2 LT 10 1 103+95.15 to 104+05.75 LT 10 105+21.61 to 105+30.99 LT 10 105+21.61 to 107+61.17 LT 11 112+954.5 to 112+99.64 LT 113+30. to 113+40. LT	ITEM NO. PAY ITEM DESCRIPTION STA. TO STA. 10 1 CLEARING AND GRUBBING 100+02.49 to 102+12.65 LT 10 1 CLEARING AND GRUBBING 100+02.49 to 102+12.65 LT 10 1 CLEARING AND GRUBBING 100+02.49 to 10445.75 LT 10 103+95.2 to 10445.75 LT LT 100+51.61 to 105+30.99 LT 112+54.5 to 112+79.58 LT 113+50. to 112+79.58 LT 113+50. to 112+79.58 LT	ITEM NO. PAY ITEM DESCRIPTION STA. TO STA. 10 1 CLEARING AND GRUBBING 100+02.49 to 102+12.65 LT 10 1 CLEARING AND GRUBBING 100+02.49 to 102+12.65 LT 10 1 CLEARING AND GRUBBING 100+02.49 to 102+12.65 LT 10 1 CLEARING AND GRUBBING 103+92.15 to 104+95.75 LT 10 103+51.01 to 104+51.17 LT LT 10 101+51.01 to 107+61.17 LT 11 112+54.5 to 112+79.58 LT 113+50. to 112+79.58 LT 113+50. to 112+79.58 LT 113+50. to 112+79.58 LT 113+50. to 112+79.58 LT	ITEM NO. PAY ITEM DESCRIPTION STA. TO STA. 10 1 CLEARING AND GRUBBING 100402.49 to 102412.65 LT 10 1 CLEARING AND GRUBBING 100402.49 to 102412.65 LT 10 1 CLEARING AND GRUBBING 103495.2 to 104456.75 LT 10 1 103455.2 to 104456.75 LT 10 107451.01 to 105450.99 LT 10 107451.01 to 107451.17 LT 11 107451.01 to 112479.58 LT 113420. 10 113440. LT LT 1134471.56 to 113440. LT 1134471.55 to 113440. LT 1134471.56 to 113440. LT	ITEM NO. PAY ITEM DESCRIPTION STA. TO STA. 10 1 CLEARING AND GRUBBING 100+02.49 to 102+12.65 LT 10 1 CLEARING AND GRUBBING 100+02.49 to 102+12.65 LT 10 1 CLEARING AND GRUBBING 103+95.2 to 104+50.75 LT 10 103+51.51 to 104+50.75 LT LT 10 107+51.01 to 107+51.07 LT 10 107+51.01 to 107+51.07 LT 11 107+51.01 to 112+92.68 LT 11 113+30. to 112+92.68 LT 113+40. 113+40. to 113+40.0 LT 113+40. 113+40.5 to 113+40.0 LT 113+40. 113+40.5 to 113+42.5 LT 113+40. 113+40.5 to 113+42.5 LT 113+40. 113+40.5 to 113+42.55 MT	ITEM NO. PAY ITEM DESCRIPTION STA. TO STA. 10 1 CLEARING AND GRUBBING 100+02.49 to 102+12.65 LT 10 1 CLEARING AND GRUBBING 100+02.49 to 102+12.65 LT 10 1 CLEARING AND GRUBBING 103+95.2 to 104-95.2 LT 10 10 103+51.51 to 104+50.75 LT 10 10 105+21.51 to 107+51.17 LT 10 107+51.01 to 107+51.17 LT 11 107+51.01 to 112+95.45 LT 11 113+40. to 113+40. LT 110 10 118+89.57 to 118+84.75 LT 110 10 100+00.12 to 120+10.66 LT 110 10 100+00.12 to 120+10.52 LT	ITEM NO. PAY ITEM DESCRIPTION STA. TO STA. 10 1 CLEARING AND GRUBBING 100+02.49 to 102+12.65 LT 10 1 CLEARING AND GRUBBING 100+02.49 to 102+12.65 LT 10 1 103+95.2 to 102+12.65 LT 10 103+95.15 to 104+65.75 LT 10 103+51.01 to 107+51.01 LT 10 107+51.01 to 107+51.01 LT 10 107+51.01 to 107+51.17 LT 112+954.5 to 112+95.48 LT 113+40. 113+40. to 113+40. LT 113+454.5 to 113+40. LT 114+71.56 to 113+40. LT 11944.5 to 118+83.57 to 118+43.75 110 4 10 RENNOVAL OF EXIST CONC 100+00.12 to 129+10.66 110 4 10 RENNOVAL OF EXIST CONC 100+00.12 to 104+05.2 110 4 10 RENNOVAL OF EXIST CONC 100+00.12 to 104+05.2 110 4 10 RENNOVAL OF EXIST CONC 100+00.12 to 104+05.2 <	ITEM NO. PAY ITEM DESCRIPTION $STA.$ TO STA. $310c$ 10 1 CLEARING AND GRUBBING 100+02.49 to 102+12.65 LT 10 1 CLEARING AND GRUBBING 100+02.49 to 102+12.65 LT 10 103+95.2 to 102+12.65 LT LT 10 103+95.1 to 104+50.75 LT LT 10 103+91.01 to 104+51.75 LT LT 10 107+51.01 to 107+41.17 LT LT 11 107+51.01 to 112+95.45 to 112+95.46 LT 11 110 11 107+51.55 to 113+40. LT 11 113+45.5 to 113+40. LT LT 110<4	ITEM NO. PAY ITEM DESCRIPTION $STA.$ TO STA. $310c$ 10 1 CLEARING AND GRUBBING 100+02.49 to 102+12.85 LT 10 1 CLEARING AND GRUBBING 100+02.49 to 102+12.85 LT 10 10 103+95.2 to 104+65.75 LT LT 10 10 103+51.01 to 104+65.75 LT LT 10 103+51.01 to 107+51.01 to 107+51.17 LT 10 10 107+51.01 to 107+51.05 LT LT 11 107+51.01 to 107+41.17 LT LT LT 11 107+51.54 to 112+95.48 LT LT LT 11 110+40.5 to 112+46.5 to 112+46.5 LT LT 110+41.0 113+46.5 to 113+46.5 to 112+41.7 LT 110 10 113+46.5 to 112+41.2 LT LT 110 110 113+46.5	ITEM NO. PAY ITEM DESCRIPTION $STA.$ TO STA. $310c$ 10 1 CLEARING AND GRUBBING 100+02.49 to 102+12.65 LT 10 1 CLEARING AND GRUBBING 100+02.49 to 102+12.65 LT 10 10 103+95.2 to 104+50.75 LT LT 10 10 103+51.51 to 104+51.75 LT LT 10 107+51.01 to 107+51.17 LT LT LT 10 10 107+51.01 to 107+51.17 LT LT 11 107+51.51 to 107+51.17 LT LT LT 11 107+51.54 to 112+95.48 LT LT LT 11 110 11 112+91.5 LT LT LT LT 110 11 113+40. LT 113+40. LT LT LT 110 110 11 113+41.2 LT LT LT LT LT 110 110	ITEM NO. PAY ITEM DESCRIPTION $STA.$ TO $STA.$ TO $STA.$ SUC SUC 10 1 CLEARING AND GRUBBING 100+02.49 to 102+12.65 LT LT 10 1 CLEARING AND GRUBBING 100+02.49 to 107+12.65 LT LT 10 10 10 103+51.51 to 104+50.75 LT LT 10 10 107+51.01 to 107+51.17 LT LT 10 10 10 107+51.01 to 107+51.17 LT LT 11 10 10 107+51.01 to 112+90.8 LT LT 11 11 11 11 11 11 LT LT LT 11 11 11 11 11 11 LT LT 11 11 11 11 11 11 LT LT 11	ITEM NO. PAY ITEM DESCRIPTION $STA.$ TO $STA.$ TO $STA.$ SUC 10 1 CLEARING AND GRUDBING 100+02.49 to 102+12.65 LT 10 1 CLEARING AND GRUDBING 100+02.49 to 107+12.65 LT 10 10 103+95.2 to 104+65.75 LT LT 10 10 107+51.01 to 107+51.17 LT LT 10 10 107+51.01 to 107+61.17 LT LT 11 107+51.01 to 107+51.01 to 117+90.2 LT LT 11 117+64.5 to 112+92.68 to 112+91.2 LT 11 113+30.57 to 112+91.25 RT LT 10 113+49.5 to 118+89.57 to 120+10.66 LT 10 110 118+89.57 to 120+10.66 LT LT 10 113+99.56 to 120+10.66 LT LT LT 10	ITEM NO. PAY ITEM DESCRIPTION $STA.$ TO $310c$ 10 1 CLEARING AND GRUDBING 100+02.49 to 102+12.65 LT 10 1 CLEARING AND GRUDBING 100+02.49 to 102+12.65 LT 10 1 103+95.2 to 104+65.75 LT LT 10 10 107+51.01 to 107+61.17 LT LT 10 10 107+51.01 to 107+61.17 LT LT 11 107+51.01 to 107+61.17 LT LT LT 113+54.5 to 113+40. to 113+40. LT LT 113+54.5 to 113+40. to 113+40. LT LT 113+54.5 to 113+40. to 113+40. LT LT 113+41.5 to 113+49.0 to 113+49.0 LT LT 10 113+491.65 to 112+415.25 K	ITEM NO. PAY ITEM DESCRIPTION $STA.$ TO STD. 10 1 CLEARING AND GRUDBING 100+02.49 to 102+12.65 LT 10 1 CLEARING AND GRUDBING 100+02.49 to 102+12.65 LT 10 1 CLEARING AND GRUDBING 100+02.49 to 107+61.17 LT 10 10 100+02.15 to 107+61.17 LT LT 10 10 107+51.01 to 107+61.17 LT LT 113+54.5 to 113+50. to 112+79.58 LT LT 113+54.5 to 113+54.5 to 113+40. LT LT 113+54.5 to 113+54.5 to 113+40. LT LT 113+54.5 to 113+54.5 to 113+40. LT LT 104 113+49.5 to 112+192.4 to 120+10.66 LT 104 113+497.5 to 112+192.55 tt LT LT 104 100+101 100+101.	ITEM NO. PAY ITEM DESCRIPTION $STA.$ TO STD. STD. 10 1 CLEARING AND GAUBTING 100+02.49 to 102+12.65 LT 10 1 CLEARING AND GAUBTING 100+02.49 to 102+12.65 LT 10 1 CLEARING AND GAUBTING 100+02.15 to 107 LT 10 1 100+02.15 to 10445.175 LT LT 10 107+51.01 to 107+51.17 LT LT LT 10 107+51.01 to 112+95.45 to 112+95.45 LT 113+20 113+20 113+20 to 112+95.45 LT 113+20 113+20 to 112+91.44 LT LT 113+20 113+20 to 112+91.44 LT LT 104 113+4155 to 112+91.05 LT LT 104 113+49 to 112+94.0 LT LT	ITEM NO. PAY ITEM DESCRIPTION STA. TO STA. TO STA. 10 1 CLEANING MO GAUGATING 100405.2 10 10145.2 17 10 11 CLEANING MO GAUGATING 100495.2 10 10145.12 17 10 11 CLEANING MO GAUGATING 100495.2 10 10145.12 17 10 100495.1 10 100495.12 10 11149.1 17 11 11 11 11 10 111449.1 17 11 11 11 11 11 11 11 11 11 11 10 11 11 11 11 11 10 11 11 11 11 11 11 11 11 11 11 11 11 11 11 11 11 11 11 11 11 11 11 11 11 11 11 11 11 11 11 11 11 11 11 11 11 11 11 11 11 11 11 11 11 11 11 <td>ITEM NO. PAY ITEM DESCRIPTION STA. TO STA. TO STD. STD. 10 1 CLEARING AND GAUBRING 100+02.49 to 102+12.65 LT LT 10 1 CLEARING AND GAUBRING 100+02.49 to 102+12.65 LT LT 10 10 100+02.15 to 104450.75 LT LT LT 10 100+51.01 to 107+51.01 to 107+51.17 LT LT 100 112+54.5 to 112+79.58 LT LT LT LT 113+50. 112+54.5 to 112+79.58 LT LT LT 113+50. 113+50. to 112+79.58 LT LT LT 113+50. 113+50. to 112+79.58 LT LT LT 1010 113+50. to 112+49.55 to 112+19.56 LT 100 110 100 112+49.157 to<</td> <td>ITEM NO. PAY ITEM DESCRIPTION STA. TO STA. D Sta 10 1 CLEARING MO GAUGNING 100+07.3 10 10 10 1 10+95.13 10 10+95.3 17 11 1 10 10+95.13 10 10 11 1 10 10+95.13 10 17 11 1 10 10 10+95.13 17 17 11 1 11 10 11+471.56 10 117 11 1 11+471.56 10 11+471.56 17 11 1 11+447.56 10 11+447.56 17 11 11+471.56 10 11+471.56 17 17 11 11+471.56 10 11+471.66 17 11 11+471.56 10 11+471.65 17 11 11+471.65 10 11+471.65 17 11 11+471.65 10<td>PAY ITEM NO. PAY ITEM NO. PAY ITEM NO. PAY ITEM NO. STA. TO STA. STA. TO STA. 010 1 1 C.G.RANDE MO GADDING 104405.13 10 104405.13 17 010 1 1 C.G.RANDE MO GADDING 104405.13 10 104405.13 17 111 111406.13 10 104405.13 17 104405.13 17 111 111406.13 10 1114406.13 17 1114406.13 17 111 111406.13 10 111446.13 17 1114406.13 17 111 111441.14 1114471.43 10 111446.13 17 111 111441.143 10 111446.143 17 111446.143 111441.145 111441.143 10 111447.143 17 111447.143 111441.145 111447.143 10 111447.143 17 111447.143 111441.145 111447.143 10 111447.143 17 111447.143 111441.145 111447.143 10 111447.143 11 111 111441.145 111447.143 10 111447.143 11 11 111441.145 111447.143 10 111447.143 11 11</td></td>	ITEM NO. PAY ITEM DESCRIPTION STA. TO STA. TO STD. STD. 10 1 CLEARING AND GAUBRING 100+02.49 to 102+12.65 LT LT 10 1 CLEARING AND GAUBRING 100+02.49 to 102+12.65 LT LT 10 10 100+02.15 to 104450.75 LT LT LT 10 100+51.01 to 107+51.01 to 107+51.17 LT LT 100 112+54.5 to 112+79.58 LT LT LT LT 113+50. 112+54.5 to 112+79.58 LT LT LT 113+50. 113+50. to 112+79.58 LT LT LT 113+50. 113+50. to 112+79.58 LT LT LT 1010 113+50. to 112+49.55 to 112+19.56 LT 100 110 100 112+49.157 to<	ITEM NO. PAY ITEM DESCRIPTION STA. TO STA. D Sta 10 1 CLEARING MO GAUGNING 100+07.3 10 10 10 1 10+95.13 10 10+95.3 17 11 1 10 10+95.13 10 10 11 1 10 10+95.13 10 17 11 1 10 10 10+95.13 17 17 11 1 11 10 11+471.56 10 117 11 1 11+471.56 10 11+471.56 17 11 1 11+447.56 10 11+447.56 17 11 11+471.56 10 11+471.56 17 17 11 11+471.56 10 11+471.66 17 11 11+471.56 10 11+471.65 17 11 11+471.65 10 11+471.65 17 11 11+471.65 10 <td>PAY ITEM NO. PAY ITEM NO. PAY ITEM NO. PAY ITEM NO. STA. TO STA. STA. TO STA. 010 1 1 C.G.RANDE MO GADDING 104405.13 10 104405.13 17 010 1 1 C.G.RANDE MO GADDING 104405.13 10 104405.13 17 111 111406.13 10 104405.13 17 104405.13 17 111 111406.13 10 1114406.13 17 1114406.13 17 111 111406.13 10 111446.13 17 1114406.13 17 111 111441.14 1114471.43 10 111446.13 17 111 111441.143 10 111446.143 17 111446.143 111441.145 111441.143 10 111447.143 17 111447.143 111441.145 111447.143 10 111447.143 17 111447.143 111441.145 111447.143 10 111447.143 17 111447.143 111441.145 111447.143 10 111447.143 11 111 111441.145 111447.143 10 111447.143 11 11 111441.145 111447.143 10 111447.143 11 11</td>	PAY ITEM NO. PAY ITEM NO. PAY ITEM NO. PAY ITEM NO. STA. TO STA. STA. TO STA. 010 1 1 C.G.RANDE MO GADDING 104405.13 10 104405.13 17 010 1 1 C.G.RANDE MO GADDING 104405.13 10 104405.13 17 111 111406.13 10 104405.13 17 104405.13 17 111 111406.13 10 1114406.13 17 1114406.13 17 111 111406.13 10 111446.13 17 1114406.13 17 111 111441.14 1114471.43 10 111446.13 17 111 111441.143 10 111446.143 17 111446.143 111441.145 111441.143 10 111447.143 17 111447.143 111441.145 111447.143 10 111447.143 17 111447.143 111441.145 111447.143 10 111447.143 17 111447.143 111441.145 111447.143 10 111447.143 11 111 111441.145 111447.143 10 111447.143 11 11 111441.145 111447.143 10 111447.143 11 11

SUMMARY OF EARTHW	SUMMARY OF QUANTITIES
SUMMARY OF EARTHW Fem DESCRIPTION TOTAL TRUCK NEASURE TOTAL TRUCK NEASURE TOTAL TRUCK NEASURE TOTAL SIS 2 LF F 15.0 15.0 15.0 15.0 15.0 15.0 15.0 15.0 15.0 15.0 15.0 15.0	
SUMMARY OF EARTHW Fem Description Tark TRUCK NEASURE Tark SUMMARY OF GUIDERAIL Tark SIS 2 LF F IS:0 F IS:0 IS IS:0 IS IS:0 F IS:0 IS IS:0 IS IS:0 IS	
TEM DESCRIPTION	0110
PAY ITEM DESCRIPT PAY ITEM DESCRIPT SUMMARY C SUMMARY C 515 1 515 1 LF 15:0 15:0 15:0 15:0	EWALK WCIAL PRDJECT 441241-1
	RONA DRIVE SI
REGULAR E BORROW E) SIDE LT TOTAL:	TROMA
PAY ITEM NO.	JAMES H. PITMAN JR. P.E. LICENSE WUMBER 42035 NFPS NFPS LAKE CITT, FLORIDA 32056 CI

DocuSign Envelope ID: 5D3013DB-2C27-40B2-8CCD-57B583C9BC42	
	DocuSign Envelope ID: 5D3013DB-2C27-40B2-8CCD-57B583C9BC42

I 0522 2 05.	SY SF	P F P F																													CITAMADY OF OUANTITES	CONTRACTOR I
1 0522 2 0527 2		FP																													DE OII	
1 0522 2 0527		FP											+	-																	ADAM	
1 0522 2	SY	£.								t	-							10.01	9.0	15.0	21.7	5.8	12.0	15.6	1.11	10.7	6.11	10.01	145			_
1 0522	SΥ							-			_												-		-		_				IDEWAL	FINANCIAL PROJECT ID 441241-1
	-												7.46	3.6	6.0	0.6	3.3											2.72	45		CITRONA DRIVE SIDEWALK	COUNTY NASSAU COUNTY
N	SY	F																													-	CITROWA DR
0522	0	٩.		102.8	5.6	r.o	5.6	14.0	5.6	13.3	50.1	78.7	84.3																368		035	S6 17 AT ION 24
1 10		F																													PITMAN JR. VSE NUMBER 421	3823 V, FLORIDA 3205 ATE OF AUTHORI
0520	1	Р	36.1			6.0																						1	42.		JANES H.	P.O. BOX
AREA ID			78217	4C189	4C18B	4C18C	40180	4C18E	4C18F	4C190	4C191	4C192	782F5	78200	40197	40190	782FC	9716	D7954	0110	9714	9715	9717	78230	9/19	9718	9718	971C	SUB-TOTAL: TOTAL:			
SIDE			RT	17	LT	11	11	LT	LT	LT	LT	IJ	LT	LT/RT	17	11	17	RI	LT	17	LT	LT	LT	11	11	1	U.	LT			WILLIAM	
		STA.	100+29.87	102+15.25	104+05.20	104+65.75	107+61.17	112+79.58	113+40.00	115+11.01	118+54.75	120+05.68	127+54.18	100+29.87	08-05+501	90-14481	20-414071	50 11+001	100+06.66	100+25.92	114+85.27	117+05.87	119+15.48	119+47.78	121+10.71	125+67.55	126+09.21	127+60.18			5 DESCI	
2		10	20	20	to	-	10	to	to	to	to	to	-	_	-	_	_	_	-	-	50	10	to	-	_	-	-	101			DAI	-
3		STA	100+00.12	100+29.87	103+95.20	104+50.75	71.12+201	112+54.50	113+30,00	114+87.06	117+64.50	118+89.67	126+14.55	100+00-01	105+21.12	114+71.62	120+03.04	ST TOTOOL	100+02.65	100+21.09	114+71.62	117+01.32	119409.4	119+41.7	121+01.1	125460.35	126+03.6	127+58.1				
	SIDE AREA ID 0520 1	SIDE AREA ID 0520 1 LF	A. SIDE AREA ID 0520 1 1 A. F	SIDE AREA ID 0520 1 . TO STA LF LF to 100429-87 RT 78217 36.1	SIDE AREA ID 0520 1 1 TO <sta< td=""> LF LF to 100+29.87 RT 78217 36.1 to 102+15.25 LT 405 1</sta<>	SIDE AREA ID 0520 1 . TO STA P LF to 100+29.87 RT 78217 36.1 to 100+29.87 RT 76219 36.1 to 102+15.25 LT 4C189 1 to 104+05.20 LT 4C186 1	SIDE AREA ID 0520 1 . TO STA P Lf Lf . to 100+29.87 Rr 78217 36.1 . to 100+29.87 Rr 78217 36.1 . to 100+29.20 LT 40189 6.1 . to 100+405.20 LT 40189 6.0 . to 104405.75 LT 40180 6.0	SIDE AREA ID 0520 1 . TO STA P LF to 100429.87 Rf 78217 36.1 to 100429.87 Rf 78217 36.1 to 100429.87 Rf 78217 36.1 to 100429.87 tr 4088 1 to 10445.25 tr 4088 1 to 10445.52 tr 4086 1 to 10445.51 tr 4086 1 to 10445.17 tr 4066 1	SIDE AREA ID 0520 1 . TO STA P LF to 100+29.87 RT 78217 36.1 to 102415.25 LT 4C189 7 to 102445.20 LT 4C189 7 to 104445.70 LT 4C189 7 to 104445.75 LT 4C189 6.0 to 107441.17 LT 4C180 6.0 to 107491.17 LT 4C180 6.0 to 107491.17 LT 4C180 6.0	SIDE AREA ID 0520 1 . TO STA P LF to 100+29.87 RT 78217 36.1 to 100+29.87 RT 78217 36.1 to 100+29.87 RT 78217 36.1 to 101425.25 LT 4C189 to 10245.25 LT 4C189 to 10445.75 LT 4C189 to 10445.75 LT 4C189 to 10445.75 LT 4C186 to 10445.75 LT 4C186 to 10749.17 LT 4C186 to 112479.58 LT 4C186 to 112479.58 LT 4C186	SIDE AREA ID 0520 1 . TO STA P LF to 100+29.87 RT 78217 36.1 to 100+29.87 RT 78217 36.1 to 100+29.87 RT 78217 36.1 to 101415.25 LT 40189 1 to 10245.25 LT 40189 1 to 10245.20 LT 40189 1 to 10244.00 LT 40180 1 to 107491.17 LT 40180 1 to 107491.17 LT 40180 1 to 112479.56 LT 40180 1 to 11141.01 <td>SIDE AREA ID 0520 1 TO<</td> TA P LF to 100+29.87 RT 78217 36.1 to 101+45.75 LT 4C189 6.0 to 104+45.75 LT 4C189 6.0 to 107+61.17 LT 4C180 6.0 to 107+61.17 LT 4C180 6.0 to 107+61.17 LT 4C180 6.0 to 107+61.10 LT 4C180 6.0 to 112+79.58 LT 4C180 7 to 113+40.00 LT 4C180 7 to 113+41.01 LT 4C190 7 to 113+47.5 LT 4C191 7	SIDE AREA ID 0520 1 TO<	SIDE AREA ID 0520 1 TO TA . P Lf to $100-29.87$ RT 78217 36.1 Lf to $100+29.87$ RT 78217 36.1 Lf to $100+29.87$ RT 78217 36.1 Lf to $100+95.26$ LT $4C189$ 6.0 $100+65.75$ LT $4C180$ 6.0 $100+65.75$ LT $4C180$ 6.0 $100+65.75$ LT $4C180$ 6.0 $100+65.75$ LT $4C180$ 6.0 $100+61.17$ LT $4C180$ 6.0 $100+61.10$ LT $4C180$ 6.0 $100+60.00$ LT $4C180$ $C0$ $100+20.81$ LT $4C180$ $C0$ $100+20.81$ LT $4C180$ LT $4C180$ $100+20.81$ LT $4C180$ LT $4C180$ LT $4C180$ LT $4C180$ LT $4C180$ LT <td< td=""><td>SIDE AREA ID 0520 1 TO TA. P LF to $100-29.87$ RT 78217 36.1 LF to $100+29.87$ RT 78217 36.1 LF to $100+29.87$ RT 78217 36.1 LF to $100+29.87$ tT $4C189$ 5.0 T to $100+45.75$ tT $4C189$ 5.0 T to $101+45.75$ tT $4C180$ 6.0 T to $107+61.17$ tT $4C180$ 6.0 T to $107+61.17$ tT $4C180$ 6.0 T to $107+61.17$ tT $4C180$ 6.0 T to $112+79.56$ tT $4C180$ T T to 111.01 tT $4C180$ T T to $112+79.56$</td><td>SIDE AREA ID 0520 1 TO TA. P LF to $100-29.87$ RT 78217 36.1 LF to $100+29.87$ RT 78217 36.1 LF to $100+29.87$ tT $4C189$ RT 36.1 LF to $100+25.76$ tT $4C189$ 6.0 RT $4C180$ RT to $100+65.76$ tT $4C180$ 6.0 RAA 6.0 RAA 6.0 RAA 6.0 $100+66.6$ $100+66.6$ $100+66.6$ 6.0 $100+66.6$ $100+20.80$ LT $4C180$ 100 <</td><td>SIDE AREA ID 0520 I TO TA. P LF to $100-29.87$ RT 78217 36.1 LF to $100+29.87$ RT 78217 36.1 LF to $100+29.87$ tT $4C189$ RT 36.1 LF to $100+25.76$ tT $4C189$ 6.0 R $4C180$ R to $100+65.76$ tT $4C180$ 6.0 R R</td><td>SIDE AREA ID 0520 I TO TA. P LF to $100-29.87$ RT 78217 36.1 to $100+29.87$ RT 78217 36.1 to $100+29.87$ tT $4C189$ LF to $100+25.75$ tT $4C189$ R to $100+65.75$ tT $4C189$ R to $100+65.75$ tT $4C180$ R to $100+65.75$ tT $4C180$ R to $100+65.75$ tT $4C180$ 6.0 to $100+65.76$ tT $4C180$ 6.0 to $112+79.58$ tT $4C180$ 6.0 to $112+79.58$ tT $4C190$ 6.0 to $112+79.58$ tT $4C190$ 6.0 to $112+79.56$ tT $4C190$ 6.0</td><td>SIDE AREA ID 0520 I TO TA. P LF to $100-29.87$ RT 78217 36.1 LF to $100+29.87$ RT 78217 36.1 LF to $100+29.87$ tT $4C189$ 5.0 LF to $100+25.75$ tT $4C189$ 5.0 LF to $100+65.75$ tT $4C180$ 5.0 L to $100+65.75$ tT $4C180$ 6.0 L to $100+65.75$ tT $4C180$ 6.0 L to $100+65.76$ tT $4C180$ L L to $11.1.01$ tT $4C190$ L L $t 0$ $112+79.58$ tT $4C190$ L L $t 0$ $112+79.58$ tT $4C190$ L L $t 0$ $112+79.58$</td><td>SIDE AREA ID 0520 1 TO TA. P LF to $100+29-87$ RT 78217 36.1 LF to $100+29-87$ RT 78217 36.1 LF to $100+29-87$ tT $4C189$ RT 36.1 LF to $100+95-30$ LT $4C189$ 6.0 RR/A 6.0 RR/A 6.0 $100+65-36$ LT $4C180$ R^{-1} R^{-1}</td><td>SIDE AREA ID 0520 1 TO TA. P LF to $100+29-87$ RT 78217 36.1 LF to $100+29-87$ RT 78217 36.1 LF to $100+29-87$ tT $4C189$ RT 36.1 LF to $100+465.75$ tT $4C180$ 6.0 RK/A 6.0 to $100+465.75$ tT $4C180$ 6.0 R/AA 6.0 R/AA to $100+465.75$ tT $4C180$ 6.0 R/AA 6.0 R/AA 6.0 R/AA 6.0 R/AA 6.0 R/AA 6.0 R/AA 6.0 $100+60.6$ $100-60.6$ $100-60.6$</td><td>SIDE AREA ID 0520 1 TO TA P LF to $100-29.87$ RT 78217 36.1 LF to $100+29.87$ RT 78217 36.1 LF to $100+29.87$ tT $4C189$ $s.0$ LF to $100+465.75$ tT $4C189$ $s.0$ R to $100+465.75$ tT $4C180$ $s.0$ R to $112+79.58$ tT $4C190$ R R to $112+79.68$ tT $4C192$ R R to $112+79.68$ tT R R R R to</td><td>SIDE AREA ID 0520 1 TO TA. P LF to $100-99.87$ RT 78217 86.1 LF to $100-99.87$ RT 78217 36.1 LF to $100+95.75$ tT $4C189$ 8.1 36.1 LF to $102+92.87$ tT $4C189$ 6.0 $RAAA$ 6.0 $RAAA$ to $102+92.86$ tT $4C180$ 6.0 $RAAA$ 6.0 $RAAA$ 6.0 $RAAA$ 6.0 $RAAA$ 6.0 $RAAA$ 8.0 $RACBA$ $RACBA$ $RACBA$ $RACBA$ $RACBA$ RAA RAA $RAAA$ $RAAA$ $RAAA$ $RAAA$ $RAAA$ $RAAA$ $RAAA$ $RAAA$ $RAAA$ $RAAAA$ $RAAAA$ $RAAAA$ $RAAAAA$ $RAAAAAAAAAAAAAAAAAAAAAAAAAAAAAAAAAAAA$</td><td>SIDE AREA ID 0520 I Γ $100-90.67$ RT 78217 R Γ Γ $100-90.67$ RT 78217 36.1 Γ Γ $100+90.67$ Γ $4C189$ R R Γ $100+90.5.76$ Γ $4C189$ R R Γ $100+90.5.76$ Γ $4C180$ 6.0 R Γ $100+90.5.76$ Γ $4C180$ 6.0 R Γ 11000 Γ $4C180$ 6.0 R R Γ 11000 Γ $4C180$ 6.0 R R Γ 11000 Γ $4C180$ 6.0 R R Γ 11000 Γ $4C180$ R R R Γ 111000</td><td>SIDE AREA ID 0520 1 TO<ta< td=""> B L L to 100-39.67 RT 78217 S6.1 L to 100-39.67 RT 78217 S6.1 L to 102-15.55 UT 4C189 S6.1 L to 102+15.55 UT 4C189 S6.1 L to 102+5.55 UT 4C189 S6.1 L to 102+5.55 UT 4C186 S6.0 L to 102+5.55 UT 4C186 S6.0 L to 102+5.55 UT 4C186 S6.0 L to 102+51.55 UT 4C186 S6.0 L to 113+40.00 UT 4C186 S6.0 L to 113+440.08 UT 4C186 L L to 113+450.18 UT 78260 L L to 112</ta<></td><td>SIDE AREA ID 0520 1 Γ Γ Γ Γ Γ Γ Γ RT 78217 36.1 Γ Γ $100-29.87$ RT 78217 36.1 Γ Γ $100+29.87$ Γ $4C189$ Γ $4C189$ Γ Γ $100+95.76$ Γ $4C180$ Γ $4C180$ Γ Γ $100+65.76$ Γ $4C180$ 6.0 Γ $4C180$ Γ Γ $100+61.17$ LT $4C180$ 6.0 Γ $4C180$ Γ Γ $100+61.17$ LT $4C180$ 6.0 Γ $4C180$ Γ $4C180$ Γ 100 1000<td>SIDE AREA ID 0520 1 T T T T T L T T T T T L 100-99.57 RT T T T L 100-99.57 RT T T S6.1 L L 100-99.57 LT 4C189 T S6.1 L L 100-95.55 LT 4C186 LT 4C186 L L L 100-95.55 LT 4C186 LT 4C186 L L L 100-95.45 LT 4C186 LT 4C186 L L L 113-400.00 LT 4C186 LT 4C186 L L L 113-400.00 LT 4C186 LT 4C186 L L L 113-440.01 LT 4C186 LT L L L L L L 113-4</td><td>SIDE AREA ID 0520 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1</td></td></td<>	SIDE AREA ID 0520 1 TO TA . P LF to $100-29.87$ RT 78217 36.1 LF to $100+29.87$ RT 78217 36.1 LF to $100+29.87$ RT 78217 36.1 LF to $100+29.87$ tT $4C189$ 5.0 T to $100+45.75$ tT $4C189$ 5.0 T to $101+45.75$ tT $4C180$ 6.0 T to $107+61.17$ tT $4C180$ 6.0 T to $107+61.17$ tT $4C180$ 6.0 T to $107+61.17$ tT $4C180$ 6.0 T to $112+79.56$ tT $4C180$ T T to 111.01 tT $4C180$ T T to $112+79.56$	SIDE AREA ID 0520 1 TO TA . P LF to $100-29.87$ RT 78217 36.1 LF to $100+29.87$ RT 78217 36.1 LF to $100+29.87$ tT $4C189$ RT 36.1 LF to $100+25.76$ tT $4C189$ 6.0 RT $4C180$ RT to $100+65.76$ tT $4C180$ 6.0 RAA 6.0 RAA 6.0 RAA 6.0 $100+66.6$ $100+66.6$ $100+66.6$ 6.0 $100+66.6$ $100+20.80$ LT $4C180$ 100 <	SIDE AREA ID 0520 I TO TA . P LF to $100-29.87$ RT 78217 36.1 LF to $100+29.87$ RT 78217 36.1 LF to $100+29.87$ tT $4C189$ RT 36.1 LF to $100+25.76$ tT $4C189$ 6.0 R $4C180$ R to $100+65.76$ tT $4C180$ 6.0 R	SIDE AREA ID 0520 I TO TA . P LF to $100-29.87$ RT 78217 36.1 to $100+29.87$ RT 78217 36.1 to $100+29.87$ tT $4C189$ LF to $100+25.75$ tT $4C189$ R to $100+65.75$ tT $4C189$ R to $100+65.75$ tT $4C180$ R to $100+65.75$ tT $4C180$ R to $100+65.75$ tT $4C180$ 6.0 to $100+65.76$ tT $4C180$ 6.0 to $112+79.58$ tT $4C180$ 6.0 to $112+79.58$ tT $4C190$ 6.0 to $112+79.58$ tT $4C190$ 6.0 to $112+79.56$ tT $4C190$ 6.0	SIDE AREA ID 0520 I TO TA . P LF to $100-29.87$ RT 78217 36.1 LF to $100+29.87$ RT 78217 36.1 LF to $100+29.87$ tT $4C189$ 5.0 LF to $100+25.75$ tT $4C189$ 5.0 LF to $100+65.75$ tT $4C180$ 5.0 L to $100+65.75$ tT $4C180$ 6.0 L to $100+65.75$ tT $4C180$ 6.0 L to $100+65.76$ tT $4C180$ L L to $11.1.01$ tT $4C190$ L L $t 0$ $112+79.58$ tT $4C190$ L L $t 0$ $112+79.58$ tT $4C190$ L L $t 0$ $112+79.58$	SIDE AREA ID 0520 1 TO TA . P LF to $100+29-87$ RT 78217 36.1 LF to $100+29-87$ RT 78217 36.1 LF to $100+29-87$ tT $4C189$ RT 36.1 LF to $100+95-30$ LT $4C189$ 6.0 RR/A 6.0 RR/A 6.0 $100+65-36$ LT $4C180$ R^{-1}	SIDE AREA ID 0520 1 TO TA . P LF to $100+29-87$ RT 78217 36.1 LF to $100+29-87$ RT 78217 36.1 LF to $100+29-87$ tT $4C189$ RT 36.1 LF to $100+465.75$ tT $4C180$ 6.0 RK/A 6.0 to $100+465.75$ tT $4C180$ 6.0 R/AA 6.0 R/AA to $100+465.75$ tT $4C180$ 6.0 R/AA 6.0 R/AA 6.0 R/AA 6.0 R/AA 6.0 R/AA 6.0 R/AA 6.0 $100+60.6$ $100-60.6$ $100-60.6$ $100-60.6$ $100-60.6$ $100-60.6$ $100-60.6$ $100-60.6$ $100-60.6$ $100-60.6$ $100-60.6$ $100-60.6$ $100-60.6$ $100-60.6$ $100-60.6$ $100-60.6$ $100-60.6$ $100-60.6$	SIDE AREA ID 0520 1 TO TA P LF to $100-29.87$ RT 78217 36.1 LF to $100+29.87$ RT 78217 36.1 LF to $100+29.87$ tT $4C189$ $s.0$ LF to $100+465.75$ tT $4C189$ $s.0$ R to $100+465.75$ tT $4C180$ $s.0$ R to $112+79.58$ tT $4C190$ R R to $112+79.68$ tT $4C192$ R R to $112+79.68$ tT R R R R to	SIDE AREA ID 0520 1 TO TA . P LF to $100-99.87$ RT 78217 86.1 LF to $100-99.87$ RT 78217 36.1 LF to $100+95.75$ tT $4C189$ 8.1 36.1 LF to $102+92.87$ tT $4C189$ 6.0 $RAAA$ 6.0 $RAAA$ to $102+92.86$ tT $4C180$ 6.0 $RAAA$ 6.0 $RAAA$ 6.0 $RAAA$ 6.0 $RAAA$ 6.0 $RAAA$ 8.0 $RACBA$ $RACBA$ $RACBA$ $RACBA$ $RACBA$ RAA RAA $RAAA$ $RAAA$ $RAAA$ $RAAA$ $RAAA$ $RAAA$ $RAAA$ $RAAA$ $RAAA$ $RAAAA$ $RAAAA$ $RAAAA$ $RAAAAA$ $RAAAAAAAAAAAAAAAAAAAAAAAAAAAAAAAAAAAA$	SIDE AREA ID 0520 I Γ $100-90.67$ RT 78217 R Γ Γ $100-90.67$ RT 78217 36.1 Γ Γ $100+90.67$ Γ $4C189$ R R Γ $100+90.5.76$ Γ $4C189$ R R Γ $100+90.5.76$ Γ $4C180$ 6.0 R Γ $100+90.5.76$ Γ $4C180$ 6.0 R Γ 11000 Γ $4C180$ 6.0 R R Γ 11000 Γ $4C180$ 6.0 R R Γ 11000 Γ $4C180$ 6.0 R R Γ 11000 Γ $4C180$ R R R Γ 111000	SIDE AREA ID 0520 1 TO <ta< td=""> B L L to 100-39.67 RT 78217 S6.1 L to 100-39.67 RT 78217 S6.1 L to 102-15.55 UT 4C189 S6.1 L to 102+15.55 UT 4C189 S6.1 L to 102+5.55 UT 4C189 S6.1 L to 102+5.55 UT 4C186 S6.0 L to 102+5.55 UT 4C186 S6.0 L to 102+5.55 UT 4C186 S6.0 L to 102+51.55 UT 4C186 S6.0 L to 113+40.00 UT 4C186 S6.0 L to 113+440.08 UT 4C186 L L to 113+450.18 UT 78260 L L to 112</ta<>	SIDE AREA ID 0520 1 Γ Γ Γ Γ Γ Γ Γ RT 78217 36.1 Γ Γ $100-29.87$ RT 78217 36.1 Γ Γ $100+29.87$ Γ $4C189$ Γ $4C189$ Γ Γ $100+95.76$ Γ $4C180$ Γ $4C180$ Γ Γ $100+65.76$ Γ $4C180$ 6.0 Γ $4C180$ Γ Γ $100+61.17$ LT $4C180$ 6.0 Γ $4C180$ Γ Γ $100+61.17$ LT $4C180$ 6.0 Γ $4C180$ Γ $4C180$ Γ 1000 <td>SIDE AREA ID 0520 1 T T T T T L T T T T T L 100-99.57 RT T T T L 100-99.57 RT T T S6.1 L L 100-99.57 LT 4C189 T S6.1 L L 100-95.55 LT 4C186 LT 4C186 L L L 100-95.55 LT 4C186 LT 4C186 L L L 100-95.45 LT 4C186 LT 4C186 L L L 113-400.00 LT 4C186 LT 4C186 L L L 113-400.00 LT 4C186 LT 4C186 L L L 113-440.01 LT 4C186 LT L L L L L L 113-4</td> <td>SIDE AREA ID 0520 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1</td>	SIDE AREA ID 0520 1 T T T T T L T T T T T L 100-99.57 RT T T T L 100-99.57 RT T T S6.1 L L 100-99.57 LT 4C189 T S6.1 L L 100-95.55 LT 4C186 LT 4C186 L L L 100-95.55 LT 4C186 LT 4C186 L L L 100-95.45 LT 4C186 LT 4C186 L L L 113-400.00 LT 4C186 LT 4C186 L L L 113-400.00 LT 4C186 LT 4C186 L L L 113-440.01 LT 4C186 LT L L L L L L 113-4	SIDE AREA ID 0520 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1					

~ ~

		CONSTRUCT I ON	CVUE																											SHEET NO.	505
_		-	-							_												-								O CERUSALUX	CONTRACTOR NO INFWINC
		DESIGN	NOIES																										_	110 40	OF CUA
CENTURNAULT TUN	NCE TURF	1 2	~	F																											INAMI
	PERFORMANCE (SOD)	0270	SY	Р	39.11	36.46	4.21	4.21	6.01	6.01	3.64	3.66	4.01	01 01	10.02	4.01	4.01	15.91	15.78	36.14	36.15	2.93	44.92	30.7		19.96	61.02	414.8	415		SUM
5	-	AREA	21		9817	9822	982C	9818	9827	9819	9828	9816	9829	0010	9826	981C	982A	9828	G186	9823	981E	981F	9824	9820	2010	9825	1725	SUB-TOTAL:	TOTAL:	JTK	FINANCIAL PROJECT ID 441241-1
		SIDE			LT	RT	RT	11	RT	LT	RT	LT	RT		RT	LT	RT	RT	17	RT	LT	11	RT	11	5	HH :	5 5	10		E SIDEWI	
				STA.	102+15.75	102+15.75	104+05.70	104+05.70	104+65.75	104+65.75	105+30.28	105+30.59	107+61.17	11.10+101	112479.58	113+40.00	113+40.00	115+11.02	115+11.02	118+54.67	118+54.67	119+09.48	120+01.70	119+88.42	17.10+021	127+44.09	127+43.21	10-101/7		CITRONA DRIVE SIDEWALK	COUNTY NASSAU COUNTY
	I OC AT LOW	FUCAL LUI		STA. TO S	to	10	to	to	to	50	03	to	to	0	to	10	00	to	to	02	50	10	02	02	10	01	10	2		CIT	CITRONA DR
				S	100+02.21	101+05.71	103+95.20	103+95.20	104+50.75	104+50.75	105+21.13	105+21.51	107+51.15	c1.1c+/01	112+54.50	113+30.00	113+30.00	114+71.34	114471.56	117+64.50	117+64.50	118+89.67	115+89.67	119+15.41	119+98.27	126+06.86	126+11.78	12/140		32	
																														JAMES H. PITMAN JR.	RFPS P.O. BOX 3823 LAKE CITY. FLORIDA 32056
																														210M	
																														DE SCRIPTIO	
																														REVISIONS	_
																														REV	
																														DESCRIPTION	
		_		_						_		_				_		_	_		_		_				_	_	_	 0,075	

1994 2004 1994 - 1996 - 199				SHEET NO.
		THIS FEA MAS REEN DIGITALLY SIGNED AND SQUEDE BT. Jarmes H Pittman Jarmes H Pittman Jarte Jarges Jarley and M The Durt ADJUAR TO THE SGM. Renarme Convies of THIS DOCUMENT ARE REMARKE STARTE AND SEELED. THE STRATTORE STARTE AND SEELED. THE STRATTORE STARTE AND SEELED. THE STRATTORE STARTE AND SEELED.	SIGNING AND PAYENENT HABACHIG PLANS ENGINEES NG RECURD. JANES N. FITTAN II. P. M. 20055 NOVELOSS NAL SERVICES, MC RET PROVID A PAYESSIONAL SERVICES, MC LAR ENT PERSON 2005 CONTANT NO. VEZ INBRAZADO VERTORN NO. VYZ INBRAZADO VERTORN NO. VYZ INBRAZADO CONTANT NO. VYZ INBRAZADO	FEAR 71
MANAGE REALECT	PLANS	A STATE OF S		
CONTRACT PLANS CITRONA DRIVE SIDEWALK FINANCIAL PROJECT ID 441241-1 WFPS PROJECT ID 1200413ELE NASSAU COUNTY COUNTY, FLORIDA	SIGNING AND PAVEMENT MARKING PLANS			
	INDEX OF SIGNING AND PAVEMENT MARKING PLANS SHEET NO. SHEET DESCRIPTION 5-1 KET SHEET 5-2 TABULATION OF QUANTIES 5-3 - S-5 SIGNING & PAVENENT MAGKINGS			
	INDEX OF sheef No. 5-1 5-3 - 5-5			







Nassau County Engineering Services Department 96161 Nassau Place Yulee, Florida 32097

Robert T. Companion, P.E. County Engineer

AS-BUILTS REQUIREMENT CHECKLIST:

The Following list is intended to highlight the majority of the as-built requirements for Construction projects in Nassau County. This list should not be considered to be all-inclusive as each project is unique in nature and may require additional information that can only be determined during the course of the project's completion. Generally, however, the AS-BUILT information shall contain the following:

- AS-BUILTS should be based on the design plans as approved through the DRC process. The submitted as-builts should be at the same scale and have the same orientation as the design files.
- AS-BUILTS should have the same page numbering as the design plans, and the cover of the AS-BUILT plans should have all sheets from the design plans listed with sheets not "AS-BUILTED" stricken through.
- AS-BUILTS should display the original design information as displayed on the plan sheets with the design information stricken through and the as-built information displayed in bold adjacent to the design information.
- AS-BUILTS shall be accompanied by an "Engineer's Certification" form from the Engineer of Record. (Exhibit 3&4)
- Three (3) sets of fully signed and sealed AS-BUILTS should be submitted along with a CADD file(s) containing the AS-BUILT information. CADD files should be in State Plane coordinates (NAD 83) with a vertical datum of NGVD 1929. Furthermore, CADD files should be submitted in ACAD 2004 or ACAD 2006 only.
- 6. Northings and Easting of all drainage structures should be included.
- 7. The plans should be clearly legible and all structure notes, distances, angles and elevations should be clearly readable.
- 8. If the plan represents a phase of a development, then that phase should be clearly identified for clarity and avoid confusion with future phases.
- 9. There should be a north arrow and scale on each sheet.

AS-BUILTS REQUIREMENT CHECKLIST Page | 2

- 10. A legend should be included explaining the symbols used in the plans.
- 11. ALL sheets must be signed and sealed by a surveyor licensed in the State of Florida.
- 12. There should be sufficient "plan" and "as-built" elevations shown to verify that the streets were constructed substantially in accordance with the approved construction plans. Generally, the County will review all low points and high points in the street and verify that the minimum grade exists for each street. On straight sections between high and low points elevations should be taken every 200 feet.
- 13. All street curve radii should be shown on the plans or in a table.
- 14. Street widths and curb type should be identified for each street on each sheet.
- 15. Whenever there are islands within the streets the as-builts should include dimensions for these islands.
- 16. The paved radii of all Cul-de-Sacs should be listed and Cul-De-Sac center and edge of pavement or gutter elevations at quarter points shall be shown.
- 17. All underdrains should be shown with size, lengths, inverts and cleanouts all shown.
- 18. Where swales are utilized there should be sufficient flow line elevations and ditch cross sections to verify capacity of the channel.
- There should be a comparison table of design and as-built pipe sizes, lengths, invert elevations, and pipe slopes.
- 20. The as-built surface area of the pond(s) at Normal Water Level (design) and Top of Bank (asbuilt) should be included.
- 21. The bottom elevation and area should be shown (2 locations min. per pond).
- 22. The surveyor shall certify by note (for each pond) that no slope is greater than 1:4 above the design NWL, unless the pond is fenced.
- 23. All structures in the pond (overflow weirs, etc.) should be included.

Phone (904) 530-6225

Fax (904) 491-3611

AS-BUILTS REQUIREMENT CHECKLIST Page | 3

- 24. All watermain and sewermain locations, dimensions, etc.
- 25. All easements required (or on a plat) should be shown on the "As- Builts" and improvement located so as to verify improvements are within the easement. Easement not recorded as part of the recorded plat including drainage and right-of-way easements shall also be identified as "as-built". For these easements the book and page of their record, property to whom easement is dedicated and date of filing should be shown on as the "as-built". All improvements intended to be within these easements shall be shown as the "as-builts" to verify the improvements are within the easement. Wetlands are not reviewed by Nassau County and need not be shown.



Nassau County Engineering Services Department 96161 Nassau Place Yulee, Florida 32097

Road Closure Policy

1. Submit a written request to the County Engineer for the road closure. In the request you must explain the following:

- a) Why the road needs to be closed and the duration of the closure?
- b) What are the benefits to public for closing the road?
- c) Include a draft detour plan.

After the road closure request and draft detour plan have been approved, the following information shall be submitted to the Nassau County Engineering Services Department for review and approval.

2. Submit to Nassau County Engineering Services Department: A Maintenance of Traffic (MOT) Plan that was prepared by or approved by Florida Advanced Maintenance of Traffic (AMOT) certified personnel referencing the latest editions of the Florida Department of Transportation (FDOT) Design Standards Section 600 and the Manual on Uniform Traffic Control Devices (MUTCD). These plans shall include the Worksite Traffic Supervisor as per the latest edition of the FDOT Standard Specifications for Road & Bridge Construction Section 105-8.3. Variable Message Boards (VMB) will be required and shall be installed 7 days prior to the detour taking place and remain throughout. Any signed detour that will be within FDOT Right-of-Way (or on portions thereof) shall have prior FDOT approval.

A Maintenance of Traffic Plan proposing a change to any approved documents, including contract documents and approved construction plans must be signed and sealed.

3. Provide a public notification advertisement to an approved local newspaper to appear no less than fourteen (14) days prior to the scheduled closure. Submit the draft notification to Nassau County Engineering Services for review and approval prior to advertising. This notification will clearly describe the name of the project and contractor, scope of the detour, scope of construction, the name of all roads that will be affected, official detour routes, duration of closure and the Worksite Traffic Supervisor's telephone number. Attach a diagram showing the surrounding area and detour routes. Diagram must be clear and depict road names and route direction. The size of advertisement in newspaper shall be a minimum of 2 columns wide by 12 inches long (tall) or one fifth (1/5) of a page with a minimum of 24 point font for the heading and 10 point for the body and detour plan. Flyers with the same information may be required. With the Engineering Services Department's written approval, local roads may not require publishing of the advertisement in the paper; however, all other criteria must be adhered to.



Road Closure Policy Page 2 of 3

- 4. Submit notification to the following departments and agencies prior to the road closure. A confirmation of notification to the Engineering Services Department must be provided prior to road closure. Depending on the location of the project/detour, notification to other agencies may be required.
- Nassau County Sheriff's Office
 - Ricky A Rowell, TAC Phone – (904) 548-4000 Fax – (904) 225-5737 Email – <u>rickyr@nassauso.com</u>
 - L. Rene Graham, Lt.
 Phone (904) 548-4028
 Fax (904) 548-4128
 Email <u>lrgraham@nassauso.com</u>
- Nassau County Emergency Management
 Martha Oberdorfer
 - Phone (904) 548-0931 Fax – (904) 548-4194 Email – <u>mloberdorfer@nassauso.com</u>
- Nassau County Fire Department
 - Chief Brady Rigdon
 Email brigdon@nassaucountyfl.com
 - Chief Greg Roland Email – <u>groland@nassaucountyfl.com</u>
 Chief Bob Ratliff
 - Email <u>tratliff@nassaucountyfl.com</u> Phone - (904) 530-6600 Fax - (904) 321-5748
- Nassau County School Bus Transportation
 - Brad Underhill
 Phone (904) 225-0127
 Fax (904) 255-9404
 Email <u>underhillbr@nassau.k12.fl.us</u>
- Nassau County Road & Bridge Department
 - Jennifer Kirkland
 Phone (904) 530-6175
 Fax (904) 530-6901
 Email <u>jkirkland@nassaucountyfl.com</u>

- Nassau County Manager's Office
 - Sabrina Robertson
 Phone (904) 530-6010
 Fax (904) 321-5784
 Email srobertson@nassaucountyfl.com
- United Postal Service (Yulee)
 Cassondra Mitchell
 Phone (904) 875-6083
 Fax (904) 225-9733
 Email cassondra.t.mitchell@usps.gov
- United Postal Service (Fernandina)
 Paul Battista Phone – (904) 557-9342 Fax – (904) 277-7947 Email – paul.d.battista@usps.gov
- United Postal Service (Callahan/Bryceville)
 Elizabeth Williams
 Phone (904) 879-2131
 Fax (904) 879-6737
 Email <u>elizabeth.s.williams@usps.gov</u>
- United Postal Service (Hilliard)
 o Bridgett Wagers
 Phone (904) 845-2151
 Fax (904) 845-7738
 Email bridgett.n.wagers@usps.gov



Road Closure Policy Page 3 of 3

- 5. Signage will be in place prior to road closure. They will be bagged/covered until needed. VMBs, will be installed and operating a minimum of seven (7) calendar days prior to any closure/detour.
- Visual inspection of the VMB by the Engineering Services Department is required at installation or relocation. Visual inspection of all road/detour signage shall be accomplished prior to road closure. Contact Engineering Services 48 hours prior to installation for inspection.

THIS BID IS SUBMITTED TO:

Invitation to Bid – NC22-013 Citrona Drive from Hickory Street to Beech Street

SECTION 00 41 15

BID FORM

Board of County Commissioners, Nassau County

PROJECT IDENTIFICATION:	CITRONA DRIVE FROM HICKORY STREET TO BEECH STREET
	Bid Number NC22-013 Nassau County, Florida
	Nassau County, Florida
BID DEADLINE:	December 22, 2022 at 10:00 a.m.

1.01 The undersigned Bidder proposes and agrees, if this Bid is accepted, to enter into an agreement with the County in the form included in the Bidding Documents to perform all Work as specified or indicated in the Bidding Documents for the prices and within the times indicated in this Bid and in accordance with the other terms and conditions of the Bidding Documents.

Company Name (typed or printed): CGC, In	<u>C.</u>	
Business address:	7036 W. 12th Street, Ja	acksonville, FL 32220	
Phone No.:	(904) 783-4119	Fax No.: (904) 783-3401	
Contact Name:	Richard C. Gaskin, Jr.		_
Contact Title:	President		
Contact email add	dress: office@cgccivil.co	om	

- 2.01 Bidder accepts all of the terms and conditions of the Advertisement or Invitation to Bid and Instructions to Bidders, including without limitation those dealing with the disposition of Bid security. The Bid will remain subject to acceptance for 90 days after the Bid opening, or for such longer period of time that Bidder may agree to in writing upon request of the County.
- 3.01 In submitting this Bid, Bidder represents, as set forth in the Agreement, that:
 - A. Bidder has examined and carefully studied the Bidding Documents, the other related data identified in the Bidding Documents, and the following Addenda, receipt of all which is hereby acknowledged:

Addendum No.	Addendum Date
One	11/28/2022
Two	12/15/2022

B. Bidder has visited the Site and become familiar with and is satisfied as to the general, local and Site conditions that may affect cost, progress, and performance of the Work.
DocuSign Envelope ID: 5D3013DB-2C27-40B2-8CCD-57B583C9BC42

	NASSAU COUNTY BOARD OF COUNTY COMMISSIO Procurement Department 96135 Nassau Place, Suite 2 Yulee, Florida 32097	Thomas O'Brie tobrien@nassaucountyfl.com
TO: FROM: SUBJECT:	904-530-6040 All Proposers Thomas O'Brien, Procurement Specialist Addendum #1	REMINDER: This addendum must be acknowledged, signed and returned with your proposal. Failure to comply may result in disqualification of your submittal.
DATE:	Invitation to Bid #NC22-013 Citrona Drive Pedestrian Improvements – 1 Rebid 11/28/22	From Hickory Street to Beech Street

This addendum is hereby incorporated into the bid documents of the project referenced above. The following items are clarifications, corrections, additions, deletions and/or revisions to, and shall take precedence over, the original documents.

Questions:

1. Is there an estimated mobilization date for this project?

Answer: Mobilization schedule is February-March 2023.

2. Is there an estimated value for this project?

Answer: The project budget is \$198,598.00.

The Bid due date and opening time remain: December 22, 2022 at 10:00 AM EST

ACKNOWLEDGMENT IS HEREBY MADE OF RECEIPT OF THIS ADDENDUM NO. 1, ISSUED DURING THE SOLICITATION PERIOD: CGC, Inc.

Vendor Signature:

Date: November 28, 2022

End of Addendum #1

DocuSign Envelope ID: 5D3013DB-2C27-40B2-8CCD-57B583C9BC42

	NASSAU COUNTY BOARD OF COUNTY COMMISSION Procurement Department 96135 Nassau Place, Suite 2 Yulee, Florida 32097	NERS
10	904-530-6040	REMINDER: This addendum must be acknowledged, signed and returned with your proposal. Failure to comply
TO: FROM:	All Proposers Thomas O'Brien, Procurement Specialist	may result in disgualification of your
SUBJECT:	Addendum #2	submittal.
SUBJECT.	Invitation to Bid #NC22-013	
	Citrona Drive Pedestrian Improvements - I	From Hickory Street to Beech Street
	Rebid	olde and the subscreep of the start of
DATE:	12/15/22	

This addendum is hereby incorporated into the bid documents of the project referenced above. The following items are clarifications, corrections, additions, deletions and/or revisions to, and shall take precedence over, the original documents.

Questions:

 Bid Item 527-2 Detectable Warnings - Please revise the unit of measure from SY to SF.

Answer: The wrong unit of SY is noted. The correct quantity is 145 SF.

Additional information:

Link to SAM.gov Wage Determinations: https://sam.gov/content/wage-determinations

The Bid due date and opening time remain: December 22, 2022 at 10:00 AM EST

ACKNOWLEDGMENT IS HEREBY MADE OF RECEIPT OF THIS ADDENDUM NO. 2, ISSUED DURING THE SOLICITATION PERIOD: CGC, Inc.

Vendor Signature: Richard C. Gaskin, Jr., President

Date: December 15, 2022

End of Addendum #2

- C. Bidder is familiar with and is satisfied as to all federal, state and local Laws and Regulations that may affect cost, progress and performance of the Work.
- D. Bidder has carefully studied all: (1) reports of explorations and tests of subsurface conditions at or contiguous to the Site and all drawings of physical conditions in or relating to existing surface or subsurface structures at or contiguous to the Site (except Underground Facilities) which have been identified in the plans and technical specifications as provided in paragraph 4.02 of the General Conditions, and (2) reports and drawings of a Hazardous Environmental Condition, if any, which has been identified in the plans and technical specifications as provided in paragraph 4.06 of the General Conditions.
- E. Bidder has obtained and carefully studied (or assumes responsibility for having done so) all additional or supplementary examinations, investigations, explorations, tests, studies, and data concerning conditions (surface, subsurface and Underground Facilities) at or contiguous to the Site which may affect costs, progress, or performance of the Work or which relate to any aspect of the means, methods, techniques, sequences, and procedures of construction to be employed by Bidder, including applying the specific means, methods, techniques, sequences, and procedures of construction expressly required by the Bidding Documents to be employed by Bidder, and safety precautions and programs incident thereto.
- F. Bidder does not consider that any further examinations, investigations, explorations, tests, studies, or data are necessary for the determination of this Bid for performance of the Work at the price(s) Bid and within the times and in accordance with the other terms and conditions of the Bidding Documents.
- G. Bidder is aware of the general nature of the Work to be performed by the County and others at the Site that relates to the Work as indicated in the Bidding Documents.
- H. Bidder has correlated the information known to Bidder, information and observations obtained from visits to the Site, reports and drawings identified in the Bidding Documents, and all additional examinations, investigations, explorations, tests, studies, and data with the Bidding Documents.
- Bidder has given Engineer of Record written notice of all conflicts, errors, ambiguities, or discrepancies that Bidder has discovered in the Bidding Documents, and the written resolution thereof by Engineer of Record is acceptable to Bidder.
- The Bidding Documents are generally sufficient to indicate and convey understanding of all terms and conditions for the performance of the Work for which this Bid is submitted.
- 4.01 Bidder further represents that this Bid is genuine and not made in the interest of or on behalf of any undisclosed individual or entity and is not submitted in conformity with any agreement or rules of any group, association, organization or corporation; Bidder has not directly or indirectly induced or solicited any other Bidder to submit a false or sham Bid; Bidder has not solicited or induced any individual or entity to refrain from Bidding; and Bidder has not sought by collusion to obtain for itself any advantage over any other Bidder or over the County.

5.01 Neither the Recipient nor any of its Contractors or their Subcontractors shall enter into any Contract, Subcontract or arrangement in connection with the Project or any property included or planned to be included in the Project in which any member, officer or employee of the Recipient or the locality during tenure or for 2 years thereafter has any interest, direct or indirect. If any such present or former member, officer or employee involuntarily acquires or had acquired prior to the beginning of tenure any such interest, and if such interest is immediately disclosed to the Recipient, the Recipient, with prior approval of the Department, may waive the prohibition contained in this paragraph provided that any such present member, officer or employee shall not participate in any action by the Recipient or the locality relating to such Contract, Subcontract or arrangement. The Recipient shall insert in all Contracts entered into in connection with the Project or any property included or planned to be included in any Project, and shall require its Contractors to insert in each of their Subcontracts, the following provision:

"No member, officer or employee of the Recipient or of the locality during his tenure or for 2 years thereafter shall have any interest, direct or indirect, in this Contract or the proceeds thereof."

The provisions of this paragraph shall not be applicable to any agreement between the Recipient and its fiscal depositories or to any agreement for utility services the rates for which are fixed or controlled by a governmental agency.

6.01 Bidder will complete the Work in accordance with the Contract Documents for the following price(s):

[remainder of page intentionally left blank]

Invitation to Bid – NC22-013 Citrona Drive from Hickory Street to Beech Street

PAY ITEM NO.	PAY ITEM DESCRIPTION	UOM	QTY	UNIT PRICE	т	DTAL COST
0101 -1	MOBILIZATION	LS	1	\$ 110,475.00	\$	110,475.00
102-1	MAINTENANCE OF TRAFFIC	LS	1	\$ 29,000.00	\$	29,000.00
102-60	WORK ZONE SIGN, F & I	ED	2275	\$ 1.00	\$	2,275.00
102-74-1	CHANNELIZING DEVICE - TYPES I, II, DI, VP, DRUM, OR LCD	ED	1300	\$ 1.00	\$	1,300.00
104-10-3	SEDIMENT BARRIER	LF	1271	\$ 2.00	\$	2,542.00
110-1-1	CLEARING & GRUBBING	AC	0.12	\$ 20,000.00	\$	2,400.00
110-4-10	REMOVAL OF EXISTING CONCRETE	SY	416	\$ 25.00	\$	10,400.00
120-1	REGULAR EXCAVATION	СҮ	16	\$ 100.00	\$	1,600.00
120-2-2	BORROW EXCAVATION, TRUCK MEASURE	CY	11	\$ 100.00	\$	1,100.00
210-1	REWORKING LIMEROCK BASE	SY	5	\$ 300.00	\$	1,500.00
334-1-13	SUPERPAVE ASPHALTIC CONCRETE, TRAFFIC C	TN	1	\$ 1,500.00	\$	1,500.00
520-1-10	CONCRETE CURB & GUTTER, TYPE F	LF	42	\$ 75.00	\$	3,150.00
522-1	CONCRETE SIDEWALK, 4" THICK	SY	368	\$ 75.00	\$	27,600.00
522-2	CONCRETE SIDEWALK, 6" THICK	SY	45	\$ 90.00	\$	4,050.00
527-2	DETECTABLE WARNINGS SF	×	145	\$ 55.00	\$	7,975.00
515-1-2	PIPE HANDRAIL - GUI DERAIL, ALUMINUM	LF	15	\$ 250.00	\$	3,750.00
570-1-2	PERFORMANCE TURF, SOD	SY	415	\$ 10.00	\$	4,150.00
711-11-123	THERMOPLASTIC, STD, WHITE SOLID FOR CROSSWALK AND ROUNDABOUT 12"	LF	835	\$ 4.00	\$	3,340.00
711-11-125	THERMOPLASTIC, STD, WHITE SOLID FOR STOP LINE AND CROSSWALK 24"	LF	520	\$ 7.00	\$	3,640.00

BID TAB CITRONA DRIVE FROM HICKORY STREET TO BEECH STREET

BASE BID TOTAL

\$_____\$221,747.00 (Use Figures)

Two Hundred Twenty One Thousand Seven Hundred Forty Seven Dollars Zero Cents (Use Words)

All specific cash allowances are included in the price(s) set forth above and have been computed in accordance with paragraph 11.02 of the General Conditions.

Unit prices have been computed in accordance with Paragraph 11.03.B of the General Conditions.

Bidder acknowledges that estimated quantities are not guaranteed and are solely for the purpose of comparison of Bids. Final payment of all Unit Price Bid Items will be based on actual quantities as determined in the Contract Documents.

- 6.02 Bidder agrees that the Work will be substantially complete within 50 calendar days after the date when the Contract Times commence to run as provided in paragraph 2.03 of the General Conditions, and completed and ready for final payment in accordance with paragraph 14.07.B of the General Conditions within 15 calendar days from the date of substantial completion. Total Contract time shall be 65 calendar days; for everyday the Work goes beyond substantial completion; a day will be removed from final completion so the total days equal 65 calendar days.
- 6.03 Bidder accepts the provisions of the Agreement as to Liquidated Damages in the event of failure to complete the Work within the times specified above, which shall be stated in the Agreement.
- 7.01 The following documents are attached to and made a condition of this Bid:
 - A. Bid Security in the form of a certified check of Bid Bond (Section 00 43 15)
 - B. Tabulation of Subcontractors & Suppliers (Section 00 43 35)
 - C. Florida Trench Safety Act Certification (Section 00 44 55)
 - D. Drug-Free Workplace Certificate (Section 00 45 20)
 - E. Public Entity Crimes Statement (Section 00 45 30)
 - F. Statement of Disputes, Litigation & Surety Completion (Section 00 45 35)
 - G. LAP Certification of Current Capacity
 - H. Bid Opportunity List for Commodities & Contractual Services
 - I. Non-Collusion Declaration and Compliance Form
 - J. Certification Regarding Debarment, Suspension, Ineligibility and Voluntary Exclusion
 - K. Certification for Disclosure of Lobbying Activities
- 8.01 The terms used in this Bid with initial capital letters have the meanings indicated in the Instructions to Bidders, the General Conditions, and the Supplementary Conditions.

SUBMITTED on _____ December 22 ____, 20 __22 ___.

State License No. ____ P15000050530 / CGC1524573

If Bidder is:

Invitation to Bid – NC22-013 Citrona Drive from Hickory Street to Beech Street

By: (Individual's Signature) Doing business as: Business address: Business address: Fax No.: Phone No.: Fax No.: mership Fax No.: Partnership Name: By: (Signature of general partner - attach evidence of authority to sign) Name (typed or printed): Business address: Phone No.: Fax No.: poration Corporation Name: CGC, Inc. Corporation Name: CGC, Inc. (SEAL) State of Incorporation: Florida Type (General Business, Professional, Service, Limited Liability): S. Corporation By: Signature - attach evidence of authority to sign) Name (typed or printed): Richard C. Gaskin, Jr. Signature - attach evidence of authority to sign) Name (typed or printed): Richard C. Gaskin, Jr. Title: President, Secretary Attest , Richard C. Gaskin, Assistant Secretary, Vi Signature of Corporate Secretary) Business address: 7036 W. 12th Street Jacksonville, FL 32220 Phone No.: (904) 783-4119 Fax No.: (904) 783-3401	Name (typed or prin	nted):	/	
Doing business as:	By:			
Dusiness address:		(Individual's	's Signature)	
Dusiness address:	Doing business as:			
Immership Partnership Name: By:	Business address:			
Immership Partnership Name: By:			/	
Partnership Name:	Phone No.:		Fax-No.:	_
Partnership Name:			\times	
By:	tnership	/		
Name (typed or printed):				
Name (typed or printed):	By:			
Business address:	(5/g	nature of general partne	er - attach evidence of authonty to sign)	
Business address:	Nama (tunad or fi	atadl		
Phone No.:				milian
Phone No.:	business address.			
poration CGC, Inc. (SEAL) State of Incorporation: Florida (SEAL) Type (General Business, Professional, Service, Limited Liability): S. Corporation By: (Signature - attach evidence of authority to sign) Name (typed or printed): Richard C. Gaskin, Jr. Title: President, Secretary Attest , Richard C. Gaskin, Assistant Secretary, VI Signature of Corporate Secretary) Business address: 7036 W. 12th Street Jacksonville, FL 32220	Bhone No :			TE .
Corporation Name: CGC, Inc. (SEAL) State of Incorporation: Florida Type (General Business, Professional, Service, Limited Liability): S. Corporation By: (Signature - attach evidence of authority to sign) Name (typed or printed): Richard C. Gaskin, Jr. Title: President, Secretary Attest , Richard C. Gaskin, Assistant Secretary, VI Signature of Corporate Secretary) Secretary Business address: 7036 W. 12th Street Jacksonville, FL 32220 Secretary			14	52
Corporation Name: CGC, Inc. (SEAL) State of Incorporation: Florida Type (General Business, Professional, Service, Limited Liability): S. Corporation By: (Signature - attach evidence of authority to sign) Name (typed or printed): Richard C. Gaskin, Jr. Title: President, Secretary Attest , Richard C. Gaskin, Assistant Secretary, VI Signature of Corporate Secretary) Secretary Business address: 7036 W. 12th Street Jacksonville, FL 32220 Secretary	noration		0.0	× 3
State of Incorporation: Florida Type (General Business, Professional, Service, Limited Liability): S. Corporation By: (Signature - attach evidence of authority to sign) Name (typed or printed): Richard C. Gaskin, Jr. Title: President, Secretary Attest , Richard C. Gaskin, Assistant Secretary, VI Signature of Corporate Secretary Signature of Corporate Secretary Business address: 7036 W. 12th Street Jacksonville, FL 32220 Signature	Corporation Name:	CGC, Inc.	(SEA	L) C
Type (General Business, Professional, Service, Limited Liability): S. Corporation By: (Signature - attach evidence of authority to sign) Name (typed or printed): Richard C. Gaskin, Jr. Title: President, Secretary Attest , Richard C. Gaskin, Assistant Secretary, VI Signature of Corporate Secretary Signature of Corporate Secretary Business address: 7036 W. 12th Street Jacksonville, FL 32220 Signature	State of Incorporati	on: Florida		
By: (Signature - attach evidence of authority to sign) Name (typed or printed): Richard C. Gaskin, Jr. Title: President, Secretary (CORPORATE SEAL) Attest , Richard C. Gaskin, Assistant Secretary, VI (Signature of Corporate Secretary) Business address: 7036 W. 12th Street Jacksonville, FL 32220	Type (General Busin	ress, Professional, Servic	e, Limited Liability): S. Corporation	5
Name (typed or printed): Richard C. Gaskin, Jr. Title: President, Secretary Attest , Richard C. Gaskin, Assistant Secretary, VI Signature of Corporate Secretary Business address: 7036 W. 12th Street Jacksonville, FL 32220	By:			
Title: President, Secretary (CORPORATE SEAL) Attest , Richard C. Gaskin, Assistant Secretary, VI Signature of Corporate Secretary) Business address: 7036 W. 12th Street Jacksonville, FL 32220	(Sig	nature - attach evidence	e of authority to sign)	
Attest (Signature of Corporate Secretary) Business address: 7036 W. 12th Street Jacksonville, FL 32220	Name (typed or prin	nted): Richard C. Gas	skin, Jr.	NTS -
(Signature of Corporate Secretary) Business address: 7036 W. 12th Street Jacksonville, FL 32220	Title: l	President, Secretary	(CORPORATE SEAL)	.33
(Signature of Corporate Secretary) Business address: 7036 W. 12th Street Jacksonville, FL 32220			(g. ,	s.Y.
Business address: 7036 W. 12th Street Jacksonville, FL 32220	Attest		, Richard C. Gaskin, Assistant Secretar	y, VE
Jacksonville, FL 32220	(Sig	nature of Corporate Sec	cretary)	04
Jacksonville, FL 32220			· · · · · · · · · · · · · · · · · · ·	······
	Business address: _	7036 VV. 12th Stree		magas
Phone No.: (904) 783-4119 Fax No.: (904) 783-3401				
	Phone No.:	(904) 783-4119	Fax No.: (904) 783-3401	-
	Date of Qualificatio	n to do business is	6/10/2015	



STATE OF FLORIDA COUNTY OF DUVAL

I HEREBY CERTIFY

THAT <u>Richard C. Gaskin, Jr.</u>, is <u>President/Secretary</u> of <u>CGC, Inc.</u>, located at <u>7036 W 12th St., Jacksonville, FL 32220</u> and he is hereby authorized to execute the Bid Documents dated, <u>December 22</u>, 2022, and that his execution thereof, attested by the Vice President/Assistant Secretary of the corporation and with corporate seal affixed, shall be the official act and deed of this corporation.

IN WITNESS WHEREOF, I have hereunto set my hand and affixed the official seal of the corporation this <u>22nd</u> day of <u>December</u>, 2022.

(sea Richard C. Gaskin, Vice President/Assistant Secretary Sworn and subscribed to me this _____22nd day of December , 2022 Notary Public Duclard Notary Public State of Florida Signature Burkard Tammy Tammy Burkard Exp.10/11/2026 Printed

My Commission Expires: 10/11/2026

Invitation to Bid – NC22-013 Citrona Drive from Hickory Street to Beech Street

Joint Venturer Name:	
By:	artner - attach evidence of authority to sign)
(Signature of joint venture p	arther - attach evidence of authority to sign)
Name (typed orpinited).	
Title:	
Business address:	
Phone No.:	Fax No.
Joint Venturer Name:	
By:	X
(Sianature - attach e	evidence of authority to sign)
Name (typed or printed):	
Title:	
Business address:	
Phone No.:	Fax No.:
Di La Martina La Malana	- for reasing of official communications:
Phone and FAX Number, and Addres	ss for receipt of official communications:

(Each joint venturer must sign. The manner of signing for each individual, partnership, and corporation that is a party to the joint venture should be in a manner indicated above.)

SECTION 00 43 15

BID BOND

Any singular reference to Bidder, Surety, Owner, or other party shall be considered plural where applicable.

BIDDER (Name and Address): CGC, Inc. 7036 W. 12th Street Jacksonville, FL 32220

SURETY (Name and Address of Principal Place of Business): Argonaut Insurance Company, 225 W Washington St., 24th Floor, Chicago, IL 60606

OWNER (Name and Address):

Board of County Commissioners Nassau County, Florida, 96135 Nassau Place, Suite 1, Nassau County, FL

BID

Bid Due Date: December 22nd, 2022 Project (Brief Description Including Location): NC22–013 Citrona Drive from Hickory Street to Beech Street Fernandina Beach, FL 32034

BOND

Bond Number: TBD Date (Not later than Bid due date): TBD 5% Penal Sum: FIVE percent of the amount bid (Words) (Figures)

Surety and Bidder, intending to be legally bound hereby, subject to the terms printed on the reverse side hereof, do each cause this Bid Bond to be duly executed on its behalf by its authorized officer, agent, or representative.

BIDDER

CGC. Inc. (Seal) Bidder's Name and Corporate Seal

By: Signature and Title Richard C. Gaskin, Jr., President

Attest: Signature and Title Richard C. Gaskin, Vice President

SURETY

Argonaut Insurance Company (Seal) Surety's Name and Corporate Sea Bv. Signature and Title Paul A. Locascio, Atlamay-in Pact & Fla. Reside (Attach Power of Attorney) Attest: Signature and Title AE

Note: Above addresses are to be used for giving required notice.

00-43-15-1

Argonaut Insurance Company Deliveries Only: 225 W. Washington, 24th Floor Chicago, IL 60606 United States Postal Service: P.O. Box 469011, San Antonio, TX 78246 POWER OF ATTORNEY

KNOW ALL MEN BY THESE PRESENTS: That the Argonaut Insurance Company, a Corporation duly organized and existing under the laws of the State of Illinois and having its principal office in the County of Cook, Illinois does hereby nominate, constitute and appoint:

L Dale Waldorff, K Wayne Walker, Pamela L, Jarman, Benjamin H French, Paul A Locascio, Rebekah F, Sharp, Trava Ridion, Ronald I Hays

Their true and lawful agent(s) and attorney(s)-in-fact, each in their separate capacity if more than one is named above, to make, execute, seal and deliver for and on its behalf as surety, and as its act and deed any and all bonds, contracts, agreements of indemnity and other undertakings in suretyship provided, however, that the penal sum of any one such instrument executed hereunder shall not exceed the sum of:

\$97,550,000.00

This Power of Attorney is granted and is signed and sealed under and by the authority of the following Resolution adopted by the Board of Directors of Argonaut Insurance Company:

"RESOLVED, That the President, Senior Vice President, Vice President, Assistant Vice President, Secretary, Treasurer and each of them hereby is authorized to execute powers of attorney, and such authority can be executed by use of facsimile signature, which may be attested or acknowledged by any officer or attorney, of the Company, qualifying the attorney or attorneys named in the given power of attorney, to execute in behalf of, and acknowledge as the act and deed of the Argonaut Insurance Company, all bond undertakings and contracts of suretyship, and to affix the corporate seal thereto."

IN WITNESS WHEREOF, Argonaut Insurance Company has caused its official seal to be hereinto affixed and these presents to be signed by its duly authorized officer on the 19th day of November, 2021. Argonaut Insurance Company



Gary E. Grose , President

STATE OF TEXAS COUNTY OF HARRIS SS:

On this 19th day of November, 2021 A.D., before me, a Notary Public of the State of Texas, in and for the County of Harris, duly commissioned and qualified, came THE ABOVE OFFICER OF THE COMPANY, to me personally known to be the individual and officer described in, and who executed the preceding instrument, and he acknowledged the execution of same, and being by me duly sworn, deposed and said that he is the officer of the said Company aforesaid, and that the seal affixed to the preceding instrument is the Corporate Scal of said Company, and the said Corporate Scal and his signature as officer were duly affixed and subscribed to the said instrument by the authority and direction of the said corporation, and that Resolution adopted by the Board of Directors of said Company, referred to in the preceding instrument is now in force.

IN TESTIMONY WHEREOF, I have hereunto set my hand, and affixed my Official Seal at the County of Harris, the day and year first above written.



tathun m. much

(Notary Public)

2022

I, the undersigned Officer of the Argonaut Insurance Company, Illinois Corporation, do hereby certify that the original POWER OF ATTORNEY of which the foregoing is a full, true and correct copy is still in full force and effect and has not been revoked.

IN WITNESS WHEREOF, I have hereunto set my hand, and affixed the Seal of said Company, on the 22 day of ______



Austin W. King Secretary

IF YOU HAVE QUESTIONS ON AUTHENTICITY OF THIS DOCUMENT CALL (833) 820 - 9137.

DAMAGES FORM

1. Bidder and Surety, jointly and severally, bind themselves, their heirs, executors, administrators, successors, and assigns to pay to the County upon default of Bidder any difference between the total amount of Bidder's Bid and the total amount of the Bid of the next lowest, responsible Bidder who submitted a responsive Bid as determined by the County for the Work required by the Contract Documents, provided that:

- 1.1. If there is no such next Bidder, and the County does not abandon the Project, then Bidder and Surety shall pay to the County the penal sum set forth on the face of this Bond, and
- 1.2. In no event shall Bidder's and Surety's obligation hereunder exceed the penal sum set forth on the race of this Bond.

2. Default of Bidder shall occur upon the failure of Bidder to deliver within the time required by the Bidding Documents (or any extension thereof agreed to in writing by the County) the executed Agreement required by the Bidding Documents and any performance and payment Bonds required by the Bidding Documents.

3. This obligation shall be null and void if:

- 3.1. The County accepts Bidder's Bid and Bidder delivers within the time required by the Bidding Documents (or any extension thereof agreed to in writing by the County) the executed Agreement required by the Bidding Documents arid any performance and payment Bonds required b) the Bidding Documents, or
- 3.2. All Bids are rejected by the County, or
- 3.3. The County fails to issue a Notice of Award to Bidder within the time specified in the Bidding Documents (or any extension thereof agreed to in writing by Bidder and, if applicable, consented to by Surety when required by Paragraph 5 hereof).

4. Payment under this Bond will be due and payable upon default by Bidder and within 30 calendar days after receipt by Bidder and Surety of written notice of default from the County, which notice will be given with reasonable promptness, identifying this Bond and the Project and including a statement of the amount due.

5. Surety waives notice of any and all defenses based on or arising out of any time extension to issue Notice of Award agreed to in writing by the County and Bidder, provided that the total time for issuing Notice of Award including extensions shall not in the aggregate exceed 120 days from Bid due date without Surety's written consent.

6. No suit or action shall be commenced under this Bond prior to 30 calendar days after the notice of default required in Paragraph 4 above is received by Bidder and Surety and in no case later than one year after Bid due date.

7. Any suit or action under this Bond shall be commenced only in a court of competent jurisdiction located in the slate in which the Project is located.

8. Notices required hereunder shall be in writing and sent to Bidder and Surety at their respective addresses shown on the face of this Bond. Such notices may be sent by personal delivery, commercial courier, or by United States Registered or Certified Mail, return receipt requested, postage pre-paid, and shall be deemed to be effective upon receipt by the party concerned.

9. Surety shall cause to be attached to this Bond a current and effective Power of Attorney evidencing the authority of the officer, agent, or representative who executed this Bond on behalf of Surety to execute, seal, and deliver such Bond and bind the Surety thereby.

10. This Bond is intended to conform to all applicable statutory requirements. Any applicable requirement of any applicable statute that has been omitted from this Bond shall be deemed to be included herein as if set forth at length. If any provision of this Bond conflicts with any applicable statute, then the provision of said statute shall govern and the remainder of this Bond that is not in conflict therewith shall continue in full force and effect.

11. The term "Bid" as used herein includes a Bid, offer, or proposal as applicable.

[Remainder of page intentionally left blank.]

SECTION 00 43 35

TABULATION OF SUBCONTRACTORS & SUPPLIERS

The undersigned states that the following is a full and complete list of the proposed Subcontractors and suppliers on this Project. Provide the class of Work and percentage of Work to be performed by each. This list will not be added to nor altered without written consent to the Owner through the Engineer.

SUBCONTRACTOR OR SUPPLIER AND ADDRESS	CLASS OF WORK OR MATERIAL/PER	CENTAGE
Safety Contractors, Inc.	Pipe Handrail	1%
5307 Wacissa Ave., Jacksonville, FL 32254		
Landmark Paving, LLC dba D.H. Striping, Co 750 Clark St., Oviedo, FL 32765	. Pavement Markings	2.5%

CGC, Inc.
Richard C. Gaskin, Jr.
President
12/22/2022

SECTION 00 44 55

FLORIDA TRENCH SAFETY ACT CERTIFICATION

Bidder acknowledges that included in the various items of the proposal and in the Total Bid Price are costs for complying with the Florida Trench Safety Act (90-96, Laws of Florida) as modified October 1, 2008. The Bidder further identifies the costs to be summarized below:

	Cost
1. Trench Safety Act Compliance	\$ 0.00
2. Special Shoring	\$_0.00

Identify method of compliance for Item #1: <u>Slope all excavations as required to meet</u> requirements of the Florida Trench Safety Act. No trenching is anticipated on this project Identify or attach a copy of Special Shoring requirements for Item #2: _____

None required on this project.

The undersigned certifies that he/she is the Contractor who will perform the trench excavation for this project, and hereby gives written assurance that Contractor will comply with the applicable trench safety standards specifically set forth in Florida's Trench Safety Act, Laws of Florida, 90-96.

BIDDER:	CGC, Inc.
By:	- la
Name:	Richard C. Gaskin, Jr.
Title:	President

STATE of Florida

COUNTY of Duval

Sworn to and subscribed before me by means of \underline{X} physical presence or _____ online notarization on this 22nd_day of _____ December____, 2022___.

- Man Birlo	J. Tammy Burkard	
Notary Public		Notary Public State of Florida
My Commission Expires:	10/11/2026	Tammy Burkard My Commission HH 200898 Exp.10/11/2026

SECTION 00 45 20

DRUG FREE WORKPLACE CERTIFICATE

I, the undersigned, in accordance with Florida Statute 287.087, hereby certify that ______ CGC, Inc._____ (print or type name of firm):

- Publishes a written statement notifying that the unlawful manufacture, distribution, dispensing, possession or use of a controlled substance in the Workplace named above that specifies actions that will be taken against violations of such prohibition.
- Informs employees about the dangers of drug abuse in the Workplace, the firm's policy
 of maintaining a drug free Working environment, and available drug counseling,
 rehabilitation, and employee assistance programs, and the penalties that may be imposed
 upon employees for drug use violations.
- Gives each employee engaged in providing commodities or Contractual services that are under Bid or proposal, a copy of the statement specified above.
- Notifies the employees that as a condition of Working on the commodities or Contractual services that are under Bid or proposal, the employee will aBide by the terms of the statement and will notify the employer of any conviction of, plea of guilty or nolo contendere to, any violation of Chapter 1893, or any controlled substance law of the State of Florida or the United States, for a violation occurring in the Work place, no later than five (5) days after such conviction, and requires employees to sign copies of such written statement to acknowledge their receipt.
- Imposes a sanction on, or requires the satisfactory participation in, a drug abuse assistance or rehabilitation program, if such is available in the employee's community, by any employee who is so convicted.
- Makes a good faith effort to continue to maintain a drug free Workplace through the implementation of a drug free Workplace program.

		Gene
		Authorized Signatu December 22, 2022
		Date Signe
State of:	Florida	
County of:		eres of X shurled seconds or coline potarization
Sworn to and on this _22nd	subscribed before me by d day of <u>December</u>	eans of <u>X</u> physical presence or <u>online notarization</u> , 20 <u>22</u> by <u>Richard C. Gaskin, Jr.</u> produced <u>N.A.</u> as identification
Sworn to and on this _22nd	subscribed before me by d day of <u>December</u>	eans of <u>X</u> physical presence oronline notarization , 20 <u>22</u> by <u>Richard C. Gaskin, Jr.</u> produced <u>N.A.</u> as identification (Specify type of Identification)
Sworn to and on this <u>22n</u> c who <u>X is per</u>	subscribed before me by d day of <u>December</u> sonally known to me or _	, 20 <u>22</u> by <u>Richard C. Gaskin, Jr.</u> produced <u>N.A.</u> as identification (Specify type of Identification)
Sworn to and on this <u>22n</u> c who <u>X is per</u>	subscribed before me by d day of <u>December</u> sonally known to me or _ my Burland , Tar	, 20 <u>22</u> by <u>Richard C. Gaskin, Jr.</u> produced <u>N.A.</u> as identification (Specify type of Identification)

SECTION 00 45 30

SWORN STATEMENT UNDER SECTION 287.133(3)(a), FLORIDA STATUTES, ON PUBLIC ENTITY CRIMES

(To be signed in the presence of a notary public or other officer authorized to administer oaths.)

STATE OF Florida

COUNTY OF Duval

Before me, the undersigned authority, personally appeared <u>Richard C. Gaskin, Jr.</u>, who being by me first duly sworn, made the following statement:

	1.	The business address of	CGC, Inc.
is		CHON PACIFIC HOUSE	(firm name of Bidder/Contractor) 7036 W. 12th Street, Jacksonville, FL 32220
	2.	My relationship to	CGC, Inc.
is			(firm name of Bidder/Contractor) President
		Inclusion in the second reasons	

(relationship such as sole proprietor, partner, president, vice president)

3. I understand that a public entity crime as defined in Section 287.133 of the Florida Statutes includes a violation of any state or federal law by a person with respect to and directly related to the transaction of business with any public entity in Florida or with an agency or political subdivision of any other state or with the United States, including, but not limited to, any Bid or Contract for goods or services to be provided to any public entity or such an agency or political subdivision and involving antitrust, fraud, theft, bribery, collusion, racketeering, conspiracy, or material misrepresentation.

4. I understand that "convicted" or "conviction" is defined by the statute to mean a finding of guilt or a conviction of a public entity crime, with or without adjudication or guilt, in any federal or state trial of record relating to charges brought by indictment or information after July 1, 1989, as a result of a jury verdict, nonjury trial, or entry of a plea of guilty or nolo contendere.

5. I understand that "affiliate" is defined by the statute to mean (1) a predecessor or successor of a person or a corporation convicted of a public entity crime; or (2) an entity under control of any natural person with is active in management of the entity and who has been convicted of a public entity crime; (3) those officers, directors, executives, partners, shareholders, employees, members, and agents who are active in the management of an affiliate; or (4) a person or corporation who knowingly entered into a joint venture with a person who has been convicted of a public entity crime in Florida during the preceding 36 months.

6. Neither the Bidder/Contractor nor any officer, director, executive, partner, shareholder, employee, member or agent who is active in the management of the Bidder/Contractor, nor any affiliate of the Bidder/Contractor, has been convicted of a public entity crime subsequent to July 1, 1989.

(Draw a line through Paragraph 6 if Paragraph 7 below applies.)

7. There has been a conviction of a public entity crime by Bidder/Contractor, or an officer, director, executive, partner, shareholder, employee, member or agent of the Bidder/Contractor who is active in the management of the Bidder/Contractor or an affiliate of the Bidder/Contractor. A determination has been made pursuant to 287.133(3) by order of the Division of Administrative Hearings that it is not in the public interest of the name of the convicted person or affiliate to appear on the convicted vendor list.

The name of the convicted person or affiliate is

(Draw a line through Paragraph 7 if Paragraph 6 above applies.)

A copy of the order of the Division of Administrative Hearings is attached to this statement.

CGC, Inc., Richard C. Gaskin, Jr., President

(Signature)

Sworn to and subscribed before me by means of X physical presence or _____ online notarization on this ______ 22nd_day of ______ December ______, 20_22___.

Tammy Burkard Notary Public

10/11/2026 My commission expires



(affix seal)

SECTION 00 45 35

BIDDER'S STATEMENT OF DISPUTES, LITIGATION, ARBITRATION, AND SURETY COMPLETION, LAST THREE (3) YEARS

Na	
ime and Address of	
Owner or Engineer	Name and Phone of
Date of	

Status

NONE

(Signed) Print Name Richard C. Gaskin, Jr., President CGC, Inc.

Sworn to and subscribed before me by means of <u>X</u> physical presence or _____online notarization on this <u>22nd</u> day of <u>_____ocember___</u>, 20<u>_22___</u>.

00-45-35-1

million Notary Public State of Florida Tammy Burkard My Commission HH 280898 Exp. 10/11/2026

My-commission expires: 10/11/2026

, Tammy Burkard

Notary Public

DocuSign Envelope ID: 5D3013DB-2C27-40B2-8CCD-57B583C9BC42

Section 00 45 40 LAP FORMS

- LAP Certification of Current Capacity
- Bid Opportunity List for Commodities & Contractual Services
- Non-Collusion Declaration and Compliance Form
- Certification Regarding Debarment, Suspension, Ineligibility and Voluntary Exclusion
- Certification for Disclosure of Lobbying Activities

[Remainder of page intentionally left blank.]

DocuSign Envelope ID: 5D3013DB-2C27-40B2-8CCD-57B583C9BC42

STATE OF FLORIDA DEPARTMENT OF TRANSPORTATION

CONFIDENTIAL per Ch 337.14(1) F.S.

For bids to be received on _

12/22/2022 (Letting Date)

	F	ill ir	ı ya	our	FD	от	Ve	end	or N	Vun	nbe	r	
VF	F	3	7	1	7	8	5	6	5	7	0	0	1

(Only applicable to FDOT pre-qualified contractors)

CERTIFICATE

I hereby certify that the amount of any proposal submitted by this bidder for the above letting does not exceed the amount of the Firm's CURRENT CAPACITY (maximum capacity rating less total uncompleted work).

The total uncompleted work as shown on the "Status of Contracts on Hand" report (page 2)

3,949,100.00

I further certify that the "Status of Contracts on Hand" report (page 2) was prepared as follows.

1. If the letting is before the 25th day of the month, the certificate and report reflect the uncompleted work as of the 15th day of the month, last preceding the month of the letting.

2. If the letting is after the 25th day of the month, the certificate and report reflects the uncompleted work in progress as of the 15th day of the month of the letting.

3. All new contracts (and subcontracts) awarded earlier than five days before the letting date are included in the report and charged against our total rating.

I certify that the information above is correct.

Sworn to and subscribed this 22nd day of _____ December ____, 20 2022

	CGC, Inc.
	NAME OF FIRM
Dur /	
By: ⊆	Richard C. Gaskin, Jr., President
	Title

525-010-46 PROGRAM MANAGEMENT 09/20 Page 1 cf 2

525-010-46 PROGRAM MANAGEMENT 09/20 Page 2 of 2

STATUS OF CONTRACTS ON HAND

(Furnish complete information about all your contracts, whether prime or subcontracts; whether in progress or awarded, but not yet begun; and regardless of whom contracted with.)

\$3,949,100.00	RKON	TOTAL UNCOMPLETED WORK ON HAND TO BE DONE BY YOU (TOTAL COLUMNS 5 AND 6)	a single item all e, amount to less than	or may consolidate and list as (al, and which, in the aggregat	amounts to be shown to nearest \$100. The Contractor may consolidate and list as a single item all contracts which, individually, do not exceed 3% of total, and which, in the aggregate, amount to less than 20% of the total.
\$587,300.00	\$3,361,800.00	TOTALS	4 to be difference amount in column 4. All	ubcontract) amounts. Column 3 to be uncompleted portion of	NOTE: Columns 2 and 3 to show total contract (or subcontract) amounts. Column 4 to be difference between columns 2 and 3. Amount in columns 5 or 6 to be uncompleted portion of amount in column 4. All
	\$450,000.00	\$500,000.00	\$1,227,700.00	\$1,727,700.00	The City of Palatka, 201 North Second Street, Palatka, FL 32177 St. Johns Avenue, Palatka FL 32177 St. Johns Avenue Streetscape Improvements
\$587,300.00		\$841,700.00	\$400,000.00	\$1,241,700.00	FBCJAX HOLDINGS LLC via Williams & Rowe Company, Inc. First Baptist Church of Jacksonville at Nocatee, Duval County, FL 32256 Silework Only for Addition
	1,509,200.00	\$1,509,200.00 \$1,509,200.00	\$1,000,000.00	\$2,509,200.00	City of Lake City Lake City Galeway Airport, Columbia County, FL32055 Realignment of Taxiway C and the Milling and Overlay of the Terminal Apron and Taxilanes
	\$1,152,500.00	\$1,152,500.00 \$	\$900,000.00	\$2,052,500.00	Keyslone Heights Airport Authonity Keyslone Heights Airport, Keystone Heights, Clay County, FL 32091 Reconstruct Existing T-Hangar Taxilanes
	\$200,000.00	\$737,100.00	\$600,000.00	\$1,337,100.00	City of Fernandina Beach, Florida Fernandina Beach, Municipal Airport Fernandina Beach, Nassau County, FL 32034 Transient Airccafl Parking Apron Rehabilitation
	\$0.00	\$654,700.00	\$500,000.00	\$1,154,700.00	Florida inland Navigation District Nocatee Parkway (30.117397N / 81.404879W), St. Johns Co., FL 32081 F.I.N.D. SJ-14 Weir and Walkway Replacement
	\$50,000.00	\$1,117,600.00	\$100,000.00	\$1,217,600.00	City of Fernandina Beach, Fernandina Beach, FL 32034, Nassau Co., FL Intersection of Ash Street, Centre Street & Alachua Street to Front Street Waterfront Roadway Improvements
	\$0.00	\$16,500.00	\$275,000.00	\$291,500.00	JEA. 21 W. Church Street, Jacksonville, FL 32202, JEA PO# 189175 Deenwood Park Blvd. Jacksonville, FL 32256, Duval County, FL Deenwood Park Blvd Roadway & Bridge Improvements (JEA PO# 189175)
	\$100.00	\$1,352,200.00	\$1,446,600.00	\$2,798,800.00	The City of Jacksonville, Florida Deerwood Park Blvd, Jacksonville, FL 32255, Duval County, FL Deerwood Park Blvd Roadway & Bridge Improvements (COJ# 1049-04)
AS SUBCONTRACTOR	AS PRIME CONTRACTOR		TO OTHERS	AMOUNT	OWNER, LOCATION AND DESCRIPTION
UNCOMPLETED AMOUNT TO BE DONE BY YOU	NCOMPLETED AMOUN BY YOU	BALANCE OF	AMOUNT	CONTRACT (OR	PROJECTS
თ	თ	4	ω	2	1

STATE OF FLORIDA DEPARTMENT OF TRANSPORTATION **BID OPPORTUNITY LIST FOR COMMODITIES & CONTRACTUAL** SERVICES

Prime Contractor:	CGC, Inc.	_
Address/Phone Number: _	7036 W. 12th Street, Jacksonville, FL 32220 / (904) 783-4119	
Procurement Number:	Financial Project ID 441241-1-58-01	_

49 CFR Part 26.11 The list is intended to be a listing of all firms that are participating, or attempting to participate, on DOT-assisted contracts. The list must include all firms that bid on prime contracts, or bid or quote subcontracts and supplies materials on DOT-assisted projects, including both DBEs and non-DBEs. This list must include all subcontractors contacting you and expressing an interest in teaming with you on a specific DOT-assisted project. Prime contractors must provide information for Numbers 1, 2, 3 and 4, and should provide any information they have available on Numbers 5, 6, and 7 for themselves, and their subcontractors.

2.	Federal Tax ID Number: 37-1785657 Firm Name: CGC, Inc. Phone: (904) 783-4119 Address: 7036 W. 12th Street Jacksonville, FL 32220	6.	DBE [.] X Non-DBE	 7. Annual Gross Receipts Less than \$1 million Between \$1 - \$5 million Between \$5 - \$10 million Between \$10 - \$15 million More than \$15 million
5.	Year Firm Established: 2015			
2.	Federal Tax ID Number: 59-3062834 Firm Name: Safety Contractors, Inc. Phone: (904) 355-6331 Address: 5307 Wacissa Ave. Jacksonville, FL 32254	6.	X DBE	7. Annual Gross Receipts Less than \$1 million Between \$1 - \$5 million Between \$5 - \$10 million Between \$10 - \$15 million More than \$15 million
5.	Year Firm Established:1991			
2.3.	Federal Tax ID Number: <u>85-3952338</u> Firm Name: <u>Landmark Paving, LLC dba D.H. Stri</u> ping, Phone: <u>(407) 359-1172</u> Address: <u>750 Clark St.</u> Oviedo, FL 32765	6. Co	DBE . 🛛 Non-DBE	7. Annual Gross Receipts Less than \$1 million Between \$1 - \$5 million Between \$5 - \$10 million Between \$10 - \$15 million More than \$15 million
5.	Year Firm Established: _2020			
2. 3.	Federal Tax ID Number: Firm Name: Phone: Address:	6.	DBE Non-DBE	 7. Annual Gross Receipts Less than \$1 million Between \$1 - \$5 million Between \$5 - \$10 million Between \$10 - \$15 million More than \$15 million
	Year Firm Established:			

REPLY (Invitation to Negotiate – ITN)

375-040-62 PROCUREMENT

01/16

	N	ON-COLLUSION	ARTMENT OF TRANSPORTATION N DECLARATION AND WITH 49 CFR § 29	575-060-13 RIGHT OF WAY 05/01 Page 1 of 3
			ITEM/SEGMENT N F.A.P. NO.: MANAGING DISTR	
			PARCEL NO.: COUNTY OF: BID LETTING OF:	Nassau County Board of County Commissioners Nassau County, Florida
l	Richard C. Gaskin, Jr.	(NAME)		, hereby declare that I am
	President	(NAME) of	CGC, Inc.	
of	(TITLE) Jacksonville, Florida			(FIRM)

(CITY AND STATE)

and that I am the person responsible within my firm for the final decision as to the price(s) and amount of this Bid on this State Project.

I further declare that:

 The prices(s) and amount of this bid have been arrived at independently, without consultation, communication or agreement, for the purpose of restricting competition with any other contractor, bidder or potential bidder.

2. Neither the price(s) nor the amount of this bid have been disclosed to any other firm or person who is a bidder or potential bidder on this project, and will not be so disclosed prior to the bid opening.

3. No attempt has been made or will be made to solicit, cause or induce any other firm or person to refrain from bidding on this project, or to submit a bid higher than the bid of this firm, or any intentionally high or non-competitive bid or other form of complementary bid.

4. The bid of my firm is made in good faith and not pursuant to any agreement or discussion with, or inducement from, any firm or person to submit a complementary bid.

5. My firm has not offered or entered into a subcontract or agreement regarding the purchase of materials or services from any firm or person, or offered, promised or paid cash or anything of value to any firm or person, whether in connection with this or any other project, in consideration for an agreement or promise by any firm or person to refrain from bidding or to submit a complementary bid on this project.

6. My firm has not accepted or been promised any subcontract or agreement regarding the sale of materials or services to any firm or person, and has not been promised or paid cash or anything of value by any firm or person, whether in connection with this or any other project, in consideration for my firm's submitting a complementary bid, or agreeing to do so, on this project.

7. I have made a diligent inquiry of all members, officers, employees, and agents of my firm with responsibilities relating to the preparation, approval or submission of my firm's bid on this project and have been advised by each of them that he or she has not participated in any communication, consultation, discussion, agreement, collusion, act or other conduct inconsistent with any of the statements and representations made in this Declaration.

8. As required by Section 337.165, Florida Statutes, the firm has fully informed the Department of Transportation in writing of all convictions of the firm, its affiliates (as defined in Section 337.165(I)(a), Florida Statutes), and all directors, officers, and employees of the firm and its affiliates for violation of state or federal antitrust laws with respect to a public contract or for violation of any state or federal law involving fraud, bribery, collusion, conspiracy or material misrepresentation with respect to a public contract. This includes disclosure of the names of current employees of the firm or affiliates who were convicted of contract crimes while in the employ of another company.

DocuSign Envelope ID: 5D3013DB-2C27-40B2-8CCD-57B583C9BC42

9. I certify that, except as noted below, neither my firm nor any person associated therewith in the capacity of owner, partner, director, officer, principal, investigator, project director, manager, auditor, and/or position involving the administration of Federal funds:

(a) is presently debarred, suspended, proposed for debarment, declared ineligible, or voluntarily excluded from covered transactions, as defined in 49 CFR §29.110(a), by any Federal department or agency;

(b) has within a three-year period preceding this certification been convicted of or had a civil judgment rendered against him or her for: commission of fraud or a criminal offense in connection with obtaining, attempting to obtain, or performing a Federal, State or local government transaction or public contract; violation of Federal or State antitrust statutes; or commission of embezzlement, theft, forgery, bribery, falsification or destruction of records, making false statements or receiving stolen property;

(c) is presently indicted for or otherwise criminally or civilly charged by a Federal, State or local governmental entity with commission of any of the offenses enumerated in paragraph 9(b) of this certification; and

(d) has within a three-year period preceding this certification had one or more Federal, State or local government public transactions terminated for cause or default.

10. I(We), certify that I(We), shall not knowingly enter into any transaction with any subcontractor, material supplier, or vendor who is debarred, suspended, declared ineligible, or voluntarily excluded from participation in this contract by any Federal Agency unless authorized by the Department.

Where I am unable to declare or certify as to any of the statements contained in the above stated paragraphs numbered (1) through (10), I have provided an explanation in the "Exceptions" portion below or by attached separate sheet.

EXCEPTIONS:

N.A.

(Any exception listed above will not necessarily result in denial of award, but will be considered in determining bidder responsibility. For any exception noted, indicate to whom it applies, initiating agency and dates of agency action. Providing false information may result in criminal prosecution and/or administrative sanctions.)

I declare under penalty of perjury that the foregoing is true and correct.

BY:		Gaskin, Jr., I		WITNESS: Jonathan B. Barton WITNESS: Man Bunland
Sume I		SIGNATURE		Tammy Burkard
Executed on this	22nd	day of	December	

575-060-13 RIGHT OF WAY 05/01 Page 3 of 3

REQUIRED CONTRACT PROVISIONS

This certification applies to subcontractors, material suppliers, vendors and other lower tier participants.

- Appendix B of 49 CFR Part 29 -

Appendix B—Certification Regarding Debarment, Suspension, Ineligibility and Voluntary Exclusion—Lower Tier Covered Transactions

Instructions for Certification

1. By signing and submitting this proposal, the prospective lower tier participant is providing the certification set out below.

2. The certification in this clause is a material representation of fact upon which reliance was placed when this transaction was entered. If it is later determined that the prospective lower tier participant knowingly rendered an erroneous certification, in addition to other remedies available to the Federal Government, the department or agency with which this transaction originated may pursue available remedies, including suspension and/or debarment.

3. The prospective lower tier participant shall provide immediate written notice to the person to which this proposal is submitted if at any time the prospective lower tier participant learns that its certification was erroneous when submitted or has become erroneous by reason of changed circumstances.

4. The terms "covered transaction," "debarred," "suspended," "ineligible," "lower tier covered transaction," "participant," "person," "primary covered transaction," "principal," "proposal," and "voluntarily excluded," as used in this clause, have the meanings set out in the Definitions and Coverage sections of rules implementing Executive Order 12549. You may contact the person to whom this proposal is submitted for assistance in obtaining a copy of those regulations.

5. The prospective lower tier participant agrees by submitting this proposal that it shall not knowingly enter into any lower tier covered transaction with a person who is debarred, suspended, declared ineligible, or voluntarily excluded from participation in this covered transaction, unless authorized by the department or agency with which this transaction originated.

6. The prospective lower tier participant further agrees by submitting this proposal that it will include this clause titled "Certification Regarding Debarment, Suspension, Ineligibility and Voluntary Exclusion--Lower Tier Covered Transaction," without modification, in all lower tier covered transactions and in all solicitations for lower tier covered transactions.

7. A participant in a covered transaction may rely upon a certification of a prospective participant in a lower tier covered transaction that it is not debarred, suspended, ineligible, or voluntarily excluded from the covered transaction, unless it knows that the certification is erroneous. A participant may decide the method and frequency by which it determines the eligibility of its principals. Each participant may, but is not required to, check the Nonprocurement List.

8. Nothing contained in the foregoing shall be construed to require establishment of a system of records in order to render in good faith the certification required by this clause. The knowledge and information of a participant is not required to exceed that which is normally possessed by a prudent person in the ordinary course of business dealings.

9. Except for transactions authorized under paragraph 5 of these instructions, if a participant in a covered transaction knowingly enters into a lower tier covered transaction with a person who is suspended, debarred, ineligible, or voluntarily excluded from participation in this transaction, in addition to other remedies available to the Federal Government, the department or agency with which this transaction originated may pursue available remedies, including suspension and/or debarrent.

Certification Regarding Debarment, Suspension, Ineligibility and Voluntary Exclusion—Lower Tier Covered Transactions

(1) The prospective lower tier participant certifies, by submission of this proposal, that neither it nor its principals is presently debarred, suspended, proposed for debarment, declared ineligible, or voluntary excluded from participation in this transaction by any Federal department or agency.

(2) Where the prospective lower tier participant is unable to certify to any of the statements in this certification, such prospective participant shall attach an explanation to this proposal.

STATE OF FLORIDA DEPARTMENT OF TRANSPORTATION CERTIFICATION REGARDING DEBARMENT, SUSPENSION, INELIGIBILITY AND VOLUNTARY EXCLUSION-LOWER TIER COVERED TRANSACTIONS FOR FEDERAL AID CONTRACTS (Compliance with 2 CFR Parts 180 and 1200)

It is certified that neither the below identified firm nor its principals are presently suspended, proposed for debarment, declared ineligible, or voluntarily excluded from participation in this transaction by any federal department or agency.

Name of Cont	htractor: CGC, Inc.	
By	, Richard C. Gaskin, Jr.	
Date:	December 22, 2022	
Title:	President	

Instructions for Certification

Instructions for Certification - Lower Tier Participants:

(Applicable to all subcontracts, purchase orders and other lower tier transactions requiring prior FHWA approval or estimated to cost \$25,000 or more - 2 CFR Parts 180 and 1200)

a. By signing and submitting this proposal, the prospective lower tier is providing the certification set out below.

b. The certification in this clause is a material representation of fact upon which reliance was placed when this transaction was entered into. If it is later determined that the prospective lower tier participant knowingly rendered an erroneous certification, in addition to other remedies available to the Federal Government, the department, or agency with which this transaction originated may pursue available remedies, including suspension and/or debarment.

c. The prospective lower tier participant shall provide immediate written notice to the person to which this proposal is submitted if at any time the prospective lower tier participant learns that its certification was erroneous by reason of changed circumstances.

d. The terms "covered transaction," "debarred," "suspended," "ineligible," "participant," "person," "principal," and "voluntarily excluded," as used in this clause, are defined in 2 CFR Parts 180 and 1200. You may contact the person to which this proposal is submitted for assistance in obtaining a copy of those regulations. "First Tier Covered Transactions" refers to any covered transaction between a grantee or subgrantee of Federal funds and a participant (such as the prime or general contract). "Lower Tier Covered Transactions" refers to any covered transaction under a First Tier Covered Transaction (such as subcontracts). "First Tier Participant" refers to the participant who has entered into a covered transaction with a grantee or subgrantee of Federal funds (such as the prime or general contractor). "Lower Tier Participant" refers any participant who has entered into a covered transaction with a Subgrantee of Federal funds (such as the prime or general contractor). "Lower Tier Participant" refers any participant who has entered into a covered transaction with a Subgrantee of Federal funds (such as the prime or general contractor). "Lower Tier Participant" refers any participant who has entered into a covered transaction with a Subgrantee of Federal funds (such as the prime or general contractor). "Lower Tier Participant" refers any participant who has entered into a covered transaction with a Subgrantee of Subgrantee of Federal funds (such as the prime or general contractor). "Lower Tier Participant" refers any participant who has entered into a covered transaction with a Subgrantee of Subgrantee of Subgrantee of Participant" refers any participant who has entered into a covered transaction with a Subgrantee of Subgrantee of Participant of the Participant or other Lower Tier Participants (such as subcontractors and suppliers).

e. The prospective lower tier participant agrees by submitting this proposal that, should the proposed covered transaction be entered into, it shall not knowingly enter into any lower tier covered transaction with a person who is debarred, suspended, declared ineligible, or voluntarily excluded from participation in this covered transaction, unless authorized by the department or agency with which this transaction originated.

f. The prospective lower tier participant further agrees by submitting this proposal that it will include this clause titled "Certification Regarding Debarment, Suspension, Ineligibility and Voluntary Exclusion-Lower Tier Covered Transaction," without modification, in all lower tier covered transactions and in all solicitations for lower tier covered transactions exceeding the \$25,000 threshold.

g. A participant in a covered transaction may rely upon a certification of a prospective participant in a lower tier covered transaction that is not debarred, suspended, ineligible, or voluntarily excluded from the covered transaction, unless it knows that the certification is erroneous. A participant is responsible for ensuring that its principals are not suspended, debarred, or otherwise ineligible to participate in covered transactions. To verify the eligibility of its principals, as well as the eligibility of any lower tier prospective participants, each participant may, but is not required to, check the Excluded Parties List System website (https://www.epls.gov/), which is compiled by the General Services Administration.

h. Nothing contained in the foregoing shall be construed to require establishment of a system of records in order to render in good faith the certification required by this clause. The knowledge and information of participant is not required to exceed that which is normally possessed by a prudent person in the ordinary course of business dealings.

i. Except for transactions authorized under paragraph e of these instructions, if a participant in a covered transaction knowingly enters into a lower tier covered transaction with a person who is suspended, debarred, ineligible, or voluntarily excluded from participation in this transaction, in addition to other remedies available to the Federal Government, the department or agency with which this transaction originated may pursue available remedies, including suspension and/or debarment.

CERTIFICATION FOR DISCLOSURE OF LOBBYING ACTIVITIES ON FEDERAL-AID CONTRACTS (Compliance with 49CFR, Section 20.100 (b))

The prospective participant certifies, by signing this certification, that to the best of his or her knowledge and belief:

(1) No federal appropriated funds have been paid or will be paid, by or on behalf of the undersigned, to any person for influencing or attempting to influence an officer or employee of any federal agency, a Member of Congress, an officer or employee of Congress, or an employee of a Member of Congress in connection with the awarding of any federal contract, the making of any federal grant, the making of any federal loan, the entering into of any cooperative agreement, and the extension, continuation, renewal, amendment, or modification of any federal contract, grant, loan, or cooperative agreement.

(2) If any funds other than federal appropriated funds have been paid or will be paid to any person for influencing or attempting to influence an officer or employee of any federal agency, a Member of Congress, an officer or employee of Congress, or an employee of a Member of Congress in connection with this federal contract, grant, loan, or cooperative agreement, the undersigned shall complete and submit Standard Form-LLL, "Disclosure of Lobbying Activities", in accordance with its instructions. (Standard Form-LLL can be obtained from the Florida Department of Transportation's Professional Services Administrator or Procurement Office.)

This certification is a material representation of fact upon which reliance was placed when this transaction was made or entered into. Submission of this certification is a prerequisite for making or entering into this transaction imposed by Section 1352, Title 31, U.S. Code. Any person who fails to file the required certification shall be subject to a civil penalty of not less than \$10,000 and not more than \$100,000 for each such failure.

The prospective participant also agrees by submitting his or her proposal that he or she shall require that the language of this certification be included in all lower tier subcontracts, which exceed \$100,000 and that all such subrecipients shall certify and disclose accordingly.

By:	Richard C. Gaskin, Jr.	Date:	12/22/2022
Authorize	ed Signature:	-	2

STATE OF FLORIDA DEPARTMENT OF TRANSPORTATION DISCLOSURE OF LOBBYING ACTIVITIES

375-030-34 PROCUREMENT 02/16

Is this form applicable to your firm?
YES D NO 🛛
If no, then please complete section 4
below for "Prime"

a, contract b, grant	itatus of Federa a. bid/offer/appli b. initial award c. post-award	cation	Date of last re (mm/dd/yyyy)	ange Only: Quarter: port:	
4. Name and Address of Reporting Entity	Address of Prime		bawardee, Enter Name and		
Congressional District, <i>if known</i> : 4c 6. Federal Department/Agency:		am Name/Descript	ion:		
8. Federal Action Number, if known:	9. Award Amount, if known: \$				
10. a. Name and Address of Lobbying (<i>if individual, last name, first nan</i>	b. Individuals Performing Services (including address if different from No. 10a) (last name, first name, MI):				
11. Information requested through this form is au U.S.C. section 1352. This disclosure of lobby material representation of fact upon which rel by the tier above when this transaction was n into. This disclosure is required pursuant to 3 This information will be available for public in person who fails to file the required disclosure to a civil penalty of not less than \$10,000 and \$100,000 for each such failure.	Print Name: <u>Ric</u> Title: <u>Pre</u>	hard C. Gaskin, Jr sident	e (mm/dd/yyyy): <u>12/22/2022</u>		
Federal Use Only:				Authorized for Local Reproduction Standard Form LLL (Rev. 7-97)	

INSTRUCTIONS FOR COMPLETION OF SF-LLL, DISCLOSURE OF LOBBYING ACTIVITIES

This disclosure form shall be completed by the reporting entity, whether subawardee or prime Federal recipient, at the initiation or receipt of a covered Federal action, or a material change to a previous filing, pursuant to title 31 U.S.C. section 1352. The filing of a form is required for each payment or agreement to make payment to any lobbying entity for influencing or attempting to influence an officer or employee of any agency, a Member of Congress, an officer or employee of Congress, or an employee of a Member of Congress in connection with a covered Federal action. Complete all items that apply for both the initial filing and material change report. Refer to the implementing guidance published by the Office of Management and Budget for additional information.

- 1. Identify the type of covered Federal action for which lobbying activity is and/or has been secured to influence the outcome of a covered Federal action.
- 2. Identify the status of the covered Federal action.
- 3. Identify the appropriate classification of this report. If this is a followup report caused by a material change to the information previously reported, enter the year and quarter in which the change occurred. Enter the date of the last previously submitted report by this reporting entity for this covered Federal action.
- 4. Enter the fullname, address, city, State and zip code of the reporting entity. Include Congressional District, if known. Check the appropriate classification of the reporting entity that designates if it is, or expects to be, a prime or subaward recipient. Identify the tier of the subawardee, e.g., the first subawardee of the prime is the 1st tier. Subawards include but are not limited to subcontracts, subgrants and contract awards under grants.
- 5. If the organization filing the report in item 4 checks "Subawardee," then enter the full name, address, city, State and zip code of the prime Federal recipient. Include Congressional District, if known.
- 6. Enter the name of the Federal agency making the award or loan commitment. Include at least one organizational level below agency name, if known. For example, Department of Transportation, United States Coast Guard.
- Enter the Federal program name or description for the covered Federal action (item 1). If known, enter the full Catalog of Federal Domestic Assistance (CFDA) number for grants, cooperative agreements, loans, and loan commitments.
- Enter the most appropriate Federal identifying number available for the Federal action identified in item 1 (e.g., Request for Proposal (RFP) number; Invitation for Bid (IFB) number; grant announcement number; the contract, grant, or loan award number; the application/proposal control number assigned by the Federal agency). Include prefixes, e.g., "RFP-DE-90-001."
- 9. For a covered Federal action where there has been an award or loan commitment by the Federal agency, enter the Federal amount of the award/loan commitment for the prime entity identified in item 4 or 5.
- (a) Enter the full name, address, city, State and zip code of the lobbying registrant under the Lobbying Disclosure Act of 1995 engaged by the reporting entity identified in item 4 to influence the covered Federal action.
 - (b) Enter the full names of the individual(s) performing services, and include full address if different from 10 (a). Enter Last Name, First Name, and Middle Initial (MI).
- 11. The certifying official shall sign and date the form, print his/her name, title, and telephone number.

According to the Paperwork Reduction Act, as amended, no persons are required to respond to a collection of information unless it displays a valid OMB Control Number. The valid OMB control number for this information collection is OMB No. 0348-0046. Public reporting burden for this collection of information is estimated to average 10 minutes per response, including time for reviewing instructions, searching existing data sources, gathering and maintaining the data needed, and completing and reviewing the collection of information. Send comments regarding the burden estimate or any other aspect of this collection of information, including suggestions for reducing this burden, to the Office of Management and Budget. Paperwork Reduction Project (0348-0046), Washington, DC 20503.



NASSAU COUNTY BOARD OF COUNTY COMMISSIONERS 96135 Nassau Place, Suite 6 Yulee, Florida 32097 John Martin Aaron C. Bell Jeff Gray Thomas R. Ford Klynt Farmer Dist. No. 1 Fernandina Beach Dist. No. 2 Amelia Island Dist. No. 3 Yulee Dist. No. 4 Bryceville/Hilliard Dist. No. 5 Callahan/West Yulee

JOHN A. CRAWFORD Ex-Officio Clerk

MICHAEL S. MULLIN County Attorney

TACO E. POPE, AICP County Manager

E-VERIFY FORM UNDER SECTION 448.095, FLORIDA STATUTES

Project Name: Citrona Drive Pedestrian Improvement from Hickory Street to Beech Street

Bid No./Contract No.: Bid No. NC22-013

DEFINITIONS:

"Contractor" means a person or entity that has entered or is attempting to enter into a contract with a public employer to provide labor, supplies, or services to such employer in exchange for salary, wages, or other remuneration. "Contractor" includes, but is not limited to, a vendor or consultant.

"Subcontractor" means a person or entity that provides labor, supplies, or services to or for a contractor or another subcontractor in exchange for salary, wages, or other remuneration.

"E-Verify System" means an internet-based system operated by the United States Department of Homeland Security that allows participating employers to electronically verify the employment eligibility of newly hired employees.

Effective January 1, 2021, Contractors, shall register with and use the E-Verify System in order to verify the work authorization status of all newly hired employees. Contractor shall register for and utilize the U.S. Department of Homeland Security's E-Verify System to verify the employment eligibility of:

- a) All persons employed by a Contractor to perform employment duties within Florida during the term of the contract; and
- b) All persons (including subvendors/subconsultants/subcontractors) assigned by Contractor to perform work pursuant to the contract with Nassau County. The Contractor acknowledges and agrees that registration and use of the U.S. Department of Homeland Security's E-Verify System during the term of the contract is a condition of the contract with Nassau County; and

(904) 530-6100

An Affirmative Action / Equal Opportunity Employer

- c) Should vendor become the successful Contractor awarded for the above-named project, by entering into the contract, the Contractor shall comply with the provisions of Section 448.095, Florida Statutes, "Employment Eligibility", as amended from time to time. This includes, but is not limited to, registration and utilization of the E-Verify System to verify the work authorization status of all newly hired employees. The Contractor shall also execute the attached affidavit (Exhibit "A") attesting that the Contractor does not employ, contract with, or subcontract with, an unauthorized alien. The Contractor shall maintain a copy of such affidavit for the duration of the contract; and
- d) Contractor shall also require all subcontractors to execute the attached affidavit (Exhibit "B") attesting that the subcontractor does not employ, contract with, or subcontract with, an unauthorized alien. The Contractor shall maintain a copy of such affidavit for the duration of the contract.

CONTRACT TERMINATION:

- a) If Nassau County has a good faith belief that a person or entity with which it is contracting has knowingly violated §448.09(1), Florida Statutes, the contract shall be terminated.
- b) If Nassau County has a good faith belief that a subcontractor knowingly violated §448.095(2), but the Contractor otherwise complied with §448.095(2), Florida Statutes, shall promptly notify the Contractor and order the Contractor to immediately terminate the contract with the subcontractor.
- c) A contract terminated under subparagraph a) or b) is not a breach of contract and may not be considered as such.
- d) Any challenge to termination under this provision must be filed in the Circuit Court no later than twenty (20) calendar days after the date of termination.
- e) If the contract is terminated for a violation of the Statute by the Contractor, the Contractor may not be awarded a public contract for a period of one (1) year after the date of termination.

EXHIBIT "A"

CONTRACTOR E-VERIFY AFFIDAVIT

I hereby certify that <u>CGC, Inc.</u> (Contractor Company Name) does not employ, contract with, or subcontract with an unauthorized alien, and is otherwise in full compliance with Section 448.095, Florida Statutes.

All employees hired on or after January 1, 2021 have had their work authorization status verified through the E-Verify system.

A true and correct copy of <u>CGC, Inc.</u> (Contractor Company Name) proof of registration in the E-Verify system is attached to this Affidavit.

Print Name: Richard C. Gaskin, Jr., President

Date: December 22, 2022

STATE OF FLORIDA

COUNTY OF Duval

The foregoing instrument was acknowledged before me by means of Aphysical presence or □online notarization, this 12/22/22(Date) by <u>Richard C. Gaskin, Jr., President (Name</u> of Officer or Agent, Title of Officer or Agent) of <u>CGC, Inc.</u> (Name of Contractor Company Acknowledging), a <u>Florida</u> (State or Place of Incorporation) Corporation, on behalf of the Corporation. <u>He/She is personally known to</u> me or has produced <u>N.A.</u> as identification.

Bidao Notary Public

Tammy Burkard Printed Name

Notary Public State of Florida Tammy Burkard My Commission HH 280898 Exp.10/11/2026

My Commission Expires: 10/11/2026

E-Verify

Welcome Tammy Burkard

Company Information

Company Name CGC, Inc.

Company ID Number 1171537

Doing Business As (DBA) Name -

DUNS Number 090136554

Physical Location

Address 1 7036 W. 12th Street

Address 2

.

city Jacksonville

State FL

Zip Code

32220

County DUVAL

Additional Information

Employer Identification Number 371785657

Total Number of Employees 10 to 19

Parent Organization

Administrator

Organization Designation

Employer Category Federal Contractor without FAR E-Verify Clause

NAICS Code 237 - HEAVY AND CIVIL ENGINEERING CONSTRUCTION

EXHIBIT "A"

SUBCONTRACTOR E-VERIFY AFFIDAVIT

I hereby certify that <u>Safety Contractors, Inc.</u> does not employ, contract with, or subcontract with an unauthorized alien, and is otherwise in full compliance with Section 448.095, Florida Statutes.

All employees hired on or after January 1, 2021 have had their work authorization status verified through the E-Verify system.

A true and correct copy of <u>Safety Contractors, Inc.</u> Subcontractor proof of registration in the E-Verify system is attached to this Affidavit.

Daniel J	Digitally signed by Daniel J Williams
Williams	Date: 2022.12.21 10:34:23 -05'00'

Print Name: Daniel J. Williams

Date: 12-22-2022

STATE OF FLORIDA

COUNTY OF

The foregoing instrument was acknowledged before me by means of physical presence or ponline notarization, this _____ (Date) by ______ (Name of Officer or Agent, Title of Officer or Agent) of ______ (Name of Contractor Company Acknowledging), a ______ (State or Place of Incorporation) Corporation, on behalf of the Corporation. He/She is personally known to me or has produced ______ as identification.

Notary Public

Printed Name

My Commission Expires: _____
Company ID Number: 170236

	INFORMATION REQUIRED FOR THE E-VERIFY PROGRAM
Information relating to your Comp	any:
Company Name:	Safety Contractors, Inc.
Company Facility Address:	5307 Wacissa Avenue Jacksonville, FL 32254
	Jacksonvine, FL 32234
Company Alternate Address:	
County or Parish:	DUVAL
Employer Identification Number:	593062834
North American Industry Classification Systems Code:	
Parent Company:	
Number of Employees:	20 to 99 Number of Sites Verified for:
Are you verifying for more than 1	site? If yes, please provide the number of sites verified for in each State.
• FLORIDA	1 site(s)

Information relating to the Program Administrator(s) for your Company on policy questions or operational problems:

Name: Telephone Number: E-mail Address:	Lori L Williams (904) 355 - 6331 ext. 402 loriwill@safetycontractors.com	Fax Number:	(904) 354 - 9840
Name: Telephone Number: E-mail Address:	Heather Baerga (904) 355 - 6331 ext. 402 heatherb@safetycontractors.com	Fax Number:	(904) 354 - 9840
Name: Telephone Number: E-mail Address:	Ann B Williams (904) 355 - 6331 ext. 405 annwill@safetycontractors.com	Fax Number:	(904) 354 - 9840

EXHIBIT "B"

SUBCONTRACTOR E-VERIFY AFFIDAVIT

I hereby certify that Landmark for ing LC (Subcontractor Company Name) does not employ, contract with, or subcontract with an unauthorized alien, and is otherwise in full compliance with Section 448.095, Florida Statutes.

All employees hired on or after January 1, 2021 have had their work authorization status verified through the E-Verify system.

A true and correct copy of Landmark for ng UC (Subcontractor Company Name) proof of registration in the E-Verify system is attached to this Affidavit.

Print Name: Date: Mana

COUNTY OF Elminole

The foregoing instrument was acknowledged before me by means of ophysical presence or conline notarization, this <u>Alaria</u> (Date) by <u>Christic Youaish</u> (Name of Officer or Agent, Title of Officer or Agent) of <u>Candmark Paving UC</u> (Name of Contractor Company Acknowledging), a <u>Florida UC</u> (State or Place of Incorporation) Corporation, on behalf of the Corporation. He/She is personally known to me or has produced _______ as identification.

Notary Public

Printed Name

My Commission Expires: 5 au/2025

Notary Public State of Florida Kelly Anna Heas My Commission HH 129350



Company ID Number: 1626345



Information Required for the E-Verify Program Information relating to your Company:								
Company Name	Landmark Paving, LLC							
Company Facility Address	750 Clark Street Oviedo, FL 32765							
Company Alternate Address								
County or Parish	SEMINOLE							
Employer Identification Number	853952338							
North American Industry Classification Systems Code	238							
Parent Company								
Number of Employees	20 to 99							
Number of Sites Verified for	1							

DocuSign Envelope ID: 5D3013DB-2C27-40B2-8CCD-57B583C9BC42

Depart	W-9 Detaber 2018) ment of the Treasury I Revenue Service		Identifica Go to www.irs.go	v/FormW9 for instr	er and Certif	est inforn		1.		1		este	r. D	the o not RS.	
-	1 Name (as shown on CGC, Inc.		tax return). Name is re	quired on this line; do	not leave this line blank	L.									
	2 Business name/disregarded entity name, if different from above														
n page 3.	fallowing saven boxes.													only to s; see	
as on		Individual/sole proprietor or C Corporation IS S Corporation Partnership Trust/estate single-member LLC										(if an	w		
Print or type. Specific Instructions	□ Limited liability company. Enter the tax classification (C=C corporation, S=S corporation, P=Partnership) ► Note: Check the appropriate box in the line above for the tax classification of the single-member owner. Do not check LLC if the LLC is classified as a single-member LLC that is disregarded from the owner unless the owner of the LLC is another LLC that is not disregarded from the owner for U.S. federal tax purposes. Otherwise, a single-member LLC that is disregarded from the owner should check the appropriate box for the tax classification of its owner.										Exemption from FATGA reporting code (if any)				
cific	Other (see instruct		r should check the app	ropnate box for the tax	classification of its ow	nor.			plapies to accounts maintained curside the U.S.						
See Spe	5 Address (number, st 7036 W.	reet, and apt		uctions.		Board	of Co	unty	Comm	ission		l)			
0	6 City, state, and ZIP code 96135 Nassau County, Jacksonville, FL 32220 Yulee, Florida 3										Place, Suite 1				
	7 List account number	(s) here (opti	onali												
Pa	ti Taxpayer	Identifi	cation Number	(TIN)							_	_	_		
back	your TIN in the appro up withholding. For inc ant alien, sole propriet as, it is your employer	fividuals, th or, or disre	his is generally your garded entity, see the	social security numl he instructions for P	ber (SSN). However, art I, later. For other	for a	Socia	Seci	urity nu	mber	-			Τ	
TIN, I	ater.						Empl	overi	dentific	ation	numl	ber	-		
Note	: If the account is in m ber To Give lhe Reque	ore than or ster for gui	idelines on whose n	umber to enter.	AISO SEE VINALINA/INE		37	-	1	7 8	L	6	5	7	

Part II Certification

Under penalties of perjury, I certify that:

- 1. The number shown on this form is my correct taxpayer identification number (or I am waiting for a number to be issued to me); and
- 2. I am not subject to backup withholding because: (a) I am exempt from backup withholding, or (b) I have not been notified by the Internal Revenue Service (IRS) that I am subject to backup withholding as a result of a failure to report all interest or dividends, or (c) the IRS has notified me that I am no longer subject to backup withholding; and
- 3. I am a U.S. cilizen or other U.S. person (defined below); and
- 4. The FATCA code(s) entered on this form (if any) indicating that I am exempt from FATCA reporting is correct.

Certification instructions. You must cross out item 2 above if you have been notified by the IRS that you are currently subject to backup withholding because you have failed to report all interest and dividends on your tax return. For real estate transactions, item 2 does not apply. For mortgage interest paid, acquisition or abandonment of secured property, cancellation of debt, contributions to an individual retirement arrangement (IRA), and generally, payments other than interest and dividends, you are not required to sign the certification, but you must provide your correct TIN. See the instructions for Part II, later.

Sign	Signature of	0	, Richard C. Ga		December 22, 2022
Here	U.S. person	 1-	President	Date P	December 22, 2022

General Instructions

Section references are to the Internal Revenue Code unless otherwise noted.

Future developments. For the latest information about developments related to Form W-9 and its instructions, such as legislation enacted after they were published, go to www.irs.gov/FormW9.

Purpose of Form

An individual or entity (Form W-9 requester) who is required to file an information return with the IRS must obtain your correct taxpayer identification number (TIN) which may be your social security number (SSN), individual taxpayer identification number (TIN), adoption taxpayer identification number (ATIN), or employer identification number (EIN), to report on an information return the amount paid to you, or other amount reportable on an information return. Examples of information return returns include, but are not limited to, the following.

· Form 1099-INT (interest earned or paid)

 Form 1099-DIV (dividends, including those from stocks or mutual funds)

 Form 1099-MISC (various types of income, prizes, awards, or gross proceeds)

 Form 1099-B (stock or mutual fund sales and certain other transactions by brokers)

- Form 1099-S (proceeds from real estate transactions)
- · Form 1099-K (merchant card and third party network transactions)
- Form 1098 (home mortgage interest), 1098-E (student loan interest), 1098-T (tuition)
- · Form 1099-C (canceled debt)
- Form 1099-A (acquisition or abandonment of secured property)

Use Form W-9 only if you are a U.S. person (including a resident alien), to provide your correct TIN.

If you do not return Form W-9 to the requester with a TIN, you might be subject to backup withholding. See What is backup withholding, later.



SAM.GOV[®]

COUNC		
Unique Entity ID WXKJCTJ1N2J5	CAGE / NCAGE 81286	Purpose of Registration All Awards
Registration Status Active Registration	Expiration Date Oct 14, 2023	
Physical Address 7036 12TH ST W Jacksonville, Florida 32220-2402 United States	Mailing Address 7036 12TH ST W Jacksonville, Florida 32220-2402 United States	
Business Information		
Doing Business as (blank)	Division Name (blank)	Division Number (blank)
Congressional District Florida 04	State / Country of Incorporation Florida / United States	URL (blank)
MPIN 		
Registration Dates		and a strange state of the
Activation Date Oct 18, 2022	Submission Date Oct 14, 2022	initial Registration Date Feb 14, 2018
Entity Dates		
Enlity Start Date Jun 10, 2015	Fiscal Year End Close Date Dec 31	
Immediate Owner		
CAGE (blank)	Legal Business Name (blank)	
Highest Level Owner		
CAGE (blank)	Legal Business Name (blank)	

Executive Compensation

In your business or organization's preceding completed fiscal year, did your business or organization (the legal entity to which this specific SAM record, represented by a Unique Entity ID, belongs) receive both of the following: 1. 80 percent or more of your annual gross revenues in U.S. federal contracts, subcontracts, loans, grants, subgrants, and/or cooperative agreements and 2. \$25,000,000 or more in annual gross revenues from U.S. federal contracts, subcontracts, loans, grants, subgrants, and/or cooperative agreements? No

Does the public have access to information about the compensation of the senior executives in your business or organization (the legal antity to which this specific SAM record, represented by a Unique Entity ID, belongs) through periodic reports filed under section 13(a) or 16(d) of the Securities Exchange Act of 1934 (15 U.S.C. 78m(a), 78o(d)) or section 6104 of the Internal Revenue Code of 1986? Not Selected

Proceedings Questions

Is your business or organization, as represented by the Unique Entity ID on this entity registration, responding to a Federal procurement opportunity that contains the provision at FAR 52.209-7, subject to the clause in FAR 52.209-9 in a current Federal contract, or applying for a Federal grant opportunity which contains the award term and condition described in 2 C.F.R. 200 Appendix XII?

Yes

Does your business or organization, as represented by the Unique Entity ID on this specific SAM record, have current active Federal contracts and/or grants with total value (including any exercised/unexercised options) greater than \$10,000,000?

No

Within the last five years, had the business or organization (represented by the Unique Entity ID on this specific SAM record) and/or any of its principals, in connection with the award to or performance by the business or organization of a Federal contract or grant, been the subject of a Federal or State (1) criminal proceeding resulting in a conviction or other acknowledgment of fault; (2) civil proceeding resulting in a finding of fault with a monetary fine, penalty, reimbursement, restitution, and/or damages greater than \$5,000, or other acknowledgment of fault; and/or (3) administrative proceeding resulting in a finding of fault with either a monetary fine or penalty greater than \$5,000 or reimbursement, restitution, or damages greater than \$100,000, or other acknowledgment of fault?

2022 FLORIDA PROFIT CORPORATION ANNUAL REPORT

DOCUMENT# P15000050530

Entity Name: CGC, INC.

Current Principal Place of Business: 7036 W 12TH ST JACKSONVILLE, FL 32220

Current Mailing Address:

7036 W 12TH ST JACKSONVILLE, FL 32220 US

FEI Number: 37-1785657

Name and Address of Current Registered Agent:

GASKIN, RICHARD CANNON JR 4260 MCGIRTS BLVD. JACKSONVILLE, FL 32210 US Certificate of Status Desired: No

The above named entity submits this statement for the purpose of changing its registered office or registered agent, or both. in the State of Florida.

SIGNATURE:

Electronic Signature of Registered Agent

Officer/Director Detail :

Title	PTSD	Tille	VPAS
Name	GASKIN, RICHARD CANNON JR	Name	GASKIN, RICHARD C.
Address	4260 MCGIRTS BLVD.	Address	4965 LONG BOW ROAD
City-State-Zip:	JACKSONVILLE FL 32210	City-Stale-Zip:	JACKSONVILLE FL 32210

I haraby certily that the information indicated on this report or supplemental report is true and accurately that I am an officer or director of the corporation or the receiver or trustae empowered to execute above, or on alternment with all other tike empowered.	ite and that my slectronic signaturo shall have the sam de this report as required by Chapter 607, Florida Statu	a lagal affect as if made under itas; and that my name appears
SIGNATURE RICHARD CANNON GASKIN, JR.	PRESIDENT	01/05/2022

Electronic Signature of Signing Officer/Director Detail

FILED Jan 05, 2022 Secretary of State 9757752244CC

Date



2022 - 2023 LOCAL BUSINESS TAX RECEIPT JIM OVERTON, DUVAL COUNTY TAX COLLECTOR

231 E. Forsylh Street, Suite 130, Jacksonville, FL 32202-3370 Phone: (904) 255-5700, option 3 Fax: (904) 255-8403 https://taxcollector.coj.net/

Note – A penalty is imposed for failure to keep this receipt exhibited conspicuously at your place of business. This business tax receipt is furnished pursuant to Municipal Ordinance Code, Chapters 770-772, for the period October 01, 2022 through September 30, 2023.

CGC, INC 7036 W 12TH ST JACKSONVILLE, FL 32220

ACCOUNT NUMBER:	273299	
BUSINESS NAME:	CGC, INC	
PHYSICAL ADDRESS:	7036 W 12TH ST JACKSONVILLE, FL	32220

CLASSIFICATION CODE: 309001 CONTRACTOR - ALL TYPES

STATE LICENSE NO: CGC1524573

 COUNTY TAX:
 11.25

 MUNICIPAL TAX:
 36.25

 COUNTY LATE PENALTY:
 0.00

 MUNICIPAL LATE PENALTY:
 0.00

 TOTAL TAX:
 47.50

RENEWAL

VALID UNTIL September 30, 2023

ATTENTION

THIS RECEIPT IS FOR BUSINESS TAX RECEIPT ONLY. CERTAIN BUSINESSES MAY REQUIRE ADDITIONAL STATE LICENSING.

This is a business tax receipt only. It does not permit the receipt holder to violate any existing regulatory or zoning laws of the County or City. It does not exempt the receipt holder from any other license or permit required by law. This is not a certification of the receipt holder's qualifications.

JIM OVERTON, TAX COLLECTOR

THIS BECOMES A RECEIPT AFTER VALIDATION. Paid 22081200005041 08/12/2022 \$ 47.50



Florida Department of Transportation

RON DESANTIS GOVERNOR 605 Suwannee Street Tallahassee, FL 32399-0450 JARED W. PERDUE, P.E. SECRETARY

May 13, 2022

CGC, INC. 7036 W 12TH STREET JACKSONVILLE, FLORIDA 32220

RE: CERTIFICATE OF QUALIFICATION

The Department of Transportation has qualified your company for the type of work indicated below.

FDOT APPROVED WORK CLASSES:

DRAINAGE, FLEXIBLE PAVING, GRADING, GRASSING, SEEDING AND SODDING, HOT PLANT-MIXED BITUM. COURSES, INTERMEDIATE BRIDGES, MINOR BRIDGES, ROADWAY SIGNING, SIDEWALK, Curb & Gutter, Driveways, Grout Filled Mat, Miscellaneous Concrete Paving, Underground Utilities, Retaining Wall, Rip Rap.

Unless notified otherwise, this Certificate of Qualification will expire 6/30/2023.

In accordance with Section 337.14(4), Florida Statutes, changes to Ability Factor or Maximum Capacity Rating will not take effect until after the expiration of the current certificate of pregualification (if applicable).

In accordance with Section 337.14(1), Florida Statutes, an application for qualification <u>must be</u> filed within (4) months of the ending date of the applicant's audited annual financial statements.

If the company's maximum capacity has been revised, it may be accessed by logging into the Contractor Prequalification Application System via the following link: HTTPS://fdotwpl.dot.state.fl.us/ContractorPreQualification

Once logged in, select "View" for the most recently approved application, and then click the "Manage" and "Application Summary" tabs.

The company may apply for a Revised Certificate of Qualification at any time prior to the expiration date of this certificate according to Section 14-22.0041(3), Florida Administrative Code (F.A.C.), by accessing the most recently approved application as shown above and choosing "Update" instead of "View." If certification in additional classes of work is desired, documentation is needed to show that the company has performed such work.

All prequalified contractors are required by Section 14-22.006(3), F.A.C., to certify their work underway monthly in order to adjust maximum bidding capacity to available bidding capacity. You can find the link to this report at the website shown above.

umas C. Taylor AA Contracts Administration Office

AA:cg

Improve Safety, Enhance Mobility, Inspire Innovation www.fdot.gov DocuSign Envelope ID: 5D3013DB-2C27-40B2-8CCD-57B583C9BC42



12/16/2022 9:12:22 AM EST

Return to Inquiry Menu

Contractor with Name CGC, INC. 1-1 of 1 contractors

Printer Friendly Version

VENDOR NAME	HOME OFFICE ADDRESS	BIDDING OFFICE ADDRESS	
CGC, INC. F371785657001 EXPIRES: 6/30/2023	7036 W 12TH STREET JACKSONVILLE, FL 32220 (904)783-4119	7036 W 12TH STREET JACKSONVILLE, FL 32220 (904)783-4119	
WORK CLASSES DRAINAGE	F	LEXIBLE PAVING	

Listing

GRADING HOT PLANT-MIXED BITUM. COURSES MINOR BRIDGES SIDEWALK * Curb & Gutter, Driveways, Grout Filled M FLEXIBLE PAVING GRASSING, SEEDING AND SODDING INTERMEDIATE BRIDGES ROADWAY SIGNING

* Curb & Gutter, Driveways, Grout Filled Mat, Miscellaneous Concrete Paving, Underground Utilities, Retaining Wall, Rip Rap.



FLORIDA DEPARTMENT OF TRANSPORTATION

Report Technical Problems to the Service Desk @ 1-866-955-4357 or email: <u>Service Desk</u> Send Prequalification Questions or Comments to <u>Contracts</u> <u>Administration Office</u> <u>Internet Privacy Policy, Disclaimers & Credits</u>



A	CERTIFICATE IS ISSUED AS A MATTER OF INFORMATION ONLY AND CONFERS NO RIGHTS UPON THE CERTIFICATE CORE NOT CONSTRUCT AND CONFERS NO RIGHTS UPON THE CERTIFICATE GENERATIVE OR PRODUCER, MAD THE COREINAGE AFFORDED BY BELOW. THIS CERTIFICATE OF INSURANCE DOES NOT CONSTRUCT A CONTRACT BETWEEN THE ISSUING INSURER(S). REPRESENTATIVE OR PRODUCER, MAD THE CERTIFICATE HOLDER. IMPORTANT: If the certificate holder is an ADDITIONAL INSURED, the policy(ies) must have ADDITIONAL INSURED provisions of issuer an endorsement. A this certificate holder is an ADDITIONAL INSURED, the policy(ies) must have ADDITIONAL INSURED provisions of issuer an endorsement. A this certificate does not confer rights to the certificate holder in lieu of such endorsement(s). DOO Riverside Ave., Suite 500 acksonville FL 32204 INSURER(3).AFFORDMON COVERA. COCINCO30 Result of the summa control of the sum and control of the issuer and the sum and control of the sum and company Insurers a. Westchester Surplus Lines Insurance Company Insurers a. Westchester Countern with RESPECT to THE INSURE ON ANY PEOP INFORMATION OF AWY CONTRACT OR OTHER COUNENT WITH RESPECT TO A MAY DETAINANCE LISTED DECKIM HAVE BEEN ISSUED TO THE INSURED ADDITION OF AWY CONTRACT OR OTHER COUNENT WITH RESPECT TO A EXCLUSIONS AND CONDITIONS OF SUCH POLICIES. LIMITS SHOWN MAY HAVE BEEN ISSUED TO THE INSURED ADDITION OF AWY CONTRACT OR OTHER COUNENT WITH RESPECT TO A EXCLUSIONS AND CONDITIONS OF SUCH POLICIES. LIMITS SHOWN MAY HAVE BEEN ISSUECT TO A EXCLUSION SUMMERY IN ANY DO CONFERCE TO A MAY PERTAINANCE LISTED DECOM MAVE BEEN ISSUED TO THE INSURED ADDITION OF AWY CONTRACT OR OTHER COUNENT WITH				(MM/DD/YYYY /21/2021							
CB	ER'	TIFICATE DO	es n Ertif	OT AFFIRMAT	FIVELY O	R NEGATIVELY AMEND, DOES NOT CONSTITU	, EXTER	ND OR ALT	ER THE CO	OVERAGE AFFORDED	BY THE	POLICIE
If	SU	BROGATION	IS W	AIVED, subjec	t to the te	erms and conditions of the	he polic	y, certain p lorsement(s	olicies may			
RO	DUC	ER	5 1101	donier rights	10 110 001	involte notaet in neu or a	CONTAG	T Missy Am	05			
				the 500			PHONE	Ext. 904-42	1-8600	FAX (A/C, No)	904-42	1-8601
a	kso	onville FL 322	204	ne 500			E-MAIL	is: mamos@	ghgins.com			
								IN	SURER(S) AFFO	RDING COVERAGE		NAIC #
							INSURE	RA: Westfiel	d Insurance	Company		2411
						CGCINC0-01				Lines Insurance Company	У	10172
);	36 V	Vest 12th Str							oup			27847
10	ksc	onville FL 322	220									
							a dela del					
כ	VER	RAGES		CEF	TIFICAT	E NUMBER: 1240464236	TINSURE	NP1	- C	REVISION NUMBER:		
N	DIC.	ATED. NOTWI	THST/ BE ISS	ANDING ANY R	EQUIREME PERTAIN,	INT, TERM OR CONDITION THE INSURANCE AFFORD	OF ANY	CONTRACT	OR OTHER	DOCUMENT WITH RESPE D HEREIN IS SUBJECT T	CT TO	WHICH TH
							DECKIN	POLICY EFF	POLICY EXP	I umi	rs	
		1					-		and a constraint of the second s	maxime and a set of a set of a	\$ 1,000	.000
j	1 -	CLAIMS-MA	DE	X OCCUR						DAMAGE TO RENTED PREMISES (Ea occurrence)	\$ 150,0	
1				2000							\$ 10,00	0
										PERSONAL & ADV INJURY	\$ 1,000,000	
1	GE			PLIES PER:						GENERAL AGGREGATE	\$ 2,000	000
		And the second se	ECT	LOC	1.0	Line - and the second				PRODUCTS - COMP/OP AGG	\$ 2,000. \$	000
	AUT	OMOBILE LIABILI	TY			CWP5147175		12/31/2021	12/31/2022		\$ 1,000	,000
	_	OWNED		SCHEDULED							-	
	x	AUTOS ONLY HIRED	Ŷ	AUTOS NON-OWNED								
	~	AUTOS ONLY	-	AUTOS ONLY		· · · · · · · · -				(Per accident)		
1	х	UMBRELLA LIAB	5	COCCUR		CWP5147175		12/31/2021	12/31/2022	EACH OCCURRENCE	\$ 1,000	000
		EXCESS LIAB						ALC: NOT	Lucio tracko		\$ 1,000,	000
ĺ		and the second se		150	1						\$	
						WFL505900801		1/1/2022	1/1/2023	X PER OTH-		
	ANY	PROPRIETOR/PART	TNERVE	XECUTIVE	N/A					E.L. EACH ACCIDENT	\$ 1,000,	000
1	(Man	datory In NH)										
4	_			NS below		000001 (7175	-	10/24/0004	40/04/0000	and the second se	\$ 1,000,	
	Insta	Ilation Floater	ent			CWP5147175		12/31/2021 12/31/2021 10/28/2021	12/31/2022 12/31/2022 10/28/2022	\$200,000	\$500 \$5,000	Ded, ACV Ded Ded
ic			ONS / LO	DCATIONS / VEHIC	LES (ACORD	AND CLOSE	le, may be	10/28/2021	10/28/2022		\$5,000) Ded
R	TIF	ICATE HOLD	ER				CANC	ELLATION				
		Nassau 96135 N	Coun	ty Board of C u Place, Suite	ounty Co	ommissioners	THE	EXPIRATION RDANCE WI	I DATE THE TH THE POLIC	ESCRIBED POLICIES BE C EREOF, NOTICE WILL I Y PROVISIONS.		
		Yulee, F	L 320)97			AUTHOR	B. G	D.			

The ACORD name and logo are registered marks of ACORD

A	Clerificate costs nor adversariative you inclarity with an concernance with recoverage arrongers and adversariative you inclarity with a concernance in the coverage arrongers in the coverage arrows are coverage arrows and coverage arrows are coverage arrows and coverage arrows are c		- N 602	MM/DD/YYYY) 20/2022												
B	ERT	IFICATE DOE W. THIS CE ESENTATIVE	S N RTIF	OT AFFIRMAT	URA ND T	Y OF	R NEGATIVELY AMEND, DOES NOT CONSTITU ERTIFICATE HOLDER.	TE A	ND OR ALT	ER THE CO BETWEEN 1	VERAGE AFFORDED E THE ISSUING INSURER	SY THE (S), AU	POLICIES			
If	SUE	BROGATION I	S W.	AIVED, subject	to I	he te	rms and conditions of the	he poli	cy, certain p	olicies may	NAL INSURED provision require an endorsemen	t. A st	endorsed atement on			
PRO	UCE	R		somer rights				CONTA	CT Abby Hins	on						
GH	GI	Isurance	Su	ite 500				PHONE (A/C, N	. Ext): 904-42	1-8613	FAX (A/C, No):	904-42	1-8601			
				118 500				É-MAIL ADDRE	ss: ahinson@	ghgins.com						
									INS	SURER(S) AFFOR	ING COVERAGE		NAIC #			
			_					INSUR	IRA: Westfiel	d Insurance C	Company		24112			
		nc					CGCINCO-DI		A bella desta de l'arge de ser	the state of the s	Lines Insurance Company	У	10172			
03	6 V	Vest 12th Stre								que			27847			
lac	kso	nville FL 322	20													
0		AGES		CER	TIEN	CATE	NUMBER: 332164179	INSUR	SRF;		REVISION NUMBER:					
Th	IIS I	S TO CERTIFY	THA	T THE POLICIES	OF	INSU	RANCE LISTED BELOW HA	WE BEE	N ISSUED TO	THE INSUR	ED NAMED ABOVE FOR T	HE POL	ICY PERIOD			
IN	DIC/	FICATE MAY B	THST.	ANDING ANY RI	PERT	REME TAIN, CIES.	NT. TERM OR CONDITION THE INSURANCE AFFORE LIMITS SHOWN MAY HAVE	OF AN	Y CONTRACT THE POLICIE REDUCED BY	OR OTHER	DOCUMENT WITH RESPE D HEREIN IS SUBJECT T	CT TO I	WHICH THIS			
SR		TYPE OF	INSUR	ANCE	ADDL	SUBR WVD	POLICY NUMBER		POLICY EFF (MM/DD/YYYY)	POLICY EXP (MM/DD/YYYY)	LIMO	15				
Α.	Х		-		1.5		CWP5147175		12/31/2022	12/31/2023		\$ 1,000	,000			
	_	CLAIMS-MA	0E	X OCCUR							PREMISES (Ea octutronce)	\$150,0	00			
	_													and the state of t	\$ 10,00	Contract of the second s
												\$ 1,000,000				
	GEN											\$ 2,000				
1	_	A CONTRACT OF A	GT	LOC							PRODUCTS - COMPIOP AGG	\$ 2.000	,000			
A	AUT	and the second se	TY		-	-	CW/P5147175		12/31/2022	12/31/2023	COMBINED SINGLE LIMIT	\$ 1,000	.000			
	-						GWI SHITTS		12.0 made	1210 112023			10 T			
	-	OTHER: OMOBILE LIABILITY ANY AUTO OWNED AUTOS ONLY X SCHE	SCHEDULED		X SCHEDULED							the second s	s			
	x	HIRED	-	NON-OWNED							PROPERTY DAMAGE	5				
	-	AUTOS ONLY	-	AUTOS CALT	-		enandure.				C. C. GUSTERNY	\$				
4	х	UMBRELLA LIAB	T	X OCCUR		-	CWP5147175		12/31/2022	12/31/2023	EACH OCCURRENCE	\$ 1,000	000			
		EXCESS LIAB	-	CLAIMS-MADE							AGGREGATE	\$ 1,000	,000			
		DED X RET	ENTIC	NSO								5	-			
c							WFL605900802		1/1/2023	1/1/2024	X STATUTE ER					
	ANY	PROPRIETCR/PAR	TNER	EXECUTIVE	NIA						E.L. EACH ACCIDENT	\$ 1,000	,000			
	(Mar	datory in NH)				1					which will be a set of the set of		and the second second			
-				ONS below	-	-			101010000	100010000		\$ 1.000	.000 0 Ded, ACV			
A 8	Inst	allation Floater	ent				CWP5147175		12/31/2022	12/31/2023	\$200,000 \$1,000,000/\$2,000,000	\$500 2,500	Ded			
ES	RIPT	TION OF OPERATIC	DNS / 1	LOCATIONS / VEHIC	LES (A	AGORE	L 101, Additional Remarks Sched	ule, may b	L attached if mor	ne spaco is requir	cdj					
E	RTIF	ICATE HOLD	ER		_			CAN	CELLATION			_				
		96135 N	lass	au Place, Suit	coun ∋6	ty Co	ommissioners	THE	EXPIRATIO	N DATE TH	EREOF, NOTICE WILL					
		Yulee, F	L 32	2097				a	:3.5	Rai	ORD CORPORATION.					

The ACORD name and logo are registered marks of ACORD

DocuSign Envelope ID: 5D3013DB-2C27-40B2-8CCD-57B583C9BC42

Tammy Burkard

From: Sent: To: Subject: TrackingUpdates@fedex.com Wednesday, December 21, 2022 2:53 PM office@cgccivil.com FedEx Shipment 770802963499: Your package has been delivered



Hi. Your package was delivered Wed, 12/21/2022 at 2:46pm.

ORIGINAL BID BOND



Delivered to 76347 VETERANS WAY 456, YULEE, FL 32097 Received by S.FORTIER

OBTAIN PROOF OF DELIVERY

TRACKING NUMBER

FROM

770802963499

CGC, Inc. 7036 W. 12th Street Jacksonville, FL, US, 32220

TO Robert M. Foster Justice Center Office of the Ex-Officio Clerk 76347 Veterans Way, Ste 456

1

ACORD [®] CERTIFICATE OF LIABILITY INSURANCE							E	DATE (MM/DD/YYYY) 12/20/2022	
C B	HIS CERTIFICATE IS ISSUED AS A M ERTIFICATE DOES NOT AFFIRMATIV ELOW. THIS CERTIFICATE OF INSU EPRESENTATIVE OR PRODUCER, AN	ELY OR	NEGATIVELY AMEND, DOES NOT CONSTITUT	EXTEN	OR ALT	R THE CO	VERAGE AFFORDED E	BY THE	POLICIES
If	MPORTANT: If the certificate holder is SUBROGATION IS WAIVED, subject t his certificate does not confer rights to	o the ter	ms and conditions of th	he policy	, certain po	olicies may			
_	DUCER	the cert	incate noider in neu or si	I CONTACT					
	IG Insurance			NAME: PHONE	Ext): 904-42		FAX	904-42	1-8601
1000 Riverside Ave., Suite 500 Jacksonville FL 32204				PHONE (A/C, No, Ext): 904-421-8613 E-MAIL ADDRESS: ahinson@ghgins.com				304-42	1-0001
a	CRSONVILLE FL 32204			ADDRESS	a second and the second s		DING COVERAGE		NAIC #
				INCLUSIO	ALC: A MARKED AND A MARKADING	ALC: NOT ALC: NOT	IDING COVERAGE		24112
INSURED CGCINCO-C				INSURER A : Westfield Insurance Company ⁻⁰¹ INSURER B : Westchester Surplus Lines Insurance Company				v	10172
				INSURER C : ICW Group				27847	
	36 West 12th Street cksonville FL 32220			INSURER	a day which many the same state	op			21011
a	CRSONVINE PL 52220			INSURER					
				INSURER					-
:0	VERAGES CERT	IFICATE	NUMBER: 332164179	THOUNER			REVISION NUMBER:		
IN C	HIS IS TO CERTIFY THAT THE POLICIES IDICATED. NOTWITHSTANDING ANY REC ERTIFICATE MAY BE ISSUED OR MAY P XCLUSIONS AND CONDITIONS OF SUCH P	QUIREME ERTAIN,	NT, TERM OR CONDITION THE INSURANCE AFFORD	OF ANY	CONTRACT	OR OTHER	DOCUMENT WITH RESPE D HEREIN IS SUBJECT T	CT TO	WHICH THIS
SR		NSD WVD	POLICY NUMBER	1	POLICY EFF	POLICY EXP (MM/DD/YYYY)	LIMI	rs	
Ą	X COMMERCIAL GENERAL LIABILITY	100 1000	CWP5147175		12/31/2022	12/31/2023	EACH OCCURRENCE	\$ 1,000	,000
	CLAIMS-MADE X OCCUR						DAMAGE TO RENTED PREMISES (Ea occurrence)	10 Carlo and	
							MED EXP (Any one person)	\$ 10,00	
							PERSONAL & ADV INJURY	La Carte Carte Carte	
	GEN'L AGGREGATE LIMIT APPLIES PER:						GENERAL AGGREGATE	\$ 2.000	.000
	POLICY X PRO- JECT LOC						PRODUCTS - COMP/OP AGG	\$ 2,000 \$,000
A	AUTOMOBILE LIABILITY		CWP5147175	12/31/2022	12/31/2023	COMBINED SINGLE LIMIT (Ea accident)	\$ 1,000,000		
	ANY AUTO					BODILY INJURY (Per person)			
	OWNED AUTOS ONLY X AUTOS						BODILY INJURY (Per accident)	s	
	X HIRED X NON-OWNED AUTOS ONLY		1			100	PROPERTY DAMAGE (Per accident)	\$	
							() of additionly	\$	
A	X UMBRELLA LIAB X OCCUR EXCESS LIAB CLAIMS-MADE		CWP5147175		12/31/2022	12/31/2023	EACH OCCURRENCE	\$ 1,000	,000
							AGGREGATE	\$ 1,000	,000
	DED X RETENTIONS 0							s	
С	WORKERS COMPENSATION	N/A	WFL505900802		1/1/2023	1/1/2024	X PER STATUTE ER		
	AND EMPLOYERS' LIABILITY ANYPROPRIETOR/PARTNER/EXECUTIVE				1.1.1.1.1.1.1		E.L. EACH ACCIDENT	\$ 1,000	,000
	(Mandatory in NH)						E.L. DISEASE - EA EMPLOYEE \$ 1,000,000		,000
	If yes, describe under DESCRIPTION OF OPERATIONS below						E.L. DISEASE - POLICY LIMIT	\$ 1,000	,000
AAB	Leased/Rented Equipment Installation Floater Pollution Liability		CWP5147175 CWP5147175 G71754157004		12/31/2022 12/31/2022 10/28/2022	12/31/2023 12/31/2023 10/28/2023	\$750,000 \$200,000 \$1,000,000/\$2,000,000	\$2,50 \$500 2,500	
ES	SCRIPTION OF OPERATIONS / LOCATIONS / VEHICLI	ES (ACORE	9 101, Additional Remarks Schedu	lule, may be	attached if mor	e space is requir	ed)		
CE	RTIFICATE HOLDER			CANC	ELLATION				
	Nassau County Board of Co 96135 Nassau Place, Suite Yulee, FL 32097	ounty Co 6	ommissioners	ACCC	EXPIRATION DRDANCE WI	N DATE TH TH THE POLIC	DESCRIBED POLICIES BE C EREOF, NOTICE WILL CY PROVISIONS.		
		1.357		li	3.5	Peri 188-2015 AC	ORD CORPORATION.	All rig	hts

The ACORD name and logo are registered marks of ACORD

Revised 01/2015

APPENDICES A and E

During the performance of this contract, the contractor, for itself, its assignees and successors in interest (hereinafter referred to as the "Contractor") agrees as follows:

- (1.) Compliance with Regulations: The Contractor shall comply with the Regulations relative to nondiscrimination in Federally-assisted programs of the U.S. Department of Transportation (hereinafter, "USDOT") Title 49, Code of Federal Regulations, Part 21, as they may be amended from time to time, (hereinafter referred to as the Regulations), which are herein incorporated by reference and made a part of this Agreement.
- (2.) Nondiscrimination: The Contractor, with regard to the work performed during the contract, shall not discriminate on the basis of race, color, national origin, sex, age, disability, religion or family status in the selection and retention of subcontractors, including procurements of materials and leases of equipment. The Contractor shall not participate either directly or indirectly in the discrimination prohibited by section 21.5 of the Regulations, including employment practices when the contract covers a program set forth in Appendix B of the Regulations.
- (3.) Solicitations for Subcontractors, including Procurements of Materials and Equipment: In all solicitations made by the Contractor, either by competitive bidding or negotiation for work to be performed under a subcontract, including procurements of materials or leases of equipment; each potential subcontractor or supplier shall be notified by the Contractor of the Contractor's obligations under this contract and the Regulations relative to nondiscrimination on the basis of race, color, national origin, sex, age, disability, religion or family status.
- (4.) Information and Reports: The Contractor shall provide all information and reports required by the Regulations or directives issued pursuant thereto, and shall permit access to its books, records, accounts, other sources of information, and its facilities as may be determined by the Florida Department of Transportation, the Federal Highway Administration, Federal Transit Administration, Federal Aviation Administration, and/or the Federal Motor Carrier Safety Administration to be pertinent to ascertain compliance with such Regulations, orders and instructions. Where any information required of a Contractor is in the exclusive possession of another who fails or refuses to furnish this information the Contractor shall so certify to the Florida Department of Transportation, the Federal Highway Administration, Federal Transit Administration, Federal Aviation Administration, and/or the Federal Motor Carrier Safety Administration as appropriate, and shall set forth what efforts it has made to obtain the information.
- (5.) Sanctions for Noncompliance: In the event of the Contractor's noncompliance with the nondiscrimination provisions of this contract, the *Florida Department of Transportation* shall impose such contract sanctions as it or the *Federal Highway Administration*, *Federal Transit Administration*, *Federal Aviation Administration*, and/or the Federal Motor Carrier Safety Administration may determine to be appropriate, including, but not limited to:
 - a. withholding of payments to the Contractor under the contract until the Contractor complies, and/or
 - b. cancellation, termination or suspension of the contract, in whole or in part.
- (6.) Incorporation of Provisions: The Contractor shall include the provisions of paragraphs (1) through (7) in every subcontract, including procurements of materials and leases of equipment, unless exempt by the Regulations, or directives issued pursuant thereto. The Contractor shall take such action with respect to any subcontract or procurement as the Florida Department of Transportation, the Federal Highway Administration, Federal Transit

Administration, Federal Aviation Administration, and/or the Federal Motor Carrier Safety Administration may direct as a means of enforcing such provisions including sanctions for noncompliance. In the event a Contractor becomes involved in, or is threatened with, litigation with a sub-contractor or supplier as a result of such direction, the Contractor may request the *Florida Department of Transportation* to enter into such litigation to protect the interests of the *Florida Department of Transportation*, and, in addition, the Contractor may request the United States to enter into such litigation to protect the interests of the United States.

(7.) Compliance with Nondiscrimination Statutes and Authorities: Title VI of the Civil Rights Act of 1964 (42 U.S.C. § 2000d et seq., 78 stat. 252), (prohibits discrimination on the basis of race, color, national origin); and 49 CFR Part 21; The Uniform Relocation Assistance and Real Property Acquisition Policies Act of 1970, (42 U.S.C. § 4601), (prohibits unfair treatment of persons displaced or whose property has been acquired because of Federal or Federal-aid programs and projects); Federal-Aid Highway Act of 1973, (23 U.S.C. § 324 et seq.), (prohibits discrimination on the basis of sex); Section 504 of the Rehabilitation Act of 1973, (29 U.S.C. § 794 et seq.), as amended, (prohibits discrimination on the basis of disability); and 49 CFR Part 27; The Age Discrimination Act of 1975, as amended, (42 U.S.C. § 6101 et seq.), (prohibits discrimination on the basis of age); Airport and Airway Improvement Act of 1982, (49 USC § 471, Section 47123), as amended, (prohibits discrimination based on race, creed, color, national origin, or sex); The Civil Rights Restoration Act of 1987, (PL 100-209), (Broadened the scope, coverage and applicability of Title VI of the Civil Rights Act of 1964, The Age Discrimination Act of 1975 and Section 504 of the Rehabilitation Act of 1973, by expanding the definition of the terms "programs or activities" to include all of the programs or activities of the Federal-aid recipients, sub-recipients and contractors, whether such programs or activities are Federally funded or not); Titles II and III of the Americans with Disabilities Act, which prohibit discrimination on the basis of disability in the operation of public entities, public and private transportation systems, places of public accommodation, and certain testing entities (42 U.S.C. §§ 12131 -- 12189) as implemented by Department of Transportation regulations at 49 C.F.R. parts 37 and 38; The Federal Aviation Administration's Non-discrimination statute (49 U.S.C. § 47123) (prohibits discrimination on the basis of race, color, national origin, and sex); Executive Order 12898, Federal Actions to Address Environmental Justice in Minority Populations and Low-Income Populations, which ensures non-discrimination against minority populations by discouraging programs, policies, and activities with disproportionately high and adverse human health or environmental effects on minority and low-income populations; Executive Order 13166, Improving Access to Services for Persons with Limited English Proficiency, and resulting agency guidance, national origin discrimination includes discrimination because of limited English proficiency (LEP). To ensure compliance with Title VI, you must take reasonable steps to ensure that LEP persons have meaningful access to your programs (70 Fed. Reg. at 74087 to 74100); Title IX of the Education Amendments of 1972, as amended, which prohibits you from discriminating because of sex in education programs or activities (20 U.S.C. 1681 et seq).

DocuSign Envelope ID: 5D3013DB-2C27-40B2-8CCD-57B583C9BC42

EXHIBIT "K"

STATE OF FLORIDA DEPARTMENT OF TRANSPORTATION

525-010-40 PROGRAM MANAGEMENT OGC/OOC- 05/21 Page 1 of 15

FPN: 441241-1-58-01	FPN: 441241-1-68-02	FPN:
Federal No (FAIN): D221-042-B	Federal No (FAIN): D221-042-B	Federal No (FAIN):
Federal Award Date:	Federal Award Date:	Federal Award Date:
Fund:	Fund:	Fund:
	Org Code:	Org Code:
FLAIR Approp:	FLAIR Approp:	FLAIR Approp:
FLAIR Obj:	FLAIR Obj:	FLAIR Obj:
County No:74	Contract No:G2717	
Recipient Vendor No: F591863042053	Recipient DUNS No: 829978514	
Catalog of Federal Domestic Assistance	(CFDA): 20.205 Highway Planning a	and Construction

THIS LOCAL AGENCY PROGRAM AGREEMENT ("Agreement"), is entered into on 3/28/2022 | 1:48 PM EDT, by and between the State of Florida Department of Transportation, an agency (This date to be entered by DOT only)

of the State of Florida ("Department"), and Nassau County ("Recipient").

NOW, THEREFORE, in consideration of the mutual benefits to be derived from joint participation on the Project, the Parties agree to the following:

- Authority: The Department is authorized to enter into this Agreement pursuant to Section 339.12, Florida Statutes. The Recipient by Resolution or other form of official authorization, a copy of which is attached as Exhibit "D" and made a part of this Agreement, has authorized its officers to execute this Agreement on its behalf.
- 2. Purpose of Agreement: The purpose of this Agreement is to provide for the Department's participation in <u>construction and CEI project management of sidewalk on Citrona Drive from Hickory Street to Beech Street</u>, as further described in Exhibit "A", Project Description and Responsibilities attached to and incorporated in this Agreement ("Project"), to provide Department financial assistance to the Recipient; state the terms and conditions upon which Department funds will be provided; and to set forth the manner in which the Project will be undertaken and completed.
- 3. Term of Agreement: The Recipient agrees to complete the Project on or before <u>12/31/2023</u>. If the Recipient does not complete the Project within this time period, this Agreement will expire on the last day of the scheduled completion as provided in this paragraph unless an extension of the time period is requested by the Recipient and granted in writing by the Department prior to the expiration of this Agreement. Expiration of this Agreement will be considered termination of the Project. The cost of any work performed after the term of this Agreement will not be reimbursed by the Department.

4. Project Cost:

- a. The estimated cost of the Project is \$ <u>198,598</u>. This amount is based upon the Schedule of Financial Assistance in Exhibit "B", attached to and incorporated in this Agreement. Exhibit "B" may be modified by mutual execution of an amendment as provided for in paragraph 5.i.
- b. The Department agrees to participate in the Project cost up to the maximum amount of \$198,598 and as more fully described in Exhibit "B". This amount includes Federal-aid funds which are limited to the actual amount of Federal-aid participation. The Department's participation may be increased or reduced upon determination of the actual bid amounts of the Project by the mutual execution of an amendment. The Recipient agrees to bear all expenses in excess of the total cost of the Project and any deficits incurred in connection with the completion of the Project.
- c. Project costs eligible for Department participation will be allowed only from the date of this Agreement. It is understood that Department participation in eligible Project costs is subject to:

525-010-40 PROGRAM MANAGEMENT OGC/OOC- 05/21 Page 2 of 15

- Legislative approval of the Department's appropriation request in the work program year that the Project is scheduled to be committed;
- ii. Availability of funds as stated in paragraphs 5.1. and 5.m. of this Agreement;
- ili. Approval of all plans, specifications, contracts or other obligating documents and all other terms of this Agreement; and
- iv. Department approval of the Project scope and budget at the time appropriation authority becomes available.

5. Requisitions and Payments

- a. The Recipient shall provide quantifiable, measurable, and verifiable units of deliverables. Each deliverable must specify the required minimum level of service to be performed and the criteria for evaluating successful completion. The Project and the quantifiable, measurable, and verifiable units of deliverables are described more fully in Exhibit "A".
- b. Invoices shall be submitted by the Recipient in detail sufficient for a proper pre-audit and post-audit based on the quantifiable, measurable and verifiable units of deliverables as established in Exhibit "A". Deliverables must be received and accepted in writing by the Department's Project Manager prior to payments. Requests for reimbursement by the Recipient shall include an invoice, progress report and supporting documentation for the period of services being billed that are acceptable to the Department. The Recipient shall use the format for the invoice and progress report that is approved by the Department.
- c. The Recipient shall charge to the Project account all eligible costs of the Project except costs agreed to be borne by the Recipient or its contractors and subcontractors. Costs in excess of the programmed funding or attributable to actions which have not received the required approval of the Department shall not be considered eligible costs. All costs charged to the Project, including any approved services contributed by the Recipient or others, shall be supported by properly executed payrolls, time records, invoices, contracts or vouchers evidencing in proper detail the nature and propriety of the charges.
- d. Supporting documentation must establish that the deliverables were received and accepted in writing by the Recipient and must also establish that the required minimum level of service to be performed based on the criteria for evaluating successful completion as specified in Exhibit "A" was met. All costs invoiced shall be supported by properly executed payrolls, time records, invoices, contracts or vouchers evidencing in proper detail the nature and propriety of charges as described in Exhibit "F", Contract Payment Requirements.
- e. Bills for travel expenses specifically authorized in this Agreement shall be submitted on the Department's Contractor Travel Form No. 300-000-06 and will be paid in accordance with Section 112.061, Florida Statutes and the most current version of the Disbursement Handbook for Employees and Managers.
- f. Payment shall be made only after receipt and approval of goods and services unless advance payments are authorized by the Chief Financial Officer of the State of Florida under Chapters 215 and 216, Florida Statutes or the Department's Comptroller under Section 334.044(29), Florida Statutes.
 - If this box is selected, advance payment is authorized for this Agreement and Exhibit "H", Alternative Advance Payment Financial Provisions is attached and incorporated into this Agreement.

If the Department determines that the performance of the Recipient is unsatisfactory, the Department shall notify the Recipient of the deficiency to be corrected, which correction shall be made within a time-frame to be specified by the Department. The Recipient shall, within thirty (30) days after notice from the Department, provide the Department with a corrective action plan describing how the Recipient will address all issues of contract non-performance, unacceptable performance, failure to meet the minimum performance levels, deliverable deficiencies, or contract non-compliance. If the corrective action plan is unacceptable to the

525-010-40 PROGRAM MANAGEMENT OGC/OOC- 05/21 Page 3 of 15

Department, the Recipient will not be reimbursed to the extent of the non-performance. The Recipient will not be reimbursed until the Recipient resolves the deficiency. If the deficiency is subsequently resolved, the Recipient may bill the Department for the unpaid reimbursement request(s) during the next billing period. If the Recipient is unable to resolve the deficiency, the funds shall be forfeited at the end of the Agreement's term.

g. Agencies providing goods and services to the Department should be aware of the following time frames. Inspection and approval of goods or services shall take no longer than 20 days from the Department's receipt of the invoice. The Department has 20 days to deliver a request for payment (voucher) to the Department of Financial Services. The 20 days are measured from the latter of the date the invoice is received or the goods or services are received, inspected, and approved.

If a payment is not available within 40 days, a separate interest penalty at a rate as established pursuant to **Section 55.03(1)**, **F.S.**, will be due and payable, in addition to the invoice amount, to the Recipient. Interest penalties of less than one (1) dollar will not be enforced unless the Recipient requests payment. Invoices that have to be returned to an Recipient because of Recipient preparation errors will result in a delay in the payment. The invoice payment requirements do not start until a properly completed invoice is provided to the Department.

A Vendor Ombudsman has been established within the Department of Financial Services. The duties of this individual include acting as an advocate for Agencies who may be experiencing problems in obtaining timely payment(s) from a state agency. The Vendor Ombudsman may be contacted at (850) 413-5516.

- h. The Recipient shall maintain an accounting system or separate accounts to ensure funds and projects are tracked separately. Records of costs incurred under the terms of this Agreement shall be maintained and made available upon request to the Department at all times during the period of this Agreement and for five years after final payment is made. Copies of these documents and records shall be furnished to the Department upon request. Records of costs incurred include the Recipient's general accounting records and the project records, together with supporting documents and records, of the contractor and all subcontractors performing work on the project, and all other records of the Contractor and subcontractors considered necessary by the Department for a proper audit of costs.
- i. Prior to the execution of this Agreement, a Project schedule of funding shall be prepared by the Recipient and approved by the Department. The Recipient shall maintain said schedule of funding, carry out the Project, and shall incur obligations against and make disbursements of Project funds only in conformity with the latest approved schedule of funding for the Project. The schedule of funding may be revised by execution of a Local Agency Program ("LAP") Supplemental Agreement between the Department and the Recipient. The Recipient acknowledges and agrees that funding for this project may be reduced upon determination of the Recipient's contract award amount.
- j. If, after Project completion, any claim is made by the Department resulting from an audit or for work or services performed pursuant to this Agreement, the Department may offset such amount from payments due for work or services done under any agreement which it has with the Recipient owing such amount if, upon demand, payment of the amount is not made within 60 days to the Department. Offsetting any amount pursuant to this paragraph shall not be considered a breach of contract by the Department.
- k. The Recipient must submit the final invoice on the Project to the Department within 120 days after the completion of the Project. Invoices submitted after the 120-day time period may not be paid.
- I. The Department's performance and obligation to pay under this Agreement is contingent upon an annual appropriation by the Legislature. If the Department's funding for this Project is in multiple fiscal years, funds approval from the Department's Comptroller must be received each fiscal year prior to costs being incurred. See Exhibit "B" for funding levels by fiscal year. Project costs utilizing these fiscal year funds are not eligible for reimbursement if incurred prior to funds approval being received. The Department will notify the Recipient, in writing, when funds are available.
- m. In the event this Agreement is in excess of \$25,000 and has a term for a period of more than one year, the provisions of Section 339.135(6)(a), Florida Statutes, are hereby incorporated:

525-010-40 PROGRAM MANAGEMENT OGC/OOC- 05/21 Page 4 of 15

"The Department, during any fiscal year, shall not expend money, incur any liability, or enter into any contract which, by its terms, involves the expenditure of money in excess of the amounts budgeted as available for expenditure during such fiscal year. Any contract, verbal or written, made in violation of this subsection is null and void, and no money may be paid on such contract. The Department shall require a statement from the comptroller of the Department that funds are available prior to entering into any such contract or other binding commitment of funds. Nothing herein contained shall prevent the making of contracts for periods exceeding 1 year, but any contract so made shall be executory only for the value of the services to be rendered or agreed to be paid for in succeeding fiscal years, and this paragraph shall be incorporated verbatim in all contracts of the Department which are for an amount in excess of \$25,000 and which have a term for a period of more than 1 year."

6. Department Payment Obligations:

Subject to other provisions of this Agreement, the Department will honor requests for reimbursement to the Recipient pursuant to this Agreement. However, notwithstanding any other provision of this Agreement, the Department may elect by notice in writing not to make a payment if:

- a. The Recipient shall have made misrepresentation of a material nature in its application, or any supplement or amendment to its application, or with respect to any document or data furnished with its application or pursuant to this Agreement;
- b. There is any pending litigation with respect to the performance by the Recipient of any of its duties or obligations which may jeopardize or adversely affect the Project, the Agreement or payments to the Project;
- c. The Recipient shall have taken any action pertaining to the Project which, under this Agreement, requires the approval of the Department or has made a related expenditure or incurred related obligations without having been advised by the Department that same are approved;
- d. There has been any violation of the conflict of interest provisions contained in paragraph 14.f.; or
- e. The Recipient has been determined by the Department to be in default under any of the provisions of the Agreement.

The Department may suspend or terminate payment for that portion of the Project which the Federal Highway Administration ("FHWA"), or the Department acting in lieu of FHWA, may designate as ineligible for Federal-aid.

In determining the amount of the payment, the Department will exclude all Project costs incurred by the Recipient prior to the Department's issuance of a Notice to Proceed ("NTP"), costs incurred after the expiration of the Agreement, costs which are not provided for in the latest approved schedule of funding in **Exhibit "B"** for the Project, costs agreed to be borne by the Recipient or its contractors and subcontractors for not meeting the Project commencement and final invoice time lines, and costs attributable to goods or services received under a contract or other arrangements which have not been approved in writing by the Department.

7. General Requirements:

The Recipient shall complete the Project with all practical dispatch, in a sound, economical, and efficient manner, and in accordance with the provisions in this Agreement, and all applicable laws. The Project will be performed in accordance with all applicable Department procedures, guidelines, manuals, standards, and directives as described in the Department's **Local Agency Program Manual** (FDOT Topic No. 525-010-300), which by this reference is made a part of this Agreement. Time is of the essence as to each and every obligation under this Agreement.

- a. A full time employee of the Recipient, qualified to ensure that the work being pursued is complete, accurate, and consistent with the terms, conditions, and specifications of this Agreement shall be in responsible charge of the Project, which employee should be able to perform the following duties and functions:
 - i. Administers inherently governmental project activities, including those dealing with cost, time,

525-010-40 PROGRAM MANAGEMENT OGC/OOC- 05/21 Page 5 of 15

adherence to contract requirements, construction quality and scope of Federal-aid projects;

- ii. Maintains familiarity of day to day Project operations, including Project safety issues;
- Makes or participates in decisions about changed conditions or scope changes that require change orders or supplemental agreements;
- iv. Visits and reviews the Project on a frequency that is commensurate with the magnitude and complexity of the Project;
- Reviews financial processes, transactions and documentation to ensure that safeguards are in place to minimize fraud, waste, and abuse;
- Directs Project staff, agency or consultant, to carry out Project administration and contract oversight, including proper documentation;
- vii. Is aware of the qualifications, assignments and on-the-job performance of the Recipient and consultant staff at all stages of the Project.
- b. Once the Department issues the NTP for the Project, the Recipient shall be obligated to submit an invoice or other request for reimbursement to the Department no less than once every 90 days (quarterly), beginning from the day the NTP is issued. If the Recipient fails to submit quarterly invoices to the Department, and in the event the failure to timely submit invoices to the Department results in the FHWA removing any unbilled funding or the loss of state appropriation authority (which may include the loss of state and federal funds, if there are state funds programmed to the Project), then the Recipient will be solely responsible to provide all funds necessary to complete the Project and the Department will not be obligated to provide any additional funding for the Project. The Recipient waives the right to contest such removal of funds by the Department, if the removal is related to FHWA's withdrawal of funds or if the removal is related to the loss of state appropriation authority. In addition to the loss of funding for the Project, the Department will also consider the de-certification of the Recipient for future LAP Projects. No cost may be incurred under this Agreement until after the Recipient has received a written NTP from the Department. The Recipient agrees to advertise or put the Project out to bid thirty (30) days from the date the Department issues the NTP to advertise the Project. If the Recipient is not able to meet the scheduled advertisement, the Department District LAP Administrator should be notified as soon as possible.
- c. If all funds are removed from the Project, including amounts previously billed to the Department and reimbursed to the Recipient, and the Project is off the State Highway System, then the Department will have to request repayment for the previously billed amounts from the Recipient. No state funds can be used on off-system projects, unless authorized pursuant to Exhibit "I", State Funds Addendum, which will be attached to and incorporated in this Agreement in the event state funds are used on the Project.
- d. In the event that any election, referendum, approval, permit, notice or other proceeding or authorization is required under applicable law to enable the Recipient to enter into this Agreement or to undertake the Project or to observe, assume or carry out any of the provisions of the Agreement, the Recipient will initiate and consummate, as provided by law, all actions necessary with respect to any such matters.
- e. The Recipient shall initiate and prosecute to completion all proceedings necessary, including Federal-aid requirements, to enable the Recipient to provide the necessary funds for completion of the Project.
- f. The Recipient shall submit to the Department such data, reports, records, contracts, and other documents relating to the Project as the Department and FHWA may require. The Recipient shall make such submissions using Department-designated information systems.
- g. Federal-aid funds shall not participate in any cost which is not incurred in conformity with applicable federal and state laws, the regulations in 23 Code of Federal Regulations (C.F.R.) and 49 C.F.R., and policies and procedures prescribed by the Division Administrator of FHWA. Federal funds shall not be paid on account of any cost incurred prior to authorization by FHWA to the Department to proceed with the Project or part thereof involving such cost (23 C.F.R. 1.9 (a)). If FHWA or the Department determines that any amount

525-010-40 PROGRAM MANAGEMENT OGC/OOC- 05/21 Page 6 of 15

claimed is not eligible, federal participation may be approved in the amount determined to be adequately supported and the Department shall notify the Recipient in writing citing the reasons why items and amounts are not eligible for federal participation. Where correctable non-compliance with provisions of law or FHWA requirements exists federal funds may be withheld until compliance is obtained. Where non-compliance is not correctable, FHWA or the Department may deny participation in parcel or Project costs in part or in total. For any amounts determined to be ineligible for federal reimbursement for which the Department has advanced payment, the Recipient shall promptly reimburse the Department for all such amounts within 90 days of written notice.

h. For any project requiring additional right-of-way, the Recipient must submit to the Department an annual report of its real property acquisition and relocation assistance activities on the project. Activities shall be reported on a federal fiscal year basis, from October 1 through September 30. The report must be prepared using the format prescribed in 49 C.F.R. Part 24, Appendix B, and be submitted to the Department no later than October 15 of each year.

8. Audit Reports:

The administration of resources awarded through the Department to the Recipient by this Agreement may be subject to audits and/or monitoring by the Department. The following requirements do not limit the authority of the Department to conduct or arrange for the conduct of additional audits or evaluations of federal awards or limit the authority of any state agency inspector general, the State of Florida Auditor General, or any other state official. The Recipient shall comply with all audit and audit reporting requirements as specified below.

- a. In addition to reviews of audits conducted in accordance with 2 CFR Part 200, Subpart F Audit Requirements, monitoring procedures may include, but not be limited to, on-site visits by Department staff and/or other procedures including, reviewing any required performance and financial reports, following up, ensuring corrective action, and issuing management decisions on weaknesses found through audits when those findings pertain to federal awards provided through the Department by this Agreement. By entering into this Agreement, the Recipient agrees to comply and cooperate fully with any monitoring procedures/processes deemed appropriate by the Department. The Recipient further agrees to comply and cooperate with any inspections, reviews, investigations, or audits deemed necessary by the Department, State of Florida Chief Financial Officer ("CFO"), or State of Florida Auditor General.
- b. The Recipient, a non-federal entity as defined by 2 CFR Part 200, as a subrecipient of a federal award awarded by the Department through this Agreement is subject to the following requirements:
 - i. In the event the Recipient expends a total amount of federal awards equal to or in excess of the threshold established by 2 CFR Part 200, Subpart F Audit Requirements, the Recipient must have a federal single or program-specific audit for such fiscal year conducted in accordance with the provisions of 2 CFR Part 200, Subpart F Audit Requirements. Exhibit "E" to this Agreement provides the required federal award identification information needed by the Recipient to further comply with the requirements of 2 CFR Part 200, Subpart F Audit Requirements. In determining federal awards expended in a fiscal year, the Recipient must consider all sources of federal awards based on when the activity related to the federal award occurs, including the federal award provided through the Department by this Agreement. The determination of amounts of federal awards expended should be in accordance with the guidelines established by 2 CFR Part 200, Subpart F Audit Requirements. An audit conducted by the State of Florida Auditor General in accordance with the provisions of 2 CFR Part 200, Subpart F Audit Requirements. An audit conducted by the State of Florida Auditor General in accordance with the provisions of 2 CFR Part 200, Subpart F Audit Requirements.
 - ii. In connection with the audit requirements, the Recipient shall fulfill the requirements relative to the auditee responsibilities as provided in 2 CFR Part 200, Subpart F Audit Requirements.

525-010-40 PROGRAM MANAGEMENT OGC/OOC- 05/21 Page 7 of 15

- iii. In the event the Recipient expends less than the threshold established by 2 CFR Part 200, Subpart F Audit Requirements, in federal awards, the Recipient is exempt from federal audit requirements for that fiscal year. However, the Recipient must provide a single audit exemption statement to the Department at <u>FDOTSingleAudit@dot.state.fl.us</u> no later than nine months after the end of the Recipient's audit period for each applicable audit year. In the event the Recipient expends less than the threshold established by 2 CFR Part 200, Subpart F Audit Requirements, in federal awards in a fiscal year and <u>elects</u> to have an audit conducted in accordance with the provisions of 2 CFR Part 200, Subpart F Audit Requirements, the cost of the audit must be paid from non-federal resources (*i.e.*, the cost of such an audit must be paid from the Recipient's resources obtained from other than federal entities).
- iv. The Recipient must electronically submit to the Federal Audit Clearinghouse ("FAC") at <u>https://harvester.census.gov/facweb/</u> the audit reporting package as required by 2 CFR Part 200, Subpart F Audit Requirements, within the earlier of 30 calendar days after receipt of the auditor's report(s) or nine months after the end of the audit period. The FAC is the repository of record for audits required by 2 CFR Part 200, Subpart F Audit Requirements, and this Agreement. However, the Department requires a copy of the audit reporting package also be submitted to <u>FDOTSingleAudit@dot.state.fl.us</u> within the earlier of 30 calendar days after receipt of the audit of 30 calendar days after receipt of the auditor's report(s) or nine months after the end of the audit period as required by 2 CFR Part 200, Subpart F Audit Requirements.
- v. Within six months of acceptance of the audit report by the FAC, the Department will review the Recipient's audit reporting package, including corrective action plans and management letters, to the extent necessary to determine whether timely and appropriate action on all deficiencies has been taken pertaining to the federal award provided through the Department by this Agreement. If the Recipient fails to have an audit conducted in accordance with 2 CFR Part 200, Subpart F Audit Requirements, the Department may impose additional conditions to remedy noncompliance. If the Department determines that noncompliance cannot be remedied by imposing additional conditions, the Department may take appropriate actions to enforce compliance, which actions may include but not be limited to the following:
 - Temporarily withhold cash payments pending correction of the deficiency by the Recipient or more severe enforcement action by the Department;
 - Disallow (deny both use of funds and any applicable matching credit for) all or part of the cost of the activity or action not in compliance;
 - 3. Wholly or partly suspend or terminate the federal award;
 - Initiate suspension or debarment proceedings as authorized under 2 C.F.R. Part 180 and federal awarding agency regulations (or in the case of the Department, recommend such a proceeding be initiated by the federal awarding agency);
 - 5. Withhold further federal awards for the Project or program;
 - 6. Take other remedies that may be legally available.
- vi. As a condition of receiving this federal award, the Recipient shall permit the Department or its designee, the CFO, or State of Florida Auditor General access to the Recipient's records including financial statements, the independent auditor's working papers, and project records as necessary. Records related to unresolved audit findings, appeals, or litigation shall be retained until the action is complete or the dispute is resolved.
- vii. The Department's contact information for requirements under this part is as follows:

Office of Comptroller, MS 24 605 Suwannee Street Tallahassee, Florida 32399-0450 FDOTSingleAudit@dot.state.fl.us

- c. The Recipient shall retain sufficient records demonstrating its compliance with the terms of this Agreement for a period of five years from the date the audit report is issued and shall allow the Department or its designee, the CFO, or State of Florida Auditor General access to such records upon request. The Recipient shall ensure that the audit working papers are made available to the Department or its designee, the CFO, or State of Florida Auditor General upon request for a period of five years from the date the audit report is issued, unless extended in writing by the Department.
- 9. Termination or Suspension of Project:

525-010-40 PROGRAM MANAGEMENT OGC/OOC-- 05/21 Page 8 of 15

The Department may, by written notice to the Recipient, suspend any or all of the Department's obligations under this Agreement for the Recipient's failure to comply with applicable law or the terms of this Agreement until such time as the event or condition resulting in such suspension has ceased or been corrected.

- a. If the Department intends to terminate the Agreement, the Department shall notify the Recipient of such termination in writing at least thirty (30) days prior to the termination of the Agreement, with instructions to the effective date of termination or specify the stage of work at which the Agreement is to be terminated.
- b. The Parties to this Agreement may terminate this Agreement when its continuation would not produce beneficial results commensurate with the further expenditure of funds. In this event, the Parties shall agree upon the termination conditions.
- c. If the Agreement is terminated before performance is completed, the Recipient shall be paid only for that work satisfactorily performed for which costs can be substantiated. Such payment, however, may not exceed the equivalent percentage of the Department's maximum financial assistance. If any portion of the Project is located on the Department's right-of-way, then all work in progress on the Department right-of-way will become the property of the Department and will be turned over promptly by the Recipient.
- d. In the event the Recipient fails to perform or honor the requirements and provisions of this Agreement, the Recipient shall promptly refund in full to the Department within thirty (30) days of the termination of the Agreement any funds that were determined by the Department to have been expended in violation of the Agreement.
- e. The Department reserves the right to unilaterally cancel this Agreement for failure by the Recipient to comply with the Public Records provisions of Chapter 119, Florida Statutes.

10. Contracts of the Recipient:

- a. Except as otherwise authorized in writing by the Department, the Recipient shall not execute any contract or obligate itself in any manner requiring the disbursement of Department funds, including consultant or construction contracts or amendments thereto, with any third party with respect to the Project without the written approval of the Department. Failure to obtain such approval shall be sufficient cause for nonpayment by the Department. The Department specifically reserves the right to review the qualifications of any consultant or contractor and to approve or disapprove the employment of such consultant or contractor.
- b. It is understood and agreed by the parties to this Agreement that participation by the Department in a project with the Recipient, where said project involves a consultant contract for engineering, architecture or surveying services, is contingent on the Recipient's complying in full with provisions of Section 287.055, Florida Statutes, Consultants' Competitive Negotiation Act, the federal Brooks Act, 23 C.F.R. 172, and 23 U.S.C. 112. At the discretion of the Department, the Recipient will involve the Department in the consultant selection process for all projects funded under this Agreement. In all cases, the Recipient shall certify to the Department that selection has been accomplished in compliance with the Consultants' Competitive Negotiation Act, and the federal Brooks Act.
- c. The Recipient shall comply with, and require its consultants and contractors to comply with applicable federal law pertaining to the use of Federal-aid funds. The Recipient shall comply with the provisions in the FHWA-1273 form as set forth in Exhibit "G", FHWA 1273 attached to and incorporated in this Agreement. The Recipient shall include FHWA-1273 in all contracts with contractors performing work on the Project.
- d. The Recipient shall require its consultants and contractors to take emergency steps to close any public road whenever there is a risk to life, health and safety of the travelling public. The safety of the travelling public is the Department's first priority for the Recipient. If lane or road closures are required by the LA to ensure the life, health, and safety of the travelling public, the LA must notify the District Construction Engineer and District Traffic Operations Engineer immediately once the travelling public are not at imminent risk. The Department expects professional engineering judgment be applied in all aspects of locally delivered projects. Defect management and supervision of LAP project structures components must be

525-010-40 PROGRAM MANAGEMENT OGC/OOC- 05/21 Page 9 of 15

proactively managed, monitored, and inspected by department prequalified structures engineer(s). The District Construction Engineer must be notified immediately of defect monitoring that occurs in LAP project construction, whether or not the defects are considered an imminent risk to life, health, or safety of the travelling public. When defects, including but not limited to, structural cracks, are initially detected during bridge construction, the engineer of record, construction engineering inspector, design-build firm, or local agency that owns or is responsible for the bridge construction has the authority to immediately close the bridge to construction personnel and close the road underneath. The LA shall also ensure compliance with the CPAM, Section 9.1.8 regarding actions for maintenance of traffic and safety concerns.

11. Disadvantaged Business Enterprise (DBE) Policy and Obligation:

It is the policy of the Department that DBE's, as defined in 49 C.F.R. Part 26, as amended, shall have the opportunity to participate in the performance of contracts financed in whole or in part with Department funds under this Agreement. The DBE requirements of applicable federal and state laws and regulations apply to this Agreement.

The Recipient and its contractors agree to ensure that DBE's have the opportunity to participate in the performance of this Agreement. In this regard, all recipients and contractors shall take all necessary and reasonable steps in accordance with applicable federal and state laws and regulations to ensure that the DBE's have the opportunity to compete for and perform contracts. The Recipient and its contractors and subcontractors shall not discriminate on the basis of race, color, national origin or sex in the award and performance of contracts, entered pursuant to this Agreement.

12. Compliance with Conditions and Laws:

The Recipient shall comply and require its contractors and subcontractors to comply with all terms and conditions of this Agreement and all federal, state, and local laws and regulations applicable to this Project. Execution of this Agreement constitutes a certification that the Recipient is in compliance with, and will require its contractors and subcontractors to comply with, all requirements imposed by applicable federal, state, and local laws and regulations, including the "Certification Regarding Debarment, Suspension, Ineligibility and Voluntary Exclusion – Lower Tier Covered Transactions," in 49 C.F.R. Part 29, and 2 C.F.R. Part 200 when applicable.

13. Performance Evaluations:

Recipients are evaluated on a project-by-project basis. The evaluations provide information about oversight needs and provide input for the recertification process. Evaluations are submitted to the Recipient's person in responsible charge or designee as part of the Project closeout process. The Department provides the evaluation to the Recipient no more than 30 days after final acceptance.

- a. Each evaluation will result in one of three ratings. A rating of Unsatisfactory Performance means the Recipient failed to develop the Project in accordance with applicable federal and state regulations, standards and procedures, required excessive District involvement/oversight, or the Project was brought in-house by the Department. A rating of Satisfactory Performance means the Recipient developed the Project in accordance with applicable federal and state regulations, standards and procedures, with minimal District involvement/oversight. A rating of Above Satisfactory Performance means the Recipient developed the Project in accordance with applicable federal and state regulations, standards and procedures, with minimal District involvement/oversight. A rating of Above Satisfactory Performance means the Recipient developed the Project in accordance with applicable federal and state regulations, standards and procedures, and the Department did not have to exceed the minimum oversight and monitoring requirements identified for the project.
- b. The District will determine which functions can be further delegated to Recipients that continuously earn Satisfactory and Above Satisfactory evaluations.

14. Restrictions, Prohibitions, Controls, and Labor Provisions:

During the performance of this Agreement, the Recipient agrees as follows, and agrees to require its contractors and subcontractors to include in each subcontract the following provisions:

a. The Recipient will comply with all the requirements imposed by Title VI of the Civil Rights Act of 1964, the regulations of the U.S. Department of Transportation issued thereunder, and the assurance by the Recipient pursuant thereto. The Recipient shall include the attached Exhibit "C", Title VI Assurances in all contracts

525-010-40 PROGRAM MANAGEMENT OGC/OOC- 05/21 Page 10 of 15

with consultants and contractors performing work on the Project that ensure compliance with Title VI of the Civil Rights Act of 1964, 49 C.F.R. Part 21, and related statutes and regulations.

- b. The Recipient will comply with all the requirements as imposed by the ADA, the regulations of the Federal Government issued thereunder, and assurance by the Recipient pursuant thereto.
- c. A person or affiliate who has been placed on the convicted vendor list following a conviction for a public entity crime may not submit a bid on a contract to provide any goods or services to a public entity; may not submit a bid on a contract with a public entity for the construction or repair of a public building or public work; may not submit bids on leases of real property to a public entity; may not be awarded or perform work as a contractor, supplier, subcontractor or consultant under a contract with any public entity; and may not transact business with any public entity in excess of the threshold amount provided in Section 287.017, Florida Statutes, for CATEGORY TWO for a period of 36 months from the date of being placed on the convicted vendor list.
- d. In accordance with Section 287.134, Florida Statutes, an entity or affiliate who has been placed on the Discriminatory Vendor List, kept by the Florida Department of Management Services, may not submit a bid on a contract to provide goods or services to a public entity; may not submit a bid on a contract with a public entity for the construction or repair of a public building or public work; may not submit bids on leases of real property to a public entity; may not be awarded or perform work as a contractor, supplier, subcontractor or consultant under a contract with any public entity; and may not transact business with any public entity.
- e. An entity or affiliate who has had its Certificate of Qualification suspended, revoked, denied or have further been determined by the Department to be a non-responsible contractor may not submit a bid or perform work for the construction or repair of a public building or public work on a contract with the Recipient.
- f. Neither the Recipient nor any of its contractors or their subcontractors shall enter into any contract, subcontract or arrangement in connection with the Project or any property included or planned to be included in the Project in which any member, officer or employee of the Recipient or the locality during tenure or for 2 years thereafter has any interest, direct or indirect. If any such present or former member, officer or employee involuntarily acquires or had acquired prior to the beginning of tenure any such interest, and if such interest is immediately disclosed to the Recipient, the Recipient, with prior approval of the Department, may waive the prohibition contained in this paragraph provided that any such present member, officer or employee shall not participate in any action by the Recipient or the locality relating to such contract, subcontract or arrangement. The Recipient shall insert in all contracts entered into in connection with the Project or any property included or planned to be included in any Project, and shall require its contractors to insert in each of their subcontracts, the following provision:

"No member, officer or employee of the Recipient or of the locality during his tenure or for 2 years thereafter shall have any interest, direct or indirect, in this contract or the proceeds thereof."

The provisions of this paragraph shall not be applicable to any agreement between the Recipient and its fiscal depositories or to any agreement for utility services the rates for which are fixed or controlled by a governmental agency.

g. No member or delegate to the Congress of the United States shall be admitted to any share or part of this Agreement or any benefit arising therefrom.

15. Indemnification and Insurance:

a. It is specifically agreed between the parties executing this Agreement that it is not intended by any of the provisions of any part of this Agreement to create in the public or any member thereof, a third-party beneficiary under this Agreement, or to authorize anyone not a party to this Agreement to maintain a suit for personal injuries or property damage pursuant to the terms or provisions of this Agreement. The Recipient guarantees the payment of all just claims for materials, supplies, tools, or labor and other just claims against the Recipient or any subcontractor, in connection with this Agreement.

525-010-40 PROGRAM MANAGEMENT OGC/OOC- 05/21 Page 11 of 15

- b. To the extent provided by law, Recipient shall indemnify, defend, and hold harmless the Department against any actions, claims, or damages arising out of, relating to, or resulting from negligent or wrongful act(s) of Recipient, or any of its officers, agents, or employees, acting within the scope of their office or employment, in connection with the rights granted to or exercised by Recipient hereunder, to the extent and within the limitations of Section 768.28, Florida Statutes. The foregoing indemnification shall not constitute a waiver of the Department's or Recipient's sovereign immunity beyond the limits set forth in Florida Statutes, Section 768.28, nor shall the same be construed to constitute agreement by Recipient to indemnify the Department for the negligent acts or omissions of the Department, its officers, agents, or employees, or for the acts of third parties. Nothing herein shall be construed as consent by Recipient to be sued by third parties in any manner arising out of this Agreement. This indemnification shall survive the termination of this Agreement.
- c. Recipient agrees to include the following indemnification in all contracts with contractors, subcontractors, consultants, or subconsultants (each referred to as "Entity" for the purposes of the below indemnification) who perform work in connection with this Agreement:

"To the extent provided by law, [ENTITY] shall indemnify, defend, and hold harmless the [RECIPIENT] and the State of Florida, Department of Transportation, including the Department's officers, agents, and employees, against any actions, claims, or damages arising out of, relating to, or resulting from negligent or wrongful act(s) of [ENTITY], or any of its officers, agents, or employees, acting within the scope of their office or employment, in connection with the rights granted to or exercised by [ENTITY].

The foregoing indemnification shall not constitute a waiver of the Department's or [RECIPIENT']'s sovereign immunity beyond the limits set forth in Florida Statutes, Section 768.28. Nor shall the same be construed to constitute agreement by [ENTITY] to indemnify [RECIPIENT] for the negligent acts or omissions of [RECIPIENT], its officers, agents, or employees, or third parties. Nor shall the same be construed to constitute agreement by [ENTITY] to indemnify the Department for the negligent acts or omissions of the Department, its officers, agents, or employees, or third parties. This indemnification shall survive the termination of this Agreement."

- d. The Recipient shall, or cause its contractor or consultant to carry and keep in force, during the term of this Agreement, a general liability insurance policy or policies with a company or companies authorized to do business in Florida, affording public liability insurance with combined bodily injury limits of at least \$200,000 per person and \$300,000 each occurrence, and property damage insurance of at least \$200,000 each occurrence, for the services to be rendered in accordance with this Agreement. The Recipient shall also, or cause its contractor or consultant to carry and keep in force Workers' Compensation Insurance as required by the State of Florida under the Workers' Compensation Law. With respect to any general liability insurance policy required pursuant to this Agreement, all such policies shall be issued by companies licensed to do business in the State of Florida. The Recipient shall provide to the Department certificates showing the required coverage to be in effect with endorsements showing the Department to be an additional insured prior to commencing any work under this Agreement. Policies that include Self Insured Retention will not be accepted. The certificates and policies shall provide that in the event of any material change in or cancellation of the policies reflecting the required coverage, thirty days advance notice shall be given to the Department or as provided in accordance with Florida law.
- 16. Maintenance Obligations: In the event the Project includes construction then the following provisions are incorporated into this Agreement:
 - a. The Recipient agrees to maintain any portion of the Project not located on the State Highway System constructed under this Agreement for its useful life. If the Recipient constructs any improvement on Department right-of-way, the Recipient

X shall

shall not

maintain the improvements located on the Department right-of-way for their useful life. If the Recipient is required to maintain Project improvements located on the Department right-of-way beyond final acceptance, then Recipient shall, prior to any disbursement of the state funding provided under this

525-010-40 PROGRAM MANAGEMENT OGC/OOC- 05/21 Page 12 of 15

Agreement, also execute a Maintenance Memorandum of Agreement in a form that is acceptable to the Department. The Recipient has agreed to the foregoing by resolution, and such resolution is attached and incorporated into this Agreement as **Exhibit "D**". This provision will survive termination of this Agreement.

17. Miscellaneous Provisions:

- a. The Recipient will be solely responsible for compliance with all applicable environmental regulations, for any liability arising from non-compliance with these regulations, and will reimburse the Department for any loss incurred in connection therewith. The Recipient will be responsible for securing any applicable permits. The Recipient shall include in all contracts and subcontracts for amounts in excess of \$150,000, a provision requiring compliance with all applicable standards, orders or regulations issued pursuant to the Clean Air Act (42 U.S.C. 7401-7671q) and the Federal Water Pollution Control Act as amended (33 U.S.C. 1251-1387).
- b. The Department shall not be obligated or liable hereunder to any individual or entity not a party to this Agreement.
- c. In no event shall the making by the Department of any payment to the Recipient constitute or be construed as a waiver by the Department of any breach of covenant or any default which may then exist on the part of the Recipient and the making of such payment by the Department, while any such breach or default shall exist, shall in no way impair or prejudice any right or remedy available to the Department with respect to such breach or default.
- d. If any provision of this Agreement is held invalid, the remainder of this Agreement shall not be affected. In such an instance, the remainder would then continue to conform to the terms and requirements of applicable law.
- e. By execution of the Agreement, the Recipient represents that it has not paid and, also agrees not to pay, any bonus or commission for the purpose of obtaining an approval of its application for the financing hereunder.
- f. Nothing in the Agreement shall require the Recipient to observe or enforce compliance with any provision or perform any act or do any other thing in contravention of any applicable state law. If any of the provisions of the Agreement violate any applicable state law, the Recipient will at once notify the Department in writing in order that appropriate changes and modifications may be made by the Department and the Recipient to the end that the Recipient may proceed as soon as possible with the Project.
- g. In the event that this Agreement involves constructing and equipping of facilities, the Recipient shall submit to the Department for approval all appropriate plans and specifications covering the Project. The Department will review all plans and specifications and will issue to the Recipient a written approval with any approved portions of the Project and comments or recommendations covering any remainder of the Project deemed appropriate. After resolution of these comments and recommendations to the Department's satisfaction, the Department will issue to the Recipient a written approval with said remainder of the Project. Failure to obtain this written approval shall be sufficient cause of nonpayment by the Department.
- h. Upon completion of right-of-way activities on the Project, the Recipient must certify compliance with all applicable federal and state requirements. Certification is required prior to authorization for advertisement for or solicitation of bids for construction of the Project, including if no right-of-way is required.
- i. The Recipient will certify in writing, prior to Project closeout that the Project was completed in accordance with applicable plans and specifications, is in place on the Recipient's facility, adequate title is in the Recipient's name, and the Project is accepted by the Recipient as suitable for the intended purpose.
- j. The Recipient agrees that no federally-appropriated funds have been paid, or will be paid by or on behalf of the Recipient, to any person for influencing or attempting to influence any officer or employee of any federal agency, a Member of Congress, an officer or employee of Congress or an employee of a Member of Congress in connection with the awarding of any federal contract, the making of any federal grant, the

525-010-40 PROGRAM MANAGEMENT OGC/OOC- 05/21 Page 13 of 15

making of any federal loan, the entering into of any cooperative agreement, and the extension, continuation, renewal, amendment or modification of any federal contract, grant, loan or cooperative agreement. If any funds other than federally-appropriated funds have been paid by the Recipient to any person for influencing or attempting to influence an officer or employee of any federal agency, a Member of Congress, an officer or employee of a Member of Congress in connection with this Agreement, the undersigned shall complete and submit Standard Form-LLL, "Disclosure Form to Report Lobbying," in accordance with its instructions. The Recipient shall require that the language of this paragraph be included in the award documents for all subawards at all tiers (including subcontracts, subgrants, and contracts under grants, loans, and cooperative agreements) and that all subrecipients shall certify and disclose accordingly. No funds received pursuant to this contract may be expended for lobbying the Legislature, the judicial branch or a state agency.

- k. The Recipient may not permit the Engineer of Record to perform Construction, Engineering and Inspection services on the Project.
- I. The Recipient shall comply with all applicable federal guidelines, procedures, and regulations. If at any time a review conducted by Department and or FHWA reveals that the applicable federal guidelines, procedures, and regulations were not followed by the Recipient and FHWA requires reimbursement of the funds, the Recipient will be responsible for repayment to the Department of all funds awarded under the terms of this Agreement.
- m. The Recipient shall:
 - utilize the U.S. Department of Homeland Security's E-Verify system to verify the employment eligibility of all new employees hired by Recipient during the term of the contract; and
 - ii. expressly require any contractor and subcontractors performing work or providing services pursuant to the state contract to likewise utilize the U.S. Department of Homeland Security's E-Verify system to verify the employment eligibility of all new employees hired by the subcontractor during the contract term.
- n. This Agreement may be executed in one or more counterparts, each of which shall be deemed an original, but all of which shall constitute the same Agreement. A facsimile or electronic transmission of this Agreement with a signature on behalf of a party will be legal and binding on such party.
- o. The Parties agree to comply with s.20.055(5), Florida Statutes, and to incorporate in all subcontracts the obligation to comply with s.20.055(5), Florida Statutes.
- p. If the Project is procured pursuant to Chapter 255 for construction services and at the time of the competitive solicitation for the Project 50 percent or more of the cost of the Project is to be paid from state-appropriated funds, then the Recipient must comply with the requirements of Section 255.0991, Florida Statutes.

18. Exhibits:

- a. Exhibits "A", "B", "C", "D", "E" and "F" are attached to and incorporated into this Agreement.
- b. If this Project includes Phase 58 (construction) activities, then Exhibit "G", FHWA FORM 1273, is attached and incorporated into this Agreement.
- c. Alternative Advance Payment Financial Provisions are used on this Project. If an Alternative Pay Method is used on this Project, then Exhibit "H", Alternative Advance Payment Financial Provisions, is attached and incorporated into this Agreement.
- d. State funds are used on this Project. If state funds are used on this Project, then Exhibit "I", State Funds Addendum, is attached and incorporated into this Agreement. Exhibit "J", State Financial Assistance (Florida Single Audit Act), is attached and incorporated into this Agreement.
- e. This Project utilizes Advance Project Reimbursement. If this Project utilizes Advance Project Reimbursement, then Exhibit "K", Advance Project Reimbursement is attached and incorporated into this Agreement.

525-010-40 PROGRAM MANAGEMENT OGC/OOC- 05/21 Page 14 of 15

- f. This Project includes funding for landscaping. If this Project includes funding for landscaping, then Exhibit "L", Landscape Maintenance, is attached and incorporated into this Agreement.
- g. This Project includes funding for a roadway lighting system. If the Project includes funding for roadway lighting system, Exhibit "M", Roadway Lighting Maintenance is attached and incorporated into this Agreement.
- h. This Project includes funding for traffic signals and/or traffic signal systems. If this Project includes funding for traffic signals and/or traffic signals systems, Exhibit "N", Traffic Signal Maintenance is attached and incorporated into this Agreement.
- i. A portion or all of the Project will utilize Department right-of-way and, therefore, Exhibit "O", Terms and Conditions of Construction in Department Right-of-Way, is attached and incorporated into this Agreement.
- j.
 The following Exhibit(s) are attached and incorporated into this Agreement: ____

k. Exhibit and Attachment List

Exhibit A: Project Description and Responsibilities Exhibit B: Schedule of Financial Assistance Exhibit C: Title VI Assurances Exhibit D: Recipient Resolution Exhibit E: Federal Financial Assistance (Single Audit Act) Exhibit F: Contract Payment Requirements * Exhibit G: FHWA Form 1273 * Exhibit H: Alternative Advance Payment Financial Provisions

- * Exhibit I: State Funds Addendum
- * Exhibit J: State Financial Assistance (Florida Single Audit Act)
- * Exhibit K: Advance Project Reimbursement
- * Exhibit L: Landscape Maintenance
- * Exhibit M: Roadway Lighting Maintenance
- * Exhibit N: Traffic Signal Maintenance
- * Exhibit O: Terms and Conditions of Construction in Department Right-of-Way

* Additional Exhibit(s):

* Indicates that the Exhibit is only attached and incorporated if applicable box is selected.

DocuSign Envelope ID: 5D3013DB-2C27-40B2-8CCD-57B583C9BC42

Contract No. CM3119

STATE OF FLORIDA DEPARTMENT OF TRANSPORTATION

By:

525-010-40 PROGRAM MANAGEMENT OGC/OOC- 05/21 Page 15 of 15

IN WITNESS WHEREOF, the parties have executed this Agreement on the day and year written above.

RECIPIENT Nassau County By: Name: Aaron C. Bell Title: Chairman

STATE OF FLORIDA, DEPARTMENT OF TRANSPORTATION

Greg Evans

Name?30125220ans Title: District 2 Secretary

Legal Review:

DocuSigned by: Angela Hensel 0D4237CB170D4C2

Alt Form 525-010-40A

STATE OF FLORIDA DEPARTMENT OF TRANSPORTATION

Contract No. CM3119

525-011-0A PROGRAM MANAGEMENT 05/21

EXHIBIT A

PROJECT DESCRIPTION AND RESPONSIBILITIES

FPN: 441241-1-58-01 & 68-02

This exhibit forms an integral part of the Agreement between the State of Florida, Department of Transportation and

Nassau County (the Recipient)

PROJECT LOCATION:

The project is on the National Highway System.

The project is on the State Highway System.

PROJECT LENGTH AND MILE POST LIMITS: .526 miles; MP 1.344 to MP 1.877

PROJECT DESCRIPTION: CEI oversight and construction of sidewalk on Citrona Drive from Hickory Street to Beech Street.

SPECIAL CONSIDERATIONS BY RECIPIENT:

The Recipient is required to provide a copy of the design plans for the Department's review and approval to coordinate permitting with the Department, and notify the Department prior to commencement of any right-of-way activities.

The Recipient shall commence the project's activities subsequent to the execution of this Agreement and shall perform in accordance with the following schedule:

- a) Study to be completed by N/A .
- b) Design to be completed by N/A .
- c) Right-of-Way requirements identified and provided to the Department by N/A
- d) Right-of-Way to be certified by N/A
- e) Construction contract to be let by 6/30/2022
- f) Construction to be completed by 6/30/2023

If this schedule cannot be met, the Recipient will notify the Department in writing with a revised schedule or the project is subject to the withdrawal of funding.

SPECIAL CONSIDERATIONS BY DEPARTMENT:

Nassau County will be managing a Construction, Engieering and Inspection (CEI) contract that has been procured by the Department.

DocuSign Envelope ID: 5D3013DB-2C27-40B2-8CCD-57B583C9BC42

Alt Form 525-010-40B

Contract No. CM3119

STATE OF FLORIDA DEPARTMENT OF TRANSPORTATION LOCAL AGENCY PROGRAM AGREEMENT

525-011-0B PROGRAM 8/21 Page 1 of 1

EXHIBIT B SCHEDULE OF FINANCIAL ASSISTANCE

RECIPIENT NAME & BILLING ADDRESS:	FINANCIAL PROJECT NUMBER:
Nassau County	441241-1-58-01 & 68-02
96161 Nassau Place	
Yulee, Florida 32097	

	MAXIMUM PARTICIPATION				
PHASE OF WORK By Fiscal Year	(1) TOTAL PROJECT FUNDS	(2) LOCAL FUNDS	(3) STATE FUNDS	(4) FEDERAL FUNDS	
Design- Phase 38 FY: (Insert Program Name) FY: (Insert Program Name) FY: (Insert Program Name) Total Design Cost	\$ \$ \$ 0.00	\$ \$ \$ \$	\$ \$ \$ 0.00	\$ \$ \$ 0.00	
Right-of-Way- Phase 48 FY: (Insert Program Name) FY: (Insert Program Name) FY: (Insert Program Name) Total Right-of-Way Cost	\$ \$ \$	\$ \$ \$_0.00	\$ \$ \$	\$ \$ \$ \$	
Construction- Phase 58 FY: 2021-2022 (<u>TALT</u>) FY: (Insert Program Name) FY: (Insert Program Name)	\$ <u>197,098.00</u> \$ \$	\$ \$	\$ \$ \$	\$ <u>197,098.00</u> \$ \$	
Total Construction Cost Construction Engineering and Inspection (CEI)- Phase 68 FY: 2021-2022 (TALT) FY: (Insert Program Name) FY: (Insert Program Name) FY: (Insert Program Name) Total CEI Cost	\$ 197,098.00 \$ <u>1,500.00</u> \$ \$ \$ 1,500.00	\$ 0.00 \$ \$ \$ 0.00	\$ 0.00 \$ \$ \$ 0.00	\$ 197,098.00 \$ <u>1,500.00</u> \$ \$ \$ 1,500.00	
Insert Phase) FY: (Insert Program Name) FY: (Insert Program Name) FY: (Insert Program Name)	\$ 	\$ \$	\$ \$	\$	
Total Phase Costs	\$ 0.00 \$ 198,598.00	\$ 0.00 \$ 0.00	\$ 0.00 \$ 0.00	\$ 0.00 \$ 198,598.00	

COST ANALYSIS CERTIFICATION AS REQUIRED BY SECTION 216.3475, FLORIDA STATUTES: I certify that the cost for each line item budget category has been evaluated and determined to be allowable, reasonable, and necessary as required by Section 216.3475, F.S. Documentation is on file evidencing the methodology used and the conclusions reached.

Kimberly Evans

Amy Roberson District Grant Manager Name

DocuSigned by: kimberly Evans

ER12EE473684449 Signature

Date

Alt Form 525-010-40C

STATE OF FLORIDA DEPARTMENT OF TRANSPORTATION

Contract No. CM3119

525-011-0C PROGRAM MANAGEMENT 05/21 Page 1 of 2

EXHIBIT C

TITLE VI ASSURANCES

During the performance of this contract, the consultant or contractor, for itself, its assignees and successors in interest (hereinafter collectively referred to as the "contractor") agrees as follows:

- (1.) Compliance with REGULATIONS: The contractor shall comply with the Regulations relative to nondiscrimination in federally-assisted programs of the U.S. Department of Transportation (hereinafter, "USDOT") *Title 49, Code of Federal Regulations, Part 21,* as they may be amended from time to time, (hereinafter referred to as the *REGULATIONS*), which are herein incorporated by reference and made a part of this contract.
- (2.) Nondiscrimination: The Contractor, with regard to the work performed by it during the contract, shall not discriminate on the basis of race, color, national origin, or sex in the selection and retention of sub-contractors, including procurements of materials and leases of equipment. The contractor shall not participate either directly or indirectly in the discrimination prohibited by Section 21.5 of the REGULATIONS, including employment practices when the contract covers a program set forth in Appendix B of the REGULATIONS.
- (3.) Solicitations for Sub-contractors, including Procurements of Materials and Equipment: In all solicitations either by competitive bidding or negotiation made by the contractor for work to be performed under sub-contract, including procurements of materials or leases of equipment, each potential sub-contractor or supplier shall be notified by the contractor of the contractor's obligations under this contract and the *REGULATIONS* relative to nondiscrimination on the basis of race, color, national origin, or sex.
- (4.) Information and Reports: The contractor shall provide all information and reports required by the REGULATIONS or directives issued pursuant thereto, and shall permit access to its books, records, accounts, other sources of information, and its facilities as may be determined by the Florida Department of Transportation or the Federal Highway Administration, Federal Transit Administration, Federal Aviation Administration, and Federal Motor Carrier Safety Administration to be pertinent to ascertain compliance with such REGULATIONS, orders and instructions. Where any information required of a contractor is in the exclusive possession of another who fails or refuses to furnish this information the contractor shall so certify to the Florida Department of Transportation, or the Federal Highway Administration, Federal Transit Administration, Federal Aviation Administration, the contractor shall so certify to the Florida Department of Transportation, or the Federal Highway Administration, Federal Transit Administration, Federal Aviation Administration, or Federal Highway Administration, Federal Transit Administration, and Shall set forth what efforts it has made to obtain the information.
- (5.) Sanctions for Noncompliance: In the event of the contractor's noncompliance with the nondiscrimination provisions of this contract, the Florida Department of Transportation shall impose such contract sanctions as it or the Federal Highway Administration, Federal Transit Administration, Federal Aviation Administration, or

Federal Motor Carrier Safety Administration may determine to be appropriate, including, but not limited to:

- a. withholding of payments to the contractor under the contract until the contractor complies, and/or
- b. cancellation, termination or suspension of the contract, in whole or in part.
- (6.) Incorporation of Provisions: The contractor shall include the provisions of paragraphs (1) through (7) in every sub-contract, including procurements of materials and leases of equipment, unless exempt by the REGULATIONS, or directives issued pursuant thereto. The contractor shall take such action with respect to any sub-contract or procurement as the Florida Department of Transportation or the Federal Highway Administration, Federal Transit Administration, Federal Aviation Administration, or Federal Motor Carrier Safety Administration may direct as a means of enforcing such provisions including sanctions for noncompliance, provided, however, that, in the event a contractor becomes involved in, or is threatened with, litigation with a sub-contractor or supplier as a result of such direction, the contractor may request the Florida Department of Transportation, and, in addition, the contractor may request the United States to enter into such litigation to protect the interests of the United States.
- (7.)Compliance with Nondiscrimination Statutes and Authorities: Title VI of the Civil Rights Act of 1964 (42 U.S.C. § 2000d et seg., 78 stat. 252), (prohibits discrimination on the basis of race, color, national origin); and 49 CFR Part 21; The Uniform Relocation Assistance and Real Property Acquisition Policies Act of 1970, (42 U.S.C. § 4601), (prohibits unfair treatment of persons displaced or whose property has been acquired because of Federal or Federal-aid programs and projects); Federal-Aid Highway Act of 1973, (23 U.S.C. § 324 et seq.), (prohibits discrimination on the basis of sex); Section 504 of the Rehabilitation Act of 1973, (29 U.S.C. § 794 et seq.), as amended, (prohibits discrimination on the basis of disability); and 49 CFR Part 27; The Age Discrimination Act of 1975, as amended, (42 U.S.C. § 6101 et seq.), (prohibits discrimination on the basis of age); Airport and Airway Improvement Act of 1982, (49 USC § 471, Section 47123), as amended, (prohibits discrimination based on race, creed, color, national origin, or sex); The Civil Rights Restoration Act of 1987, (PL 100-209), (Broadened the scope, coverage and applicability of Title VI of the Civil Rights Act of 1964, The Age Discrimination Act of 1975 and Section 504 of the Rehabilitation Act of 1973, by expanding the definition of the terms "programs or activities" to include all of the programs or activities of the Federal-aid recipients, sub-recipients and contractors, whether such programs or activities are Federally funded or not); Titles II and III of the Americans with Disabilities Act, which prohibit discrimination on the basis of disability in the operation of public entities, public and private transportation systems, places of public accommodation, and certain testing entities (42 U.S.C. §§ 12131 -- 12189) as implemented by Department of Transportation regulations at 49 C.F.R. parts 37 and 38; The Federal Aviation Administration's Non-discrimination statute (49 U.S.C. § 47123) (prohibits discrimination on the basis of race, color, national origin, and sex); Executive Order 12898, Federal Actions to Address Environmental Justice in Minority Populations and Low-Income Populations, which ensures nondiscrimination against minority populations by discouraging programs, policies, and activities with disproportionately high and adverse human health or environmental effects on minority and low-income populations; Executive Order 13166, Improving Access to Services for Persons with Limited English Proficiency, and resulting agency guidance, national origin discrimination includes discrimination because of limited English proficiency (LEP). To ensure compliance with Title VI, you must take reasonable steps to ensure that LEP persons have meaningful access to your programs (70 Fed. Reg. at 74087 to 74100); Title IX of the Education Amendments of 1972, as amended, which prohibits you from discriminating because of sex in education programs or activities (20 U.S.C. 1681 et seq).

Contract No. CM3119

Alt Form 525-010-40D

STATE OF FLORIDA DEPARTMENT OF TRANSPORTATION

525-011-0D PROGRAM MANAGEMENT 05/21 Page 1 of 1

EXHIBIT D

RECIPIENT RESOLUTION

The Recipient's Resolution authorizing entry into this Agreement is attached and incorporated into this Agreement.
RESOLUTION NO. 2022- 04

A RESOLUTION AUTHORIZING THE EXECUTION OF THE AGREEMENT BETWEEN THE STATE OF FLORIDA DEPARTMENT OF TRANSPORTATION AND THE BOARD OF COUNTY COMMISSIONERS OF NASSAU COUNTY, FLORIDA

WHEREAS, the County Engineer has recommended that the Board of County Commissioners of Nassau County, Florida, execute a Local Agency Program Agreement between the State of Florida Department of Transportation and Nassau County, Florida regarding the construction and Construction Engineering Inspection of sidewalk improvements on Citrona Drive from Hickory Street to Beech Street (Financial Project ID No. 441241-1-58-01);

NOW, THEREFORE, BE IT RESOLVED, this <u>10th</u> day of <u>January</u>, 2022, by the Board of County Commissioners of Nassau County, Florida as follows:

1. The Local Agency Program Agreement between the State of Florida Department of Transportation and Nassau County is hereby approved, and the Chairman is authorized to execute said agreement.





Resolution No. 2022-04

BOARD OF COUNTY COMMISSIONERS NASSAU COUNTY, FLORIDA

Aaron C. Bell, Chairman

Attest as to Chairman's

Signature: John A. Crawford

Approved as to form by the Nassau County Attorney:

Michael S. Mullin

Alt Form 525-010-40E

STATE OF FLORIDA DEPARTMENT OF TRANSPORTATION LOCAL AGENCY PROGRAM AGREEMENT

525-011-0E PROGRAM MANAGEMENT 05/21 Page 1 of 1

Contract No. CM3119

EXHIBIT E

FEDERAL FINANCIAL ASSISTANCE (SINGLE AUDIT ACT)

FEDERAL RESOURCES AWARDED PURSUANT TO THIS AGREEMENT ARE AS FOLLOWS:

CFDA No.:	20.205
CFDA Title:	Highway Planning and Construction
Sector Sectors	Federal-Aid Highway Program, Federal Lands Highway Program
CFDA Program	https://beta.sam.gov/fal/1093726316c3409a8e50f4c75f5ef2c6/view?keywords=20.205&sort=-
Site:	relevance&index=cfda&is_active=true&page=1
Award Amount:	\$198,598.00
Awarding Agency:	Florida Department of Transportation
Award is for R&D:	Νο
Indirect Cost Rate:	N/A

FEDERAL RESOURCES AWARDED PURSUANT TO THIS AGREEMENT ARE SUBJECT TO THE FOLLOWING:

2 CFR Part 200 – Uniform Administrative Requirements, Cost Principles & Audit Requirements for Federal Awards

http://www.ecfr.gov/cgi-bin/text-idx?node=2:1.1.2.2.1

FEDERAL RESOURCES AWARDED PURSUANT TO THIS AGREEMENT MAY ALSO BE SUBJECT TO THE FOLLOWING:

Title 23 – Highways, United States Code http://uscode.house.gov/browse/prelim@title23&edition=prelim

Title 49 – Transportation, United States Code http://uscode.house.gov/browse/prelim@title49&edition=prelim

Map-21 – Moving Ahead for Progress in the 21st Century, Public Law 112-141 http://www.gpo.gov/fdsys/pkg/PLAW-112publ141/pdf/PLAW-112publ141.pdf

Federal Highway Administration – Florida Division http://www.fhwa.dot.gov/fldiv/

Federal Funding Accountability and Transparency Act (FFATA) Sub-award Reporting System (FSRS) https://www.fsrs.gov/ Alt Form 525-010-40F

STATE OF FLORIDA DEPARTMENT OF TRANSPORTATION

525-011-0F PROGRAM MANAGEMENT 05/21 Page 1 of 2

EXHIBIT F

CONTRACT PAYMENT REQUIREMENTS Florida Department of Financial Services, Reference Guide for State Expenditures Cost Reimbursement Contracts

Invoices for cost reimbursement contracts must be supported by an itemized listing of expenditures by category (salary, travel, expenses, etc.). Supporting documentation shall be submitted for each amount for which reimbursement is being claimed indicating that the item has been paid. Documentation for each amount for which reimbursement is being claimed must indicate that the item has been paid. Check numbers may be provided in lieu of copies of actual checks. Each piece of documentation should clearly reflect the dates of service. Only expenditures for categories in the approved agreement budget may be reimbursed. These expenditures must be allowable (pursuant to law) and directly related to the services being provided.

Listed below are types and examples of supporting documentation for cost reimbursement agreements:

Salaries: Timesheets that support the hours worked on the project or activity must be kept. A payroll register, or similar documentation should be maintained. The payroll register should show gross salary charges, fringe benefits, other deductions and net pay. If an individual for whom reimbursement is being claimed is paid by the hour, a document reflecting the hours worked times the rate of pay will be acceptable.

Fringe benefits: Fringe benefits should be supported by invoices showing the amount paid on behalf of the employee, e.g., insurance premiums paid. If the contract specifically states that fringe benefits will be based on a specified percentage rather than the actual cost of fringe benefits, then the calculation for the fringe benefits amount must be shown. Exception: Governmental entities are not required to provide check numbers or copies of checks for fringe benefits.

Travel: Reimbursement for travel must be in accordance with s. 112.061, F.S., which includes submission of the claim on the approved state travel voucher along with supporting receipts and invoices.

Other direct costs: Reimbursement will be made based on paid invoices/receipts and proof of payment processing (cancelled/processed checks and bank statements). If nonexpendable property is purchased using state funds, the contract should include a provision for the transfer of the property to the State when services are terminated. Documentation must be provided to show compliance with DMS Rule 60A-1.017, F.A.C., regarding the requirements for contracts which include services and that provide for the contractor to purchase tangible personal property as defined in s. 273.02, F.S., for subsequent transfer to the State.

Indirect costs: If the contract stipulates that indirect costs will be paid based on a specified rate, then the calculation should be shown. Indirect costs must be in the approved agreement budget and the entity must be able to demonstrate that the costs are not duplicated elsewhere as direct costs. All indirect cost rates must be evaluated for reasonableness and for allowability and must be allocated consistently.

Contracts between state agencies may submit alternative documentation to substantiate the reimbursement request, which may be in the form of FLAIR reports or other detailed reports.

The Florida Department of Financial Services, online Reference Guide for State Expenditures can be found at this web address <u>https://www.myfloridacfo.com/Division/AA/Manuals/documents/ReferenceGuideforState</u> Expenditures.pdf.

Alt Form 525-010-40G

STATE OF FLORIDA DEPARTMENT OF TRANSPORTATION

Contract No. CM3119

525-011-0G PROGRAM MANAGEMENT 05/21 Page 1 of 1

EXHIBIT G

FHWA FORM 1273 FEDERAL RESOURCES AWARDED PURSUANT TO THIS AGREEMENT ARE AS FOLLOWS:

LEGAL REQUIREMENTS AND RESPONSIBILITY TO THE PUBLIC – COMPLIANCE WITH FHWA 1273.

The FHWA-1273 version dated May 1, 2012 is appended in its entirety to this Exhibit. FHWA-1273 may also be referenced on the Department's website at the following URL address: http://www.fhwa.dot.gov/programadmin/contracts/1273/1273.pdf

Sub-recipients of federal grants awards for Federal-Aid Highway construction shall take responsibility to obtain this information and comply with all provisions contained in FHWA-1273.

RESOLUTION NO. 2022-04

A RESOLUTION AUTHORIZING THE EXECUTION OF THE AGREEMENT BETWEEN THE STATE OF FLORIDA DEPARTMENT OF TRANSPORTATION AND THE BOARD OF COUNTY COMMISSIONERS OF NASSAU COUNTY, FLORIDA

WHEREAS, the County Engineer has recommended that the Board of County Commissioners of Nassau County, Florida, execute a Local Agency Program Agreement between the State of Florida Department of Transportation and Nassau County, Florida regarding the construction and Construction Engineering Inspection of sidewalk improvements on Citrona Drive from Hickory Street to Beech Street (Financial Project ID No. 441241-1-58-01);

NOW, THEREFORE, BE IT RESOLVED, this <u>10th</u> day of <u>January</u>, 2022, by the Board of County Commissioners of Nassau County, Florida as follows:

 The Local Agency Program Agreement between the State of Florida Department of Transportation and Nassau County is hereby approved, and the Chairman is authorized to execute said agreement.





Resolution No. 2022-04

BOARD OF COUNTY COMMISSIONERS NASSAU COUNTY, FLORIDA

Aaron C. Bell, Chairman

Attest as to Chairman's Signature:

John A. Crawford

Approved as to form by the Nassau County Attorney:

Michael S. Mullin

To: kimberly.evans@dot.state.fl.us

FLORIDA DEPARTMENT OF TRANSPORTATION FUNDS APPROVAL

G2717

3/15/2022

CONTRACT INFORMATION

G2717
GD - GRANT DISBURSEMENT (GRANT)
G - GOVERMENTAL AGENCY (287.057, F.S.)
NASSAU COUNTY BOARD OF COUNTY C
F591863042053
03/14/2022
12/31/2023
ct = \$198,598.00
cst and cei project management of sidewalk on Citrona Drivefrom Hickory Street to Beech Street

FUNDS APPROVAL INFORMATION

FUNDS APPROVED/REVIEWED FOR ROBIN M. NAITOVE, CPA, COMPTROLLER ON 3/15/2022

Action:	Original	Original
Reviewed or Approved:	APPROVED	APPROVED
Organization Code:	55024010206	55024010206
Expansion Option:	АВ	A8
Object Code:	780000	780000
Amount:	\$197,098.00	\$1,500.00
Financial Project:	44124115801	44124116802
Work Activity (FCT):	215	215
CFDA:	20.205	20.205
Fiscal Year:	2022	2022
Budget Entity:	55150200	55150200
Category/Category Year:	088717/22	088718/22
Amendment ID:	O001	O001
Sequence:	00	01
User Assigned ID:		
Enc Line (6s)/Status:	0001/04	0002/04

Total Amount: \$198,598.00

Certificate Of Completion

Envelope Id: 2ADA506067E54967BBF1A99D20219F1B Subject: Please DocuSign: G2717 Citrona.pdf Contract Number (ex. C9A12, optional): G2717 Document Contains Confidential Information?: No Fin Proj Num (ex.123456-1-32-01, Optional): 441241-1-58-01 Office (contact Procurement if add is needed): Local Programs HR Action ?: No Source Envelope: Signatures: 2 Document Pages: 29 Initials: 0 Certificate Pages: 2 AutoNav: Enabled Envelopeld Stamping: Enabled Time Zone: (UTC-05:00) Eastern Time (US & Canada)

Record Tracking

Status: Original 3/28/2022 11:20:33 AM

Signer Events

Angela Hensel angela.hensel@dot.state.fl.us **District Legal Counsel** Florida Department of Transportation Security Level: Email, Account Authentication (None)

Electronic Record and Signature Disclosure: Not Offered via DocuSign

Greg Evans greg.evans@dot.state.fl.us **D2** Secretary FLDOT Security Level: Email, Account Authentication (None)

Electronic Record and Signature Disclosure: Not Offered via DocuSign

Holder: Kimberly Evans kimberly.evans@dot.state.fl.us

Signature

DocuSigned by: Angela Hensel 0D4237CB170D4C2

Signature Adoption: Pre-selected Style Using IP Address: 156.75.252.6

DocuSigand by: Greg Evans BRADSECTAAA

Signature Adoption: Pre-selected Style Using IP Address: 156.75.252.6

Sent: 3/28/2022 1:38:52 PM Viewed: 3/28/2022 1:48:12 PM Signed: 3/28/2022 1:48:27 PM

In Person Signer Events	Signature	Timestamp	
Editor Delivery Events	Status	Timestamp	
Agent Delivery Events	Status	Timestamp	
Intermediary Delivery Events	Status	Timestamp	
Certified Delivery Events	Status	Timestamp	
Carbon Copy Events	Status	Timestamp	
Witness Events	Signature	Timestamp	

DocuSign

Envelope Originator: **Kimberly Evans** 605 Suwannee Street MS 20 Tallahassee, FL 32399-0450 kimberly.evans@dot.state.fl.us IP Address: 156.75.252.6

Location: DocuSign

Timestamp

Sent: 3/28/2022 11:24:51 AM Viewed: 3/28/2022 1:34:57 PM Signed: 3/28/2022 1:38:51 PM

Status: Completed

Notary Events	Signature	Timestamp
Envelope Summary Events	Status	Timestamps
Envelope Sent	Hashed/Encrypted	3/28/2022 11:24:51 AM
Certified Delivered	Security Checked	3/28/2022 1:48:12 PM
Signing Complete	Security Checked	3/28/2022 1:48:27 PM
Completed	Security Checked	3/28/2022 1:48:27 PM
Payment Events	Status	Timestamps

REQUIRED CONTRACT PROVISIONS FEDERAL-AID CONSTRUCTION CONTRACTS

General

- II. Nondiscrimination
- III. Non-segregated Facilities
- IV. Davis-Bacon and Related Act Provisions
- V. Contract Work Hours and Safety Standards Act
- Provisions
- VI. Subletting or Assigning the Contract
- VII. Safety: Accident Prevention
- VIII. False Statements Concerning Highway Projects
- Implementation of Clean Air Act and Federal Water Pollution Control Act
- X. Certification Regarding Debarment, Suspension, Ineligibility and Voluntary Exclusion
- XI. Certification Regarding Use of Contract Funds for Lobbying
- XII. Use of United States-Flag Vessels:

ATTACHMENTS

A. Employment and Materials Preference for Appalachian Development Highway System or Appalachian Local Access Road Contracts (included in Appalachian contracts only)

I. GENERAL

1, Form FHWA-1273 must be physically incorporated in each construction contract funded under title 23, United States Code, as required in 23 CFR 633,102(b) (excluding emergency contracts solely intended for debris removal). The contractor (or subcontractor) must insert this form in each subcontract and further require its inclusion in all lower tier subcontracts (excluding purchase orders, rental agreements and other agreements for supplies or services). 23 CFR 633.102(e).

The applicable requirements of Form FHWA-1273 are incorporated by reference for work done under any purchase order, rental agreement or agreement for other services. The prime contractor shall be responsible for compliance by any subcontractor, lower-tier subcontractor or service provider. 23 CFR 633.102(e).

Form FHWA-1273 must be included in all Federal-aid designbuild contracts, in all subcontracts and in lower tier subcontracts (excluding subcontracts for design services, purchase orders, rental agreements and other agreements for supplies or services) in accordance with 23 CFR 633.102. The design-builder shall be responsible for compliance by any subcontractor, lower-tier subcontractor or service provider.

Contracting agencies may reference Form FHWA-1273 in solicitation-for-bids or request-for-proposals documents, however, the Form FHWA-1273 must be physically incorporated (not referenced) in all contracts, subcontracts and lower-tier subcontracts (excluding purchase orders, rental agreements and other agreements for supplies or services related to a construction contract). 23 CFR 633.102(b).

Subject to the applicability criteria noted in the following sections, these contract provisions shall apply to all work performed on the contract by the contractor's own organization and with the assistance of workers under the contractor's immediate superintendence and to all work performed on the contract by piecework, station work, or by subcontract. 23 CFR 633.102(d).

3. A breach of any of the stipulations contained in these Required Contract Provisions may be sufficient grounds for withholding of progress payments, withholding of final payment, termination of the contract, suspension / debarment or any other action determined to be appropriate by the contracting agency and FHWA.

4. Selection of Labor: During the performance of this contract, the contractor shall not use convict labor for any purpose within the limits of a construction project on a Federal-aid highway unless it is labor performed by convicts who are on parole, supervised release, or probation. 23 U.S.C. 114(b). The term Federal-aid highway does not include roadways functionally classified as local roads or rural minor collectors. 23 U.S.C. 101(a).

II. NONDISCRIMINATION (23 CFR 230.107(a); 23 CFR Part 230, Subpart A, Appendix A; EO 11246)

The provisions of this section related to 23 CFR Part 230, Subpart A, Appendix A are applicable to all Federal-aid construction contracts and to all related construction subcontracts of \$10,000 or more. The provisions of 23 CFR Part 230 are not applicable to material supply, engineering, or architectural service contracts.

In addition, the contractor and all subcontractors must comply with the following policies: Executive Order 11246, 41 CFR Part 60, 29 CFR Parts 1625-1627, 23 U.S.C. 140, Section 504 of the Rehabilitation Act of 1973, as amended (29 U.S.C. 794), Title VI of the Civil Rights Act of 1964, as amended (42 U.S.C. 2000d et seq.), and related regulations including 49 CFR Parts 21, 26, and 27; and 23 CFR Parts 200, 230, and 633.

The contractor and all subcontractors must comply with: the requirements of the Equal Opportunity Clause in 41 CFR 60-1.4(b) and, for all construction contracts exceeding \$10,000, the Standard Federal Equal Employment Opportunity Construction Contract Specifications in 41 CFR 60-4.3.

Note: The U.S. Department of Labor has exclusive authority to determine compliance with Executive Order 11246 and the policies of the Secretary of Labor including 41 CFR Part 60, and 29 CFR Parts 1625-1627. The contracting agency and the FHWA have the authority and the responsibility to ensure compliance with 23 U.S.C. 140, Section 504 of the Rehabilitation Act of 1973, as amended (29 U.S.C. 794), and Title VI of the Civil Rights Act of 1964, as amended (42 U.S.C. 2000d et seq.), and related regulations including 49 CFR Parts 21, 26, and 27; and 23 CFR Parts 200, 230, and 633.

The following provision is adopted from 23 CFR Part 230, Subpart A, Appendix A, with appropriate revisions to conform to the U.S. Department of Labor (US DOL) and FHWA requirements. 1. Equal Employment Opportunity: Equal Employment Opportunity (EEO) requirements not to discriminate and to take affirmative action to assure equal opportunity as set forth under laws, executive orders, rules, regulations (*see* 28 CFR Part 35, 29 CFR Part 1630, 29 CFR Parts 1625-1627, 41 CFR Part 60 and 49 CFR Part 27) and orders of the Secretary of Labor as modified by the provisions prescribed herein, and imposed pursuant to 23 U.S.C. 140, shall constitute the EEO and specific affirmative action standards for the contractor's project activities under this contract. The provisions of the Americans with Disabilities Act of 1990 (42 U.S.C. 12101 et seq.) set forth under 28 CFR Part 35 and 29 CFR Part 1630 are incorporated by reference in this contract. In the execution of this contract, the contractor agrees to comply with the following minimum specific requirement activities of EEO:

a. The contractor will work with the contracting agency and the Federal Government to ensure that it has made every good faith effort to provide equal opportunity with respect to all of its terms and conditions of employment and in their review of activities under the contract. 23 CFR 230.409 (g)(4) & (5).

b. The contractor will accept as its operating policy the following statement:

"It is the policy of this Company to assure that applicants are employed, and that employees are treated during employment, without regard to their race, religion, sex, sexual orientation, gender identity, color, national origin, age or disability. Such action shall include: employment, upgrading, demotion, or transfer; recruitment or recruitment advertising; layoff or termination; rates of pay or other forms of compensation; and selection for training, including apprenticeship, pre-apprenticeship, and/or on-the-job training."

2. EEO Officer: The contractor will designate and make known to the contracting officers an EEO Officer who will have the responsibility for and must be capable of effectively administering and promoting an active EEO program and who must be assigned adequate authority and responsibility to do so.

3. Dissemination of Policy: All members of the contractor's staff who are authorized to hire, supervise, promote, and discharge employees, or who recommend such action or are substantially involved in such action, will be made fully cognizant of and will implement the contractor's EEO policy and contractual responsibilities to provide EEO in each grade and classification of employment. To ensure that the above agreement will be met, the following actions will be taken as a minimum:

a. Periodic meetings of supervisory and personnel office employees will be conducted before the start of work and then not less often than once every six months, at which time the contractor's EEO policy and its implementation will be reviewed and explained. The meetings will be conducted by the EEO Officer or other knowledgeable company official.

b. All new supervisory or personnel office employees will be given a thorough indoctrination by the EEO Officer, covering all major aspects of the contractor's EEO obligations within thirty days following their reporting for duty with the contractor.

c. All personnel who are engaged in direct recruitment for the project will be instructed by the EEO Officer in the contractor's procedures for locating and hiring minorities and women. d. Notices and posters setting forth the contractor's EEO policy will be placed in areas readily accessible to employees, applicants for employment and potential employees.

e. The contractor's EEO policy and the procedures to implement such policy will be brought to the attention of employees by means of meetings, employee handbooks, or other appropriate means.

4. Recruitment: When advertising for employees, the contractor will include in all advertisements for employees the notation: "An Equal Opportunity Employer." All such advertisements will be placed in publications having a large circulation among minorities and women in the area from which the project work force would normally be derived.

a. The contractor will, unless precluded by a valid bargaining agreement, conduct systematic and direct recruitment through public and private employee referral sources likely to yield qualified minorities and women. To meet this requirement, the contractor will identify sources of potential minority group employees and establish with such identified sources procedures whereby minority and women applicants may be referred to the contractor for employment consideration.

b. In the event the contractor has a valid bargaining agreement providing for exclusive hiring hall referrals, the contractor is expected to observe the provisions of that agreement to the extent that the system meets the contractor's compliance with EEO contract provisions. Where implementation of such an agreement has the effect of discriminating against minorities or women, or obligates the contractor to do the same, such implementation violates Federal nondiscrimination provisions.

c. The contractor will encourage its present employees to refer minorities and women as applicants for employment. Information and procedures with regard to referring such applicants will be discussed with employees.

5. Personnel Actions: Wages, working conditions, and employee benefits shall be established and administered, and personnel actions of every type, including hiring, upgrading, promotion, transfer, demotion, layoff, and termination, shall be taken without regard to race, color, religion, sex, sexual orientation, gender identity, national origin, age or disability. The following procedures shall be followed:

 a. The contractor will conduct periodic inspections of project sites to ensure that working conditions and employee facilities do not indicate discriminatory treatment of project site personnel.

b. The contractor will periodically evaluate the spread of wages paid within each classification to determine any evidence of discriminatory wage practices.

c. The contractor will periodically review selected personnel actions in depth to determine whether there is evidence of discrimination. Where evidence is found, the contractor will promptly take corrective action. If the review indicates that the discrimination may extend beyond the actions reviewed, such corrective action shall include all affected persons.

d. The contractor will promptly investigate all complaints of alleged discrimination made to the contractor in connection with its obligations under this contract, will attempt to resolve such complaints, and will take appropriate corrective action within a reasonable time. If the investigation indicates that the discrimination may affect persons other than the complainant, such corrective action shall include such other persons. Upon completion of each investigation, the contractor will inform every complainant of all of their avenues of appeal.

6. Training and Promotion:

a. The contractor will assist in locating, qualifying, and increasing the skills of minorities and women who are applicants for employment or current employees. Such efforts should be aimed at developing full journey level status employees in the type of trade or job classification involved.

b. Consistent with the contractor's work force requirements and as permissible under Federal and State regulations, the contractor shall make full use of training programs (i.e., apprenticeship and on-the-job training programs for the geographical area of contract performance). In the event a special provision for training is provided under this contract, this subparagraph will be superseded as indicated in the special provision. The contracting agency may reserve training positions for persons who receive welfare assistance in accordance with 23 U.S.C. 140(a).

 c. The contractor will advise employees and applicants for employment of available training programs and entrance requirements for each.

d. The contractor will periodically review the training and promotion potential of employees who are minorities and women and will encourage eligible employees to apply for such training and promotion.

7. Unions: If the contractor relies in whole or in part upon unions as a source of employees, the contractor will use good faith efforts to obtain the cooperation of such unions to increase opportunities for minorities and women. 23 CFR 230.409. Actions by the contractor, either directly or through a contractor's association acting as agent, will include the procedures set forth below:

a. The contractor will use good faith efforts to develop, in cooperation with the unions, joint training programs almed toward qualifying more minorities and women for membership in the unions and increasing the skills of minorities and women so that they may qualify for higher paying employment.

b. The contractor will use good faith efforts to incorporate an EEO clause into each union agreement to the end that such union will be contractually bound to refer applicants without regard to their race, color, religion, sex, sexual orientation, gender identity, national origin, age, or disability.

c. The contractor is to obtain information as to the referral practices and policies of the labor union except that to the extent such information is within the exclusive possession of the labor union and such labor union refuses to furnish such information to the contractor, the contractor shall so certify to the contracting agency and shall set forth what efforts have been made to obtain such information.

d. In the event the union is unable to provide the contractor with a reasonable flow of referrals within the time limit set forth in the collective bargaining agreement, the contractor will, through independent recruitment efforts, fill the employment vacancies without regard to race, color, religion, sex, sexual orientation, gender identity, national origin, age, or disability; making full efforts to obtain qualified and/or qualifiable minorities and women. The failure of a union to provide sufficient referrals (even though it is obligated to provide exclusive referrals under the terms of a collective bargaining agreement) does not relieve the contractor from the requirements of this paragraph. In the event the union referral practice prevents the contractor from meeting the obligations pursuant to Executive Order 11246, as amended, and these special provisions, such contractor shall immediately notify the contracting agency.

8. Reasonable Accommodation for Applicants /

Employees with Disabilities: The contractor must be familiar with the requirements for and comply with the Americans with Disabilities Act and all rules and regulations established thereunder. Employers must provide reasonable accommodation in all employment activities unless to do so would cause an undue hardship.

9. Selection of Subcontractors, Procurement of Materials and Leasing of Equipment: The contractor shall not discriminate on the grounds of race, color, religion, sex, sexual orientation, gender identity, national origin, age, or disability in the selection and retention of subcontractors, including procurement of materials and leases of equipment. The contractor shall take all necessary and reasonable steps to ensure nondiscrimination in the administration of this contract.

 The contractor shall notify all potential subcontractors, suppliers, and lessors of their EEO obligations under this contract.

b. The contractor will use good faith efforts to ensure subcontractor compliance with their EEO obligations.

10. Assurances Required:

a. The requirements of 49 CFR Part 26 and the State DOT's FHWA-approved Disadvantaged Business Enterprise (DBE) program are incorporated by reference.

b. The contractor, subrecipient or subcontractor shall not discriminate on the basis of race, color, national origin, or sex in the performance of this contract. The contractor shall carry out applicable requirements of 49 CFR part 26 in the award and administration of DOT-assisted contracts. Fallure by the contractor to carry out these requirements is a material breach of this contract, which may result in the termination of this contract or such other remedy as the recipient deems appropriate, which may include, but is not limited to:

(1) Withholding monthly progress payments;

- (2) Assessing sanctions;
- (3) Liquidated damages; and/or

(4) Disqualifying the contractor from future bidding as nonresponsible.

c. The Title VI and nondiscrimination provisions of U.S. DOT Order 1050.2A at Appendixes A and E are incorporated by reference. 49 CFR Part 21.

11. Records and Reports: The contractor shall keep such records as necessary to document compliance with the EEO requirements. Such records shall be retained for a period of three years following the date of the final payment to the contractor for all contract work and shall be available at reasonable times and places for inspection by authorized representatives of the contracting agency and the FHWA.

a. The records kept by the contractor shall document the following:

 The number and work hours of minority and nonminority group members and women employed in each work classification on the project;

(2) The progress and efforts being made in cooperation with unions, when applicable, to increase employment opportunities for minorities and women; and

(3) The progress and efforts being made in locating, hiring, training, qualifying, and upgrading minorities and women.

b. The contractors and subcontractors will submit an annual report to the contracting agency each July for the duration of the project indicating the number of minority, women, and nonminority group employees currently engaged in each work classification required by the contract work. This information is to be reported on Form FHWA-1391. The staffing data should represent the project work force on board in all or any part of the last payroll period preceding the end of July. If on-the-job training is being required by special provision, the contractor will be required to collect and report training data. The employment data should reflect the work force on board during all or any part of the last payroll period preceding the end of July.

III. NONSEGREGATED FACILITIES

This provision is applicable to all Federal-aid construction contracts and to all related construction subcontracts of more than \$10,000. 41 CFR 60-1.5.

As prescribed by 41 CFR 60-1.8, the contractor must ensure that facilities provided for employees are provided in such a manner that segregation on the basis of race, color, religion, sex, sexual orientation, gender identity, or national origin cannot result. The contractor may neither require such segregated use by written or oral policies nor tolerate such use by employee custom. The contractor's obligation extends further to ensure that its employees are not assigned to perform their services at any location under the contractor's control where the facilities are segregated. The term "facilities" includes waiting rooms, work areas, restaurants and other eating areas, time clocks, restrooms, washrooms, locker rooms and other storage or dressing areas, parking lots, drinking fountains, recreation or entertainment areas, transportation, and housing provided for employees. The contractor shall provide separate or single-user restrooms and necessary dressing or sleeping areas to assure privacy between sexes.

IV. DAVIS-BACON AND RELATED ACT PROVISIONS

This section is applicable to all Federal-aid construction projects exceeding \$2,000 and to all related subcontracts and lower-tier subcontracts (regardless of subcontract size), in accordance with 29 CFR 5.5. The requirements apply to all projects located within the right-of-way of a roadway that is functionally classified as Federal-aid highway. 23 U.S.C. 113. This excludes roadways functionally classified as local roads or rural minor collectors, which are exempt. 23 U.S.C. 101. Where applicable law requires that projects be treated as a project on a Federal-aid highway, the provisions of this subpart will apply regardless of the location of the project. Examples include: Surface Transportation Block Grant Program projects funded under 23 U.S.C. 133 [excluding recreational trails projects], the Nationally Significant Freight and Highway Projects funded under 23 U.S.C. 117, and National Highway Freight Program projects funded under 23 U.S.C. 167.

The following provisions are from the U.S. Department of Labor regulations in 29 CFR 5.5 "Contract provisions and related matters" with minor revisions to conform to the FHWA-1273 format and FHWA program requirements.

1. Minimum wages (29 CFR 5.5)

a. All laborers and mechanics employed or working upon the site of the work, will be paid unconditionally and not less often than once a week, and without subsequent deduction or rebate on any account (except such payroll deductions as are permitted by regulations issued by the Secretary of Labor under the Copeland Act (29 CFR part 3)), the full amount of wages and bona fide fringe benefits (or cash equivalents thereof) due at time of payment computed at rates not less than those contained in the wage determination of the Secretary of Labor which is attached hereto and made a part hereof, regardless of any contractual relationship which may be alleged to exist between the contractor and such laborers and mechanics.

Contributions made or costs reasonably anticipated for bona fide fringe benefits under section 1(b)(2) of the Davis-Bacon Act on behalf of laborers or mechanics are considered wages paid to such laborers or mechanics, subject to the provisions of paragraph 1.d. of this section; also, regular contributions made or costs incurred for more than a weekly period (but not less often than quarterly) under plans, funds, or programs which cover the particular weekly period, are deemed to be constructively made or incurred during such weekly period. Such laborers and mechanics shall be paid the appropriate wage rate and fringe benefits on the wage determination for the classification of work actually performed, without regard to skill, except as provided in 29 CFR 5.5(a)(4). Laborers or mechanics performing work in more than one classification may be compensated at the rate specified for each classification for the time actually worked therein: Provided, That the employer's payroll records accurately set forth the time spent in each classification in which work is performed. The wage determination (including any additional classification and wage rates conformed under paragraph 1.b. of this section) and the Davis-Bacon poster (WH-1321) shall be posted at all times by the contractor and its subcontractors at the site of the work in a prominent and accessible place where it can be easily seen by the workers.

b.(1) The contracting officer shall require that any class of laborers or mechanics, including helpers, which is not listed in the wage determination and which is to be employed under the contract shall be classified in conformance with the wage determination. The contracting officer shall approve an additional classification and wage rate and fringe benefits therefore only when the following criteria have been met:

 (i) The work to be performed by the classification requested is not performed by a classification in the wage determination; and

 (ii) The classification is utilized in the area by the construction industry; and (iii) The proposed wage rate, including any bona fide fringe benefits, bears a reasonable relationship to the wage rates contained in the wage determination.

(2) If the contractor and the laborers and mechanics to be employed in the classification (if known), or their representatives, and the contracting officer agree on the classification and wage rate (including the amount designated for fringe benefits where appropriate), a report of the action taken shall be sent by the contracting officer to the Administrator of the Wage and Hour Division, U.S. Department of Labor, Washington, DC 20210. The Administrator, or an authorized representative, will approve, modify, or disapprove every additional classification action within 30 days of receipt and so advise the contracting officer or will notify the contracting officer within the 30-day period that additional time is necessary.

(3) In the event the contractor, the laborers or mechanics to be employed in the classification or their representatives, and the contracting officer do not agree on the proposed classification and wage rate (including the amount designated for fringe benefits, where appropriate), the contracting officer shall refer the questions, including the views of all interested parties and the recommendation of the contracting officer, to the Administrator for determination. The Administrator, or an authorized representative, will issue a determination within 30 days of receipt and so advise the contracting officer or will notify the contracting officer within the 30-day period that additional time is necessary.

(4) The wage rate (including fringe benefits where appropriate) determined pursuant to paragraphs 1.b.(2) or 1.b.(3) of this section, shall be paid to all workers performing work in the classification under this contract from the first day on which work is performed in the classification.

c. Whenever the minimum wage rate prescribed in the contract for a class of laborers or mechanics includes a fringe benefit which is not expressed as an hourly rate, the contractor shall either pay the benefit as stated in the wage determination or shall pay another bona fide fringe benefit or an hourly cash equivalent thereof.

d. If the contractor does not make payments to a trustee or other third person, the contractor may consider as part of the wages of any laborer or mechanic the amount of any costs reasonably anticipated in providing bona fide fringe benefits under a plan or program, Provided, That the Secretary of Labor has found, upon the written request of the contractor, that the applicable standards of the Davis-Bacon Act have been met. The Secretary of Labor may require the contractor to set aside in a separate account assets for the meeting of obligations under the plan or program.

2. Withholding (29 CFR 5.5)

The contracting agency shall upon its own action or upon written request of an authorized representative of the Department of Labor, withhold or cause to be withheld from the contractor under this contract, or any other Federal contract with the same prime contractor, or any other federallyassisted contract subject to Davis-Bacon prevailing wage requirements, which is held by the same prime contractor, so much of the accrued payments or advances as may be considered necessary to pay laborers and mechanics, including apprentices, trainees, and helpers, employed by the contractor or any subcontractor the full amount of wages required by the contract. In the event of failure to pay any laborer or mechanic, including any apprentice, trainee, or helper, employed or working on the site of the work, all or part of the wages required by the contract, the contracting agency may, after written notice to the contractor, take such action as may be necessary to cause the suspension of any further payment, advance, or guarantee of funds until such violations have ceased.

3. Payrolls and basic records (29 CFR 5.5)

a. Payrolls and basic records relating thereto shall be maintained by the contractor during the course of the work and preserved for a period of three years thereafter for all laborers and mechanics working at the site of the work. Such records shall contain the name, address, and social security number of each such worker, his or her correct classification, hourly rates of wages paid (including rates of contributions or costs anticipated for bona fide fringe benefits or cash equivalents thereof of the types described in section 1(b)(2)(B) of the Davis-Bacon Act), daily and weekly number of hours worked, deductions made and actual wages paid. Whenever the Secretary of Labor has found under 29 CFR 5.5(a)(1)(iv) that the wages of any laborer or mechanic include the amount of any costs reasonably anticipated in providing benefits under a plan or program described in section 1(b)(2)(B) of the Davis-Bacon Act, the contractor shall maintain records which show that the commitment to provide such benefits is enforceable, that the plan or program is financially responsible, and that the plan or program has been communicated in writing to the laborers or mechanics affected, and records which show the costs anticipated or the actual cost incurred in providing such benefits. Contractors employing apprentices or trainees under approved programs shall maintain written evidence of the registration of apprenticeship programs and certification of trainee programs, the registration of the apprentices and trainees, and the ratios and wage rates prescribed in the applicable programs.

b.(1) The contractor shall submit weekly for each week in which any contract work is performed a copy of all payrolls to the contracting agency. The payrolls submitted shall set out accurately and completely all of the information required to be maintained under 29 CFR 5.5(a)(3)(i), except that full social security numbers and home addresses shall not be included on weekly transmittals. Instead the payrolls shall only need to include an individually identifying number for each employee (e.g., the last four digits of the employee's social security number). The required weekly payroll information may be submitted in any form desired. Optional Form WH-347 is available for this purpose from the Wage and Hour Division Web site. The prime contractor is responsible for the submission of copies of payrolls by all subcontractors. Contractors and subcontractors shall maintain the full social security number and current address of each covered worker, and shall provide them upon request to the contracting agency for transmission to the State DOT, the FHWA or the Wage and Hour Division of the Department of Labor for purposes of an investigation or audit of compliance with prevailing wage requirements. It is not a violation of this section for a prime contractor to require a subcontractor to provide addresses and social security numbers to the prime contractor for its own records, without weekly submission to the contracting agency.

(2) Each payroll submitted shall be accompanied by a "Statement of Compliance," signed by the contractor or subcontractor or his or her agent who pays or supervises the payment of the persons employed under the contract and shall certify the following:

(i) That the payroll for the payroll period contains the information required to be provided under 29 CFR 5.5(a)(3)(ii), the appropriate information is being maintained under 29 CFR 5.5(a)(3)(i), and that such information is correct and complete;

(ii) That each laborer or mechanic (including each helper, apprentice, and trainee) employed on the contract during the payroll period has been paid the full weekly wages earned, without rebate, either directly or indirectly, and that no deductions have been made either directly or indirectly from the full wages earned, other than permissible deductions as set forth in 29 CFR part 3;

(iii) That each laborer or mechanic has been paid not less than the applicable wage rates and fringe benefits or cash equivalents for the classification of work performed, as specified in the applicable wage determination incorporated into the contract.

(3) The weekly submission of a properly executed certification set forth on the reverse side of Optional Form WH–347 shall satisfy the requirement for submission of the "Statement of Compliance" required by paragraph 3.b.(2) of this section.

(4) The falsification of any of the above certifications may subject the contractor or subcontractor to civil or criminal prosecution under 18 U.S.C. 1001 and 31 U.S.C. 231.

c. The contractor or subcontractor shall make the records required under paragraph 3.a. of this section available for inspection, copying, or transcription by authorized representatives of the contracting agency, the State DOT, the FHWA, or the Department of Labor, and shall permit such representatives to interview employees during working hours on the job. If the contractor or subcontractor fails to submit the required records or to make them available, the FHWA may, after written notice to the contractor, the contracting agency or the State DOT, take such action as may be necessary to cause the suspension of any further payment, advance, or guarantee of funds. Furthermore, failure to submit the required records upon request or to make such records available may be grounds for debarment action pursuant to 29 CFR 5.12.

4. Apprentices and trainees (29 CFR 5.5)

a. Apprentices (programs of the USDOL).

Apprentices will be permitted to work at less than the predetermined rate for the work they performed when they are employed pursuant to and individually registered in a bona fide apprenticeship program registered with the U.S. Department of Labor, Employment and Training Administration, Office of Apprenticeship Training, Employer and Labor Services, or with a State Apprenticeship Agency recognized by the Office, or if a person is employed in his or her first 90 days of probationary employment as an apprentice in such an apprenticeship program, who is not individually registered in the program, but who has been certified by the Office of Apprenticeship Training, Employer and Labor Services or a State Apprenticeship Agency (where appropriate) to be eligible for probationary employment as an apprentice.

The allowable ratio of apprentices to journeymen on the job site in any craft classification shall not be greater than the ratio permitted to the contractor as to the entire work force under the registered program. Any worker listed on a payroll at an apprentice wage rate, who is not registered or otherwise employed as stated above, shall be paid not less than the applicable wage rate on the wage determination for the classification of work actually performed. In addition, any apprentice performing work on the job site in excess of the ratio permitted under the registered program shall be paid not less than the applicable wage rate on the wage determination for the work actually performed. Where a contractor is performing construction on a project in a locality other than that in which its program is registered, the ratios and wage rates (expressed in percentages of the journeyman's hourly rate) specified in the contractor's or subcontractor's registered program shall be observed.

Every apprentice must be paid at not less than the rate specified in the registered program for the apprentice's level of progress, expressed as a percentage of the journeymen hourly rate specified in the applicable wage determination. Apprentices shall be paid fringe benefits in accordance with the provisions of the apprenticeship program. If the apprenticeship program does not specify fringe benefits, apprentices must be paid the full amount of fringe benefits listed on the wage determination for the applicable classification. If the Administrator determines that a different practice prevails for the applicable apprentice classification, fringes shall be paid in accordance with that determination.

In the event the Office of Apprenticeship Training, Employer and Labor Services, or a State Apprenticeship Agency recognized by the Office, withdraws approval of an apprenticeship program, the contractor will no longer be permitted to utilize apprentices at less than the applicable predetermined rate for the work performed until an acceptable program is approved.

b. Trainees (programs of the USDOL).

Except as provided in 29 CFR 5.16, trainees will not be permitted to work at less than the predetermined rate for the work performed unless they are employed pursuant to and individually registered in a program which has received prior approval, evidenced by formal certification by the U.S. Department of Labor, Employment and Training Administration.

The ratio of trainees to journeymen on the job site shall not be greater than permitted under the plan approved by the Employment and Training Administration.

Every trainee must be paid at not less than the rate specified in the approved program for the trainee's level of progress, expressed as a percentage of the journeyman hourly rate specified in the applicable wage determination. Trainees shall be paid fringe benefits in accordance with the provisions of the trainee program. If the trainee program does not mention fringe benefits, trainees shall be paid the full amount of fringe benefits listed on the wage determination unless the Administrator of the Wage and Hour Division determines that there is an apprenticeship program associated with the corresponding journeyman wage rate on the wage determination which provides for less than full fringe benefits for apprentices. Any employee listed on the payroll at a trainee rate who is not registered and participating in a training plan approved by the Employment and Training Administration shall be paid not less than the applicable wage rate on the wage determination for the classification of work actually performed. In addition, any trainee performing work on the job site in excess of the ratio permitted under the registered program shall be paid not less than the applicable wage rate on the wage determination for the work actually performed.

In the event the Employment and Training Administration withdraws approval of a training program, the contractor will no longer be permitted to utilize trainees at less than the applicable predetermined rate for the work performed until an acceptable program is approved.

c. Equal employment opportunity. The utilization of apprentices, trainees and journeymen under this part shall be in conformity with the equal employment opportunity requirements of Executive Order 11246, as amended, and 29 CFR part 30.

d. Apprentices and Trainees (programs of the U.S. DOT).

Apprentices and trainees working under apprenticeship and skill training programs which have been certified by the Secretary of Transportation as promoting EEO in connection with Federal-aid highway construction programs are not subject to the requirements of paragraph 4 of this Section IV. 23 CFR 230.111(e)(2). The straight time hourly wage rates for apprentices and trainees under such programs will be established by the particular programs. The ratio of apprentices and trainees to journeymen shall not be greater than permitted by the terms of the particular program.

5. Compliance with Copeland Act requirements. The contractor shall comply with the requirements of 29 CFR part 3, which are incorporated by reference in this contract as provided in 29 CFR 5.5.

6. Subcontracts. The contractor or subcontractor shall insert Form FHWA-1273 in any subcontracts and also require the subcontractors to include Form FHWA-1273 in any lower tier subcontracts. The prime contractor shall be responsible for the compliance by any subcontractor or lower tier subcontractor with all the contract clauses in 29 CFR 5.5.

7. Contract termination: debarment. A breach of the contract clauses in 29 CFR 5.5 may be grounds for termination of the contract, and for debarment as a contractor and a subcontractor as provided in 29 CFR 5.12.

8. Compliance with Davis-Bacon and Related Act requirements. All rulings and interpretations of the Davis-Bacon and Related Acts contained in 29 CFR parts 1, 3, and 5 are herein incorporated by reference in this contract as provided in 29 CFR 5.5.

9. Disputes concerning labor standards. As provided in 29 CFR 5.5, disputes arising out of the labor standards provisions of this contract shall not be subject to the general disputes clause of this contract. Such disputes shall be resolved in accordance with the procedures of the Department of Labor set forth in 29 CFR parts 5, 6, and 7. Disputes within the meaning of this clause include disputes between the contractor (or any of its subcontractors) and the contracting agency, the U.S. Department of Labor, or the employees or their representatives.

10. Certification of eligibility (29 CFR 5.5)

a. By entering into this contract, the contractor certifies that neither it (nor he or she) nor any person or firm who has an interest in the contractor's firm is a person or firm ineligible to be awarded Government contracts by virtue of section 3(a) of the Davis-Bacon Act or 29 CFR 5.12(a)(1).

b. No part of this contract shall be subcontracted to any person or firm ineligible for award of a Government contract by virtue of section 3(a) of the Davis-Bacon Act or 29 CFR 5.12(a)(1).

c. The penalty for making false statements is prescribed in the U.S. Criminal Code, 18 U.S.C. 1001.

V. CONTRACT WORK HOURS AND SAFETY STANDARDS ACT

Pursuant to 29 CFR 5.5(b), the following clauses apply to any Federal-aid construction contract in an amount in excess of \$100,000 and subject to the overtime provisions of the Contract Work Hours and Safety Standards Act. These clauses shall be inserted in addition to the clauses required by 29 CFR 5.5(a) or 29 CFR 4.6. As used in this paragraph, the terms laborers and mechanics include watchmen and guards.

1. Overtime requirements. No contractor or subcontractor contracting for any part of the contract work which may require or involve the employment of laborers or mechanics shall require or permit any such laborer or mechanic in any workweek in which he or she is employed on such work to work in excess of forty hours in such workweek unless such laborer or mechanic receives compensation at a rate not less than one and one-half times the basic rate of pay for all hours worked in excess of forty hours in such workweek. 29 CFR 5.5.

2. Violation; liability for unpaid wages; liquidated damages. In the event of any violation of the clause set forth in paragraph 1 of this section, the contractor and any subcontractor responsible therefor shall be liable for the unpaid wages. In addition, such contractor and subcontractor shall be liable to the United States (in the case of work done under contract for the District of Columbia or a territory, to such District or to such territory), for liquidated damages. Such liquidated damages shall be computed with respect to each individual laborer or mechanic, including watchmen and guards, employed in violation of the clause set forth in paragraph 1 of this section, in the sum currently provided in 29 CFR 5.5(b)(2)* for each calendar day on which such individual was required or permitted to work in excess of the standard workweek of forty hours without payment of the overtime wages required by the clause set forth in paragraph 1 of this section. 29 CFR 5.5.

* \$27 as of January 23, 2019 (See 84 FR 213-01, 218) as may be adjusted annually by the Department of Labor; pursuant to the Federal Civil Penalties Inflation Adjustment Act of 1990). 3. Withholding for unpaid wages and liquidated damages. The FHWA or the contacting agency shall upon its own action or upon written request of an authorized representative of the Department of Labor withhold or cause to be withheld, from any moneys payable on account of work performed by the contractor or subcontractor under any such contract or any other Federal contract with the same prime contractor, or any other federally-assisted contract subject to the Contract Work Hours and Safety Standards Act, which is held by the same prime contractor, such sums as may be determined to be necessary to satisfy any liabilities of such contractor or subcontractor for unpaid wages and liquidated damages as provided in the clause set forth in paragraph 2 of this section. 29 CFR 5.5.

4. Subcontracts. The contractor or subcontractor shall insert in any subcontracts the clauses set forth in paragraphs 1 through 4 of this section and also a clause requiring the subcontractors to include these clauses in any lower tier subcontracts. The prime contractor shall be responsible for compliance by any subcontractor or lower tier subcontractor with the clauses set forth in paragraphs 1 through 4 of this section. 29 CFR 5.5.

VI. SUBLETTING OR ASSIGNING THE CONTRACT

This provision is applicable to all Federal-aid construction contracts on the National Highway System pursuant to 23 CFR 635.116.

1. The contractor shall perform with its own organization contract work amounting to not less than 30 percent (or a greater percentage if specified elsewhere in the contract) of the total original contract price, excluding any specialty items designated by the contracting agency. Specialty items may be performed by subcontract and the amount of any such specialty items performed may be deducted from the total original contract price before computing the amount of work required to be performed by the contractor's own organization (23 CFR 635.116).

a. The term "perform work with its own organization" in paragraph 1 of Section VI refers to workers employed or leased by the prime contractor, and equipment owned or rented by the prime contractor, with or without operators. Such term does not include employees or equipment of a subcontractor or lower tier subcontractor, agents of the prime contractor, or any other assignees. The term may include payments for the costs of hiring leased employees from an employee leasing firm meeting all relevant Federal and State regulatory requirements. Leased employees may only be included in this term if the prime contractor meets all of the following conditions: (based on longstanding interpretation)

 the prime contractor maintains control over the supervision of the day-to-day activities of the leased employees;

(2) the prime contractor remains responsible for the quality of the work of the leased employees;

(3) the prime contractor retains all power to accept or exclude individual employees from work on the project; and

(4) the prime contractor remains ultimately responsible for the payment of predetermined minimum wages, the submission of payrolls, statements of compliance and all other Federal regulatory requirements.

b. "Specialty Items" shall be construed to be limited to work that requires highly specialized knowledge, abilities, or

equipment not ordinarily available in the type of contracting organizations qualified and expected to bid or propose on the contract as a whole and in general are to be limited to minor components of the overall contract. 23 CFR 635.102.

2. Pursuant to 23 CFR 635.116(a), the contract amount upon which the requirements set forth in paragraph (1) of Section VI is computed includes the cost of material and manufactured products which are to be purchased or produced by the contractor under the contract provisions.

3. Pursuant to 23 CFR 635.116(c), the contractor shall furnish (a) a competent superintendent or supervisor who is employed by the firm, has full authority to direct performance of the work in accordance with the contract requirements, and is in charge of all construction operations (regardless of who performs the work) and (b) such other of its own organizational resources (supervision, management, and engineering services) as the contracting officer determines is necessary to assure the performance of the contract.

4. No portion of the contract shall be sublet, assigned or otherwise disposed of except with the written consent of the contracting officer, or authorized representative, and such consent when given shall not be construed to relieve the contractor of any responsibility for the fulfillment of the contract. Written consent will be given only after the contracting agency has assured that each subcontract is evidenced in writing and that it contains all pertinent provisions and requirements of the prime contract. (based on longstanding interpretation of 23 CFR 635.116).

5. The 30-percent self-performance requirement of paragraph (1) is not applicable to design-build contracts; however, contracting agencies may establish their own self-performance requirements. 23 CFR 635.116(d).

VII. SAFETY: ACCIDENT PREVENTION

This provision is applicable to all Federal-aid construction contracts and to all related subcontracts.

1. In the performance of this contract the contractor shall comply with all applicable Federal, State, and local laws governing safety, health, and sanitation (23 CFR Part 635). The contractor shall provide all safeguards, safety devices and protective equipment and take any other needed actions as it determines, or as the contracting officer may determine, to be reasonably necessary to protect the life and health of employees on the job and the safety of the public and to protect property in connection with the performance of the work covered by the contract. 23 CFR 635.108.

2. It is a condition of this contract, and shall be made a condition of each subcontract, which the contractor enters into pursuant to this contract, that the contractor and any subcontractor shall not permit any employee, in performance of the contract, to work in surroundings or under conditions which are unsanitary, hazardous or dangerous to his/her health or safety, as determined under construction safety and health standards (29 CFR Part 1926) promulgated by the Secretary of Labor, in accordance with Section 107 of the Contract Work Hours and Safety Standards Act (40 U.S.C. 3704). 29 CFR 1926.10.

3. Pursuant to 29 CFR 1926.3, it is a condition of this contract that the Secretary of Labor or authorized representative thereof, shall have right of entry to any site of contract performance to inspect or investigate the matter of compliance

with the construction safety and health standards and to carry out the duties of the Secretary under Section 107 of the Contract Work Hours and Safety Standards Act (40 U.S.C. 3704).

VIII. FALSE STATEMENTS CONCERNING HIGHWAY PROJECTS

This provision is applicable to all Federal-aid construction contracts and to all related subcontracts.

In order to assure high quality and durable construction in conformity with approved plans and specifications and a high degree of reliability on statements and representations made by engineers, contractors, suppliers, and workers on Federalaid highway projects, it is essential that all persons concerned with the project perform their functions as carefully, thoroughly, and honestly as possible. Willful falsification, distortion, or misrepresentation with respect to any facts related to the project is a violation of Federal law. To prevent any misunderstanding regarding the seriousness of these and similar acts, Form FHWA-1022 shall be posted on each Federal-aid highway project (23 CFR Part 635) in one or more places where it is readily available to all persons concerned with the project:

18 U.S.C. 1020 reads as follows:

"Whoever, being an officer, agent, or employee of the United States, or of any State or Territory, or whoever, whether a person, association, firm, or corporation, knowingly makes any false statement, false representation, or false report as to the character, quality, quantity, or cost of the material used or to be used, or the quantity or quality of the work performed or to be performed, or the cost thereof in connection with the submission of plans, maps, specifications, contracts, or costs of construction on any highway or related project submitted for approval to the Secretary of Transportation; or

Whoever knowingly makes any false statement, false representation, false report or false claim with respect to the character, quality, quantity, or cost of any work performed or to be performed, or materials furnished or to be furnished, in connection with the construction of any highway or related project approved by the Secretary of Transportation; or

Whoever knowingly makes any false statement or false representation as to material fact in any statement, certificate, or report submitted pursuant to provisions of the Federal-aid Roads Act approved July 11, 1916, (39 Stat. 355), as amended and supplemented;

Shall be fined under this title or imprisoned not more than 5 years or both."

IX. IMPLEMENTATION OF CLEAN AIR ACT AND FEDERAL WATER POLLUTION CONTROL ACT (42 U.S.C. 7606; 2 CFR 200.88; EO 11738)

This provision is applicable to all Federal-aid construction contracts in excess of \$150,000 and to all related subcontracts. 48 CFR 2.101; 2 CFR 200.326.

By submission of this bid/proposal or the execution of this contract or subcontract, as appropriate, the bidder, proposer, Federal-aid construction contractor, subcontractor, supplier, or vendor agrees to comply with all applicable standards, orders or regulations issued pursuant to the Clean Air Act (42 U.S.C. 7401-7671q) and the Federal Water Pollution Control Act, as amended (33 U.S.C. 1251-1387), Violations must be reported to the Federal Highway Administration and the Regional Office of the Environmental Protection Agency. 2 CFR Part 200, Appendix II.

The contractor agrees to include or cause to be included the requirements of this Section in every subcontract, and further agrees to take such action as the contracting agency may direct as a means of enforcing such requirements. 2 CFR 200.326.

X. CERTIFICATION REGARDING DEBARMENT, SUSPENSION, INELIGIBILITY AND VOLUNTARY EXCLUSION

This provision is applicable to all Federal-aid construction contracts, design-build contracts, subcontracts, lower-tier subcontracts, purchase orders, lease agreements, consultant contracts or any other covered transaction requiring FHWA approval or that is estimated to cost \$25,000 or more – as defined in 2 CFR Parts 180 and 1200. 2 CFR 180.220 and 1200.220.

1. Instructions for Certification - First Tier Participants:

a. By signing and submitting this proposal, the prospective first tier participant is providing the certification set out below.

b. The inability of a person to provide the certification set out below will not necessarily result in denial of participation in this covered transaction. The prospective first tier participant shall submit an explanation of why it cannot provide the certification set out below. The certification or explanation will be considered in connection with the department or agency's determination whether to enter into this transaction. However, failure of the prospective first tier participant to furnish a certification or an explanation shall disqualify such a person from participation in this transaction. 2 CFR 180.320,

c. The certification in this clause is a material representation of fact upon which reliance was placed when the contracting agency determined to enter into this transaction. If it is later determined that the prospective participant knowingly rendered an erroneous certification, in addition to other remedies available to the Federal Government, the contracting agency may terminate this transaction for cause of default. 2 CFR 180.325.

d. The prospective first tier participant shall provide immediate written notice to the contracting agency to whom this proposal is submitted if any time the prospective first tier participant learns that its certification was erroneous when submitted or has become erroneous by reason of changed circumstances, 2 CFR 180.345 and 180.350.

e. The terms "covered transaction," "debarred," "suspended," "ineligible," "participant," "person," "principal," and "voluntarily excluded," as used in this clause, are defined in 2 CFR Parts 180, Subpart I, 180,900-180,1020, and 1200. "First Tier Covered Transactions" refers to any covered transaction between a recipient or subrecipient of Federal funds and a participant (such as the prime or general contract). "Lower Tier Covered Transactions" refers to any covered transaction under a First Tier Covered Transaction (such as subcontracts). "First Tier Participant" refers to the participant who has entered into a covered transaction with a recipient or subrecipient of Federal funds (such as the prime or general contractor). "Lower Tier Participant" refers any participant who has entered into a covered transaction with a First Tier Participant or other Lower Tier Participants (such as subcontractors and suppliers).

f. The prospective first tier participant agrees by submitting this proposal that, should the proposed covered transaction be entered into, it shall not knowingly enter into any lower tier covered transaction with a person who is debarred, suspended, declared ineligible, or voluntarily excluded from participation in this covered transaction, unless authorized by the department or agency entering into this transaction. 2 CFR 180.330.

g. The prospective first tier participant further agrees by submitting this proposal that it will include the clause titled "Certification Regarding Debarment, Suspension, Ineligibility and Voluntary Exclusion-Lower Tier Covered Transactions," provided by the department or contracting agency, entering into this covered transaction, without modification, in all lower tier covered transactions and in all solicitations for lower tier covered transactions exceeding the \$25,000 threshold. 2 CFR 180.220 and 180.300.

h. A participant in a covered transaction may rely upon a certification of a prospective participant in a lower tier covered transaction that is not debarred, suspended, ineligible, or voluntarily excluded from the covered transaction, unless it knows that the certification is erroneous. 2 CFR 180.300; 180.320, and 180.325. A participant is responsible for ensuring that its principals are not suspended, debarred, or otherwise ineligible to participate in covered transactions. 2 CFR 180.335. To verify the eligibility of its principals, as well as the eligibility of any lower tier prospective participants, each participant may, but is not required to, check the System for Award Management website (https://www.sam.gov/). 2 CFR 180.300, 180.320, and 180.325.

i. Nothing contained in the foregoing shall be construed to require the establishment of a system of records in order to render in good faith the certification required by this clause. The knowledge and information of the prospective participant is not required to exceed that which is normally possessed by a prudent person in the ordinary course of business dealings.

j. Except for transactions authorized under paragraph (f) of these instructions, if a participant in a covered transaction knowingly enters into a lower tier covered transaction with a person who is suspended, debarred, ineligible, or voluntarily excluded from participation in this transaction, in addition to other remedies available to the Federal Government, the department or agency may terminate this transaction for cause or default. 2 CFR 180.325.

....

2. Certification Regarding Debarment, Suspension, Ineligibility and Voluntary Exclusion – First Tier Participants:

a. The prospective first lier participant certifies to the best of its knowledge and belief, that it and its principals:

 Are not presently debarred, suspended, proposed for debarment, declared ineligible, or voluntarily excluded from participating in covered transactions by any Federal department or agency, 2 CFR 180.335;. (2) Have not within a three-year period preceding this proposal been convicted of or had a civil judgment rendered against them for commission of fraud or a criminal offense in connection with obtaining, attempting to obtain, or performing a public (Federal, State, or local) transaction or contract under a public transaction; violation of Federal or State antitrust statutes or commission of embezzlement, theft, forgery, bribery, falsification or destruction of records, making false statements, or receiving stolen property, 2 CFR 180.800;

(3) Are not presently indicted for or otherwise criminally or civilly charged by a governmental entity (Federal, State or local) with commission of any of the offenses enumerated in paragraph (a)(2) of this certification, 2 CFR 180.700 and 180.800; and

(4) Have not within a three-year period preceding this application/proposal had one or more public transactions (Federal, State or local) terminated for cause or default. 2 CFR 180.335(d).

(5) Are not a corporation that has been convicted of a felony violation under any Federal law within the two-year period preceding this proposal (USDOT Order 4200.6 implementing appropriations act requirements); and

(6) Are not a corporation with any unpaid Federal tax liability that has been assessed, for which all judicial and administrative remedies have been exhausted, or have lapsed, and that is not being paid in a timely manner pursuant to an agreement with the authority responsible for collecting the tax liability (USDOT Order 4200.6 implementing appropriations act requirements).

b. Where the prospective participant is unable to certify to any of the statements in this certification, such prospective participant should attach an explanation to this proposal. 2 CFR 180.335 and 180.340.

3. Instructions for Certification - Lower Tier Participants:

(Applicable to all subcontracts, purchase orders, and other lower tier transactions requiring prior FHWA approval or estimated to cost \$25,000 or more - 2 CFR Parts 180 and 1200). 2 CFR 180.220 and 1200.220.

 By signing and submitting this proposal, the prospective lower tier participant is providing the certification set out below.

b. The certification in this clause is a material representation of fact upon which reliance was placed when this transaction was entered into. If it is later determined that the prospective lower tier participant knowingly rendered an erroneous certification, in addition to other remedies available to the Federal Government, the department, or agency with which this transaction originated may pursue available remedies, including suspension and/or debarment.

c. The prospective lower tier participant shall provide immediate written notice to the person to which this proposal is submitted if at any time the prospective lower tier participant learns that its certification was erroneous by reason of changed circumstances. 2 CFR 180.365.

d. The terms "covered transaction," "debarred," "suspended," "ineligible," "participant," "person," "principal," and "voluntarily excluded," as used in this clause, are defined in 2 CFR Parts 180, Subpart I, 180.900 – 180.1020, and 1200. You may contact the person to which this proposal is submitted for assistance in obtaining a copy of those regulations. "First Tier Covered Transactions" refers to any covered transaction between a recipient or subrecipient of Federal funds and a participant (such as the prime or general contract). "Lower Tier Covered Transactions" refers to any covered transaction under a First Tier Covered Transaction (such as subcontracts). "First Tier Participant" refers to the participant who has entered into a covered transaction with a recipient or subrecipient of Federal funds (such as the prime or general contractor). "Lower Tier Participant" refers any participant who has entered into a covered transaction with a First Tier Participant or other Lower Tier Participants (such as subcontractors and suppliers).

e. The prospective lower tier participant agrees by submitting this proposal that, should the proposed covered transaction be entered into, it shall not knowingly enter into any lower tier covered transaction with a person who is debarred, suspended, declared ineligible, or voluntarily excluded from participation in this covered transaction, unless authorized by the department or agency with which this transaction originated. 2 CFR 1200.220 and 1200.332.

f. The prospective lower tier participant further agrees by submitting this proposal that it will include this clause titled "Certification Regarding Debarment, Suspension, Ineligibility and Voluntary Exclusion-Lower Tier Covered Transaction," without modification, in all lower tier covered transactions and in all solicitations for lower tier covered transactions exceeding the \$25,000 threshold. 2 CFR 180.220 and 1200.220.

g. A participant in a covered transaction may rely upon a certification of a prospective participant in a lower tier covered transaction that is not debarred, suspended, ineligible, or voluntarily excluded from the covered transaction, unless it knows that the certification is erroneous. A participant is responsible for ensuring that its principals are not suspended, debarred, or otherwise ineligible to participate in covered transactions. To verify the eligibility of its principals, as well as the eligibility of any lower tier prospective participants, each participant may, but is not required to, check the System for Award Management website (https://www.sam.gov/), which is compiled by the General Services Administration. 2 CFR 180.300, 180.320, 180.330, and 180.335.

h. Nothing contained in the foregoing shall be construed to require establishment of a system of records in order to render in good faith the certification required by this clause. The knowledge and information of participant is not required to exceed that which is normally possessed by a prudent person in the ordinary course of business dealings.

I. Except for transactions authorized under paragraph e of these instructions, if a participant in a covered transaction knowingly enters into a lower tier covered transaction with a person who is suspended, debarred, ineligible, or voluntarily excluded from participation in this transaction, in addition to other remedies available to the Federal Government, the department or agency with which this transaction originated may pursue available remedies, including suspension and/or debarment. 2 CFR 180.325.

.....

Certification Regarding Debarment, Suspension, Ineligibility and Voluntary Exclusion--Lower Tier Participants:

1. The prospective lower tier participant certifies, by submission of this proposal, that neither it nor its principals: (a) is presently debarred, suspended, proposed for debarment, declared ineligible, or voluntarily excluded from participating in covered transactions by any Federal department or agency, 2 CFR 180.355;

(b) is a corporation that has been convicted of a felony violation under any Federal law within the two-year period preceding this proposal (USDOT Order 4200.6 implementing appropriations act requirements); and

(c) is a corporation with any unpaid Federal tax fiability that has been assessed, for which all judicial and administrative remedies have been exhausted, or have lapsed, and that is not being paid in a timely manner pursuant to an agreement with the authority responsible for collecting the tax liability. (USDOT Order 4200.6 implementing appropriations act requirements)

 Where the prospective lower tier participant is unable to certify to any of the statements in this certification, such prospective participant should attach an explanation to this proposal.

....

XI. CERTIFICATION REGARDING USE OF CONTRACT FUNDS FOR LOBBYING

This provision is applicable to all Federal-aid construction contracts and to all related subcontracts which exceed \$100,000. 49 CFR Part 20, App. A.

 The prospective participant certifies, by signing and submitting this bid or proposal, to the best of his or her knowledge and belief, that:

a. No Federal appropriated funds have been paid or will be paid, by or on behalf of the undersigned, to any person for influencing or attempting to influence an officer or employee of any Federal agency, a Member of Congress, an officer or employee of Congress, or an employee of a Member of Congress in connection with the awarding of any Federal contract, the making of any Federal grant, the making of any Federal loan, the entering into of any cooperative agreement, and the extension, continuation, renewal, amendment, or modification of any Federal contract, grant, loan, or cooperative agreement.

b. If any funds other than Federal appropriated funds have been paid or will be paid to any person for influencing or attempting to influence an officer or employee of any Federal agency, a Member of Congress, an officer or employee of Congress, or an employee of a Member of Congress in connection with this Federal contract, grant, loan, or cooperative agreement, the undersigned shall complete and submit Standard Form-LLL, "Disclosure Form to Report Lobbying," in accordance with its instructions.

2. This certification is a material representation of fact upon which reliance was placed when this transaction was made or entered into. Submission of this certification is a prerequisite for making or entering into this transaction imposed by 31 U.S.C. 1352. Any person who fails to file the required certification shall be subject to a civil penalty of not less than \$10,000 and not more than \$100,000 for each such failure.

 The prospective participant also agrees by submitting its bid or proposal that the participant shall require that the language of this certification be included in all lower tier subcontracts, which exceed \$100,000 and that all such recipients shall certify and disclose accordingly.

XII. USE OF UNITED STATES-FLAG VESSELS:

This provision is applicable to all Federal-aid construction contracts, design-build contracts, subcontracts, lower-tier subcontracts, purchase orders, lease agreements, or any other covered transaction. 46 CFR Part 381.

This requirement applies to material or equipment that is acquired for a specific Federal-aid highway project. 46 CFR 381.7. It is not applicable to goods or materials that come into inventories independent of an FHWA funded-contract.

When oceanic shipments (or shipments across the Great Lakes) are necessary for materials or equipment acquired for a specific Federal-aid construction project, the bidder, proposer, contractor, subcontractor, or vendor agrees:

1. To utilize privately owned United States-flag commercial vessels to ship at least 50 percent of the gross tonnage (computed separately for dry bulk carriers, dry cargo liners, and tankers) involved, whenever shipping any equipment, material, or commodities pursuant to this contract, to the extent such vessels are available at fair and reasonable rates for United States-flag commercial vessels. 46 CFR 381.7.

2. To furnish within 20 days following the date of loading for shipments originating within the United States or within 30 working days following the date of loading for shipments originating outside the United States, a legible copy of a rated, 'on-board' commercial ocean bill-of-lading in English for each shipment of cargo described in paragraph (b)(1) of this section to both the Contracting Officer (through the prime contractor in the case of subcontractor bills-of-lading) and to the Office of Cargo and Commercial Sealift (MAR-620), Maritime Administration, Washington, DC 20590. (MARAD requires copies of the ocean carrier's (master) bills of lading, certified onboard, dated, with rates and charges. These bills of lading may contain business sensitive information and therefore may be submitted directly to MARAD by the Ocean Transportation Intermediary on behalf of the contractor). 46 CFR 381.7. ATTACHMENT A - EMPLOYMENT AND MATERIALS PREFERENCE FOR APPALACHIAN DEVELOPMENT HIGHWAY SYSTEM OR APPALACHIAN LOCAL ACCESS ROAD CONTRACTS (23 CFR 633, Subpart B, Appendix B) This provision is applicable to all Federal-aid projects funded under the Appalachian Regional Development Act of 1965.

 During the performance of this contract, the contractor undertaking to do work which is, or reasonably may be, done as on-site work, shall give preference to qualified persons who regularly reside in the labor area as designated by the DOL wherein the contract work is situated, or the subregion, or the Appalachian counties of the State wherein the contract work is situated, except:

 a. To the extent that qualified persons regularly residing in the area are not available.

b. For the reasonable needs of the contractor to employ supervisory or specially experienced personnel necessary to assure an efficient execution of the contract work.

c. For the obligation of the contractor to offer employment to present or former employees as the result of a lawful collective bargaining contract, provided that the number of nonresident persons employed under this subparagraph (1c) shall not exceed 20 percent of the total number of employees employed by the contractor on the contract work, except as provided in subparagraph (4) below.

2. The contractor shall place a job order with the State Employment Service indicating (a) the classifications of the laborers, mechanics and other employees required to perform the contract work, (b) the number of employees required in each classification, (c) the date on which the participant estimates such employees will be required, and (d) any other pertinent information required by the State Employment Service to complete the job order form. The job order may be placed with the State Employment Service in writing or by telephone. If during the course of the contract work, the information submitted by the contractor in the original job order is substantially modified, the participant shall promptly notify the State Employment Service.

3. The contractor shall give full consideration to all qualified job applicants referred to him by the State Employment Service. The contractor is not required to grant employment to any job applicants who, in his opinion, are not qualified to perform the classification of work required.

4. If, within one week following the placing of a job order by the contractor with the State Employment Service, the State Employment Service is unable to refer any qualified job applicants to the contractor, or less than the number requested, the State Employment Service will forward a certificate to the contractor indicating the unavailability of applicants. Such certificate shall be made a part of the contractor's permanent project records. Upon receipt of this certificate, the contractor may employ persons who do not normally reside in the labor area to fill positions covered by the certificate, notwithstanding the provisions of subparagraph (1c) above.

 The provisions of 23 CFR 633.207(e) allow the contracting agency to provide a contractual preference for the use of mineral resource materials native to the Appalachian region.

6. The contractor shall include the provisions of Sections 1 through 4 of this Attachment A in every subcontract for work which is, or reasonably may be, done as on-site work. "General Decision Number: FL20220193 02/25/2022

Superseded General Decision Number: FL20210193

State: Florida

Construction Type: Highway

Counties: Baker and Nassau Counties in Florida.

HIGHWAY CONSTRUCTION PROJECTS

Note: Contracts subject to the Davis-Bacon Act are generally required to pay at least the applicable minimum wage rate required under Executive Order 14026 or Executive Order 13658. Please note that these Executive Orders apply to covered contracts entered into by the federal government that are subject to the Davis-Bacon Act itself, but do not apply to contracts subject only to the Davis-Bacon Related Acts, including those set forth at 29 CFR 5.1(a)(2)-(60).

If the contract is entered into on or after January 30, 2022, or the contract is renewed or extended (e.g., an option is exercised) on or after January 30, 2022:	 Executive Order 14026 generally applies to the contract. The contractor must pay all covered workers at least \$15.00 per hour (or the applicable wage rate listed on this wage determination, if it is higher) for all hours spent performing on the contract in 2022.
If the contract was awarded on or between January 1, 2015 and January 29, 2022, and the contract is not renewed or extended on or after January 30, 2022:	이 아님 가지 않는 것 같아요. 것 같아요. 그가 앉아지 말았던 것 같아요. 아이지 않는 것이 같아요. 이 이 것 같아.

The applicable Executive Order minimum wage rate will be adjusted annually. If this contract is covered by one of the

Executive Orders and a classification considered necessary for performance of work on the contract does not appear on this wage determination, the contractor must still submit a conformance request.

Additional information on contractor requirements and worker protections under the Executive Orders is available at https://www.dol.gov/agencies/whd/government-contracts.

Modification Number	Publication	Date
0	01/07/2022	
1	02/25/2022	

* SUFL2013-054 08/19/2013

F	Rates	Fringes
CARPENTER, Includes Form Work\$	14.14 *	* 0.00
CEMENT MASON/CONCRETE FINISHER\$	14.35 *	* 0.00
ELECTRICIAN\$	21.87	0.00
FENCE ERECTOR\$	11.41 *	* 0.00
HIGHWAY/PARKING LOT STRIPING: Operator (Striping Machine)\$	12,37 *	* 0.32
HIGHWAY/PARKING LOT STRIPING: Painter\$	12,13 *	* 0.00
INSTALLER - GUARDRAIL\$	11.94 *	* 0.00
IRONWORKER, ORNAMENTAL\$	13.48 *	* 0.00
IRONWORKER, REINFORCING\$	15.77	0.00
IRONWORKER, STRUCTURAL\$	17.50	0.00
LABORER (Traffic Control Specialist)\$	10.94 *	* 0.00
LABORER: Asphalt, Includes Raker, Shoveler, Spreader and		
Distributor\$	13.61 *	* 0.00
LABORER: Common or General\$	10.70 *	* 0.00
LABORER: Concrete Saw (Hand		

Held/Walk	Behind)\$	12.04	**	0.00
LABORER:	Flagger\$	12.02	**	0.00
LABORER:	Grade Checker\$	13.64	**	0.00
LABORER:	Landscape &			
Irrigation	1\$	11.48	**	0.00
LABORER:	Mason Tender -			
Cement/Cor	ncrete\$	12.81	**	0.00
LABORER:	Pipelayer\$	14.42	**	0.00
OPERATOR:	Auger\$	12.43	**	0.00
OPERATOR:				
	kcavator/Trackhoe\$	15.06		0.00
OPERATOR:	Bobcat/Skid			
Steer/Skie		12.86	**	0.00
OPERATOR:	Boom\$	16.50		0.00
OPERATOR:	Boring Machine\$	17.18		0.00
OPERATOR:	Broom/Sweeper\$	11.60	**	0.00
OPERATOR:	Bulldozer\$	15,76		0.00
OPERATOR:	Concrete Finishing			
		15.44		0.00
OPERATOR:	Concrete Pump\$	19.57		0.00
OPERATOR :	Concrete Saw\$	15.09		0.00
OPERATOR :	Crane\$	20.62		0.00
OPERATOR:	Curb Machine\$	19.21		0.00
OPERATOR:	Distributor\$	15.01		0.00
OPERATOR:	Drill\$	14.71	**	0.00
OPERATOR:	Forklift\$	12.02	**	0.00
OPERATOR:	Gradal1\$	14.71	**	0.00
OPERATOR:	Grader/Blade\$	18,21		0.00

OPERATOR: Machine	Grinding/Grooving \$	16.07		0.00
OPERATOR :	Loader\$	14.07	**	0.00
OPERATOR:	Mechanic\$	18.20		0.00
OPERATOR:	Milling Machine\$	15.27		0.00
OPERATOR:	Oiler\$	14.92	**	0.00
OPERATOR: Aggregate,	Paver (Asphalt, and Concrete)\$	15.32		0.00
OPERATOR:	Piledriver\$	17.23		0.00
OPERATOR: (Guardrail,	Post Driver Fences)\$	20.22		0.00
OPERATOR:	Roller\$	12.61	**	0.00
OPERATOR:	Scraper\$	12.01	**	0.00
OPERATOR:	Screed\$	14.97	**	0.00
OPERATOR:	Tractor\$	12.91	**	0.00
OPERATOR:	Trencher\$	20.17		0.00
PAINTER: :	Spray\$	19.57		0.00
	SNALIZATION: gnal Installation\$	16.36		0.00
TRUCK DRIV	R: Dump Truck\$	13.47	**	0.00
TRUCK DRIV	R: Flatbed Truck\$	14.28	**	0.00
TRUCK DRIV	R: Lowboy Truck\$	15,85		0.00
TRUCK DRIV	R: Slurry Truck\$	11.96	**	0.00
TRUCK DRIV	R: Vactor Truck\$	13,70	**	0.00
TRUCK DRIV	R: Water Truck\$	14.23	**	0.00

WELDERS - Receive rate prescribed for craft performing operation to which welding is incidental.

** Workers in this classification may be entitled to a higher minimum wage under Executive Order 14026 (\$15.00) or 13658 (\$11.25). Please see the Note at the top of the wage determination for more information.

Note: Executive Order (EO) 13706, Establishing Paid Sick Leave for Federal Contractors applies to all contracts subject to the Davis-Bacon Act for which the contract is awarded (and any solicitation was issued) on or after January 1, 2017. If this contract is covered by the EO, the contractor must provide employees with 1 hour of paid sick leave for every 30 hours they work, up to 56 hours of paid sick leave each year. Employees must be permitted to use paid sick leave for their own illness, injury or other health-related needs, including preventive care; to assist a family member (or person who is like family to the employee) who is ill, injured, or has other health-related needs, including preventive care; or for reasons resulting from, or to assist a family member (or person who is like family to the employee) who is a victim of, domestic violence, sexual assault, or stalking. Additional information on contractor requirements and worker protections under the EO is available at

https://www.dol.gov/agencies/whd/government-contracts.

Unlisted classifications needed for work not included within the scope of the classifications listed may be added after award only as provided in the labor standards contract clauses (29CFR 5.5 (a) (1) (ii)).

The body of each wage determination lists the classification and wage rates that have been found to be prevailing for the cited type(s) of construction in the area covered by the wage determination. The classifications are listed in alphabetical order of ""identifiers"" that indicate whether the particular rate is a union rate (current union negotiated rate for local), a survey rate (weighted average rate) or a union average rate (weighted union average rate).

Union Rate Identifiers

A four letter classification abbreviation identifier enclosed in dotted lines beginning with characters other than ""SU"" or ""UAVG"" denotes that the union classification and rate were prevailing for that classification in the survey. Example: PLUM0198-005 07/01/2014. PLUM is an abbreviation identifier of the union which prevailed in the survey for this classification, which in this example would be Plumbers. 0198 indicates the local union number or district council number where applicable, i.e., Plumbers Local 0198. The next number, 005 in the example, is an internal number used in processing the wage determination. 07/01/2014 is the effective date of the most current negotiated rate, which in this example is July 1, 2014.

Union prevailing wage rates are updated to reflect all rate changes in the collective bargaining agreement (CBA) governing this classification and rate.

Survey Rate Identifiers

Classifications listed under the "SU"" identifier indicate that no one rate prevailed for this classification in the survey and the published rate is derived by computing a weighted average rate based on all the rates reported in the survey for that classification. As this weighted average rate includes all rates reported in the survey, it may include both union and non-union rates. Example: SULA2012-007 5/13/2014. SU indicates the rates are survey rates based on a weighted average calculation of rates and are not majority rates. LA indicates the State of Louisiana. 2012 is the year of survey on which these classifications and rates are based. The next number, 007 in the example, is an internal number used in producing the wage determination. 5/13/2014 indicates the survey completion date for the classifications and rates under that identifier.

Survey wage rates are not updated and remain in effect until a new survey is conducted.

Union Average Rate Identifiers

Classification(s) listed under the UAVG identifier indicate that no single majority rate prevailed for those classifications; however, 100% of the data reported for the classifications was union data. EXAMPLE: UAVG-OH-0010 08/29/2014. UAVG indicates that the rate is a weighted union average rate. OH indicates the state. The next number, 0010 in the example, is an internal number used in producing the wage determination. 08/29/2014 indicates the survey completion date for the classifications and rates under that identifier.

A UAVG rate will be updated once a year, usually in January of each year, to reflect a weighted average of the current negotiated/CBA rate of the union locals from which the rate is based.

WAGE DETERMINATION APPEALS PROCESS

1.) Has there been an initial decision in the matter? This can be:

- * an existing published wage determination
- * a survey underlying a wage determination
- * a Wage and Hour Division letter setting forth a position on a wage determination matter
- * a conformance (additional classification and rate) ruling

On survey related matters, initial contact, including requests for summaries of surveys, should be with the Wage and Hour National Office because National Office has responsibility for the Davis-Bacon survey program. If the response from this initial contact is not satisfactory, then the process described in 2.) and 3.) should be followed.

With regard to any other matter not yet ripe for the formal process described here, initial contact should be with the Branch of Construction Wage Determinations. Write to:

Branch of Construction Wage Determinations Wage and Hour Division U.S. Department of Labor 200 Constitution Avenue, N.W. Washington, DC 20210

2.) If the answer to the question in 1.) is yes, then an interested party (those affected by the action) can request review and reconsideration from the Wage and Hour Administrator (See 29 CFR Part 1.8 and 29 CFR Part 7). Write to:

Wage and Hour Administrator U.S. Department of Labor 200 Constitution Avenue, N.W. Washington, DC 20210

The request should be accompanied by a full statement of the interested party's position and by any information (wage payment data, project description, area practice material, etc.) that the requestor considers relevant to the issue.

3.) If the decision of the Administrator is not favorable, an

interested party may appeal directly to the Administrative Review Board (formerly the Wage Appeals Board). Write to:

> Administrative Review Board U.S. Department of Labor 200 Constitution Avenue, N.W. Washington, DC 20210

4.) All decisions by the Administrative Review Board are final.

END OF GENERAL DECISIO"



Requisition Form

NASSAU COUNTY BOARD OF COUNTY COMMISSIONERS 96135 Nassau Place Suite 1 Yulee, FL 32097

VENDOR NAME/ADDRESS

DEPARTMENT Capital Improvement Project Mana

CGC, Inc.	
7036 W. 12th Street	
Jacksonville, Florida 32220	
	1

REQUESTED BY

				Robert T. Companion		
	Citrona Pedestrian Improvemen 63470541 DESCRIPTION	-331492-C0054	\$ 223,247.00	Encum	ber Contract	CM3333
EM NO.			UNIT PRICE	AMOUNT		
	CGC, Inc. Construction Contract	1.00	\$ 221,747.00	\$ 221,747.00	Original Grant	for \$198,598 + pproved SA from 649 = \$223,247
				\$ 0.00	FDOT for \$24,	649 = \$223,247
				\$ 0.00		
				\$ 0.00		
		\mathbf{VO}		\$ 0.00		
		vp .		\$ 0.00	3.1	
				\$ 0.00		
				\$ 0.00		
				\$ 0.00		
				\$ 0.00		
		1.1		\$ 0.00		
				\$ 0.00		
				\$ 0.00		
				\$ 0.00		
				\$ 0.00		
				\$ 0.00		
				\$ 0.00		
				\$ 0.00		1

COPY - DEPARTMENT

Department Head

I attest that, to the best of my knowledge, this requisition reflects accurate information, has been reviewed, budgeted for and follows the Nassau County Purchasing Policy.

Robert Companion

3/1/2023

Office of Management and Budget (signature required if over Department Head signature authority or \$5,000, whichever is less.) I attest that, to the best of my knowledge, funds are available for payment.

Procurement Director (signature required if over Department Head signature authority or \$5,000, whichever is less.) I attest that, to the best of my knowledge, this requisition is accurate and necessary and is consistent with the Nassau County Purchasing Policy.

County Manager (signature required if over Department Head signature authority or \$5,000, whichever is less.) I attest that, to the best of my knowledge, the appropriate staff have reviewed and approved this Requisition and no other conditions would prevent approval.

Clerk:	_
Date:	

\$ 221,747.00

Total



VENDOR NUMBER

ITEM NO.

Requisition Form

NASSAU COUNTY BOARD OF COUNTY COMMISSIONERS 96135 Nassau Place Suite 1 Yulee, FL 32097

AMOUNT A

\$ 223,24

UNIT P

QUANTITY

FUNDING SOURCE

Citrona Pedestrian Improvemen 63470541-563202 C0054

DESCRIPTION

VENDOR NAME/ADDRESS

PROJECT NAME

CGC, Inc. 7036 W. 12th Street Jacksonville, Florida 32220

DEPARTMENT Capital Improvement Project Mana

			REQUESTED BY	
			Robert T. C	ompanion
AILABLE		STANDARD F	O OR ENCUMBER ONLY	CONTRACT NO.
7.00		Encumber	Contract	CM3333
RICE	AMOUNT	CONTRACTOR PARTY	Creating and the	
\$ 221,747.00		Original Grant for \$198,598 + plus recently approved SA from EDOT for \$24 649 = \$223 247		

Total

1	CGC, Inc. Construction Contract	1.00	\$ 221,747.00	\$ 221,747.00	Original Grant for \$198,598 + plus recently approved SA from FDOT for \$24,649 = \$223,247
				\$ 0.00	FDOT for \$24,649 = \$223,247
				\$ 0.00	
				\$ 0.00	
				\$ 0.00	
,				\$ 0.00	
			1.2.2.1	\$ 0.00	
		· · · · · · · · · · · · · · · · · · ·		\$ 0.00	
				\$ 0.00	
			· · · · · · · · · · · · · · · · · · ·	\$ 0.00	
				\$ 0.00	
				\$ 0.00	
				\$ 0.00	
				\$ 0.00	
				\$ 0.00	
				\$ 0.00	
				\$ 0.00	-
				\$ 0.00	
ORIGINAL	- FINANCE			Shippi	ng \$ 0.00

COPY - DEPARTMENT

Department Head

I attest that, to the best of my knowledge, this requisition reflects accurate information, has been reviewed, budgeted for and follows the Nassau County

Purchasing Bolicy. Kobert Companion

3/1/2023

Office of Management and Budget (aignature required if over Department Head signature authority or \$5,000, whichever is less.) l attest that, to the best of my knowledge, funds are available for payment. CUNS Lacambra 3/1/2023

Procurement Director (signature required if over Department Head signature authority or \$5,000, whichever is less.)

1 attest that, to the best of my knowledge, this requisition is accurate and necessary and is consistent with the Nassau County Purchasing Policy. 3/2/2023

County Manager (signature required if over Department Head signature authority or \$5,000, whichever is less.)

I affest that, to the best of my knowledge, the appropriate staff have reviewed and approved this Requisition and no other conditions would prevent approval. 3/7/2023

Clerk:	_
Date:	

\$ 221,747.00

DocuSign

Certificate Of Completion

Envelope Id: 5D3013DB2C2740B28CCD57B583C9BC42 Status: Completed Subject: Complete with DocuSign: Citrona Drive Road Construction Contract - DRAFT 02062023.pdf, CGC Requ... Source Envelope: Document Pages: 274 Signatures: 10 Envelope Originator: Initials: 2 **Robert Companion** Certificate Pages: 4 RCompanion@nassaucountyfl.com AutoNav: Enabled IP Address: 50.238.237.26

Envelopeld Stamping: Enabled Time Zone: (UTC-05:00) Eastern Time (US & Canada)

Record Tracking

Status: Original 2/7/2023 3:34:47 PM

Signer Events

Robert Companion rcompanion@nassaucountyfl.com Deputy County Manager - County Engineer Nassau County BOCC Security Level: Email, Account Authentication (None)

Electronic Record and Signature Disclosure: Not Offered via DocuSign

Lanaee Gilmore lgilmore@nassaucountyfl.com **Procurement Director** Nassau County BOCC Security Level: Email, Account Authentication (None)

Electronic Record and Signature Disclosure: Not Offered via DocuSign

Tracy Poore tpoore@nassaucountyfl.com **OMB** Admin

Nassau County BOCC

Security Level: Email, Account Authentication (None)

Electronic Record and Signature Disclosure: Not Offered via DocuSign

chris lacambra clacambra@nassaucountyfl.com **OMB** Director Nassau County BOCC Security Level: Email, Account Authentication (None)

Electronic Record and Signature Disclosure: Not Offered via DocuSign

RCompanion@nassaucountyfl.com

Signature

Robert Companion.

Signature Adoption: Pre-selected Style Using IP Address: 50.238.237.26

Lanare Helmore

Signature Adoption: Pre-selected Style Using IP Address: 50.238.237.26

Signature Adoption: Pre-selected Style

Using IP Address: 50.238.237.26

clivis lacambra

TP

Signature Adoption: Pre-selected Style Using IP Address: 50.238.237.26

Sent: 2/7/2023 3:51:48 PM Viewed: 2/8/2023 4:09:12 PM Signed: 2/8/2023 4:09:38 PM

Location: DocuSign

Sent: 2/7/2023 3:51:20 PM

Viewed: 2/7/2023 3:51:30 PM

Signed: 2/7/2023 3:51:42 PM

Timestamp

Sent: 2/8/2023 3:53:45 PM Resent: 2/8/2023 4:09:46 PM Viewed: 2/17/2023 2:45:16 PM Signed: 2/17/2023 2:45:28 PM

Sent: 2/17/2023 2:45:35 PM Viewed: 2/18/2023 7:35:01 PM Signed: 2/18/2023 7:35:08 PM

Holder: Robert Companion

Signer Events

Abigail Jorandby ajorandby@nassaucountyfl.com Assistant County Attorney Nassau BOCC Security Level: Email, Account Authentication (None)

Electronic Record and Signature Disclosure: Not Offered via DocuSign

Robert Companion rcompanion@nassaucountyfl.com Deputy County Manager - County Engineer Nassau County BOCC Security Level: Email, Account Authentication (None)

Electronic Record and Signature Disclosure: Not Offered via DocuSign

Robert Companion rcompanion@nassaucountyfl.com Deputy County Manager - County Engineer Nassau County BOCC Security Level: Email, Account Authentication (None)

Electronic Record and Signature Disclosure: Not Offered via DocuSign

chris lacambra clacambra@nassaucountyfl.com OMB Director Nassau County BOCC Security Level: Email, Account Authentication (None)

Electronic Record and Signature Disclosure: Not Offered via DocuSign

Lanaee Gilmore IglImore@nassaucountyfl.com Procurement Director Nassau County BOCC Security Level: Email, Account Authentication (None)

Electronic Record and Signature Disclosure: Not Offered via DocuSign

Denise C. May dmay@nassaucountyfl.com Assistant County Attorney

Nassau County BOCC Security Level: Email, Account Authentication (None)

Electronic Record and Signature Disclosure: Not Offered via DocuSign

Signature

as

Signature Adoption: Pre-selected Style Using IP Address: 50,238,237.26

Robert Companion

Signature Adoption: Pre-selected Style Using IP Address: 50.238.237.26

Robert Companion.

Signature Adoption: Pre-selected Style Using IP Address: 50.238.237.26

clivis lacambra

Signature Adoption: Pre-selected Style Using IP Address: 50.238.237.26

Kanaco Almolo

Signature Adoption: Pre-selected Style Using IP Address: 50.238.237.26

Denise C. May

Signature Adoption: Pre-selected Style Using IP Address: 50.238.237.26

Timestamp

Sent: 2/18/2023 7:35:16 PM Viewed: 3/1/2023 10:51:09 AM Signed: 3/1/2023 10:51:26 AM

Sent: 3/1/2023 10:51:34 AM Vlewed: 3/1/2023 10:54:00 AM Signed: 3/1/2023 10:54:18 AM

Sent: 3/1/2023 12:59:28 PM Viewed: 3/1/2023 12:59:59 PM Signed: 3/1/2023 1:00:12 PM

Sent: 3/1/2023 10:54:25 AM Resent: 3/1/2023 1:00:21 PM Viewed: 3/1/2023 1:02:05 PM Signed: 3/1/2023 1:02:51 PM

Sent: 3/1/2023 1:03:02 PM Viewed: 3/2/2023 4:28:28 PM Signed: 3/2/2023 4:28:35 PM

Sent: 3/2/2023 4:28:44 PM Viewed: 3/7/2023 8:13:20 AM Signed: 3/7/2023 8:14:01 AM

Signer Events	Signature	Timestamp
Taco Pope, AICP tpope@nassaucountyfl.com County Manager Nassau County BOCC	72	Sent: 3/7/2023 8:14:07 AM Viewed: 3/7/2023 8:16:18 AM Signed: 3/7/2023 8:16:26 AM
Security Level: Email, Account Authentication (None) Electronic Record and Signature Disclosure:	Signature Adoption: Drawn on Device Using IP Address: 50.238.237.26	
Not Offered via DocuSign		
In Person Signer Events	Signature	Timestamp
Editor Delivery Events	Status	Timestamp
Agent Delivery Events	Status	Timestamp
Intermediary Delivery Events	Status	Timestamp
Certified Delivery Events	Status	Timestamp
Carbon Copy Events	Status	Timestamp
Clerk Admin clerkservices@nassaucountyfl.com Security Level: Email, Account Authentication (None)	COPIED	Sent: 3/7/2023 8:16:33 AM
Electronic Record and Signature Disclosure: Not Offered via DocuSign		
Procurement Staff BOCCProcurement@nassaucountyfl.com Security Level: Email, Account Authentication	COPIED	Sent: 3/7/2023 8:16:36 AM
(None) Electronic Record and Signature Disclosure: Not Offered via DocuSign		
Witness Events	Signature	Timestamp
Notary Events	Signature	Timestamp
Envelope Summary Events	Status	Timestamps
Envelope Cont	A STATE AND A STATE AN	

Envelope Sent Envelope Updated Envelope Updated

Hashed/Encrypted Security Checked Security Checked

2/7/2023 3:51:20 PM 2/8/2023 3:53:44 PM 2/8/2023 3:53:44 PM 2/8/2023 3:53:44 PM 2/8/2023 3:53:44 PM 2/8/2023 3:56:05 PM 2/8/2023 4:08:17 PM 2/28/2023 4:21:21 PM 2/28/2023 4:21:21 PM 2/28/2023 4:21:21 PM 2/28/2023 4:21:21 PM

Envelope Summary Events	Status	Timestamps	
Envelope Updated	Security Checked	2/28/2023 4:21:21 PM	
Envelope Updated	Security Checked	2/28/2023 4:21:21 PM	
Envelope Updated	Security Checked	2/28/2023 4:21:21 PM	
Envelope Updated	Security Checked	2/28/2023 4:21:21 PM	
Envelope Updated	Security Checked	2/28/2023 4:21:21 PM	
Envelope Updated	Security Checked	3/1/2023 12:56:10 PM	
Envelope Updated	Security Checked	3/1/2023 12:56:10 PM	
Envelope Updated	Security Checked	3/1/2023 12:56:10 PM	
Envelope Updated	Security Checked	3/1/2023 12:59:27 PM	
Envelope Updated	Security Checked	3/1/2023 12:59:27 PM	
Envelope Updated	Security Checked	3/1/2023 12:59:27 PM	
Envelope Updated	Security Checked	3/1/2023 12:59:27 PM	
Envelope Updated	Security Checked	3/1/2023 12:59:27 PM	
Envelope Updated	Security Checked	3/1/2023 12:59:27 PM	
Envelope Updated	Security Checked	3/1/2023 12:59:27 PM	
Envelope Updated	Security Checked	3/1/2023 12:59:27 PM	
Envelope Updated	Security Checked	3/1/2023 12:59:27 PM	
Envelope Updated	Security Checked	3/1/2023 12:59:27 PM	
Envelope Updated	Security Checked	3/1/2023 12:59:27 PM	
Certified Delivered	Security Checked	3/7/2023 8:16:18 AM	
Signing Complete	Security Checked	3/7/2023 8:16:26 AM	
Completed	Security Checked	3/7/2023 8:16:36 AM	
Payment Events	Status	Timestamps	